ACCELERATING THE BIODEGRADATION OF POLY(LACTIC ACID) AT MESOPHILIC TEMPERATURES

By

Pooja C Mayekar

A DISSERTATION

Submitted to Michigan State University in partial fulfillment of the requirements for the degree of

Packaging—Doctor of Philosophy

2024

ABSTRACT

As plastic production and consumption increase globally, so does the amount of plastic waste to be disposed of and managed. Most worldwide plastic waste ends in landfills, open dumps, or leaks into the environment. Hence, new alternatives, such as developing biobased and biodegradable polymers, have been proposed to tackle the ever-growing plastic waste problem and help reduce the amount of plastic coupled with organic waste reaching landfills or incineration facilities. It is essential to understand the degradation behavior of these novel polymers to guarantee their ultimate biodegradation together with organic waste. Among them, poly(lactic acid) (PLA), a popular biobased and compostable polymer produced from renewable sources has garnered much interest due to its low environmental footprint and ability to replace conventional polymers and be disposed of in industrial compost environments. Although PLA is industrially compostable when subjected to a suitable set of conditions (i.e., aerobic thermophilic conditions for an extended period), its acceptance in industrial composting facilities is affected adversely due to longer timeframes to degrade than the readily biodegradable organic fraction of waste. So, PLA's requirement to be fully exposed to thermophilic conditions for prolonged periods to biodegrade has restricted its adoption and hindered its acceptance in industrial composting facilities. This dissertation proposes three different approaches to improve PLA biodegradation under mesophilic conditions to open new avenues of biodegradation, such as home/backyard composting and guaranteeing industrial composting biodegradation at similar times as that of readily biodegradable materials.

For the first approach, a reactive blend of PLA with thermoplastic starch (TPS) was produced and evaluated for biodegradation under mesophilic (37°C) and thermophilic (58°C) conditions. Films were tested for biodegradation by analysis of evolved CO₂ for 180 days in simulated composting conditions in an in-house built direct measurement respirometer (DMR). The films' average molecular weight (M_n) and crystallinity (X_c) were tracked throughout the test duration, and the kinetic degradation rate was calculated. The introduction of TPS positively affected accelerating PLA hydrolysis during the lag phase in both mesophilic and thermophilic conditions due to increased chain mobility, resulting in faster degradation of PLA at both biodegradation conditions.

The second approach involved the introduction of biostimulants in compost to target different stages of biodegradation and enhance the enzymatic activity of microorganisms. PLA and biostimulants, Fe₃O₄ nano-powder, skim milk, gelatin, and ethyl lactate were introduced into the compost media at 37°C. The CO₂ evolution, M_n , and X_c of PLA, PLA added with single biostimulants, and PLA added with a combination of biostimulants were evaluated to investigate the degradation of PLA. To attain M_n values of ≤ 10 kDa for PLA, PLA added with skim milk experienced a biodegradation acceleration of 15%, 25% with gelatin, and 22% with ethyl lactate. Fe₃O₄ enhanced the biodegradation rate by 17% whereas the combination of gelatin and Fe₃O₄ resulted in a substantial increase of biodegradation rate of around 30%.

The last approach explores the use of enzymatic pretreatments. PLA films were pretreated with proteinase K enzyme at 37°C for 7 and 10 days and at 58°C for 2 and 5 days. These films were later introduced in inoculated vermiculite at 37°C and 58°C in the DMR to investigate the effect of pretreatment by simulating home and industrial composting settings. The results showed a higher CO₂ evolution and visible degradation for PLA films pretreated with proteinase K compared to the untreated control PLA films.

This dissertation presented three innovative methods to speed up PLA film biodegradation in composting. It provides potential solutions to remove the barrier for degrading PLA in home and industrial composting conditions and to help address the plastic pollution challenge by effectively degrading biodegradable polymers with organic waste. Copyright by POOJA C MAYEKAR 2024

ACKNOWLEDGEMENTS

When it comes to crafting a thesis, the acknowledgment section stands out as one of the most daunting tasks. It brings to mind a quote from my advisor, Dr. Rafael Auras, who often says, "It's never about the destination, but it's about the journey," and indeed, what an extraordinary journey it has been.

I want to kick off my acknowledgments by expressing my deepest appreciation to Dr. Rafael Auras for his unwavering support, guidance, and belief in me throughout this endeavor. Not only has he been an exceptional advisor, but he has also served as a remarkable role model, a guide, and a mentor who always showed faith in me, guiding me towards the right path. From granting us the autonomy to work at our own pace to caring about our well-being as graduate students by checking if we've had enough rest, he has gone above and beyond the call of duty.

I also extend my heartfelt gratitude to my committee members, Dr. Maria Rubino, Dr. Eva Almenar, and Dr. Ramani Narayan (College of Engineering, Michigan State University), for their invaluable support and insightful feedback during my research journey.

No research endeavor can succeed without access to proper laboratory equipment and a competent lab manager. Therefore, I would like to acknowledge Mr. Aaron Walworth (School of Packaging, Michigan State University) for graciously allowing me to utilize the lab resources. He is always ready to help and teaches things with a smile no matter how many times it is needed.

Additionally, I am immensely grateful to my RAA research group members: particularly Dr. Anibal Bher, and Dr. Wanwarang Limsukon for being my friends and colleagues and for their unwavering guidance and support.

I would also like to express my gratitude to my SOP friends for their constant encouragement, transforming into a family away from home. A special mention goes to my partner Dr. HenryManoj Dsouza for his caring nature, encouragement, for creating a safe space for me, and for his endless support through the most difficult journey. I am truly grateful to my best friend Kajal Rajwal for donning so many hats – sister, soulmate, unpaid therapist, and more. I would like to extend my heartfelt appreciation to Dr. Kaustubh Rane for all the love, support, and companionship, for making me laugh and helping me find my spark back. You three inspire me every day to keep going and always have my back. I am profoundly grateful for all the moments we have shared and thank you for being my rock and for believing in me. A special mention to my oldest friends, Dhruv Ghiya & Amey Talekar for endlessly listening to me go on about my research and keeping it light and fun when I needed it.

Finally, none of this would have been possible without the love and support of my family, my father Chandrashekhar Mayekar, my mother Chhaya Mayekar, my brother Prasad Mayekar, and my fur baby my loyal companion Rhea (I miss you very dearly). I am deeply thankful to the Universe for granting me the strength and blessing me with the opportunity to be a Spartan! and aiming for the impossible.

TABLE OF CONTENTS

CHAPTER 1: INTRODUCTION 1 REFERENCES 6 CHAPTER 2: LITERATURE REVIEW 8 REFERENCES 153 CHAPTER 3: BREAKING IT DOWN: HOW THERMOPLASTIC STARCH ENHANCES POLY(LACTIC ACID) BIODEGRADATION IN COMPOST – A COMPARATIVE ANALYSIS OF REACTIVE BLENDS 205 REFERENCES 227 APPENDIX 3A: MATERIAL PROCESSING 234 APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS 236 APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS 237 REACTIVE BLEND 237 APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS 238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION 238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION 237
CHAPTER 1: INTRODUCTION 6 REFERENCES
CHAPTER 2: LITERATURE REVIEW. .8 REFERENCES. .153 CHAPTER 3: BREAKING IT DOWN: HOW THERMOPLASTIC STARCH ENHANCES POLY(LACTIC ACID) BIODEGRADATION IN COMPOST – A COMPARATIVE ANALYSIS OF REACTIVE BLENDS .205 REFERENCES. .227 APPENDIX 3A: MATERIAL PROCESSING .234 APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS .236 APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS .237 REACTIVE BLEND .237 APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS
CHAPTER 2: LITERATURE REVIEW
REFERENCES 153 CHAPTER 3: BREAKING IT DOWN: HOW THERMOPLASTIC STARCH ENHANCES POLY(LACTIC ACID) BIODEGRADATION IN COMPOST – A COMPARATIVE ANALYSIS OF REACTIVE BLENDS 205 REFERENCES 227 APPENDIX 3A: MATERIAL PROCESSING 234 APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS 236 APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS 237 APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS 238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION 238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION 238
CHAPTER 3: BREAKING IT DOWN: HOW THERMOPLASTIC STARCH ENHANCES POLY(LACTIC ACID) BIODEGRADATION IN COMPOST – A COMPARATIVE ANALYSIS OF REACTIVE BLENDS
POLY(LACTIC ACID) BIODEGRADATION IN COMPOST – A COMPARATIVE ANALYSIS OF REACTIVE BLENDS
ANALYSIS OF REACTIVE BLENDS
REFERENCES 227 APPENDIX 3A: MATERIAL PROCESSING 234 APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS 236 APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS 237 REACTIVE BLEND 237 APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS 238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC
APPENDIX 3A: MATERIAL PROCESSING
APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS
APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS REACTIVE BLEND
REACTIVE BLEND237 APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC
APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS238 APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC
APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC
PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC
CONDITIONS
APPENDIX 3F: CRYSTALLINITY MEASUREMENTS
CHAPTER 4: ACCELERATING BIODEGRADATION: ENHANCING POLY(LACTIC
ACID) BREAKDOWN AT MESOPHILIC ENVIRONMENTAL CONDITIONS WITH
BIOSTIMULANTS 244
REFERENCES
CHAPTER 5: SPEEDING IT UP: DUAL EFFECTS OF BIOSTIMULANTS AND IRON ON
BIODEGRADATION OF POLY(LACTIC ACID) AT MESOPHILIC CONDITIONS272
REFERENCES
APPENDIX 5A: PHYSICOCHEMICAL CHARACTERISTICS
APPENDIX 5B: CO ₂ EVOLUTION AND MINERALIZATION OF PLA IN THE
PRESENCE OF SKIM MILK, GELATIN, AND ETHYL LACTATE
APPENDIX 5C: CO ₂ EVOLUTION AND MINERALIZATION OF PLA IN THE DREGENCE OF FE O. NANODOWDED, CRIM MILK FE, AND ETHYL
I ACTATE + FE 204 NANOPOWDER, SKIM MILK + FE, AND EITIL 204
LACTATE \pm FE
CHAPTER 6: ENHANCING BIODEGRADATION OF POLY(LACTIC ACID) IN
MESOPHILIC AND THERMOPHILIC ENVIRONMENTAL CONDITIONS: THE ROLE
OF PROTEINASE K AS A PRETREATMENT
REFERENCES
APPENDIX 6A: PHYSICOCHEMICAL CHARACTERISTICS
CHAPTER 7: CONCLUSIONS AND RECOMMENDATIONS FOR FUTURE WORK323

LIST OF ABBREVIATIONS

BA	bonds of 1,4-butanediol of adipic acid
BOD	biochemical oxygen demand
BT	bonds of 1,4-butanediol and terephthalic acid
CO_2	carbon dioxide
EFP	environmental footprint
EPS	extracellular polymeric substances
GHG	greenhouse gas
MSW	municipal solid waste
X_c	degree of crystallinity
M_{no}	initial number-average molecular weight
M_n	number-average molecular weight
M_w	weight-average molecular weight
MWD	molecular weight distribution
ОН	hydroxyl group
PBAT	poly(butylene adipate-co-terephthalate)
PBS	poly(butylene succinate)
PBSA	poly(butylene succinate-co-adipate)
PBSe	poly(butylene sebacate)
PBSeT	poly(butylene sebacate terephthalate)
PBST	poly(butylene succinate terephthalate)
PBT	poly(butylene terephthalate)
PDLA	poly(D-lactide)
PDLLA	poly(D, L-lactide)
PE	poly(ethylene)

PEA	poly(ethylene adipate)
PES	poly(ethylene succinate)
PET	poly(ethylene terephthalate)
PGA	poly(glycolic acid)
PHA	poly(hydroxyalkanoates)
PHB	poly(hydroxybutyrate)
PHU	polyhydroxyurethane
PLA	poly(lactic acid)
PLLA	poly(L, L-lactide)
PS	polystyrene
PU	polyurethane
PVOH	poly(vinyl alcohol)
SEC	size exclusion chromatography
SUP	single-use plastic
TCA	tricarboxylic acid
THF	tetrahydrofuran
TPS	thermoplastic starch
UV	ultraviolet radiation
VOC	volatile organic compounds

CHAPTER 1: INTRODUCTION

1.1 Background and motivation

The last two decades have seen a significant revolution in the plastic industry concerning the fate of polymers in the environment post-consumer use [1]. As the global production of plastic increases, so does the attention towards its disposal and waste management scenario [2]. The current problem of plastic waste finding its way to oceans and remote locations, such as the alps, arctic has revealed the lack of waste management systems and policies targeting their responsible use and consumption [3–5]. The growing concern and environmental awareness of the problem of white pollution have been a driving force to develop green, biodegradable, recyclable, or compostable polymers with a low environmental footprint [6,7]. Replacing the fossil-based polymers with biodegradable polymers such as poly(lactic acid) (PLA), thermoplastic starch (TPS), polycaprolactone (PCL), and polyhydroxyalkanoate (PHA) is perceived as one of the many solutions to tackle the plastic pollution problem [8,9]. Implementing these polymers can expand their commercial market and help divert organic waste from reaching landfills.

PLA is a bio-based polymer that can be synthesized from the fermentation of dextrose obtained from crops such as corn, sugarcane, and cassava [10,11]. PLA is compostable in nature and can be biodegraded under the right conditions to produce carbon dioxide, water, and biomass, which is cycled back and non-toxic to the environment. In addition, PLA displays a moderate barrier to gas and flavor, is easily processed using extrusion, has high clarity and good stiffness equivalent to traditional polymers such as polystyrene [12].

PLA has found its applications in food packaging such as containers and trays [13– 16]. It can be collected with the organic fraction of municipal solid waste (MSW). The organic collection of MSW can be treated in industrial composting treatment facilities. These organic composting treatment facilities were created keeping in mind the considerable volume of organic waste that comes from food waste streams. However, they were not designed to process bioplastics such as PLA, whose degradation rate is lower than readily biodegradable feedstocks such as food waste.

Since PLA undergoes chemical hydrolytic degradation at temperatures higher than its glass transition temperature ($T_s \sim 58^{\circ}$ C), which has been reported as the primary controlling degradation mechanism, this brings a considerable change in the mechanical properties and reduction in molecular weight to a point where it can be easily assimilated by microorganisms present in the given environment [13]. PLA shows a more extended biodegradation lag phase at lower temperatures and degrades at a slower rate than other readily biodegradable materials such as starch and cellulose. This dependence of PLA on degrading at elevated temperatures reduces its disposal scenario to industrial composting and limits degradation on home or backyard composting [14,15].

Furthermore, since industrial composting facilities handle large volumes and the turnaround time can be faster than the current standard requirements, PLA may not be wholly degraded at the end of the composting process [16]. Any variation in the process related to the frequency of turning the compost pile, uniform mixing to aerate the piles, maintaining optimal temperature range, and avoiding the formation of pockets of anaerobic conditions through the compost heaps can easily derail the biodegradation of PLA, thereby leaving half degraded or at times whole packages of the same by the end of the composting cycle [17]. This defeats the purpose of compostable polymers and produces low-quality, contaminated, and unmarketable compost (soil conditioner and fertilizer). Overall, this negative effect has discouraged composting facilities from accepting biodegradable polymers like PLA [15,16,18].

Since biodegradation of PLA is a complex phenomenon and involves the action of extracellular enzymes released by the microbes at the same time, researchers have studied and advocated that apart from chemical hydrolysis, the action of the hydrolases class of enzymes plays an essential role in the breakdown of the high molecular weight PLA by severing the backbone chain [19–23]. Since PLA is marketed as a compostable polymer, inspecting different paths for improving PLA degradation in composting and ambient conditions is essential. So, enhancing the degradation of PLA under lower temperature conditions such as mesophilic range can open new avenues to dispose of PLA inhome/backyard composting and provide some assistance towards guaranteeing the degradation at industrial composting facilities.

Biostimulation, which is the addition of selective compounds in the degradation environment, is one way of inducing enzymatic degradation and accelerating the degradation of PLA. Researchers have studied the prospective of biostimulation by introducing different groups of nutrients and enhancing the degradation activity. Modifying the PLA structure through a process such as blending PLA with other biodegradable polymers to tailor its properties such as hydrophobicity, tensile strength, and elongation at the break while still retaining/ improving the biodegradability can be another option to advance its slow degradation rate. Finally, bioaugmentation is another favorable option that involves introducing specific microbial strains known to degrade PLA in the current environment and boosting its degradation rate. These three methods, modification of PLA structure (blending), biostimulation, and bioaugmentation, can open the venue to make PLA easily compostable in industrial facilities and provide an opportunity for home composting.

1.2 Overall goal and objectives

This dissertation aims to elucidate the degradation process of PLA at mesophilic temperatures and investigate different pathways to accelerate the aerobic biodegradation of PLA in simulated composting conditions. Three objectives have been outlined to accomplish this goal: 1) To modify PLA structure by blending PLA with other biobased biodegradable polymers to increase its biodegradation.

2) To assess and understand the effect of adding biostimulants to enhance the biodegradation of PLA by altering different steps involved in the biodegradation process.

3) To investigate the effect of pretreating PLA with proteinase K enzyme prior to introducing the films in a compost environment.

1.3 Dissertation overview

This dissertation is organized as follows.

The current chapter, Chapter 1, details the background and motivation for this dissertation study, including the main goal and the specific objectives outlined to accomplish this goal.

Chapter 2 lays out an extensive literature review regarding the main polymer classification, abiotic and biotic polymer degradation mechanism, mesophilic environments, polymer properties affecting the biodegradation rate, different assessment techniques to measure biodegradation, and polymers that can be blended with PLA to improve its properties and biodegradation. The chapter also covers the main class of enzymes active in PLA biodegradation and concludes by highlighting the biodegradation pathway of PLA and other biodegradable polymers.

Chapter 3 investigates the effect of reactively blending thermoplastic starch with PLA and compares biodegradation at two different simulated composting settings: industrial (thermophilic temperature range ~ 58°C) and home/backyard (mesophilic temperature range ~ 37°C). The study explores how the addition of thermoplastic starch improves PLA biodegradation.

Chapter 4 investigates the biostimulation technique to improve PLA biodegradation at mesophilic conditions. Different biodegradable biostimulants were added to the compost environment to stimulate the native microbial population and target different steps involved in the biodegradation process.

Chapter 5 explores the effect of a combination of biostimulants on PLA biodegradation when introduced in the compost environment. The combination of biostimulants targets two different steps (i.e., adding compounds to trigger enzyme generation and electron donor/acceptor compounds) in the PLA biodegradation process.

Chapter 6 presents the results of the bioaugmentation technique wherein PLA films are pretreated with proteinase K enzyme and subjected to industrial and home composting simulated degradation setting. A comparative analysis of PLA degradation at two different temperatures, 37°C and 58°C, for two different pretreatment time intervals was investigated.

Chapter 7 summarizes all the works in this dissertation and concludes with future work recommendations.

REFERENCES

- [1] Welden N. A. and Lusher A., Microplastics, in *Plastic Waste and Recycling*, vol. 32, no. 3, Elsevier, 2020, pp. 223–249.
- [2] Ahmed T., Shahid M., Azeem F., et al., Biodegradation of plastics: current scenario and future prospects for environmental safety, *Environ. Sci. Pollut. Res.*, vol. 25, no. 8, pp. 7287–7298, Mar. 2018, doi: 10.1007/s11356-018-1234-9.
- [3] Bergmann M., Mützel S., Primpke S., *et al.*, White and wonderful? Microplastics prevail in snow from the Alps to the Arctic, *Sci. Adv.*, vol. 5, no. 8, pp. 1–11, Aug. 2019, doi: 10.1126/sciadv.aax1157.
- Kaza, Silpa; Yao, Lisa C.; Bhada-Tata, Perinaz; Van Woerden F., What a Waste 2.0 A Global Snapshot of Solid Waste Management to 2050, Urban Development;. Washington, DC: World Bank. © World Bank., 2018. https://openknowledge.worldbank.org/handle/10986/30317 (accessed Jan. 31, 2022).
- [5] United Nations, Transforming Our World: The 2030 Agenda for Sustainable Development, 2015. https://sdgs.un.org/2030agenda (accessed Jan. 31, 2022).
- [6] Hatti-Kaul R., Nilsson L. J., Zhang B., et al., Designing Biobased Recyclable Polymers for Plastics, *Trends Biotechnol.*, vol. 38, no. 1, pp. 50–67, Jan. 2020, doi: 10.1016/j.tibtech.2019.04.011.
- [7] RameshKumar S., Shaiju P., O'Connor K. E., and P R. B., Bio-based and biodegradable polymers - State-of-the-art, challenges and emerging trends, *Curr. Opin. Green Sustain. Chem.*, vol. 21, pp. 75–81, Feb. 2020, doi: 10.1016/j.cogsc.2019.12.005.
- [8] Chen G.-Q. and Patel M. K., Plastics Derived from Biological Sources: Present and Future: A Technical and Environmental Review, *Chem. Rev.*, vol. 112, no. 4, pp. 2082– 2099, Apr. 2012, doi: 10.1021/cr200162d.
- [9] Zheng J. and Suh S., Strategies to reduce the global carbon footprint of plastics, *Nat. Clim. Chang.*, vol. 9, no. 5, pp. 374–378, May 2019, doi: 10.1038/s41558-019-0459-z.
- [10] Auras R., Harte B., and Selke S., An overview of polylactides as packaging materials, *Macromol. Biosci.*, vol. 4, no. 9, pp. 835–864, 2004, doi: 10.1002/mabi.200400043.
- Kabasci S., Biobased plastics, in *Plastic Waste and Recycling*, vol. 25, no. 11, Elsevier, 2020, pp. 67–96.
- [12] Castro-Aguirre E., Iñiguez-Franco F., Samsudin H., et al., Poly(lactic acid)—Mass production, processing, industrial applications, and end of life, Adv. Drug Deliv. Rev., vol. 107, pp. 333–366, Dec. 2016, doi: 10.1016/j.addr.2016.03.010.
- [13] Castro-Aguirre E., Auras R., Selke S., et al., Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.

- [14] Rosengren C., Some facilities stop accepting compostable packaging as contamination debate persists | Waste Dive, 2019. https://www.wastedive.com/news/compostablepackaging-rexius-US-Composting-Council-Conference/550012/ (accessed Mar. 04, 2022).
- [15] Bioplastics news, What Happens to Bioplastics at Industrial Composting Sites?, 2020. https://bioplasticsnews.com/2020/06/22/compostable-plastics-industrial-compostingsite/ (accessed Mar. 04, 2022).
- [16] Association of Recyclers of Oregon, A Message from Composters Serving Oregon, 2020. https://oregonrecyclers.org/blog/composters-say-no-compostable-packaging-andserviceware (accessed Jan. 31, 2022).
- [17] Kale G., Kijchavengkul T., Auras R., et al., Compostability of bioplastic packaging materials: An overview, Macromol. Biosci., vol. 7, no. 3, pp. 255–277, 2007, doi: 10.1002/mabi.200600168.
- [18] EcoEnclose, Composters Don't Want Your Packaging, Especially Your Bioplastic, 2021. https://www.ecoenclose.com/blog/composters-dont-want-your-packaging-especiallyyour-bioplastic-/ (accessed Mar. 31, 2022).
- [19] Jarerat A. and Tokiwa Y., Degradation of Poly(L-lactide) by a Fungus, *Macromol. Biosci.*, vol. 1, no. 4, pp. 136–140, Jun. 2001, doi: 10.1002/1616-5195(20010601)1:4<136::AID-MABI136>3.0.CO;2-3.
- [20] Masaki K., Kamini N. R., Ikeda H., and Iefuji H., Cutinase-Like Enzyme from the Yeast Cryptococcus sp. Strain S-2 Hydrolyzes Polylactic Acid and Other Biodegradable Plastics, *Appl. Environ. Microbiol.*, vol. 71, no. 11, pp. 7548–7550, Nov. 2005, doi: 10.1128/AEM.71.11.7548-7550.2005.
- [21] Zafar U., Houlden A., and Robson G. D., Fungal Communities Associated with the Biodegradation of Polyester Polyurethane Buried under Compost at Different Temperatures, *Appl. Environ. Microbiol.*, vol. 79, no. 23, pp. 7313–7324, Dec. 2013, doi: 10.1128/AEM.02536-13.
- [22] Butbunchu N. and Pathom-Aree W., Actinobacteria as Promising Candidate for Polylactic Acid Type Bioplastic Degradation, *Front. Microbiol.*, vol. 10, no. December, 2019, doi: 10.3389/fmicb.2019.02834.
- [23] Satti S. M., Shah A. A., Auras R., and Marsh T. L., Isolation and characterization of bacteria capable of degrading poly(lactic acid) at ambient temperature, *Polym. Degrad. Stab.*, vol. 144, pp. 392–400, Oct. 2017, doi: 10.1016/j.polymdegradstab.2017.08.023.

CHAPTER 2: LITERATURE REVIEW

2.1 Abstract

Finding alternatives to diminish plastic pollution has become one of the main challenges of modern life. A few alternatives have gained potential for a shift towards a more circular and sustainable relationship with plastics. Biodegradable polymers, derived from both bio- and fossil-based sources, have emerged as one feasible alternative to overcome inconveniences associated with the use and disposal of non-biodegradable polymers. Biodegradation of biodegradable polymers depends on the environments, abiotic and biotic factors, and the polymer bulk and surface properties, resulting in a plethora of parameters that create a complex process whereby biodegradation times and rates can vary immensely. The intent of this review is to provide background and a comprehensive, systematic, and critical assessment of the factors affecting the biodegradation of biodegradable polymers, with special focus on the mesophilic range. To accomplish this goal, the literature on biodegradable polymers since 1990 was assessed to create a holistic understanding of the main proxies responsible for biodegradation such as abiotic and biotic mechanisms of degradation, environments, factors, microbial populations, and polymer properties. Insights gained and remarks for potential future research are provided with focus on mesophilic aerobic environments and the main biodegradable polymers produced.

2.2 Introduction

Plastics are pervasive and have become an indispensable part of our everyday life. The nature of plastics and their easy processability, durability, low cost, and availability favor their use, opening up an array of opportunities in market segments such as consumer goods, food and medical packaging, plastic films and pots for the agriculture sector, construction, and automotive parts [1,2]. Between 1950 and 2020, global plastic production reached an accumulated amount of c. 9,500¹ million metric tons [1,3]. With annual production of c. 370 million metric tons in 2020, estimates for 2030 are c. 600² million metric tons. The global value of the plastic industry in 2020 was around USD 580 billion [4,5], and is expected to reach USD 800³ billion by 2030. Flexible and rigid single-use plastic (SUP) represents c. 40% of the total global plastic production [1].

The benefits of plastics and SUPs are numerous, and include lower water and energy consumption during production and use than glass and metal, which in turn reduces the cost of resources used [6]. However, the ability of plastics to persist, even in harsh environments, has led to white pollution (i.e., leakage and accumulation of plastics in the environment). SUPs have been blamed as one of the main offenders for white pollution and are a growing concern for our modern society since increasing amounts end up in landfills as a portion of municipal solid waste (MSW), as litter on land and in drainage systems worldwide, and ultimately leaking into rivers and oceans [7–9]. At present, c. 8 million metric tons of plastic end up in our oceans annually, in addition to the 150 million metric tons that are already circulating in marine environments since the dawn of the plastic era [10,11]. A recent prediction, if business continues as usual without mitigation measures, is that c. 90 million metric tons of plastic waste will reach the world's aquatic environments by 2030 [12].Plastics ending up in the environment mostly start as macromolecular structures formed as films, bottles, trays, etc. (macroplastics) and then break down into smaller fragments called microplastics, which are formed due to mechanical abrasion, radiation light, and heat, and

¹ Estimations were obtained from references [1,3]. Results are based on production estimated from reference [3] until 2015 and the addition of production for the 2016–2020 period from reference [1].

² Estimation was obtained based on a linear projection growth rate from 2006–2018 from each global region from references [1,3] and extrapolated to 2030.

³ Estimation obtained based on a linear global projection growth rate from 2016–2020 from references [4,5] and extrapolated to 2030.

can even be reduced to nanoplastics. Microplastics are a concern due to the ability to concentrate pollutants and become a channel for bioaccumulation, while nanoplastics are also a health concern since they can potentially translocate in cell membranes of living organisms and become a source of transporting toxic chemicals [13–15].

Most of the plastic waste in the ocean comes from land-based sources, such as agricultural soils, open dumps and industries, or mismanaged plastic waste from land litter and incomplete collection, and then ending up in sludge, sewage, and polluted streams, finding its way through river pathways and leading to global marine pollution [11,16,17]. Apart from rivers [16], wind and snow have also been identified as responsible for transporting airborne plastic debris to locations perceived uninhabitable and remote such as the polar regions and the French and Swiss Alps [18,19]. So, plastic pollution has called attention worldwide in the form of a global crisis leading to ecological imbalance [7,20].

A consumer paradigm shift about plastics has occurred due to the growing amount of unmanaged disposal of flexible SUPs, pushing industries to embrace the long-term circular economy of plastics [21–23]. As part of this circular economy, new challenges have been highlighted such as *novel policies* targeting responsible consumption, a push for *worldwide waste management infrastructure* creation to recover plastics, and the development, production and use of *highly recyclable or biodegradable* plastics with lower environmental footprint (EFP) [24–26]. Novel policies targeting responsible consumption have been developed, such as the 2030 Agenda for Sustainable Development by the United Nations establishing the seventeen Sustainable Development Goals (SDGs) to achieve a better and more sustainable future for all [27]. Specifically, Goal 12 stipulates sustainable consumption and production, which has been adopted by countries around the world to create novel policies about use of materials such as plastics [28]. For example, various U.S. states have established "extended producer responsibility" for packaging and have banned plastics bags [29–31]. Furthermore, bans or extra fees for some SUPs are already effective in the European Union and countries in Asia such as China and Indonesia [32–34], and are in development in New Zealand and Australia [35].

The need for worldwide waste management infrastructure has been noted: in 2016, the world generated c. 2 billion metric tons of MSW, and is expected to generate c. 2.6 billion metric tons of waste by 2030 if no measures are taken to curb the growing generation of waste [36]. High and upper-middle income economies generated 66% of the global MSW in 2016, representing c. 1.34 billion metric tons, and are expected to generate c. 1.6 billion metric tons by 2030; lower-middle and low-income economies generated 34% of global MSW in 2016, representing c. 0.7 billion metric tons, and are expected to generate 1.0 billion metric tons by 2030 [36]. Inadequate or scarce waste management solutions are prevalent in lower-middle and low-income economies. In 2016, high and upper-middle income economies had MSW collection rates of 96% and 82%, respectively, in urban areas; whereas lower-middle and lowincome economies had collection rates of 51% and 39%, respectively, in urban areas [36]. Therefore, as economies move from low and lower-middle to upper-middle and high-income, MSW generation will increase, and waste management systems and collection must improve accordingly. Concentrated efforts are being directed to improve material recovery facilities around the world, with special emphasis on the lower-middle and low-income economies [37,38]. However, without MSW collection, white pollution will not be deterred.

To tackle the problem of plastic pollution, cradle approaches to deal with the production of highly recyclable and biodegradable polymers with lower EFP are increasingly being considered [39,40]. Replacing fossil-based plastics with bio-based plastics is one strategy shown to reduce the greenhouse gas (GHG) emissions produced by plastics [41–43]. For example, substitution of c. 66% of the world's fossil-based plastics with bio-based plastics has been estimated to reduce GHG emissions by 241 to 316 Mt CO₂-eq. per year [44]. To date,

mechanical recycling has been the main recovery method for polymers such as poly(ethylene terephthalate) (PET) and poly(ethylene) (PE). However, novel chemical upcycling methods for polymers are being explored, which entail the use of rejected plastic (waste) and converting it to a high-value resource in the form of fuels, chemicals and novel polymers, thus avoiding the buildup of plastic and reducing the use of fossil fuels [45]. Most of these new recovery methods are also being explored for flexible SUPs [46].

Among the cradle approaches to deal with littering and recovery of plastics, the production of biodegradable polymers is a promising solution, primarily since they can be recovered by traditional waste management options, including mechanical and chemical recycling and energy recovery, and the additional recovery route through aerobic industrial and home composting or anaerobic digestion. If enough volume of isotropic biodegradable polymers is collected and treated, they can be recycled. However, if volumes collected are low and/or the materials are contaminated, they can be routed towards biodegradation recovery scenarios such as industrial and home composting, with soil biodegradation being a special route for agriculture films.

The efficacy of biodegradation is conditioned by drastically different environmental conditions, such as heat, humidity, and acidic or alkaline media, and by the polymer characteristics such as chemical structure and physical properties.

Previous reviews on the biodegradation of polymers in some of these environments and conditions have focused on biodegradable polymers in general [47–49], biodegradable polyesters [50,51], and mechanisms of degradation [52,53]. Furthermore, recently works have reviewed and identified gaps and research needs in this area [54,55]. This comprehensive review expands on these previous works, and aims to provide an overview of the mechanisms, environments, and factors affecting the biodegradation of biodegradable polymers, giving special attention to the mesophilic degradation range (20 - 45°C). The specific goals are: 1) to provide a transdisciplinary background on the aspects affecting the biodegradation of biodegradable polymers,

2) to describe the different methods used for assessing biodegradation at mesophilic temperatures, and

3) to provide insights on the degradation pathway followed by polymers susceptible to biodegradation.

2.3 Bio- and fossil-based biodegradable polymer classification

Figure 2.1 provides a general classification of polymers according to their feedstock source and their ability to experience biodegradation. The first group of polymers are biobased in nature and non-biodegradable, such as bio-based poly(ethylene) (Bio-PE) and biobased poly(ethylene terephthalate) (Bio-PET). The second group of polymers are bio-based and biodegradable, such as poly(lactic acid) (PLA), poly(hydroxyalkanoate)s (PHAs), cellulose, and starch. The third group includes polymers that are derived from fossil-based sources, but also present biodegradable characteristics, such as poly(butylene adipate-*co*terephthalate) (PBAT) and poly(caprolactone) (PCL). The fourth and final group corresponds to the conventional group of polymers that are derived from fossil-based sources and are nonbiodegradable, such as low- and high-density PE (LDPE, and HDPE) and PET. This classification is very general since the characteristics of the material, the environment, and the rate of biodegradation for polymers vary widely among these groups.

As shown in **Figure 2.1**, biodegradability of a polymer in regular environmental conditions is not strictly related to the source of the polymer, and factors such as its chemical structure and physical properties are essential [56]. Some bio-based polymers, such as bio-PE and bio-PET, are more difficult to degrade than their fossil-based counterparts (i.e., PE and PET); whereas some fossil-based polymers, such as fossil-poly(butylene succinate) (PBS)

and PBAT, are as easily biodegradable as some bio-based polymers, such as bio-PBS and PHAs, when assessed under standard conditions [57,58].



Figure 2.1 Classification of polymers considering their bio-based or fossil-based feedstock and ability to be biodegradable or non-biodegradable in environments as compost, soil, and aquatic; and carbon cycle of bio-based and fossil-based polymers; adapted from [59,60].

Considering the carbon used to produce polymers, the main benefits of biodegradable polymers can be obtained when the polymers are produced from renewable resources since they can replenish the carbon cycle (i.e., the times needed to create them and to convert them to biomass are equivalent) (**Figure 2.1**). Fossil-based polymers can also be considered renewable like the bio-based polymers, but the main difference between both is the amount of time needed to convert to biomass and then back to their original form. Biodegradable polymers produced from bio-based resources take far less time to be converted to biomass whereas the fossil-based polymers take millions of years to achieve the same. The longer time frames are due to the imbalance between the rate of consumption and the replenishment rate, which further leads to mass imbalance in the carbon cycle. There is no additional carbon footprint associated with renewable-carbon feedstock used to produce biodegradable polymers, such as starch-heavy crops not intended for human consumption (e.g., starches from field/feed corn), due to quite similar time frames for consumption and conversion to biomass [43,60,61].

2.4 Abiotic and biotic polymer degradation mechanisms

Polymer degradation is defined as an irreversible change of the chemical structure, physical properties, and visual appearance due to the chemical cleavage of the polymer's constitutive macromolecules by one or more mechanisms acting concurrently [49]. More than one mechanism can simultaneously take place due to the action of external factors, and one mechanism can be more dominant than others at any time [48]. External factors associated with the environment, such as heat, humidity, radiation, and acidic or alkaline conditions, could modify the degradation process and its rate. The degradation process can alter polymer properties such as mechanical, optical, electrical, discoloration, phase separation or delamination, erosion, cracking and crazing [49]. The four main abiotic mechanisms associated with polymer degradation are mechanical, thermal (or thermo-oxidative), photo (photo-oxidative), and hydrolytic (chemical) degradation, some of which can be assisted by catalysis. In addition, ozone degradation (chemical) is considered a mechanism of degradation for polymers but is less common (Figure 2.2). Mechanical and thermal degradation can occur at the early stage of polymer processing. Photodegradation is observed when polymers are exposed to radiation like ultraviolet (UV) light or gamma rays. Abiotic hydrolytic (chemical) degradation is considered the most significant mechanism for biodegradable polymers such as polyesters. These abiotic mechanisms of degradation can be combined, accelerating the final biotic enzymatic process, for example, increasing the

exposed surface area for interaction with the microbial population [48]. We will be discussing hydrolytic and biotic degradation in detail below.



Figure 2.2 Main abiotic and biotic mechanisms of polymer degradation.

2.4.1 Hydrolytic degradation

Chemical hydrolytic degradation or abiotic hydrolysis is one of the main abiotic degradation mechanisms for biodegradable polymers, especially for aliphatic and aliphatic/aromatic polyesters. In this review, we refer to this mechanism also as chemical hydrolysis. With the uptake of water, susceptible chemical bonds in polymers can undergo chain scission, resulting in a reduction of M_w , loss of mass and mechanical properties, and increased surface area of the polymer, thereby increasing the available sites for attack by enzymatic activity, the biotic step initiated by microorganisms [48,52,62].

Chemical hydrolysis proceeds via two mechanisms when considering the macro structure: bulk or surface erosion. Depending on the conditions, these mechanisms can occur independently or combined. Bulk erosion is the dominant mechanism when the water diffusion is higher than the hydrolysis reaction rate, and surface erosion is dominant when the water diffusion into the polymer matrix or bulk is lower than the hydrolysis reaction rate [48,50,63].

When bulk erosion is the dominant mechanism, the M_w of the polymer bulk is reduced so that the polymer loses its mechanical properties in a short period of time. Due to the M_w reduction and higher mobility of shorter polymeric chain segments, crystallinity may change. Loss of mass and change/s in geometric shape take more time. The byproducts of bulk erosion are first accumulated; when the polymer chains are short enough, and reaching *n*-mers size, they can start to diffuse out. When the polymer undergoes surface erosion, the loss of mass is mostly from the surface while the bulk remains intact. As degradation advances, the loss of mass happens faster at the surface and the polymer gets smaller in size. When compared with bulk erosion, the mechanical properties and M_w are preserved for an extended period, and release of byproducts from the surface occurs from the beginning [63].

The kinetic rate of the chemical hydrolysis – surface or bulk dominant – depends on and can be affected by several factors associated with the polymer itself and the environment. The roles of some factors are discussed in the next sections and additional information can be found elsewhere [48,50,63,64].

In terms of environmental factors, increases in temperature and moisture intensify the rate of chemical hydrolysis [65]. Polymer chain mobility increases as the temperature increases. Hence, the susceptibility of hydrolysable bonds to undergo chain scission increases. The chemical potential of water on the surrounding media plays a significant role in hydrolysis of polymers [66]. Hydrolysis in acidic or alkaline conditions can occur through different mechanisms so the byproducts of the reactions can differ [67]. Finally, catalysts can modify the rate of the hydrolytic process [68,69].

In terms of polymer factors, hydrophilic polymers are more susceptible to hydrolytic degradation than are hydrophobic polymers [48]. Hydrolysis depends on the presence of

hydrolysable covalent bonds, such as esters, ethers, anhydrides, carbamide (urea), and ester amide (urethane), which increase the rate of chemical hydrolysis [52,70]. **Table 2.1** compares the half-lives of hydrolysable bonds in various polymers and shows that poly(anhydride)s are subjected to rapid hydrolysis due to the presence of hydrolysable bonds of very low half-life. By contrast, polyamides are very resistant to hydrolysis due to the resistance of the amide bonds to hydrolysis. The kinetics of the hydrolysable bond half-life presented in **Table 2.1** can increase or decrease due to the influence of neighboring groups.

The presence of amorphous regions increases the chemical hydrolysis rate due to easy diffusion of water into the polymer matrix, compared to semi- and crystalline polymers showing well-organized structure where diffusion is limited, even at temperature higher than T_g [71–73]. So, for polymers with lower or similar values of T_g than the mesophilic range, the diffusion likely is mostly controlled by the amorphous region where chemical hydrolysis is dominant.

Polymer	Chemical structure	Half-life*
Poly(anhydrides)	$- \begin{bmatrix} \mathbf{O} & \mathbf{O} \\ \ & \ \\ \mathbf{R} - \mathbf{C} - \mathbf{O} - \mathbf{C} \end{bmatrix}_{\mathbf{O}}$	0.1 hours
Poly(ketal)		3 hours
Poly(ortho esters)		4 hours

Table 2.1 Half-lives of hydrolysable bonds (in water at pH 7 and 25 °C). Adapted from [74].

Table 2.1 (cont'd)



*Half-life: time required for 50% hydrolysis in water at pH = 7 and 25 °C for the low M_w (methyl, ethyl) model compounds.

The macro structure properties, such as size and shape of the polymer, are factors that condition whether the dominant mechanism will be either surface or bulk erosion. In this way, a material can go from surface to bulk erosion when its thickness is reduced to a value lower than a critical value, called critical sample thickness (L_{crit}) [64,75].

Polyesters that contain ester groups are degraded mainly by chemical hydrolysis. Bulk degradation is predominant for aliphatic polyesters, such as poly(glycolic acid) (PGA), PLA, PCL, and PBS. The main stages of the hydrolytic degradation of polyesters undergoing bulk erosion can be summarized as 1) diffusion of water in the polymer matrix (amorphous regions); 2) water reacting with random ester linkage to produce shorter chains; 3) autocatalysis due to the presence of acid chain ends in the medium; and 4) release of watersoluble oligomers and monomers creating a void core and subsequent reduction in M_w [48,76]. The duration of the chemical hydrolysis process depends mainly on the initial M_w , crystallinity, temperature, and pH [53].

PLA is an example for chemical hydrolysable polymer degradation that does not require enzymes to catalyze the hydrolytic degradation. In this sense, the media to which the material is exposed and factors such as temperature, pH, and moisture play major roles in delaying or speeding up the hydrolytic degradation rate. In an industrial composting process (~58 °C and ~60% RH), PLA can absorb water and undergo chemical hydrolytic degradation. However, at lower temperatures, such as in agricultural soil environments (~25 $^{\circ}$ C and ~45% RH), the rate of chemical hydrolysis is low, increasing the time for the enzymatic hydrolysis process to start. One of the main differences between bulk and surface erosion mechanisms can be recognized in the diffusion of the degradation byproducts. During bulk degradation of polyesters, these hydrolysis-formed oligomers and monomer byproducts, such as carboxylic acid and hydroxyl groups, are trapped and accumulated inside the bulk, leading to an autocatalytic degradation that tends to accelerate the degradation kinetics [50,77]. Burkersroda et al. [75] reported that hydrolytic degradation of PLA, evaluated at 37 °C, follows a bulk erosion mechanism for thicknesses between 0.5 and 2 mm, a core-accelerated erosion for thicknesses between 2 mm and 74 mm, and surface erosion for thicknesses greater than 74 mm. Houglund et al. [78] reported that the hydrolysis of 100% PLLA increased upon the addition of a low percentage of d-Lactide units, due to reduction of the polymer order structure, showing the effect of tacticity and optical purity on the hydrolytic degradation of PLA. PLA hydrolysis is delayed in comparison to PGA hydrolysis due to the presence of the methyl group in PLA that blocks the attack of water to interact with the hydrolysable bonds

[48,63]. A review of the hydrolysis of PLA at mesophilic and thermophilic conditions has been reported by Tsuji [64].

PBAT, due to the presence of an aromatic group in its polyester chain experiences a lower hydrolytic degradation rate than polyester with only aliphatic units as PLA and PGA [79]. The presence of the aromatic group reduces chain flexibility, provides less susceptible bonds, and creates an steric interference effect to the access of the susceptible ester bonds [80]. The soft aliphatic domain bonds consisting of 1,4-butanediol and adipic acid monomers (BA) are more susceptible to hydrolysis than the hard aromatic bonds of 1,4-butanediol and terephthalic acid monomers (BT). In this sense, PBAT displays good biodegradability when the aromatic moiety concentration is kept below 55 mol% [81]. Kijchavengkul et al. [82] also demonstrated that the increase of crosslinking on PBAT has a detrimental effect, not only on chemical hydrolysis but also in enzymatic hydrolysis.

Polymers that undergo surface erosion are desirable when designing medical devices and for drug release, since the retention of mechanical properties and capacity for a controlled release of drugs can be achieved by mass loss without compromising the M_w . Polymers that can mostly undergo surface erosion are polyanhydrides, some poly(ortho esters), and some polycarbonates [83–85].

2.4.2 Biotic enzymatic degradation

Biotic enzymatic degradation is the mechanism of degradation where microorganisms break down organic substances through an enzymatic process. The four main stages of biotic degradation are shown in **Figure 2.3**. The main outcome of biotic degradation is reduction of the polymer to small molecules that are utilized by the microorganisms as a source of carbon and energy, resulting in final products like CO₂ and H₂O in aerobic conditions or CO₂, H₂O and CH₄ in anaerobic conditions. Microorganisms like bacteria and fungi are actively involved in the biodegradation process. These microorganisms have their own optimal growth conditions; for this reason, biotic degradation is a complex process where several factors associated with the polymer, microorganisms, and the environment come into play [65].

Abiotic chemical hydrolysis degradation, and biotic enzymatic degradation are the two main processes for the cleavage of polymer bonds and degradation [63]. The biodegradation process involves several steps, some of them are abiotic, and some are biotic and act synergistically to decompose the organic matter. Some of the abiotic mechanisms described above, such as photo, hydrolytic, or even mechanical degradation, can enhance the biotic degradation process by increasing the surface area for biofilm formation or by reducing the M_w [53]. However, the dominant mechanism in the biotic degradation process is related to biotic agents.

The first stage of the biodegradation process, as shown in **Figure 2.3**, is biofilm formation. In the second stage, depolymerization, microorganisms secrete extracellular enzymes. These agents can cleave the molecules by random or end chain scission, reducing the M_w and resulting in the generation of oligomers, dimers, and monomers. The susceptibility of a polymer to microbial attack depends on enzyme availability, availability of sites in the polymer for enzyme attack, enzyme specificity for the polymer, and the presence of cofactors [53]. In the third stage, bioassimilation, the molecules transported from the depolymerization stage are assimilated into the microorganism's metabolism and are used to produce energy, new biomass, and primary and secondary metabolites. In the final stage, mineralization, simple molecules such as CO_2 , CH_4 , H_2O , N_2 , and different salts from intracellular metabolites, which are completely oxidized, are released to the environment [86]. These non-toxic components are redistributed through the carbon, nitrogen, and sulfur cycles in nature. Some simple and complex metabolites, such as organic acids, terpenes, aldehydes, and antibiotics, can reach the extracellular surroundings by excretion [52].



Figure 2.3 Main stages of the biotic degradation process: (1) biofilm formation, (2) depolymerization, (3) bioassimilation, and (4) mineralization. Adapted from [87].

The literature has also described these processes as biodeterioration. Biodeterioration is commonly described as the undesirable degradation of materials by microorganisms, and it is considered the mechanism responsible for causing several irreparable damages. In this review, to avoid any confusion regarding the stages of the biodegradation, we will not use this term to describe any particular process. The term biodeterioration is mostly used to describe a combined mechanism when microorganisms are in contact with a polymer surface and the polymer experiences fouling, degradation of leaching components (as additives), biocorrosion (enzymes and radicals attack additives and polymer backbone), hydration and penetration in the bulk polymer, and changes in color of the polymer matrix [88,89].

2.4.3 Biofilm formation

Biofilm structure and formation have been identified as the dominant phase of life for microorganisms on Earth. Studies have shown that microorganisms, in general, live in aggregates or mixed species rather than as single cells in pure cultures [90,91]. When a biodegradation process occurs, biofilm formation is considered the first step and a necessary step in the process. However, the formation of a biofilm on a surface does not necessarily imply that biodegradation will occur [89].

In biofilm formation, a microbial community is established on a surface. These surfaces, such as metals, sediments, or polymers, can exist in different forms, have different properties, and different compositions. Biofilms are considered highly sophisticated and complex synergistic structures originated by the selective attachment of phylogenetically and functionally diverse communities of bacteria, fungi, protozoans, or algae [92]. The organization of microorganisms on a surface is specific for the material and dependent on that material's surface properties and the environmental conditions. Biofilms can be developed in solid/liquid and solid/air interfaces [89]. The first step of biofilm formation for bacteria is the microorganism's initial attachment to the surface via the cell pole or the flagellum (within minutes after the first contact with the substrate) (Figure 2.4a). The initial attachment is a reversible step. The second step of biofilm formation is the microorganisms' irreversible attachment to the surface using a glue-like substance and taillike structures. The attached microorganisms start producing slimy extracellular polymeric substances (EPS), formed by proteins, polysaccharides, nucleic acids, lipids, and humic substances, and develop clusters of cells in contact with each other and with the substrate. EPS production allows the microbial community to develop a complex structure highly influenced by the environmental factors, and is the main factor responsible for the adhesion to surfaces and for the integrity of the biofilm [93]. During this second step, the growth of microbial communities can occur in a matter of hours. Biofilm maturation occurs in the third step, when cell clusters embedded in the EPS become mature and layered. A high level of biofilm maturation is achieved as cell clusters and microcolonies reach their maximum average thickness in the fourth step. In the final step, as the maturation of the colonies progresses, the complex structures weaken, detach from the substrate, release, and

propagate. This variable sized group of cells can now attach to a different zone of the surface, or another previously optimally developed biofilm. The detachment step is characterized by cells evacuating from the interior of the clusters, forming void spaces [90,94].

In the case of fungi population, the development of filamentous fungi biofilm has been proposed (Figure 2.4b) [95]. The first step, similar for bacteria biofilm, implies the deposition and adsorption of spores and/or hyphal fragments. The second step implies the development of a fungal EPS for active attachment to the surface. In the third step a microcolony is formed with branching of a monolayer hyphal and extension of the EPS for better adherence of the microcolony to the surface of the substrate. In the fourth step a colony is formed, a hyphal compacted network is developed, and the maturation of the colony occurs. Finally, in the fifth step the dispersal or release of new cells takes place. These new cells can start a new cycle. A detailed discussion of the biofilm formation mechanism can be found elsewhere [96–98].



Figure 2.4 a) bacteria biofilm formation, steps: (1) attachment of microorganisms to the surface using a specialized glue-like substance and tail-like structures, (2) colonization, (3) growth, (4) maturation, and (5) detachment; b) Fungi biofilm formation, steps: (1) attachment of microorganisms to the surface using a specialized glue-like substance and tail-like structures, (2) colonization, (3) growth, (4) maturation, and (5) detachment. Adapted from [91,95,99].

Table 2.2	Functions	of	extracellular	polymeric	substances	(EPS)	in	biofilm	formation.
Reproduced	d with pern	niss	sion from [93].						

Function	Relevance for biofilms
Adhesion	Allows the initial steps in the colonization of abiotic and biotic surfaces by planktonic cells, and the long-term attachment of whole biofilms to surfaces
Aggregation of bacterial cells	Enables bridging between cells, the temporary immobilization of bacterial populations, the development of high cell densities and cell-cell recognition

Cohesion of biofilms	Forms a hydrated polymer network, mediating the mechanical stability of biofilms, determines biofilm architecture, and allows cell-cell communication
Retention of water	Maintains a highly hydrated microenvironment for biofilm microorganisms, leading to their tolerance of desiccation in water-deficient environments
Sorption of organic compounds	Allows the accumulation of nutrients from the environment and the sorption of xenobiotics
Sorption of inorganic ions	Promotes polysaccharides gel formation, ion exchange, mineral formation, and the accumulation of toxic metal ions
Enzymatic activity	Enables the digestion of exogenous macromolecules for nutrient acquisition and the degradation of structural EPS, allowing the release of cells from biofilms
Nutrient source	Provides a source of carbon, nitrogen, and phosphorus containing compounds for utilization by the biofilm community
Electron donor or acceptor	Allows redox activity in the biofilm matrix
Export of cell components	Releases cellular material as a result of metabolic turnover
Sink for excess energy	Stores excess carbon under unbalanced carbon-to-nitrogen ratios
Binding of enzymes	Results in the accumulation, retention, and stabilization of enzymes through their interaction with polysaccharides

Many of the identified extracellular enzymes, such as oxidoreductases and hydrolases, are responsible for the degradation of biopolymer substrates. Extracellular enzymes can break down substrates available on the EPS or from the surface to which the biofilm is attached; the resulting low molecular mass products are used as a source of carbon and energy by the microorganisms. Hence, enzymes capable of degrading substrates turn the EPS into an external digestive system for the microorganism [93]. Extracellular enzymes are key for the breakdown of water-soluble substrates (e.g., polysaccharides, proteins, and nucleic acids) or water-insoluble substrates (e.g., cellulose, lipids, and bio- or fossil-based polymers chains), leading to depolymerization [93,100].
2.4.4 Depolymerization

The enzymatic activity that occurs after biofilm formation is the main contributor to the depolymerization step. Enzymatic activity can occur via a hydrolytic or oxidative route (**Figure 2.5**), involving either random or end chain scission [53,65].

The oxidative mechanism is called enzymatic oxidation. In the case of nonhydrolysable polymers, due to the absence of hydrolysable groups, redox reactions are the most effective way to break the backbone made of C-C bonds. However, extracellular enzymes must have redox potentials high enough to allow the electron extraction from non-reactive C-H or C-C bonds. A high redox potential requirement could be an important obstacle for ultimate polymer biodegradation [101].

Enzymatic hydrolysis mirrors abiotic chemical hydrolysis. As chemical hydrolysis progresses, the M_w is reduced and consequently the polymer becomes available for enzymatic hydrolysis, which starts dominating the depolymerization step. For hydrolysable polymers, with ester, carbonate or amide groups, the hydrolytic enzymatic degradation by extracellular hydrolases has been reported and is presented and discussed in further section in detail.

Within the major enzyme classes (**Table 2.3**), hydrolases (EC 3) and oxidoreductases (EC 1) are the main groups of enzymes linked to depolymerization.



Figure 2.5 Enzymatic hydrolysis or oxidation routes for depolymerization. Adapted from [101].

Table 2.3 International Union of Biochemistry and Molecular Biology (IUBMB) classification	on
of enzymes by function of the reactions they catalyze. Adapted from [102].	

EC Number	Enzyme class	Reaction
1	Oxidoreductases	Oxidation-reduction
2	Transferases	Chemical group transfers
3	Hydrolases	Hydrolytic bond cleavages
4	Lyases	Nonhydrolytic bond cleavages
5	Isomerases	Changes in arrangements of atoms in molecules
6	Ligases	Joining together of two or more molecules

The lower activation energy needed for enzymatic hydrolysis of ester linkages, such as those in aliphatic and semi-aromatic polyesters, appears to facilitate the depolymerization of polyesters in comparison to polyolefins, where non-hydrolysable linkages are present. However, large differences have been reported in the rates of biodegradation for polyesters as a function of their morphology and chemical structure. For example, the aromatic polyester PBT is considered non-biodegradable; however, the copolymer obtained from terephthalic acid and adipic acid, PBAT, is biodegradable. Besides the presence of the aromatic ring in both structures as well as hydrolysable bonds, the presence of the adipic acid component in PBAT improves the flexibility of the polymer structure, making it more susceptible to attack by extracellular enzymes [51].

Enzymes are macromolecules made up mostly of proteins, which are complex chemical structures, with high M_w and hydrophilic groups acting as biocatalysts that accelerate the depolymerization reaction rates by lowering the activation energy of the reaction [103]. The simplest enzymes consist entirely of amino acids while conjugated enzymes contain a non-protein component, a cofactor (or co-enzyme) along with a protein component.

Extracellular enzymes are released when optimal conditions are present between the polymer surface and the attached biofilm. Enzymes bind to a substrate by their active site and transform the substrate into a product. **Figure 2.6** shows the steps of this process. First, an enzyme binds to its substrate and positions it properly in its active site to catalyze the reaction. In the second step, the enzyme-substrate complex is formed. In the third step, the enzyme-substrate complex aligns reactive groups in the substrate and places strain on specific bonds, reducing the activation energy required for making the reaction to occur. In the fourth step, the cleaved products are released. Finally, in the fifth step, the enzyme is ready to begin the catalytic cycle again.

The main factors influencing the susceptibility of a polymer towards microbial attack by extracellular enzymes are:

30

Enzyme availability. Availability is determined by the type of microorganisms and the environment.

Available sites on the polymer for enzyme attack. Extracellular enzymes are classified as exo- and endo-enzymes. Exo-enzymes are responsible for chain end scission, while endoenzymes are responsible for random chain scission [104].

Enzyme specificity. Enzymes are known as catalysts of biochemical reactions with high substrate specificity. This means that an enzyme catalyzes a special reaction with high efficiency. Therefore, many different reactions catalyzed by different enzymes can run in parallel simultaneously. The specificity is a function of the three-dimensional structure of the enzyme [104].

Presence of cofactors. Cofactors are additional chemical groups incorporated to the structure of the active site of the enzyme to facilitate a biochemical reaction. Cofactors can be metal ions (e.g., calcium, magnesium, potassium, sodium, or zinc) or co-enzymes (organic cofactors). A common function of cofactors is to provide a geometric place for the substrate to bound to the enzyme by maintaining the stability and activity of the enzyme at the active site [105].



Figure 2.6 Catalytic cycle of an enzyme. Adapted from [86].

The priority of extracellular enzymes is obtaining carbon to ensure a supply of resources. Additionally, the microbial community is able to shift enzyme production between groups of substrate-specific enzymes and non-specific enzymes, to match substrate requirements. In other words, enzymes are selectively produced to increase the supply of the most limiting element and to target the most available substrates [106]. From an energy point of view, enzyme production is energy intensive. For this reason, microorganisms produce enzymes at the expense of growth and metabolism when nutrients are scarce. Furthermore, when available nutrients are scarce, microorganisms can produce adaptive enzymes to obtain resources from complex sources [107]. On the other hand, when assimilable nutrients are available and abundant, the production of constitutive enzymes may be decreased [108].

For polymer degradation, depolymerases are the extracellular enzymes secreted by microorganisms that cleave complex polymeric substrates into oligomers, dimers, and monomers. The hydrolytic cleavage can be by exo-attack or endo-attack. Exo-attacks occur at the end of the polymer chain, and the byproducts are oligomers or monomers that can be assimilated by the cell. Endo-attacks occur randomly along the polymer chain, reducing the M_{w} ; hence, products are not assimilable without further depolymerization [109,110]. An important characteristic of extracellular enzymes is that they are too large to penetrate deeper into the polymer material. For this reason, enzymes can only act on the polymer surface, making depolymerization by enzymatic activity a surface erosion process [104]. Increasing the surface area can increase the rate of depolymerization by extracellular enzymes [111]. Fragments small enough to go through the membrane cells as monomers are transported inside the cell and transformed to obtain energy for the growth process by the action of intracellular enzymes. Usually these are oxidative enzymes, and the process is called bioassimilation or assimilation.

2.4.5 Bioassimilation

Bioassimilation is related to the acquisition or uptake of substances for the microbial metabolic process. Compounds small enough to pass the semi-permeable membrane after the depolymerization step can be potentially processed by the metabolism of the microorganism and finally mineralized to CO_2 (dissimilation) or be used for biosynthesis of new products through metabolic pathways (assimilation) (**Figure 2.7**). In general, the periplasmic space – the cell membrane – is where the cleavage takes place, and from where oligomers can be

transported across the cytoplasmic membrane for further oxidation in the β -oxidation cycle. Oligomers can be internalized with the aid of surfactants produced by microorganisms during biofilm formation and be used as carbon and energy sources by the action of intracellular enzymes. The presence of water for the transport of components is a critical factor during the bioassimilation step [86].

2.4.6 Mineralization

Mineralization, or ultimate biodegradation, refers to the degradation of polymer fragments to the mineralized components and biomass, plus CO₂ and H₂O in aerobic conditions or CO₂, H₂O, and CH₄ in anaerobic conditions (**Figure 2.7**). Depending on the polymer composition, other compounds also can be released, including sulphide, sulphate or sulphite, ammonia, nitrite or nitrate, phosphate or phosphite, chloride, and fluoride. By measuring the mineralization levels (*i.e.*, CO₂ released or evolved), biodegradation rates and % mineralization can be quantified. Bioassimilated monomers are part of the catabolism cycle. During this step, organic compounds, such as carbohydrates and proteins, are used as metabolites of the tricarboxylic (TCA) cycle or Krebs cycle by aerobic and anaerobic respiring microorganisms to produce energy [93,112]. Insights on external factors affecting mineralization of biodegradable polymers can be found elsewhere [113].



Figure 2.7 Microbial bioassimilation and mineralization during the polymer biodegradation process. Adapted from [114].

2.5 Biodegradation environments

There are many feasible waste management recovery processes for polymers, ranging from recycling, biodegradation included as defined by the U.S. Environmental Protection Agency (EPA), waste-to-energy conversion, and landfill, each with trade-off environmental impacts. Littering or leakage to the environment must not be considered waste management processes. Each of these waste management environments present specific conditions that can tailor the degradation rate of polymers. So, evaluation of degradation of a polymer in different environments may reveal different rates due to the influence of external abiotic and biotic factors [115–117].

Heat (temperature) and other factors in the environment, such as aeration, acidic/alkaline media, and water content, play crucial roles in how degradation of polymers takes place. Temperature is one fundamental parameter affecting the rate of degradation and has been studied in detail [76,118–120]. Besides affecting the abiotic mechanisms of degradation, temperature drives chemical and microbial changes through the different phases of biodegradation. Microbial activity is temperature dependent since each microorganism's optimal population and growth is driven by different temperature regimes. Marine environments and some lakes and rivers are associated with temperatures in the psychrophilic range (0–20 °C); however, depending on the geographical region, rivers and lakes can also reach the mesophilic range (20–45 °C). Soil and home composting environments are mostly in the mesophilic range (45–60 °C). **Table 2.4** presents a summary of the temperature ranges and the main environments where biodegradation occurs.

Temperature range, °C	Environment	General description	Management
20–30	Soil/Agricultural soils	Large scale. Soil structure (texture, porosity), moisture, aeration, radiation	Uncontrolled
20-45	Home composting	Small scale. C/N ratio, moisture, aeration, heat, pH	Controlled
45-60	Industrial composting	Medium scale. C/N ratio, moisture, aeration, pH	Controlled

Table 2.4 Typical temperature ranges and conditions for different environments where polymers can be subjected to aerobic biodegradation.

2.5.1 Soil environments

Soils provide diverse environments where the biodegradation of polymers can take place. Soil is a typical disposal scenario for biodegradable and non-biodegradable polymers employed as agricultural mulch films [121,122]. For several decades, non-biodegradable fossil-based polymers, such as PE, have been employed as mulch films for crops. However, in the last 15 years, bio-based and fossil-based biodegradable polymers have gained market momentum since their use can avoid the removal of the plastic film after harvest and reduces the leakage of plastic debris [123,124]. The study of plastic use in agriculture is known as "plasticulture" and includes products such as drip irrigation tape, greenhouse covers, hoophouse covers, silage bags, row covers, hay bale wraps, plastic trays and pots, and mulch films [125,126].

Soil is a diverse and typical habitat for microorganisms [127,128], and biodegradation usually takes place in the mesophilic range of temperature. Biodegradation of polymers in soil is affected by both biotic and abiotic components such as microorganisms, solid particles (i.e., organic matter and inorganic minerals), water, and air. Solid, liquid, and gaseous phases make up the soil environment, along with different organisms. The liquid and gaseous phases, water and air, vary with the climatic conditions and human activity, whereas the solid phase generally is resistant to these activities [127]. The main parameters used to classify a soil are based on its granularity and porosity due to the amount of clay, sand, and silt (**Figure 2.8**). The texture and structure of soils is determined by the relative proportions of clay, sand, and silt and their relative sizes [129,130]. Silty soils can possess high water retention capacity, but clay soils possess the highest water retention capacity. Hence, fungal populations can be supported by a dry, sandy, well-ventilated soil, whereas an insufficiently aerated clay soil provides habitat for facultative bacteria [131].



Figure 2.8 USDA soil texture classes determined according to the relative proportions of sand, clay, and silt. Adapted from [129,132].

The chemical and biological properties of soils are characterized by acidic/alkaline media, cation exchange capacity, organic carbon concentration, and soil respiration [133]. These properties control the formation and activity of the microbial diversity, and the combination of the mentioned factors create habitats where only certain microorganisms can grow [70]. The distribution of the different particles creates pores of different sizes that can retain water or surrounding living organic material. The soil connectivity determines the circulation of nutrients, soluble organic compounds, and water and is ultimately tied to the pore geometry and network [134]. Thus, the size of the pores is a factor that determines and helps to explain the spatial separation of living organisms [127].

The biodegradation process occurring in soil environments should consider the surface layer and underground matrix [135]. The surface layer of the soil is highly affected by abiotic factors. On the other hand, the underground matrix is associated with the microbial population and factors for its optimal activity [70]. The factors playing major roles in the biodegradation process in soil are soil texture and structure, water content, organic matter, pH, temperature, O₂, and sunlight [131].

Water content and retention is a function of the texture and structure of the soil, as discussed above. While a dry soil encourages the formation of fungal populations, a wet soil promotes the genesis of bacterial populations [131]. Fungi spread through the soil using hyphae, which are thin filaments forming the mycorrhizal network. Under dry conditions, while in search of water and nutrients, the hyphae spread and take different routes. The fungi continue enlarging this network and bridge the gaps between different small pockets of water and nutrients, thus enabling survival and growth in soil, where the moisture content may be low [136].

Microorganisms can adapt to specific ranges of pH values. Thus, the soil pH is a factor that can limit the growth of microorganisms. Alkaline to neutral pH favors bacterial growth whereas acid pH favors fungal development [131]. The pH influences the availability of nutrients and concentrations of trace metals such as zinc, iron, calcium, magnesium, and phosphorus. Fungi take in these molecules across their membranes by creating a proton gradient; this proton gradient affects the ability to take up the nutrients when exposed to extreme pH conditions [137]. In acidic media, certain nutrients, such as phosphorus, become less available and other nutrients like magnesium and aluminum can become more toxic, thus creating a hostile environment for helpful soil bacteria.

The O_2 content of the medium determines whether the microbial population expressed is aerobic or anaerobic. Soil temperature governs the physical, chemical, and biological processes in the soil. Changes in soil respiration rate, due to the fluctuation in temperatures also affect the bioactivity. Microbial activity is inhibited or reduced drastically with lowering temperatures [125]. Radiation, mostly from UV light, can inhibit the growth of microbial populations, depending on the intensity of the radiation. The optimal conditions of temperature, organic matter, aeration and O_2 , and water content are in the first 30 cm of the soil layer [129,135].

Agricultural soils can be considered as a particular type of soil environment, and have been extensively studied in the plasticulture field [123,126]. One of the most studied applications has been polymeric mulch films, which undergo several steps in biodegradation. This process involves a period of intense photodegradation when the mulch film starts crosslinking and eroding, followed by an intensive period of biodegradation [125,138–140]. **Figure 2.9** shows a typical life cycle of polymeric mulch films in agriculture soils.



Figure 2.9 Biodegradable mulch film cycle, starting from raising the bed and applying herbicide in spring, to harvesting and the disposal of the films in late fall, and the associated degradation processes. [141] (Copyright 2008. Reproduced with permission from Elsevier Science Ltd.).

2.5.2 Home and industrial composting

Home composting could become a common disposal scenario for polymers and polymer blends used for packaging (food and beverages) [142]. Home composting is garnering interest since it can be very instrumental in diverting the household organic fraction from going to landfill [143]. Additionally, as consumers are becoming more aware of plastic pollution, home composting has also become important as a potential methodology to reduce organic waste and contaminated packages that cannot be efficiently recovered or diverted through the MSW management system. The US Composting Council describes home composting as the natural aerobic decomposition of organic wastes or materials, usually in small-scale composters by "slow-stack" treatment methods where temperatures are in the psychrophilic (0–20 °C) to mesophilic (20–45 °C) range [144]. As per the US Composting Council, home composting can also be labeled as "backyard" or "composting at home". However, the terminology varies in different geographical regions worldwide since "composting at home" may imply composting in designed vessels inside the apartment or house [145,146], and "backyard" composting may refer to uncontrolled composting units outside the house subject to the environmental conditions.

The typical matrix for home composting includes biowastes, which are generated in the kitchen, and garden waste such as weeds and leaves (**Figure 2.10**). Many key factors, such as temperature, pH, moisture, composter efficiency, substrate, C/N ratio, and microbial populations, affect the home and industrial composting process [147,148]. Home composting is a far less controlled process in comparison to industrial composting, since the process takes place in the backyard and hence on a smaller scale. Usually, it never reaches the high temperatures in the thermophilic range for long periods of time, as seen in industrial composting. The small size of the installation, accompanied with difficulties in reaching an optimum control of factors, result in home composting requiring a longer time to achieve a mature compost [131]. The material volumes that can be handled and the abundance of microorganisms are lower for home composting settings. In addition, seasonal changes can influence "backyard composting" depending on the geographical location, and hence lower and more variable temperatures are inevitable.

Regardless of whether the composting is done at home, in a community backyard, or in an industrial facility, the composting process must ensure a succession of microbial communities (i.e., mesophiles-thermophiles-mesophiles) and the corresponding temperature regimes to operate. These factors are necessary to guarantee the safety and quality of the compost process and the final product.

One advantage of home composting is that it can be helpful in rural and suburban areas where collection of organics is limited or there is no infrastructure for industrial composting [143,149,150].



Figure 2.10 Home composting representation. [60] (Copyright 2008. Reproduced/adapted with permission from Wiley & Sons, Ltd.).

Industrial composting is a process designed to handle large volumes of yard, food, and manure waste [60,148,151]. By employing better aeration, moisture control, and higher temperatures the biodegradation in industrial composting is accelerated significantly in comparison to natural and home composting processes. The industrial composting process requires a proper system in place for collection of wastes and a good infrastructure (e.g., windrow, aerated static piles, and in-vessel composting) [152]. Biodegradation in industrial composting takes place mostly in the thermophilic temperature range. **Figure 2.11** shows a representation of an industrial composting process.



Figure 2.11 Industrial composting process. [60] (Copyright 2008. Reproduced with permission from Wiley & Sons, Ltd.).

The industrial composting process follows four main stages. The first stage is the mesophilic stage (20–45°C), where microorganisms decompose the simplest organic, degradable substances into CO_2 and water in an exothermic reaction. The high amount of substrate ensures high microbial activity, which leads to the generation of large quantities of metabolic heat energy that causes the temperature to rise swiftly. The second stage is the thermophilic stage (45–60 °C) where bacteria and fungi mesophiles become less active and are replaced by thermophiles. As the temperature rises above 55 °C, microorganisms such as pathogens are destroyed. For safety reasons, several certifying entities require that the

temperature must reach above a certain temperature, such as 55 °C, and remain at that level for a set period of time, such as 15 days, to ensure that the resulting compost is pathogen free [153]. Temperatures in some industrial compost facilities during the early stages commonly reach values of c. 70 °C [154]. Such high temperatures expedite the disintegration process of high energy carbohydrates and structurally complex molecules. As the disintegration process comes to an end, there is no longer any supply of these high energy compounds, and the third stage kicks into action where the mesophiles take over once again. The third stage is a transition stage from high to low temperature. The final stage, also called "curing" or maturation, can take several months to result in stabilized compost [131]. The total composting time varies in systems used worldwide, from two to more than six months; thus, certified compostable packages can encounter difficulties to fully disintegrate in some operations [155].

2.5.3 Aquatic environment

Natural aquatic environments (i.e., oceans, rivers, and lakes) unfortunately are environments where discarded polymers from activities such as fishing and shoreline recreation are commonly found [11,16,156]; however, these are not formal waste management scenarios and must not be considered as such. The natural aquatic environment is a non-desired end-of-life scenario due to the creation of white pollution and a lack of proper conditions for biodegradation and control of the process due to its complexity [157]. Biodegradation in the aquatic environment can happen in lakes, rivers, and oceans as well as in reservoirs, and wastewater facility treatments (aerobic or anaerobic); however, our discussion is focused on the natural aquatic environments.

Geographical considerations of the aquatic environment play an important role in understanding the presence and flow of plastics. Lakes are generally low-flow environments and act as a point for accumulation for plastics and microplastics [158]. Rivers are considered the essential route for transporting plastics to the ocean. Considering their proximity to urban and industrial areas, rivers become an easy access point to the marine environment with respect to plastic pollution. Plastics are extensively carried out during floods in cities with poor waste management systems [12]. For the marine environment (seawater), three main habitats can be considered when addressing the degradation of plastics (Figure 2.12): the pelagic zone, an illuminated and aerated column of water; the littoral zone, which is the beach sediment periodically covered by water due to waves or tide; and the sublittoral zone, which is the seabed interface up to 200 meters in depth that is aerated and photosynthetically active. The physical and chemical properties of seawater, including the essential nutrients for living organisms, vary with the depth, latitude, and proximity to land. Because of this variation, the microbial populations within seawater also vary. Furthermore, the degradation process of the plastics entering this environment can be altered by agitation and turbulence caused by ocean currents, salinity, temperature gradients, and solar radiation among others [159,160]. Biodegradation of polymers in aquatic environments is described in terms of scarce evolution for synthetic biodegradable polymers. However, high efficiency has been reported for natural polymers as cellulose, starch, and PHAs regardless of the low temperatures reached (**Table 2.9**). The "plastisphere," the development of biofilms on the surface of polymers present in water, has been extensively studied to elucidate the main components and behavior of microorganisms during the colonization and depolymerization of polymers in aquatic environments [161–163].



Figure 2.12 Main habitats to consider when addressing the degradation of polymers in the marine environment. The pelagic zone, an illuminated and aerated column of water; the littoral zone, which is the beach sediment periodically covered by water due to waves or tide; and the sublittoral zone, which is the seabed interface up to 200 meters in depth that is aerated and photosynthetically active. Adapted from [164].

2.6 Factors and properties that affect degradation rate

The rate of polymer degradation is affected by the degradation mechanisms, the environments, and the polymer properties. This framework creates a complex interplay governing what is reflected in the rate and efficiency of the whole degradation process.

In the next subsections, we selectively provide a discussion of factors important for mesophilic biodegradation and correlate these factors to the information already provided about mechanisms and environments. Detailed discussions of these factors are also provided in selected reviews [48,50,53].

2.7 Environmental factors

Factors that can affect the degradation rate of a polymer are related to the environment where the degradation process takes place, and include thermal energy (heat), acidic/alkaline media, moisture, aeration, and microbial populations. Some of these factors are more relevant or critical than others and are important during the abiotic and biotic degradation stages affecting both the polymer's properties and the microbial activity.

2.7.1 Heat

The amount of thermal energy, identified as the temperature of the system, is one of the main factors affecting the rate of abiotic and biotic degradation mechanisms and varies with the environment (**Table 2.4**). In this section, we are not expanding on the thermal degradation mechanism; instead, we briefly discuss temperature as a factor that can modify the rate of other mechanisms such as chemical hydrolysis and microbial activity. At an early stage in the degradation process, mechanisms such as chemical hydrolysis can be dominant and the temperature plays a crucial role on the rate [64,165]. For example, for PLA, the chemical hydrolytic degradation is dependent on the temperature since a large initial reduction of the M_w is needed before microorganisms can assimilate the byproducts [64]. Higher temperatures activate chain mobility, increasing free volume and polymer rearrangements. If the temperature is higher than the T_g of the polymer, mobility and reaction are accelerated, increasing the rate of polymer degradation (**Table 2.5**). Furthermore, the presence and potential growth of different microorganisms depends on the environment temperature, and a change in temperature regulates both presence and activity [86,131]. Biodegradation rate is a function of temperature, mostly described by the Arrhenius equation above and below T_g [120].

2.7.2 Moisture

The presence of water plays a crucial role in the degradation of hydrolysable chemical bonds, such as in polyesters, since they are susceptible to chain scission reactions [166,167]. Furthermore, microorganisms need water for transport of nutrients through the cell membrane and for growth. The amount of water in the different environments, such as soil and home or industrial composting, can create different surroundings for the microorganisms. Low levels of moisture can lead to dry environments with low biological activity [168]. High values of moisture will lead to loss of the porosity of the matrix (soil or compost), turning the process into one with anaerobic conditions [169]. Pore spaces are essential for the normal air flow and aerobic regimen; the optimal humidity range for microbial activity is a function of the percentage of pore space needed that does not obstruct the air flow required for the microbial activity [170]. For example, for the composting process, an optimal range of moisture content is 45 to 65% [148].

2.7.3 Acidic and alkaline media

Acidic or alkaline media can modify the rate of reactions and the mechanism of hydrolytic degradation [64]. For example, for PLA in acidic conditions the hydrolysis proceeds via a chain-end scission, while in alkaline solution the hydrolysis takes place via backbiting [64]. In the case of PCL films evaluated at extreme pH values (1 and 13) at 37 °C, different behavior was observed for reduction of M_w and crystallinity, suggesting a surface erosion process in alkaline media and bulk erosion in acidic media [171]. During the biodegradation process, pH values close to neutral are highly favorable for the growth of microbial populations. In soil environments, a pH range close to alkaline-neutral values is favorable for bacteria populations, whereas fungi are more tolerant to acidic and alkaline media; fluctuations of pH are considered a harmful situation for living organisms [131,172].

2.7.4 Light and UV radiation

If sufficient energy is absorbed by light and UV radiation, polymers can be subjected to photodegradation, experiencing changes in their chemical structure and physical properties. Light and UV radiation is important in agricultural soils and aquatic environments. So, photodegradation can be the precursor of the degradation process before microorganisms can use byproducts [82,140].

2.7.5 C/N ratio

Microorganisms need carbon as a source of energy and nitrogen for synthesis of amino acids, proteins and nucleic acids [148]. The C/N ratio is a key parameter in environments such as compost and soil. Optimal values for the C/N ratio in compost and soil are in the range of 15:1 to 30:1. During the active aerobic phase of breakdown, microorganisms use around 30 parts of carbon for each part of nitrogen, due to the high energy requirement. If carbon levels are higher, microorganisms need to undergo several life cycles to oxidize the excess carbon, slowing down the biodegradation process. If carbon levels are low, microorganisms do not have sufficient energy source to survive [148].

2.7.6 Oxygen flow and porosity

Aeration and porosity are key factors for the normal activity of the microbial population in soil and compost environments. To maintain aerobic conditions the porosity should allow O₂ concentrations of around 5%. Porosity is highly correlated with the air flow within a matrix. Low porosity hinders air flow, whereas high porosity can lead to excessive aeration and low water retention capacity. The shape, size, and structure of particles of the matrix (soil or compost) affects its texture. Therefore, a tight packing arrangement reduces the porosity and the compressed matrix impacts the air flow [148].

2.8 **Polymer properties**

The factors affecting degradation associated with the bulk polymer matrix can be categorized as chemical structure and physical properties such as morphology, crystallinity, constitutional unit, flexibility, crosslinking, M_w , tacticity, density, shape, and polarity. The surface properties affecting degradation are related mostly to hydrophobic/hydrophilic ratio, roughness, surface energy, and available surface area.

2.8.1 Bulk properties

Chain flexibility. A polymer chain that is highly flexible is more accessible to attack by microorganisms. Longer aliphatic chains can exhibit high biodegradation rates. However, aromatic rings can act as obstacles, providing steric hindrance to the enzyme attacking the ester bonds, thereby lowering the rate of biodegradation [70]. During the depolymerization step, enzyme binding is favored by high flexibility of the polymer chains. In this sense, it is aptly recognized that microorganisms are more likely to start the biodegradation process in the amorphous region of the polymer [48,49]. Polymers with T_g values in or below the mesophilic range, such as PCL, PBS, PBAT, PHAs, and PGA, will be more flexible in favoring chemical and enzymatic hydrolysis in the mesophilic range (see **Table 2.6**). Flexibility and mobility are enhanced by copolymerization, blending, or by increasing the temperature, and are reduced by crystalline domains [80,173].



Table 2.6 Polymer structure and thermal properties (T_g, T_m) of the biodegradable polymers discussed in this work.





Chemical structure (functional units and functional groups). Chemical structure is an inherent property of a material and determines whether the polymer is prone to undergo biodegradation. The chemical structure depicts the spatial arrangement of chemical bonds and atoms in the molecule influencing the molecular geometry and governs how the molecules are packed together allowing the formation of crystalline or amorphous regions. The presence of bulky groups in the main chain, such as aromatic rings, restricts the free movement of the polymer molecule, reducing chain flexibility such as in PBT. However, when the linear copolymer of adipic acid and 1,4-butanediol is added to the main chain of PBT to obtain PBAT, polymer flexibility improves and susceptible hydrolysable bonds are introduced, so the polymer is more flexible and prone to biodegradation [81]. Modifications such as inclusion of functional groups by copolymerization in the main chain of initially nonbiodegradable chemical structures can make a polymer more prone to biodegradation [56]. The addition of functional groups also can impart a hydrophilic nature to a hydrophobic polymer thus improving its likelihood of undergoing biodegradation [109].

Chain structure configuration (side chains and crosslinking). The length of side chains influences the degradation process. For example, Li et al. [174] concluded that the enzymatic degradation of PHA was dependent on the length of side chain in the PHA structure. Crosslinking can occur and play a significant role in polymer mass transfer properties and chain flexibility hindering biodegradation. Kijchavengkul et al. demonstrated that increasing the amount of crosslinking reduces the biodegradation of PBAT [175].

Crystallinity. Crystallinity can increase the stiffness and density of a polymer [48]. A high crystalline fraction decreases the abiotic and biotic degradation rates. The amorphous region is more susceptible to chemical hydrolysis due to the ease of water diffusion. A characteristic of the crystalline region is its low mass transfer to gases and vapors, decreasing the rate of the hydrolytic degradation [176,177]. Extracellular enzymes mainly attack the amorphous region of the polymer structure [104,178]. Biodegradable polymers are, in general, semicrystalline polymers with a crystalline and amorphous region.

Molecular weight (M_w) . To obtain polymers with usable thermal, mechanical, and barrier properties a high M_w is required. However, microorganisms assimilate polymers when selected thresholds of low M_w fractions of the polymer are reached. The higher the M_w value of the polymer residue, the harder it is for microorganisms to assimilate the chain segments and assimilate that to their cell, which reduces the rate of the biodegradation. So, a critical threshold low M_w value must be reached to kick off the degradation by enzymatic attack [70]. Generally, this M_w is attainable by a precursor degradation mechanism such as photodegradation or chemical hydrolysis, as with polyesters. In the case of PLA, the polymer first undergoes primarily chemical hydrolysis, accelerated under industrial composting conditions, until reaching a $M_w \leq 10$ kDa, and then enzymatic activity becomes the dominant degradation mechanism, with a high mineralization rate [113].

Density and porosity. Denser and more compact polymers have lower chances to experience water diffusion. For polyesters, chemical hydrolysis is generally the initial trigger mechanism of degradation, mostly through a bulk erosion process, so water diffusivity of the polymer plays a crucial role. One way to modify the diffusion or the hydrophilicity of a polymer matrix is by blending different polymers. So, biodegradable blends and copolymers can be used to tailor some of these bulk properties. Blends of PLA and TPS have shown higher biodegradation rates [179].

2.8.2 Surface properties

Hydrophobic/hydrophilic ratio, surface roughness, surface energy, and surface/volume ratio are the more relevant factors during the degradation process. Chemical hydrolysis is highly affected by the hydrophobic/hydrophilic ratio of the polymer surface. Furthermore, enzyme activity, biofilm formation, and colonization are also linked to surface properties.

Hydrophobic/hydrophilic ratio. In the case of isotropic polymers, surface and bulk water sensitivity plays a major role in the degradation process. Hydrophobic surfaces will not allow water to be adsorbed and will delay water uptake, so that any degradation mechanism triggered by water diffusion will be delayed. Table 2.7 shows that polymers with hydrophobic surface and high-water diffusion, such as the polyester PLA, mostly degrade under a bulk degradation process [63]. So, by tailoring the surface and bulk hydrophobicity and the water diffusion of the polymer matrix, the overall chemical hydrolysis can be controlled, as shown for PLA [166]. In terms of enzymatic activity, a hydrophobic/hydrophilic balance allows the presence of necessary water for optimal microbial activity [162]. Some studies have demonstrated that biofilms develop faster on hydrophobic nonpolar surfaces [180]. However, Tsuji et al. reported an alkaline treatment to increase the hydrophilicity of PLLA and PCL to improve enzymatic attack. The effect was important for PLLA films, where enzymatic attack by Proteinase K was higher on hydrophilic surfaces [181,182]; however, the attack by lipases on PCL films remained unchanged [182]. The fact that lipases need a hydrophobic surface to be active could be an important conditioning of the scarce activity on PCL films. Furthermore, the exposure to hydrophobic surfaces has been reported to be a relevant signal for the production of extracellular enzyme cutinases by fungi to act on the surface of polyesters such as PCL, PBS, and PBSA, among others [183]. Tribedi et al. reported the effect

of cell hydrophobicity when comparing enzymatic esterase activity of two strains of *Pseudomonas* on the surface of PES. The strain with higher hydrophobicity also showed higher microbial activity, which is indicative that the interaction and hydrophobic balance between the microorganism and polymer surface is also relevant for microbial and enzymatic activity [184].

Water diffusion	Surface	Degradation process	Example
Low	Hydrophilic	Surface	PHAs
High	Hydrophilic	Bulk/surface	Starch, TPS, Cellulose
High	Hydrophobic	Bulk	PLA, PCL, PBS, PCL
Low	Hydrophobic	Surface (depending on the ratio of hydrophobic depletion and water diffusion)	PLA with chain extender

 Water diffusion
 Surface
 Degradation process
 Example

Surface roughness. Surface roughness is a measure of the finely spaced microirregularities on the surface texture and depicts the irregularities on the polymer surface. Some researchers have used surface roughness as an indicator of surface biodegradation [125,185]. The types of microbes able to colonize a surface and the formation of biofilms depend on the surface roughness. Increased roughness favors bacterial adhesion because of the greater area of contact between the polymeric material and the bacterial cells [186]. A rough surface offers micro- and nano-irregularities in the range of 0.5 to 2 μ m, which appear as voids and can provide sites for microorganisms to attach and eventually access the polymer chains, increasing the rate of biodegradation [187,188].

Surface area. The shape (e.g., film, pellet, powder, and fiber) and size (macro, micro, and nano) of the polymer play important roles during the degradation process [53]. For example, thicker polyester samples take more time to biodegrade [75]. The surface area

available has a high effect on the rate of biodegradation: as the surface to volume ratio increases with time, so does the speed at which biodegradation occurs. Pits and cracks continue to increase as time proceeds, and gradually the sample shape and size change, enabling access to the inside of the matrix [189]. Extracellular enzymes are highly active on the surface of a polymer since they are relatively large to penetrate the bulk. Hence, increasing the surface area available for enzymatic attack translates into an increase in the kinetics of the biodegradation process. Herzog et al. [111] showed that the enzymatic degradation of a polyester by *Candida cylindracea* at 40 °C was more effective on nanoparticles (100 nm diameter) than on thick films (110 µm thickness) of the same polyester. The effect of morphology on water biodegradation of PHBV was evaluated by Komiyama et al. [190]. Samples evaluated in powder form showed the faster biodegradation due to the larger surface area available for biofilm formation in comparison to film, undrawn fiber, and fivefold drawn fiber.

2.9 Biodegradation assessment

The misuse of the terms "biodegradable" or "biodegradation" has given rise to inflated and unsubstantiated claims. Claims about general biodegradable products that are used to deceive consumers into believing that products are environmentally friendly have been coined as "greenwashing." It is essential to avoid such false claims, guarantee transparency to consumers, and stop the unqualified use of vague terms. Certification for biodegradation, per se, does not exist worldwide. Some polymer and paper materials are certified for biodegrading in specific environments such as home and industrial composting, soil, and water [142]. Standards and methods have been developed to aid certification, to avoid confusion, and to define the environment and conditions in which the samples can be biodegraded [191].

2.10 Standards for evaluation of biodegradation at mesophilic conditions

Several organizations are associated with developing the standards for biodegradability of materials in different environments for different countries and world regions [191]. Various reviews and reports have provided the standards available for biodegradation in soils [192], aquatic environments [193], or home and industrial composting [194,195]. In this review we specifically summarize, in **Table 2.8**, the different standards used to assess biodegradability under aerobic conditions for mesophilic temperatures and tracking evolution of CO_2 and O_2 demand and cite published works that reported the use of these standards. Furthermore, standards with specifications of materials to be evaluated and for certification are described in ASTM 6400, ISO 17088, and EN13432. The environmental conditions in which the biodegradation takes place are an important aspect since the biodegradability of a material differs from one environment to another. Development of standards for assessing biodegradation in different environments is essential [196,197].

Standard	Name	Parameter evaluated	Biodegradation requirement	Environment	Temperature range	Time frame	Selected published works
ISO 14852:2018	Determination of the ultimate aerobic biodegradability of plastic materials in an aqueous medium — Method by analysis of evolved carbon dioxide	Measure CO ₂ evolved	> 60% for reference material (end of test)	Natural aqueous medium (inoculum from activated sludge, compost, or soil)	20–25 °C (± 1 °C)	6 months	[198–200]
ISO 14851:2019	Determination of the ultimate aerobic biodegradability of plastic materials in an aqueous medium — Method by measuring the oxygen demand in a closed respirometer	Measure O ₂ demand	> 60% for reference material (end of test)	Natural aqueous medium (inoculum from activated sludge, compost, or soil)	20–25 °C (± 1 °C)	6 months	[115,201– 207]

Table 2.8 Standards for assessing aerobic biodegradation of polymers at mesophilic conditions in different environments, and selected studies that used the standards to conduct their biodegradation tests.

ISO 17556:2019	Plastics — Determination of the ultimate aerobic biodegradability of plastic materials in soil by measuring the oxygen demand in a respirometer or the amount of carbon dioxide evolved	$\begin{array}{c} Measure \\ O_2 \\ demand, \\ CO_2 \\ evolved \end{array}$	> 60% for reference material (plateau phase or end of test)	Soil	20–28 °C (preferably 25 °C, ± 2 °C)	6 months	[115,208– 210]
ISO 19679:2019	Plastics — Determination of aerobic biodegradation of non-floating plastic materials in a seawater/sediment interface — Method by analysis of evolved carbon dioxide	Measure CO ₂ evolved	> 60% for reference material after 180 days	Seawater / sandy sediment interface	15–25 °C (don't exceed 28 °C, ± 2 °C)	≤24 months.	[211]

Table 2.8 (cont'd)

ISO 18830:2016	Plastics — Determination of aerobic biodegradation of non-floating plastic materials in a seawater/sandy sediment interface — Method by measuring the oxygen demand in closed respirometer	Measure O2 demand	> 60% for reference material (after 180 days)	Seawater / sandy sediment interface	15–25 °C (don't exceed 28 °C, ± 2 °C)	≤ 24 months.
ISO 22403:2020	Plastics — Assessment of the intrinsic biodegradability of materials exposed to marine inocula under mesophilic aerobic laboratory conditions — Test methods and requirements	Measure CO ₂ evolved	≥ 90% for reference material (within 2 years)	Marine	15–25 °C (don't exceed 28 °C, ± 2 °C)	24 months.

ISO 22404:2019	Plastics — Determination of the aerobic biodegradation of non-floating materials exposed to marine sediment — Method by analysis of evolved carbon dioxide	Measure CO ₂ evolved	> 60% for reference material (after 180 days)	Marine sediment	15–25 °C (don't exceed 28 °C, ± 2 °C)	≤ 24 months.
ISO 23977- 1:2020	Plastics — Determination of the aerobic biodegradation of plastic materials exposed to seawater — Part 1: Method by analysis of evolved carbon dioxide	Measure CO2 evolved		Sea water	15–25 °C	≤ 24 months

ISO 23977- 2:2020	Plastics — Determination of the aerobic biodegradation of plastic materials exposed to seawater — Part 2: Method by measuring the oxygen demand in closed respirometer	Measure O ₂ demand		Sea water	15–25 °C	≤24 months	
EN 17033:2018	Plastics – Biodegradable mulch films for use in agriculture and horticulture – Requirements and test methods	Measure CO ₂ evolved	> 90% conversion	Agriculture soil	20–28 °C (25 °C preferred, ± 2 °C)	24 months	[120]
ASTM D5988-18	Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials in Soil	Measure CO ₂ evolved	> 70% for reference material after 180 days (starch or cellulose)	Soil and mature compost	25 ± 2 °C	6 months	[120,189,218– 227,208,228,229,209,212– 217]

ASTM D6691- 17	Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials in the Marine Environment by a Defined Microbial Consortium or Natural Sea Water Inoculum	Measure CO ₂ evolved	> 70% for reference material	Marine (seashore and open ocean). Synthetic seawater with pre-grown population of at least 10 aerobic marine micro-organisms. Natural seawater with inorganic nutrients	30 ± 2 °C	10–90 days	[115,230– 232]
ASTM D7991- 15	Standard Test Method for Determining Aerobic Biodegradation of Plastics Buried in Sandy Marine Sediment under Controlled Laboratory Conditions	Measure CO ₂ evolved	> 60% for reference material (after 180 days)	Marine (tidal zone, sandy sediment + seawater)	15–25 °C (do not exceed 28 °C, ± 2 °C)	24 months	[230,233]
ASTM D5929-18	Standard Test Method for Determining Biodegradability of Materials Exposed to Source-Separated Organic Municipal Solid Waste Mesophilic Composting Conditions by Respirometry	Measure O ₂ uptake, Measure CO ₂ evolved	Total O ₂ uptake > 80g Volatile fatty acids > 2g/kg (invalid test)	Municipal solid waste inoculated with compost	40 ± 2 °C	45 days	
---------------------	---	---	--	--	---------------------------	--	-------
AS 5810- 2010	Biodegradable plastics— Biodegradable plastics suitable for home composting	Measure CO ₂ evolved	≥ 90% (dry weight) degradation of test sample.	Organic waste, kitchen waste	25 ± 5 °C (< 30 °C)	12 months	[234]
NF U52- 001:2005	Biodegradable materials for use in agriculture and horticulture – Mulching products – Requirements and test methods	Measure CO ₂ evolved	60% for reference (cellulose) in soil, 90% for cellulose in compost or water media	Soil, compost, and water	28 ± 5 °C	12 months in soil, 6 months in compost, 6 months in water	

2.11 Methods for biodegradation assessment

Different methodologies, both quantitative and qualitative, are used to determine the biodegradation process. When used in combination, the different methodologies help to recognize if there is any disagreement among the achieved results. Also, supporting quantitative methodologies, such as CO_2 evolution and M_w reduction, with qualitative methodologies, such as scanning electron microscopy (SEM), visual observation, and spectroscopy, is helpful in corroborating the biodegradation of the material under study. The main methodologies to assess and report the degree of biodegradation in aerobic conditions have been summarized in several reviews [49,195,235-237]. The oldest and most common methodology is the gravimetric reduction in weight or mass loss of the material under biodegradation. Significant deterioration in mechanical properties has also been reported as a degree of biodegradation. Macro visualization, mass loss, and deterioration of mechanical properties are methods for the approximate assessment of biodegradation. These methods are more related to physical degradation of the material and not to the biological process conducted by a population of microorganisms. In general, are more useful for gaining insights during the early step of polymer biodegradation as during abiotic degradation or during biofilm formation on the surface of the polymer.

For enzymatic activity clear zone formation, turbidimetric assays, and techniques that monitor the release of soluble products into the supernatant solution as Total Organic Carbon (TOC), and spectroscopy combined with chromatography have been reported. Nowadays the use of microbalance with dissipation monitoring measurements constitutes an additional analytical technique to evaluate the evolution of the enzymatic hydrolysis of hydrolysable polymers.

For tracking CO₂ evolution and mineralization, respirometric methods has been developed and are supported by standards for assessing the conversion of the C present in the polymer to CO_2 . Furthermore, standards also describe for specific environments the measure of biochemical oxygen demand (BOD) instead of CO_2 . Radio labeling and tracking of C has been reported as an adequately technique to complement with respirometric methods.

Associated with each of the main evaluation methodologies are several techniques used to quantify the degree of biodegradation. **Table 2.9** lists published studies conducted to measure biodegradation using techniques to measure CO_2 and/or O_2 under aerobic conditions at mesophilic temperatures. This section provides a brief description of the methodologies used and the published studies using those methodologies in the mesophilic range.

Parameter	Polymer (shape, initial M_w , initial X_c)	Environment	Temperature, °C	Main result (test duration)	Published studies
CO_2	Cellulose (powder)	Soil	15, 20, 28	-	[219]
CO_2	Cellulose (paper mulch)	Soil in laboratory conditions	27	-	[237]
CO_2	PBS (dumbbell, 21.2 kDa, 57.6%)	Soil compost in laboratory conditions	25 ± 2	65% CO ₂ evolution (180 days)	[226]
CO_2	PCL (powder, 100 kDa)	Compost in laboratory composting conditions	40	20% mineralization (180 days)	[246]
CO_2	PLA (films, 100–200 kDa), starch (powder)	Soil in laboratory conditions	28, 40	PLA (100kDa): 10–40% mineralization (28 °C, 180 days), PLA (200 kDa): 30– 95% mineralization (40 °C, 180 days)	[247]
CO_2	PLA (sheets, 170 and 180 kDa)	Soil inoculated in laboratory conditions	30	5–40% mineralization (60 days)	[222]
CO ₂	PLLA (film, 100 kDa, 30– 35%)	Aquatic laboratory conditions	25, 37	PLA (25 °C): 10% mineralization (180 days), PLA (37 °C): 12% mineralization (180 days)	[248]
CO_2	PLA (films, 163 kDa)	Soil in laboratory conditions	30	10–25% mineralization (150 days)	[236]

Table 2.9 Biotic degradation of polymers at mesophilic conditions measuring CO_2 evolution or O_2 demand. Polymer details as shape, initial molecular weight (M_w), and initial crystallinity (X_c); environment in which the biodegradation study is conducted, testing temperature, the extent of biodegradation with the time frame, and the corresponding selected studies are mentioned.

CO ₂	PHB (powder and film), PCL (powder), starch (powder)	Soil in laboratory conditions	22 ± 3	PHB powder: 91% mineralization (90 days), PCL powder: 102% mineralization (270 days), PHB films: 26% mineralization (210 days)	[225]
CO_2	PHBV (powder, -, 68.9%), cellulose (powder)	Marine in laboratory conditions	25	PHBV: 90% mineralization (450 days)	[241]
CO_2	PHBV (film), cellulose (powder)	Soil in laboratory conditions	28	PHBV: 90% mineralization (120 days)	[224]
CO ₂	PHB (film), PBSe (film), PBSeT (film)	Marine in laboratory conditions	25	PHB: 70% mineralization (360 days) and 95% mineralization (200 days), PBSe: 95% mineralization (365 and 200 days), PBSeT: 85% mineralization (360 days) and 90% mineralization (200 days)	[218]
CO ₂	PLLA (powder and film, 5, 11, 34, 256 kDa, 0, 18, 42%)	Compost	30, 37	PLA (5 kDa): 70% mineralization (40 days), PLA (11 kDa): 55% mineralization (40 days), PLA (34 kDa): 35% mineralization (40 days), PLA (256 kDa): 20% mineralization (40 days)	[249]

CO ₂	PHA, PBS, cellulose (powder)	Soil in laboratory conditions	25, 37	PHA (25 °C): 95% mineralization (150 days), PHA (37 °C): 90% mineralization (180 days), PBS (25 °C): 90% mineralization (200 days), PBS (37 °C): 75% mineralization (180 days)	[217]
$\overline{\mathrm{CO}_2}$	PU (films)	Soil/Sturm test	30	10 g CO_2 evolution (30 days)	[250]
$\rm CO_2$	PBAT (films, -, 9%)	Soil	25	5% mineralization (100 days)	[251]
CO_2	PBSe (powder), cellulose (powder)	Soil	28	55–90% mineralization (140 days)	[195]
CO_2	PHB (film), PBSe (film), PBSeT (film), cellulose (powder)	Soil	25	PHB: 95% mineralization (360 days), PBSe: 90% mineralization (360 days), PBSeT: 90% mineralization (360 days)	[216]
$\rm CO_2$	Cellulose (powder)	Soil	25 ± 2	-	[234]
CO ₂	UV irradiated PLA (powder, 198 kDa)	Inoculated sterilized compost, Sturm test	37	PLA (compost): 35–45% mineralization (40 days), PLA (Sturm test): 10–20% mineralization (40 days)	[73]
CO_2	PHB (powder, 470 kDa))	Sturm test	27	10–80% mineralization (28 days)	[252]
CO ₂	PLA (film, -, 20.8), PHBV (film, -, 72.6), cellulose	Soil	23-25	PLA: 5% mineralization (190 days), PHBV: 25% mineralization (190 days)	[227]
CO_2	PHA (films), PHB (films)	Soil	23 ± 4	PHA: 0.2 mM/mg CO ₂ (90 days), PHB: 0.3 mM/mg CO ₂ (90 days)	[221]

CO ₂	PHB (film, 175–225 kDa, 48–52%) PHBV (films, 400–300 kDa, 48–52%) with 1% nucleating agent	Microorganisms from marine environment in simulated laboratory conditions	30	PHB: 80–95% mineralization (115 days), PHBV: 90–100% mineralization (115 days)	[239]
$\rm CO_2$	PHBV (films, 455 kDa, 47%), cellulose (powder)	Marine (foreshore sand, sand & seawater, seawater) in laboratory conditions	25	PHBV (foreshore sand): 90% mineralization (250 days)	[253]
CO ₂	PHA (film), PLA (bag, bottle)	Marine	30	PHA: 38–45% mineralization (180 days), PLA (bag): 4.5% mineralization (180 days), PLA (bottle): 3.1% mineralization (180 days)	[254]
CO_2	PHBV (film, 500–600 kDa, 14–58%), cellulose, starch	Soil	25	PHBV: 90% mineralization (250 weeks)	[223]
CO_2	PHA (film), cellulose (paper)	Soil	20 ± 2	PHA: 70% mineralization (660 days)	[235]
CO_2	PLA with chain extender (films sheets, 449 kDa, 0.9%), PBAT (films sheet, 44 kDa, 15.2%), cellulose (powder)	Soil in laboratory conditions	28	PLA: 10% mineralization (180 days), PBAT: 20% mineralization (180 days)	[220]
$\rm CO_2$	PLA (sheets), PHB (sheets), PBS (sheets), TPS (sheets), PCL (sheets), cellulose (powder)	Soil, home composting*, marine pelagic, and fresh water	$25 \pm 2, 28 \pm 2,$ $30 \pm 1, \text{ and } 21 \pm 1$	PLA (soil): negligible (141 days), PLA (home composting): <20% mineralization (365 days), PLA (marine water): <10% relative biodegradation	[128]

PU (films, 48.7 kDa)	Sturm test	35, 30	$7.6-8.6 \text{ g/l CO}_2$	[255 - 257]
PBAT (films)	Soil	30	15% mineralization (120	[258]
			days)	
PU (films)	Sturm test	35	4.46 g/l CO_2	[259]
PBSA (films)	Sturm test	37	78% mineralization (40 days)	[260]
PLA (sheets)	Sterilized soil,	30	PLA inoculated: 20%	[228]
	non-sterilized soil,		mineralization (60 days)	
	non-sterilized			
	inoculated soil in			
	laboratory			
	conditions			
Cellulose (foil)	Respirometer	20	-	[215]
PBS (sheets, 90 kDa,	Sturm test	25	PBS: 18% mineralization (40	[261]
58.9%), PEA (sheets, 88	(activated sludge)		days), PEA: 12%	
kDa, 40.6%)	~		mineralization (50 days)	
PBSA (films), cellulose	Compost	25	70% mineralization (55 days)	[242]
(powder)	~			
PHA (films), PVOH (films)	Sea water	30	PHA: 100% mineralization	[240]
			(100 days), PVOH: 85%	
			mineralization (100 days)	500 7 0001
PCL, PHBV, PBSA,	Aqueous solution	30	PCL: 26% mineralization,	[205, 206]
PVOH, PEA, starch,			PHBV: 53% mineralization,	
cellulose			PBSA: 3% mineralization,	
			PVOH: 5% mineralization,	
			PEA: 36% mineralization (2	
$DI = 0.001 D (f^{*}) = 0.000 D$	A	20	weeks)	[007]
PLA 3001D (films, -, 7.7%),	Aqueous mineral	30	5% mineralization (115 days)	[207]
centulose (powder)	solution (including			
PRAT (films 56 38 kDs)	Soil insubation	95	7 15% minoralization (G	[140]
i DAI (iiiiis, 50–56 KDa)	Son meusation	20	veeks)	[140]
	PU (films, 48.7 kDa) PBAT (films) PU (films) PBSA (films) PLA (sheets) Cellulose (foil) PBS (sheets, 90 kDa, 58.9%), PEA (sheets, 88 kDa, 40.6%) PBSA (films), cellulose (powder) PHA (films), PVOH (films) PCL, PHBV, PBSA, PVOH, PEA, starch, cellulose PLA 3001D (films, -, 7.7%), cellulose (powder) PBAT (films, 56–38 kDa)	PU (films, 48.7 kDa)Sturm testPBAT (films)SoilPU (films)Sturm testPBSA (films)Sturm testPLA (sheets)Sterilized soil, non-sterilized soil, non-sterilized soil in laboratory conditionsCellulose (foil)RespirometerPBS (sheets, 90 kDa, 58.9%), PEA (sheets, 88 kDa, 40.6%)Sturm test (activated sludge) kDa, 40.6%)PBSA (films), cellulose (powder)Compost (activated sludge)PHA (films), PVOH (films)Sea waterPCL, PHBV, PBSA, PVOH, PEA, starch, celluloseAqueous solutionPLA 3001D (films, -, 7.7%), cellulose (powder)Aqueous mineral solution (including wastewater)PBAT (films, 56–38 kDa)Soil incubation	PU (films, 48.7 kDa)Sturm test35, 30PBAT (films)Soil30PU (films)Sturm test35PBSA (films)Sturm test37PLA (sheets)Sterilized soil, non-sterilized soil, inoculated soil in laboratory conditions30Cellulose (foil)Respirometer20PBS (sheets, 90 kDa, kDa, 40.6%)Sturm test259BSA (films), cellulose (powder)Compost259PEA (films), cellulose 	PU (films, 48.7 kDa)Sturm test35, 307.6–8.6 g/l CO2PBAT (films)Soil30115% mineralization (120 days)PU (films)Sturm test354.46 g/l CO2PBSA (films)Sturm test3778% mineralization (40 days)PLA (sheets)Sterilized soil, non-sterilized soil, inoculated soil in laboratory conditions30PLA inoculated: 20% mineralization (60 days)Cellulose (foil)Respirometer20-PBS (sheets, 90 kDa, Stern testSturm test25PBS: 18% mineralization (40 days), PEA: 12% mineralization (50 days)PBS (films), cellulose (powder)Compost2570% mineralization (50 days)PHA (films), PVOH (films)Sea water30PHA: 100% mineralization (100 days), PVOH: 85% mineralization (100 days)PCL, PHBV, PBSA, celluloseAqueous solution30PCL: 26% mineralization, PBSA: 3% mineralization, PBSA: 3% mineralization, PEA: 36% mineralizatio

CO_2	PU (foam)	Soil	21 ± 2	43% mineralization (192 days)	[229]
CO_2	PU (foam), cellulose (paper)	Soil	27 ± 1	10% mineralization (320 days)	[230]
CO_2	PU (foam)	Sewage water/modified Sturm test	22 ± 2	32–45.6% mineralization (60 days)	[262]
CO ₂	Non-isocyanate polyurethane (NIPU) polyhydroxyurethane (PHU) (film)	Soil	20–28	40% mineralization (120 days)	[231]
O_2	PCL (powder), cellulose (powder)	Aqueous environment	25	30–35% BOD (150 days)	[208]
O2	 PHB (film, 735 kDa, 65%), PHBV (film 484 kDa, 46%), PCL (films, 187 kDa, 63%), PES (film, 87 kDa, 61%), PEA (film, 144 kDa, 74%), PBS (film, 79 kDa, 63%), PBA (film, 81 kDa, 70%), PBSe (films, 31.5 kDa, 68%) 	Freshwater (river)	25	PHB: 75 ± 16% BOD, PHBV: 76 ± 2% BOD, PCL: 75 ± 8% BOD, PES: 83 ± 2% BOD, PEA: 70 ± 3% BOD, PBS: 3 ± 1% BOD, PBA: 20 ± 4% BOD, PBSe: 6 ± 3% BOD (28 days)	[263,264]
O ₂	 PHB (film, 735 kDa, 65%), PHBV (film 484 kDa, 46%), PCL (films, 187 kDa, 63%), PES (film, 87 kDa, 61%), PEA (film, 144 kDa, 74%), PBS (film, 79 kDa, 63%), PBA (film, 81 kDa, 70%) 	Freshwater (lake)	25	PHB: 52 ± 7% BOD, PHBV: 71 ± 1% BOD, PCL: 77 ± 1% BOD, PES: 77 ± 1% BOD, PEA: 68 ± 8% BOD, PBS: 12 ± 8% BOD, PBA: 80 ± 13% BOD (28 days)	[263]

O ₂	PHB, PHBV, PCL, PES, PEA, PBS, PBA	Seawater (bay)	25	PHB: $27 \pm 10\%$ BOD, PHBV: $84 \pm 2\%$ BOD, PCL: $79 \pm 2\%$ BOD, PES: $1 \pm 1\%$ BOD, PEA: $65 \pm 3\%$ BOD, PBS: $1 \pm 1\%$ BOD, PBA: $20 \pm 2\%$ BOD (28 days)	[263]
O ₂	PHB, PHBV, PCL, PES, PEA, PBS, PBA	Seawater (ocean)	25	PHB: 14 ± 10% BOD, PHBV: 78 ± 5% BOD, PCL: 43 ± 14% BOD, PES: 3 ± 2% BOD, PEA: 46 ± 13% BOD, PBS: 2 ± 0% BOD, PBA: 10 ± 5% BOD (28 days)	[263]
O ₂	Cellulose (filter paper)	Seawater (pelagic, eulittoral, sublittoral, supralittoral, deep sea, buried under sediments)	11–26	-	[213]
O ₂	PLA (film), PBAT (film), PCL (film and powder), cellulose (powder)	Inoculum from activated sludge	30 ± 2	PLA: 3.7% BOD, PBAT: 15.1% BOD, PCL (film): 34.8% BOD, PCL (powder): 37.7% BOD (28 days)	[209]
O ₂	PLA (films, fibers), PHA (films)	Soil	30, 40	PLA (films, 30 °C, 20 days): 9.8–10.3% BOD, PLA (films, 40 °C, 10 days): 11.8–17.9% BOD, PLA (fiber, 30 °C, 20 days): 9% BOD, PLA (fiber, 40 °C, 10 days): 16% BOD, PHA (films, 30 °C, 20 days): 26.3% BOD, PHA (films, 40 °C, 12 days): 49.5% BOD	[265]

O_2	PBS (sheets), cellulose	Inoculum from	25	PBS: 31% BOD (80 days)	[212]
O_2	PHBV (powder, 376 kDa, 58.5%), cellulose (powder)	Aqueous conditions	20	PHBV: 80 % BOD (28 days)	[214]
O ₂	PLA (film)	Lake water, compost, soil in laboratory conditions	20	PLA (lake water): ~5 mgO ₂ /dm ³ water, PLA (compost): ~25 mgO ₂ /kg compost, PLA (soil): ~100 mgO ₂ /kg soil (28 days)	[266]
O_2	PCL (film), PLA (film)	Compost, activated sludge, river water, sea water	20	PCL (compost): 140 mgO ₂ /dm ³ , PLA (compost): 125 mgO ₂ /dm ³ , PCL (activated sludge): 120 mgO ₂ /dm ³ , PLA (activated sludge): 115 mgO ₂ /dm ³ , PCL (river water): 10 mgO ₂ /dm ³ , PLA (river water): 8 mgO ₂ /dm ³ , PCL (sea water): 5 mgO ₂ /dm ³ , PLA (sea water): 5 mgO ₂ /dm ³ (7 days)	[267]
O_2	PBAT (film, 16 kDa)	Mineral medium	25	10% BOD (22 days), 45% BOD (45 days)	[268,269]
O ₂	PHBV (powder, film, undrawn fiber, fivefold- drawn fiber, 250 kDa)	Freshwater, seawater	25	Powder: 18% BOD, film: 18% BOD, undrawn fiber: 18% BOD, fivefold-drawn fiber: 8% BOD (28 days)	[196]
O_2	PCL (powder), cellulose (powder)	Activated sludge	25	PCL: 20–100% (100 days)	[210]
O_2	PLA (powder), PCL (powder)	Aqueous conditions	30	PLA: 35% (40 days), PCL: 100% (days)	[211]

O_2	PLA (film, particle), PBAT	Seawater in	27	PLA: 0.3 % BOD, PBAT: 1-	[270]
	(film, particle), PBS (film,	laboratory		1.4% BOD, PBS: 0.1–1.3%	
	particle), PBSA (film,	conditions		BOD, PBSA: 0.4–29.2 %	
	particle), PCL (film,			BOD, PCL: 14.5–40.9% BOD,	
	particle), PHB (particle)			PHB: 44–60.4% BOD (4	
				weeks)	

BOD: biochemical oxygen demand; *using ISO 14855; *relative to the reference material

2.12 Mass loss and mechanical properties deterioration

Measurement of mass loss is the most commonly used method to indicate the extent of degradation and is indicated as mass loss measured from the samples retrieved during the degradation test [195]. Mass loss is used mostly to designate the degradation occurring on the polymer surface and is contingent on the disintegration phenomena. Many researchers have reported the use of mass loss determination to indicate that the material has undergone degradation. Furthermore, deterioration of mechanical properties (assessed on films, sheets, or dumbbell specimens) indicative of degradation by the action of abiotic mechanisms has been reported along with mass loss.

Quartz crystal microbalance with dissipation (QCM-D). In terms of enzymatic degradation, microbalance weight loss technique in the nanogram scale has been reported during enzymatic hydrolysis of aliphatic and aromatic polyesters as PCL and PBAT. This is a unique approach to monitor the dynamic of the enzymatic hydrolysis and has showed high sensitivity [262–266].

2.13 Macro and micro visual analysis of the polymer surface

Macro visual analysis is the second most commonly used technique after mass loss. Macro visual changes of the polymer do not necessarily indicate biodegradation, but these changes are usually the first evidence of microbial colonization and biofilm formation.

Micro visual inspection using microscopic techniques like SEM, transmission electron microscopy (TEM) or atomic force microscopy (AFM) can impart more knowledge regarding the biodegradation process at the early stage, specifically biofilm formation and the structure of the sample [49]. The topographical changes occurring in the polymer are usually seen as the formation of holes, cracks, cavities (material erosion), discoloration, or surface roughness [267]. Kijchavengkul et al. studied the surface evolution during biodegradation of PBAT films and demonstrated the consequences of the degradation by using SEM methodology among

other techniques [268]. For PBAT samples with c. 30% or less crosslinking, biofilm formation was observed. Large number of microbes consumed PBAT samples, creating pits in the film surface. For samples with more than 30% crosslinking, no cavities were observed on the PBAT film surface, indicating that increased crosslinking results in reduced biodegradation [268]. Shah et al. reported changes in surface morphology, such as pit formation and erosion, due to the biodegradation of PHBV films in a basal salt medium after two weeks of immersion [269]. Techniques as TEM and AFM were extensively used for identification of chemical and enzymatic degradation of polyesters [270–273].

2.13.1 Chromatography

Size exclusion chromatography (or gel permeation chromatography). It is used to study the reduction of M_w . Reduction and distribution in M_w are a preferred parameter that provides evidence of the biodegradation process. When accompanied with mineralization, the M_w reduction can provide more insights into understanding the process.

High performance liquid chromatography is widely used for qualitative and quantitative analysis of soluble compounds derived from enzymatic activity that are released into solution.

Lu et al. examined biodegradability of PPC/starch composites in soil at room temperature; the study of M_w change for unburied, 40 and 180 days along with weight loss and other qualitative techniques like FTIR, SEM, and photographs helped the researchers conclude that PPC was the last component to biodegrade post microbial colonization and starch degradation [274]. Reduction of Mw by chemical hydrolysis has been reported for aliphatic and aromatic polyesters as PLA, PCL, PHB, and PBAT, among others [64,275,276].

2.13.2 Spectroscopy

A qualitative way to assert biodegradation is by identifying the chemical changes in the polymer structure [195]. These changes could translate into the formation of low M_w compounds resulting from the polymer degradation. Changes in the molecular structure can be identified by various spectroscopic analysis methods.

Nuclear magnetic resonance (NMR): The nuclei of any given type (C, H, N, P, or O) resonate at different energies. The information from the NMR signal (position and pattern) gives critical information about the nuclei environment and presence [79,277]. The use of NMR has been reported for the degradation of different polymers in different environments. Kijchavengkul et al. studied the biodegradation of PBAT in compost and tracked the evolution of the BT and BA dimers using ¹H NMR and showed that the soft aliphatic portion and the amorphous region are more susceptible to hydrolysis and biodegradation than the rigid aromatic portion and the crystalline region [79].

Fourier-transform infrared spectroscopy (FTIR): FTIR analysis of any given material provides a specific fingerprint spectrum for that material, and the appearance and disappearance of peaks associated with the functional groups can help explain the changes happening in the material structure [79,195]. Mass spectroscopy is an analytical technique widely used for identification of products during enzymatic degradation of polymers. In general, it is used along with techniques as liquid chromatography.

Weng et al. studied the biodegradation of PHB/PLA blends buried in soil at different depths at c. 20 °C; the FTIR spectra showed that the peaks in the 4000 to 3000 cm⁻¹ region were broad in nature due to the formation of -OH and -COOH groups after degradation [278]. Furthermore, Mbarki et al., conducted both the FTIR and NMR analysis on PDLA samples immersed in the soil/liquid culture at 37 °C and found no significant difference in the chemical structure before and after immersion (45 days for FTIR and 28 days NMR); the conclusion derived was that the biodegradation phenomena was only surface and not bulk [279].

2.13.3 Plate (clear zone formation) and turbidimetry assays

Plate tests were originally designed to gauge the resistance of plastics to degradation via microorganisms. However, in addition to testing resistance they are now used to see if the polymer can support the growth of microorganisms through biofilm formation. The polymeric material is dispersed in a petri dish containing a mineral salts agar medium that serves as the sole carbon source. The polymer in the surface, suspended in the medium, is then inoculated with microorganisms and held for a predetermined amount of time at a constant temperature to allow the microorganisms to grow. The formation of a halo or clear zone around the microorganism colony marks an end for this test, since the clear zone indicates that the microorganism can at least depolymerize the polymeric material. The test is also used in screening, isolating, and identifying the potential degrading microorganisms for any given polymer [280,281]. Urbanek et al. isolated, screened, and assessed the degrading capability of Antarctic soil microorganisms on PCL, PBS, and PBSA at low temperature with the help of the clear zone formation technique [282].

2.13.4 Respirometric tests for CO₂ evolution and biochemical O₂ demand

These tests involve measuring the consumption of O_2 or formation of CO_2 under aerobic conditions. The CO_2 evolved can be measured by three different techniques [60]: in cumulative measurement respirometry (CMR) the evolved CO_2 (trapped in basic solution such sodium hydroxide, barium hydroxide) is quantified by titration method [283]; in gravimetric measurement respirometry (GMR) the evolved CO_2 is trapped in absorption columns and the weight increase is used to quantify the amount of CO_2 [283]; and in direct measurement respirometry (DMR) the evolved CO_2 is quantified by means of an inline nondispersive infrared gas analyzer or gas chromatograph [113]. Kale et al. compared the use of CMR, GMR, and DMR to assess the biodegradation of PLA under simulated composting conditions, and found similar evolution of biodegradation [283]. They concluded that the biodegradation process is further dependent on various factors, including shape, size, thickness, and sample/compost ratio, among others. The advantages and disadvantages associated with these techniques are explained in detail elsewhere [268].

Techniques measuring O_2 consumption, reported as BOD, are assessed in specific aquatic environments as sewage sludge and wastewater. However, standards have been also developed for assessing O_2 consumption in soil environments (**Table 2.8**).

2.13.5 Radiolabeling

An understudied approach for assessing the degree of biodegradation is the use of radiolabeled carbon. This is one of the absolute tests to determine biodegradation and involves tracking carbon from biodegradable polymers into CO_2 and biomass. The approach is based on labeling the carbon atoms in the polymer backbone with carbon isotopes: ¹³C (stable in nature) and ¹⁴C (radioactive) [284,285].

Early works conducted by Albertsson et al. showed that the technique of radiolabeling polymers using ¹⁴C were useful not only for detecting the biotic stage of the biodegradation process but also the abiotic stage [286–289]. In this sense, PE films produced using a ¹⁴C marker showed ¹⁴CO₂ evolution of the carbonyl oxidized byproducts when the films were exposed to soil [286].

Studies in the area of biodegradable polymers addressed in this review are insufficient. Zumstein et al. employed the use of ¹³C labeled polymer along with isotopespecific analytical methods (i.e., cavity ring-down spectroscopy) to track the biodegradation of PBAT in soil [140]. This technique allowed for tracking of the basic biodegradation steps by distinguishing the labelled PBAT CO₂ from the CO₂ evolved due to the mineralization of organic matter in the soil. In summary, two or more methods are commonly employed together to determine biodegradation. The change in M_w , weight loss, and surface analysis are used widely but these alone do not guarantee biodegradation, and at most hint towards disintegration of the material under study. The evolution of CO_2 and radiolabeling represents the complete assessment of the breakdown of the material into biomass and need to be employed on a more regular basis for biodegradation studies. Though the respirometry method gives the mineralization value, radiolabeling is far more advanced by showing the actual integration of polymer carbon into the microbial biomass.

With respect to standards, there is no international standard specifying how home composting should be conducted for effective biodegradation of biodegradable polymers. Also, many standards for determining polymer biodegradation in aquatic environments, as listed in **Table 2.8**, mention temperatures (laboratory simulated settings conditions) that are much higher than the actual conditions encountered in real-world environments. In general, adaptations of the international standards to specific conditions are implemented to assess and report, for example, biodegradation of polymers in home composting or at mesophilic temperatures.

2.14 Microorganisms and enzymes able to biodegrade polymers

The ability to degrade biodegradable polymers is widely distributed among bacteria, fungi, and actinomycetes, and there is much variation in ability. **Table 2.10** lists extracellular enzymes and/or microorganisms able to biodegrade polymers in different mesophilic environments, as reported in the published literature.

2.14.1 Microbial Population

Some microorganisms can digest several polymer structures in different environments, and degradation rate efficiency can differ. A high portion of the published works have reported the digestion activity of a specific microorganism in the highly controlled conditions of incubated or culture media; under these conditions, the polymer substrate is mostly the only source of nutrient for the microorganism. In contrast, in less restricted environments, such as soil, home composting, industrial composting, or aquatic environments, the complexity of the biological activity process increases, and several sources of substrates and microorganisms may be available. In natural soil or aquatic environments, an active population of microorganisms with different requirements, in terms of nutrients and optimal growth conditions, are competing or working cooperatively for the resources available.

The presence of microorganisms and the formation of a biofilm, due to the colonization

of the polymer surface, creates an effect that sometimes can alter the abiotic degradation of the polymer. For example, when PCL biodegradation was evaluated under low stirring, the impediment of biofilm formation resulted in a higher weight loss [298]. **Table 2.10** Enzymes and/or microorganisms with activity for degrading biodegradable polymers when tested in mesophilic conditions. Different parameters such as the enzymes released, microbial species used, the environment from which the microorganisms were isolated / testing media, polymer studied, the temperature and pH for conducting the biodegradation study, optimal conditions for the microorganisms, and studies reporting them are mentioned.

Enzymes*	Microorganism*	Environment	Polymer	Т (°С), рН	Optimal conditions of T (°C) and pH	Reference
Alcalase (3.4.21.62)	Bacillus licheniformis (B)	Buffer solution	PLA	40, 8.0	60, 9.5	[290]
Amidase (3.5.14)/esterase (55 kDa)	Rhodococcus equi strain TB-60	Soil/culture	PU	30, 7	45, 5.5	[291]
Carboxyl esterase (3.1.1.1)	Alcanivorax borkumensis (B), Rhodopseudomonas palustris (B)	Culture	PCL, PDLLA, PBSA	30, 8.0	30-37, 9.5-10	[292]
Carboxyl esterase	Alcanivorax borkumensis (B)	Culture	PES, PHBV, PDLLA	30, 8.0	55-60, 9.5-10	[292]
Chymotrypsin (3.4.21.1)	-	Culture	PLLA, PEA	37, 7.0	-, -	[293]
Cutinase (3.1.1.74) (21.6 kDa)	Aspergillus oryzae RIB40 (F)	Culture	PBS, PBSA, PLA	37, 8.0	35–55, 9.0	[183]
Cutinase	Alternaria brassicicola (F), Aspergillus fumigatus (F), Aspergillus oryzae (F), Humicola insolens (F), Fusarium solani (F)	Culture	PCL	40, 3, 5, 8	-, -	[294]
Cutinase	Fusarium solani (F)	Buffer solution	PBAT	30, -	-, -	[140]

Cutinase (21 kDa)	Cryptococcus magnus (F)	Larval midgut of stag beetle (Aegus laevicollis)/culture	PBS, PBSA, PCL, PDLLA, PLLA	30, 7.4	40, 7.5	[295]
Cutinase	Fusarium solani (F)	Buffer solution	PCL	37, 7.2	-, -	[296]
Cutinase (20 kDa)	Fusarium sp. FS1301 (F)	Soil/liquid culture	PBS, PCL	30, -	50, 8.0	[297]
Cutinase (19.7 kDa)	Paraphoma-related fungal strain B47-9 (F)	Barely phyllophane/liquid culture	PBAT, PBS, PBSA, PCL, PDLLA	30, 7.2	45, 7.2	[298]
Cutinase	Pichia pastoris (F)	Buffer solution	PBS	37, 7.4	-, -	[299]
Cutinase	-	Culture	PBS, PBA	37, 7.4	-, -	[300]
Cutinase (20.3 kDa)	Pseudozyma antarctica JCM 10317 (Y)	Culture	PBS, PBSA, PCL, PLLA, PDLLA	30	40, 9.5	[301,302]
Cutinase	Fusarium solani (F), Fusarium moniliforme (F)	Culture	PCL	22	9– 10	[303]
Cutinase	Bacillus sp. KY0701	Culture	PCL	30, 7	$50, \\7$	[304]
Cutinase	Aspergillus oryzae (F)	Buffer solution	PCL	40, 8	-, -	[305]
Cutinase	Pseudozyma jejuensis OL71 (F)	Leaves of Citrus unshiu/culture	PCL	30, -	-, -	[306]
Cutinase-like enzyme (22 kDa)	Cryptococcus flavus GB-1 (Y)	Culture	PBSA	30, 6.8	45, 7.8	[307]

Cutinase-like enzyme	Cryptococcus sp. Strain S-2 (F)	Liquid culture	PBS, PLA, PCL	30, -	37, 7.0	[308]
Close related to Cutinase	Pseudomonas pachastrellae JCM12285 ^T (B)	Marine, coastal seawater/culture	PCL	30, -	-, -	[309]
Elastase	-	Culture	PLA	37, 7.0	-, -	[293]
Esterase (3.1.1.1)	Aspergillus sp. strain S45 (F)	Solid waste dumpsite/liquid culture	PU	30, 7.0	-, -	[240]
Esterase	Bacillus sp. AF8 (B), Pseudomonas sp. AF9 (B), Micrococcus sp. 10 (B), Arthrobacter sp. AF11 (B), Corynebacterium sp. AF12 (B)	Soil/culture	PU	30–35	-, -	[250]
Esterase	Hog liver	Buffer solution	PGA	37, 7.5	-, -	[310]
Esterase	Bacillus subtilis (B)	Buffer solution	PCL, PLA	37, -	-, -	[258]
Esterase	Aspergillus tubingensis (F)	Soil/solid and liquid culture	PU	(30, 37, 40), (5 - 9)	37, 7.0	[311]
Esterase	Bacillus licheniformis (B)	Compost/liquid culture	PLLA	32, 7.4	-, -	[312]
Esterase	Alicycliphilus sp. (B)	Culture	PU	37, 7	-, -	[313]
Esterase	Leptothrix sp. TB-71 (B)	Soil, fresh water/culture	PBSA, PES, PCL	30, -	-, -	[314]

Esterase (62 kDa)	Comamonas acidovorans strain TB- 35 (B)	Soil/liquid culture	PU	30, 7.2	45, 6.5	[315– 317]
Esterase (28 kDa)	Curvularia senegalensis (F)	Soil/liquid culture	PU	(21–25), 30, 35, 45, (4.0– 8.0)	-, 7–8	[318]
Esterase (42 kDa)	Comamonas acidovorans (B)	Culture	PU	30, 5–8	-, -	[319]
Esterase	Penicillium verrucosum (F), Aspergillus ustus (F)	Compost soil/culture	PLA	30, 5.6	-, -	[320]
Esterase	Pseudomonas aeruginosa MZA-85 (B), Bacillus subtilis MZA-75 (B)	Soil/liquid culture	PU	37, 7.0	-, -	[246– 248]
Esterase	Pseudomonas aeruginosa strain S3 (B)	Culture	PLA	30–37, 8	37, 8	[321]
Esterase	Pseudomonas (B)	Soil/Culture	PES	30, -	-, -	[184]
Esterase	Porcine liver	Buffer solution	PLA	40, 8.0	40, 8.0	[290]
Close related to esterase	Bacillus pumilus strain KT1012 (B)	Soil, water/culture	PES, PCL	30, 7.0	40– 45, -	[322]
Lipase (3.1.1.3)	Rhizopus delemar (F)	Buffer solution	PLA	37, 7.2	- ,-	[323]
Lipase	Acidovorax delafieldii Strain BS-3 (B)	Soil/solid and emulsified substrate	PBS, PBSA	30, 7.0	-, -	[324]
Lipase	Rhizopus oryzae (F), Burkholderia sp. (B)	Liquid culture	PCL	30, -	-, -	[308]

Lipase	Candida rugosa (F)	Buffer solution	PCL, PLA	37, -	-, -	[258]
Lipase (36 kDa)	Aspergillus niger MTCC 2594 (F)	Liquid culture	PCL, PLA	30, 7	37, 7.0	[325]
Lipase	Aspergillus oryzae (F)	Buffer solution	PCL	37, 7.0	-, -	[326]
Lipase	Aspergillus tubingensis (F)	Soil/solid and liquid culture	PU	(30, 37, 40), (5–9)	37, 5.0	[311]
Lipase	Burkholderia cepacia PBSA-1 (B), Pseudomonas aeruginosa PBSA-2 (B)	Soil/culture	PBSA	27, 37	-,	[251]
Lipase	Candida cylindracea (F)	Buffer solution	PLA	40, 8.0	40, 8.0	[290]
Lipase	Candida antarctica (F)	Buffer solution	PCL, PBS	45, 7.2	-, -	[296,327,328]
Lipase	Candida rugosa (F)	Liquid culture	PU	(20–50), (4–9)	35, 7.0	[329]
Lipase	Chromobacterium viscosum (B), Rhizopus orizae (F), Rhizopus niveus (F)	Culture	PCL, PBS, PBSA	37, 7.0	-, -	[330]
Lipase (23 kDa)	Cryptococcus sp. MTCC 5455 (F)	Liquid culture	PBAT	25, -	-, -	[331]
Lipase	Cryptococcus sp. MTCC 5455 (F)	Buffer solution	PU	30, 7.0	37, (7.0– 8.0)	[332]

Lipase	Lactobacillus plantarum (B)	Culture	PCL	37, 8.0	-, -	[333]
Lipase (25 kDa)	Penicillium sp. strain 14-3 (F)	Soil/liquid culture	PEA	30, 6.0	45, 4.5	[334]
Lipase	Pseudomonas (B)	Buffer solution	PLLA, PCL, PDLLA	37, 7.0	-, -	[335,336]
Lipase	Pseudomonas cepacia (B)	Buffer solution	PCL	37, 7,0	-, -	[201]
Lipase	Pseudomonas cepacia (B), Rhizopus delemar (F)	Buffer solution	PCL, PPS	30, 7.2	-, -	[337]
Lipase	Pseudomonas fluorescens (B)	Buffer solution	PCL	37, 7.4	-, -	[338]
Lipase (22 kDa)	Cryptococcus sp. (Y)	Buffer solution	PBS, PBSA	30, 7	-, -	[339]
Lipase	Fusarium solani (F)	Culture	PCL	22, 6.8	-, -	[340]
Lipase (34 kDa)	Pseudomonas sp. strain DS04-T (B)	Activated Sludge/liquid medium	PLLA, PCL, PHB	37, 8	50, 8.5	[341]
Lipase	Rhizopus oryzae (F)	Solution	PBS, PLLA, PBA	40, 5	40, 7	[263]
Lipase	Rhizopus arrhizus (F)	Buffer solution	PCL	30, 7	-, -	[182]
Lipase	Pseudomonas (B)	Buffer solution	PCL	25, 37, 7	-, -	[342]
Lipase	Rhizopus oryzae (F)	Buffer solution	PBAT	30, -	-, -	[140]
Lipase	Rhizopus delemar (F)	Buffer solution	PU	37, -	-, -	[343]

Lipase	Pseudomonas (B)	Buffer solution	PCL	37, 7	-, -	[344]
Lipase	Achromobacter sp (B), Candida cylindracea (F), Rhizopus arrhizus (F), Rhizopus delemar (F), Geotrichum candidum (F)	Buffer solution	PEA, PCL	37, 7.0	-, -	[345]
Lipase	Bacillus sp. (B)	Soil/culture buffer solution	PBAT	30– 37, 7.4	-, -	[346]
Lipase	Pseudomonas sp. (B)	Buffer solution	PEA	37, 7.0	-, -	[347]
Lipase	Stenotrophomonas sp. YCJ1	Soil/culture	PBAT	30, 7.2	37, 7.5	[348]
Lipase	Candida Antarctica (F)	Buffer solution	PBAT	45, 7.2	-, -	[349]
PBAT hydrolase (close related to lipase)	Rhodococcus fascians NKCM 2511 (B)	Soil/liquid culture	PBAT, PCL, PBSA, PES, PBS (low activity)	25, -	-, -	[259]
PBAT hydrolase (close related to cutinase) (18.9 kDa)	Rhodococcus fascians (B)	Liquid culture	PBAT, PCL, PBSA, PES, PBS	30, 7	-, -	[350]
PBAT hydrolase (close related to Lipase)	Bacillus pumilus (B) (NKCM3101, NCKM3201, NCKM3202, KT1012), Brevibacillus choshinensis PBATH (B)	Soil/liquid culture	PBAT (low activity), PBSA, PBS, PES, PCL	30, 7.0	-, -	[81]
PLA depolymerase (related to lipase)	Paenibacillus amylolyticus Strain TB-13 (B)	Soil/culture	PBS, PBSA, PDLLA, PCL, PES	37, 8	45– 55, 10.0	[351]

PBAT hydrolase	Isaria fumosorosea strain NKCM1712 (F)	Soil/culture	PBAT, PBA, PBS, PBSA, PES, PHB, PCL	25-45, 7.0	-, -	[260]
PBS-degrading enzyme (44.7 kDa)	Aspergillus sp. XH0501-a (F)	Soil/culture	PBSA	30	40, 8.6	[352]
PCL depolymerase (63.5 kDa) (esterase)	<i>Brevundimonas</i> sp. strain MRL-AN1 (B)	Liquid culture	PCL, data not shown for PLA, PES, PHB, and PHBV	37, 7	30, 6– 8	[353]
PCL depolymerase	Penicillium oxalicum strain DSYD05-1 (F)	Soil/liquid culture	PCL, PHB, PBS	30, 6.8	-, -	[354]
PCL depolymerase	Alcaligenes faecalis TS22 (B)	Culture	PCL	30, -	-, -	[355]
PCL depolymerase	Paecilomyces lilacinus strain D218 (F)	Soil/solid culture	PCL	30, 5.2	$30, \\ 3.5- \\ 4.5$	[356]
PLA depolymerase (58 kDa)	Pseudomonas tamsuii TKU015 (B)	Soil/culture	PLLA	30, 7.0	60, 10	[357]
PLLA degrading enzyme	Actinomadura keratinilytica T16-1 (B)	Culture	PLLA	45, 7	45, 6– 8	[358]
PHA depolymerase (3.1.1.76)	Alcaligenes faecalis (B)	Buffer solution	PHB, PHBV, PHA	37, 7.4	-, -	[359]
PHA depolymerase (48 kDa)	Pseudomonas stutzeri YM1414 (B)	Fresh water/buffer solution	PHB	37, 7.4	55, 9.5	[360]
PHA depolymerase	Ralstonia pickettii T1 (B)	Buffer solution	PHB, PHBV	37, 7.5	-, -	[174]

PHA depolymerase	Ralstonia pikettii T1 (B), Acidovorax sp. TP4 (B)	Buffer solution	РНА	37, 38, 7.5, 8.0	-, -	[361]
PHA depolymerase	Comamonas sp. DSM 6781 (B), Pseudomonas lemoignei LMG 2207 (B), Pseudomonas fluorescens GK13 DSM 7139 (B)	Liquid culture	PHB, PHV, PHBV	30, 7.2	-, -	[362]
PHA depolymerase (50 kDa)	Comamonas testosteroni (B)	Buffer solution	PHB, PHBV	37, 7.4	-, 9.5– 10	[363]
PHA depolymerases (33.8 and 59.4 kDa)	Pseudomona mendocina DS04-T (B)	Mineral medium	PHB, PHBV	37, -	50, 8 and 8.5	[364]
PHA depolymerase (intracellular)	Pseudomonas putida LS46 (B)	Culture	PHB, PCL, PES	30, 7	-, -	[365]
PHB depolymerase (3.1.1.75)	Alcaligenes faecalis (B)	Culture	PHB	37, 7.4	-, -	[366]
PHB depolymerase	Alcaligenes faecalis (B), Pseudomonas stutzeri (B), Comamonas acidovorans (B)	Buffer solution	PHB, PEA, PES	37, 7.4	-, -	[367]
PHB depolymerase (57 kDa)	Aspergillus fumigatus (F)	Buffer solution	PHB, PHBV, PEA, PES	45, 8.0	70, 8	[368,369]
PHB depolymerase (49 kDa)	Comamonas testosteroni strain ATSU (B)	Soil/culture	PHB, PHBV	37, 7.4	$70, \\ 8.5$	[370]
PHB depolymerase (42.7)	Aureobacterium saperdae (B)	Buffer solution	PHB	37, 7	45, 8	[371]

PHB depolymerase (57 kDa)	Aspergillus fumigatus 76T-3		PHB, PES, PBS	45, -	55, 6.4	[372]
PHB depolymerase (50–48 kDa)	Emericellopsis minima W2 (F)	Wastewater/liquid culture	PHB, PHBV	30, 8.0	55, 9.0	[373]
PHB depolymerase (40 kDa)	Microbacterium paraoxydans RZS6 (B)	Dumping yard/culture	PHB	30, -	30, 7	[374]
PHB depolymerase (46.8 kDa)	Penicillium sp. DS9701-D2 (F)	Activated sludge/culture	PHB	28– 30, 6.8	30, 5	[375]
PHB depolymerase	Streptoverticillium kashmirense AF1 (A)	Sewage sludge/culture	PHBV	30, 8	-, -	[376]
PHB depolymerase (50 kDa)	Acidovorax sp. strain TP4 (B)	Pond water, river water, farm soil/culture	PHB	$\begin{array}{c} 30,\\ 8.5\end{array}$	-, -	[377]
PHB depolymerase (47 kDa)	Arthrobacter sp. strain W6 (B)	Soil/culture broth	PHB, PHBV	30, 7	50, 8.5	[378]
PHB depolymerase (85 kDa)	Fusarium solani Thom (F)	Wastewater/culture	PHB	25, 8	$55, \\7$	[379]
PHB depolymerase (62.3 kDa)	Bacillus megaterium N-18-25- 9 (B)	Culture	PHB	30– 37, 9	65, 9	[380]
PHB depolymerase (44.8 kDa)	Penicillium sp. (F)	Culture	PHB	40, 4– 6	50, 5	[381]
PHB depolymerase (61.8–70 kDa)	Marinobacter sp. NK-1 (B)	Culture	PHB	37, 7.4	-, 8	[382,383]
PHB depolymerase	Nocardiopsis aegyptia sp. nov. DSM 44442 ^T (B)	Marine seashore sediments/culture	PHB, PHBV	30, 7	-, -	[384]

PHB depolymerase (33 kDa)	Penicillium funiculosum (F)	Culture	PHB	$\begin{array}{c} 30,\\ 7.5\end{array}$	-, 6.5	[385]
PHB depolymerase (36 kDa)	Penicillium simplicissimum LAR13 (F)	Soil/culture	РНВ	25, 30, 37, -	45, 5.0	[386]
PHB depolymerase	Paecilomyces lilacinus D218 (F)	Soil/liquid culture	PHB, PCL	30, 6.0	50, 6.5 - 7.5	[356]
PHB depolymerase	Pseudomonas fluorescens (B), Pseudomonas aeruginosa (B), Pseudomonas putida (B)	Contaminated soil/culture	PHB, PHBV	30, 7.9	-, -	[105]
PHB depolymerase (48 kDa)	Comamonas acidovorans YM1609 (B)	Freshwater/culture	PHB, PHBV	37, 7.4	-, -	[387]
PHB depolymerase	Pseudomonas stutzeri (B)	Sea water/Buffer solution	РНВ	30-45, 7.4	-, 7–7.5	[388]
PHB depolymerases (44, 46 kDa)	Agrobacterium sp. K-03 (B)	Culture	PHB, PHBV	30, 8	45, 7,9 and 8.1	[389]
PHB depolymerase (49 kDa)	Streptomyces exfoliatus K10 (B)	Culture	PHB	25- 37, 8	40, 8.5– 9	[390]
PHB depolymerase (40 kDa)	Pseudomonas pickettii (B)	Culture	PHB	37, 7.4	40, 5.5	[391]
PHB depolymerase (53 kDa)	Comamonas sp. (B)	Solid culture	PHB	37, 8	-, -	[392]

PHB depolymerase (65 kDa)	Alcaligenes faecalis AE122 (B)	Seawater/culture	PHB	37,	-, -	[393]
PHB depolymerase (95.5 kDa)	Alcaligenes faecalis AE122 (B)	Seawater/culture	PHB	30, 6.8–7.5	$55, \\9$	[394]
PHB depolymerase (40 kDa)	Aspergillus fumigatus (F)	Culture	PHB	30–32, 8	-, -	[395]
PHB depolymerase (48 kDa)	Alcaligenes faecalis T ₁ (B)	Activated sludge/culture	PHB	30, 7.5	-, 7.5	[396]
PHB depolymerase	Ralstonia pikettii (B)	Culture	PHB, PHBV	20, 7.5	-, -	[270]
PHB depolymerase (45 kDa)	Paecilomyces lilacinus F4-5 (F)	Soil/culture	PHB, PHBV	27–37, 7	$50, \\7$	[397]
PHB depolymerase (52.2 kDa)	Diaphorobacter sp. PCA039 (B)	Culture	PHB, PHBV	30, -	45, 8	[398]
PHB depolymerase (63.7 kDa)	Aspergillus fumigatus 202 (F)	Soil/culture	PHB	30, 37, 45, 7	45, 7	[399]
PHB depolymerase (20 kDa)	Penicillium expansum (F)	Wastewater/culture	PHB	30, 5	50, 5	[400]
PHB depolymerase	Streptomyces sp. SNG9 (B)	Marine/liquid culture	PHB, PHBV	30, 7	-, -	[401]

PHB depolymerase	Bacillus (B), Clostridium (B),	Soil, lake water,	PHB,	4-58,	29– 25	[402]
(45 KDa)	Comamonas (B), Acaligenes (B), Zoogloea (B)	air/liquid culture	PHBV	4.8– 10.6	55, 9.4	
PHB depolymerase (37 kDa)	Penicillium funiculosum (F)	Culture	PHB	30, 5	-, 6	[403]
PHB depolymerase (48 kDa)	Paecilomyces lilacinus D218	Buffer solution	PHB, PHBV	30, 6.8	45, 7	[404]
PHB depolymerase	Aspergillus clavatus strain NKCM1003 (F)	Soil/culture	PES, PHB, PCL, PBS	30, -	-, -	[405]
PHBV depolymerase (36, 68, 72, 90 kDa)	Aspergillus sp. NA-25 (F)	Soil/solid culture	PHBV	30, 7.0	45, 7.0	[406]
PHBV depolymerase (43.4 kDa)	Acidovorax sp. HB01	Activated sludge/	PHBV, PHB, PCL	37, 6.8	50, 7	[407]
PHBV depolymerase (51 kDa)	Streptomyces sp. strain AF-111 (B)	Sewage sludge/culture	PHBV	30– 37,	35-55, 7-8	[408]
PHV depolymerase (43.6 kDa)	Pseudomonas lemoignei (B)	Liquid culture	PHB, PHV	37, 8	-, -	[409,410]
Polyurethanase – lipase (28 kDa)	Bacillus subtilis (B)	Soil/liquid culture	PU	30, 7	-, -	[411]
Polyurethanase esterase (27 kDa)	Pseudomonas chlororaphis (B)	Liquid culture	PU	30, 7.2	-, 7– 8	[412]

Polyurethanase esterase/protease (63 kDa), Polyurethanase esterase (31 kDa)	Pseudomonas chlororaphis (B)	Yeast extract salts medium	PU	30, -	-, 8.5 and 7	[413]
Polyurethanase protease (29 kDa)	Pseudomonas fluorescens (B)	Liquid culture	PU	30, 7.2	25, 5.0	[414]
Polyurethanase lipase	Pseudomonas protegens strain Pf-5 (B)	Liquid culture	PU	27, 7.4	-, -	[415]
Polyurethanase (66 kDa)	Acinetobacter gerneri P7 (B)	Liquid culture	PU	30, 7.0	37, 8.0	[416]
Polyurethanase – protease	Alternaria solani Ss1-3 (F)	Soil/liquid culture	PU	(20– 35), (4.0– 8.0)	30, 7.0	[417]
Polyurethanase – esterase and amidase	Alicycliphilus sp. BQ8 (B)	Liquid culture	PU	37, 7.0	-, -	[418]
Polyurethanase serine hydrolase family (21 kDa)	Pseudomonas chlororaphis (B), Pestalotiopsis microspora (E2712A, 3317B) (F), Lasiodiplodia sp. E2611A (F), Bionectria sp. strain E2910B (F), Aspergillus niger (F), Pleosporales sp. E2812A (F)	Soil/liquid culture	PU	30, -	-, -	[419]
Protease (3.4.21)	Amycolatopsis orientalis (A)	Liquid culture	PLLA	30–40, 7.0	-, -	[420]
Protease	Bacillus licheniformis (B)	Buffer solution	PLA	37, -	-, -	[258]

Protease	Tritirachium album (F), Lentzea waywayandensis (A), Amycolatopsis orientalis (A)	Culture	PLLA	30, 7	-, -	[421]
PLA degrading enzyme close related to Protease (40–42 kDa)	<i>Amycolatopsis</i> sp. strain 41 (A)	Soil/liquid culture	PLLA	37, 7.0	37 - 45, 6.0	[422]
Protease, esterase, and lipase	Amycolatopsis sp. strain SCM_MK2-4 (A)	Soil/liquid, solid culture	PLA, PCL	30, 7.0	-, -	[423]
Protease, PLA degrading enzyme	Stenotrophomonas pavanii CH1 (B), Pseudomonas geniculata WS3 (B)	Soil, wastewater sludge/liquid culture	PLA	30, -	30, 7.5 30, 8.0	[424]
Proteinase K (3.4.21.64)	-	Buffer solution	PLLA	37, 8.6	-, -	[182]
Proteinase K	÷	Buffer solution	Amorphous PLLA (not crystalline PLLA)	37, 8.6	-, -	[335]
Proteinase K	Tritirachium album	Liquid culture	PLA	30, -	-, -	[308]
Proteinase K	-	Culture	PLLA, PES, PEA, PBS, PBSA, PCL	37, 7.0	-, -	[293]
Proteinase K	-	Culture	PLLA	37, 8.6.	-, -	[181,425]

Proteinase K	Tritirachium album	Buffer solution	PLA	37, -	-, -	[258]
(PVAase)-Cu ₃ (PO ₄) ₂	Bacillus niacini (B)	Culture	PVOH	30, 8.0	30, 7	[426]
PVOH oxidase (1.1.3.30)	Sphingomonas sp. (B)	Activated sludge/culture	PVOH	25, 7.5	-, -	[427]
PVOH oxidase	Sphingopyxis sp. PVA3 (B)	Activated sludge/culture	PVOH	30, 7.2	-, -	[428]
PVOH degrading enzyme (30 kDa)	Pseudomonas (B)	Buffer solution	PVOH	27, 7.3	40, 7–9	[429]
PVOH degrading enzyme	Streptomyces venezuelae GY1	Culture	PVOH	30, 8	-, -	[430]
PVOH degrading enzyme	Penicillium sp. WSH0-21 (F)	Activated sludge/culture	PVOH	30, 7	-, -	[431]
PVOH degrading enzyme (67 kDa)	Alcaligenes faecalis KK314	River water/culture	PVOH	30, 7.2	-, -	[432]
Serine enzyme (3.4.21) (24 kDa)	Amycolatopsis sp. strain K104-1 (A)	Soil/liquid medium	PLLA	37, 7.0	55-60, 9.5	[433]
Subtilisin (3.4.21.62)	-	Culture	PLA, PEA, PBS, PBSA, PCL	37, 7.0	-, -	[293]

Trypsin (3.4.21.4)	-	Culture	PLA, PEA	37, 7.0	-, -	[293]
Aliphatic-aromatic co-polyester degrading enzyme (27–31 kDa)	Roseateles depolymerans TB-87 (B)	Soil, fresh water/culture	PBS, PBSA, PCL, PBST, PES	20–40, 6–11	35, 7	[434,435]
Esterase and protease activity	Paenibacillus amylolyticus TB-13 (B)	Soil/culture	PLA, PBSA, PBS, PCL, PES	30, -	-, -	[436]
Esterase and amidase	-	Buffer solution	PU	37, 7	-, -	[437]
PU esterase (48 kDa)	Pseudomonas fluorescens (B)	Culture	PU	37, -	-, -	[438]
Lipase, manganese peroxidase, laccase	Penicillium brevicompactum OVR-5 (F)	Liquid medium	PVOH	28, -	30, 7	[439]
Fungal peroxidase (1.11.1.7), Laccase (1.10.3.2)	Aspergillus sp. (F)	Buffer solution	PU	30, 7	-, -	[440]
Esterase deacetylase (3.5.1.)	Comamonas sp. strain NyZ500	Activated sludge/culture	PVOH	37, -	-, -	[441]
-	Pseudomonas aeruginosa (B)	Culture	PU	37, -	-, -	[442]
-	Nocardioides OK12	Culture	PHB, PHBV	30, -	-, -	[443]
---	---	--	--------------	---------------------------	---------	-------
-	Aspergillus flavus (F)	Culture	PU	28, 6- 6.5	-, -	[444]
-	Aspergillus versicolor (F)	Culture	PBSA	30, 7.2	-, -	[445]
-	Pseudomonas chlororaphis ATCC 55729 (B)	Culture	PU (foam)	29, -	-, -	[446]
-	Aspergillus fumigatus (F), Paecilomyces farinosus (F), Fusarium solani (F), Penicillium simplicissimum (F), Penicillium minioluteum (F), Penicillium pinophilum (F), Penicillium funiculosum (F)	Activated sludge soil/farm soil	PHB	28, 37, -	-, -	[243]
-	<i>Pseudonocardia</i> sp. RM423 (A)	Culture	PLA	30, 7	-, -	[226]
-	Fusarium solani (F), Candida ethanolica (F)	Compost, Soil	PU	25, 45	-, -	[447]
-	Enterobacter sp. IBP-VN1 (B), Bacillus sp. IBP-VN2 (B), Gracilibacillus sp. IBP-VN3 (B), Enterobacter sp. IBP-VN4 (B), Enterobacter sp. IBP-VN5 (B), Enterobacter sp. IBP-VN6 (B)	Seawater/culture	PHB, PHBV	27.1- 30.4, 7.0-7.5	-, -	[448]
-	Acidovorax delafieldii (B7-7, B7-21, B7-28) (B), Streptomyces acidiscabies A2–21 (A), Streptomyces griseus A2–10 (A), Fusarium oxysporium F1–3 (F), Paecilomyces lilacinus F4–5 (F), Paecilomyces farinosus F4–7 (F)	Natural Soil/incubated artificial soil	PHBV	30, -	-, -	[449]

-	Pseudomonas aeruginosa (B)	Soil/liquid	culture	PD	LA	37, -	-, -	[279]
-	Fusarium solani WF-6 (F)	Soil/cul	ture	Pl	BS	30, -	-, -	[450]
-	Flammulina velutipes (F)	Cultu	re	PV	ОН	28, -	-, -	[451]
-	Aspergillus flavus (F), Aspergillus oryzae (F), Aspergillus parasiticus (F), Aspergillus racemosus spp. (F)	Soil/cul	ture	PH PH	IB, BV	28 - 30, 6 - 7	-, -	[452]
-	Azospirillum brasilense BCRC 12270 (B)	Liquid c	ulture	PBSA	30,	7.0 -,		[453]
-	Aspergillus fumigatus (F)	Compost/cult	zure media	PCL	23, 30, 5	25, -, - 37, .5	. []
-	Aspergillus fumigatus (F) strain NKCM1706	Soil/culture	PBS, PBSA, PHB, PC	PES, CL	30, 7	30, -		[455]
-	Leptothrix sp. TB-71 (B)	Culture nutrient broth	PBST, PB	AT	30, -	-, -		[456]
-	Burkholderia cepacia (B)	Culture	PLLA		35, 7	-, -		[457]
-	Bacillus pumilus strain 1-A (B)	Soil/Culture	PBSA, PBS,	PCL	30, 7.0	-, -		[458]
-	Bacillus sp. JY14 (B)	Marine/culture	PHB, PH	BV	30, -	-, -		[459]
-	<i>Pseudomonas</i> sp. (B)	Marine water/culture	PCL		25, -	-, -		[460]
-	Actinomadura AF-555 (A)	Soil/culture	PHBV		37, -	-, -		[269]

- Trichoderma viride (F)	Soil/liquid culture	PLA	28, -	-, -	[461]
- Chryseobacterium S1 (B), Sphingobacterium S2 (B), Pseudomonas aeruginosa (S3, S4) (B)	Compost/liquid culture	PLA	30, 7.2	-, -	[462]
- Amycolatopsis sp. (SST, SNC, SO1.2, SO1.1) (A)	Soil/basal medium	PLLA	30, 7	-, -	[463]
- Amycolatopsis sp. (A)	Culture	PLLA, PCL, PHB	30, 7.3	-, -	[464]
- <i>Amycolatopsis</i> sp strain 3118 (A)	Soil/liquid medium	PLLA	(30, 37, 43, 48), 7.0	43, 7.0	[465]
- <i>Amycolatopsis</i> sp. strain HT-32 (A)	Soil/liquid culture	PLLA	30, 7.0	-, -	[466]
- <i>Amycolatopsis</i> sp. strain KT-s-9 (A)	Soil/liquid medium	PLLA	30, -	-, -	[467]
 Acidovorax facilis (B), Varivorax paradoxus (B), Pseudomonas syringae (B), Comamonas testosteroni (B), Cytophaga jhonsonae (B), Bacillus megaterium (B), Bacillus polymyxia (B), Streptomyces spp. (B), Aspergillus fumigatus (F), Paecilomyces marquandii (F), Penicillium daleae (F), Penicillium simplicissimum (F), Penicillium ochrochloron (F), Penicillium adametzii (F), Penicillium chermisimun (F), Penicillium restrictum (F), Acremonium sp. (F) 	Soil/incubated	PHB, PHBV	(15, 28, 40), (3.5, 3.9, 6.3, 6.5, 7.1)	-, -	[172]

 Acinetobacter calcoaceticus, Arthrobacter artocyaneus, Bacillus aerophilus, Bacillus megaterium, Bacillus sp., Brevibacillus agri, Brevibacillus invocatus, Chromobacterium violaceum, Cupriavidus gilardii, Mycobacterium fortuitum, Ochrobactrum anthropi, Staphylococcus arlettae, Staphylococcus haemoliticus, Staphylococcus pasteuri, Pseudomonas acephalitica, Rodococcus equi, Bacillus cereus, Bacillus megaterium, Bacillus mycoides, B. agri, Gordonia 	Soil	PHB, PHBV	(26– 31), -	-, -	[468]
terrari, Microbacterium paraoxydans, Burkholderia sp, Streptomyces, Mycobacterium spp, Nocardiopsis, Gongronella butleri, Penicillium, Acremonium recifei, Paecilomyces lilacinus, Trichoderma pseudokoningii,					
- <i>Amycolatopsis thailandensis</i> strain CMU-PLA07 ^T (A)	Soil/liquid culture	PLLA	30, -	-, -	[469]
- Bacillus pumilus B12 (B)	Soil/minimal salt medium agar	PLA	30, -	-, -	[470]
- Kibdelosporangium aridum (B)	Solid/liquid culture	PLLA	$30, \\ 6.6- \\ 7.8$	-, -	[471]
 Lentzea (B), Saccharothrix (A), Amycolaptosis (B), Kibdelosporangium (B), Streptoalloteichus (B) 	Culture	PLLA	30, 7	-, -	[472]
- Pseudonocardia alni AS4.1531 ^T (A)	Soil	PLA	30, -	-, -	[473]
- Saccharothrix waywayandensis (A)	Culture	PLLA	30, 7	-, -	[474]
- Tritirachium album ATCC 22563 (F)	Liquid culture with gelatin	PLLA	30, -	-, -	[475]

-	Parengyodontium (F), Aspergillus (F), Penicillium (F), Fusarium (F)	Soil/agar medium	PLLA, PCL	25, 7.0, 6.0	-, -	[476]
-	Stenotrophomonas maltophilia LB 2-3 (B)	Compost/sturm test	PLLA exposed to UV irradiation	37, 7	-, -	[242]
-	Mortierella sp. (F), Doratomyces microsporus (F), Fusarium solani (F), Fennellomyces sp. (F), Aspergillus fumigatus (F), Verticillium sp. (F), Lecanicillium saksenae (F), Cladosporium sp. (F), Trichoderma sp. (F)	Compost, soil	PLLA	25, 7.2	-, -	[477]
-	Bordetella petrii PLA-3 (B)	Compost	PLLA	30, 37, 7.0	-, -	[239]
-	Flammulina velutipes (F)	Quartz sand/culture	PVOH	28, -	-, -	[451]
-	Bacillus cereus RA 23 (B)	Oil sludge/culture	PVOH	30, 7.0	28, 7	[478]
-	Bacillus sp. (B), Curtobacterium sp. (B)	Sewage sludge/culture	PVOH	$\begin{array}{c} 35,\\ 8.0 \end{array}$	-, -	[479]
-	<i>Eutypella</i> sp. BJ (F)	Soil compost/culture	PVOH	30, -	-, -	[480]
-	Geomyces pannorum (F), Phoma sp. (F)	Soil/solid culture	PU	< 25, 5.5, 6.7	-, -	[481]
-	Geomyces sp. B10I (F), Fusarium sp. B3'M (F), Sclerotinia sp. B11IV (F)	Antarctic soil/liquid culture	PCL, PBS	(14, 20, 28), -	-, -	[282]

2.15 Extracellular enzymes

Figure 2.13 depicts the main extracellular enzymes reported for depolymerization of aliphatic and aliphatic/aromatic polyesters, PUs derived from ester, where the ester bond cleavage is considered as the rate-determining step, [104]; and PVOH. These enzymes belong to the esterase (EC: 3.1) and peptidase (EC: 3.4) groups of the main group hydrolases (EC: 3); and oxidoreductases (EC 1).

Enzymes like cutinases, esterases, lipases, and PHA/PHB depolymerases are the main extracellular enzymes for enzymatic hydrolysis of the ester group and belong to the α/β hydrolase family that are structurally similar but with diverse functionality [173,482,483]. The natural activity of the esterase group of enzymes is the hydrolysis of lipids. Proteases are the main group for enzymatic degradation of the peptidase group. For polyurethanes, various esterases, proteases, amidases (EC: 3.5.1.4), and ureases (EC: 3.5.1.5) also have been reported to induce enzymatic degradation. In this case, esterases are involved in ester scission, and ureases are more inclined to scission of urethane bonds and are more resistant to chemical and enzymatic hydrolysis [101]. In the case of PVOH, and also some PU, an oxidative pathway prior to the hydrolytic enzymatic degradation has been reported, and the main extracellular enzymes are the oxidoreductases (EC: 1).



Figure 2.13 Classification of the main extracellular enzymes reported for enzymatic activity of aliphatic and aliphatic/aromatic polyesters, PUs derived from esters, and PVOH. The numbers in parentheses are the enzyme codes according to the Enzyme Commission (EC) nomenclature [102].

2.15.1 Carboxylesterases

In general, carboxylesterases (3.1.1.1) are reported as esterases, creating some confusion in the literature since the main group classification (3.1) is esterases. Since the natural function of esterases is the hydrolysis of lipids, for polymer attack they need a hydrophobic surface to be activated for scission of ester bonds. Carboxylesterases, in general, act hydrolyzing short-chains (C < 10) and present a lid domain that covers the active site. The lid domain (present also in lipases), when binding to the hydrophobic substrate, opens the active site to promote the catalysis. The lid domain structure is important since some differences can determine the specificity of the enzymes toward some substrates. Disulfide bonds is not present in carboxylesterases [484]. Hajighasemi et al. reported the action of carboxylesterases from *Alcanivorax borkumensis* and *Rhodopseudomonas palustris* on PLA and other polyesters; the enzymatic endo and exo activity resulted in the production of oligomers, dimers, and monomers of PDLLA but did not show activity for PDLA or PLLA [292].

2.15.2 Lipases

Lipases (3.1.1.3) are water-soluble extracellular enzymes reported to show enzymatic hydrolysis activity for several biodegradable polymers such as PLA (PLLA and PDLA), PCL, PBS, PBSA, PBAT, PBA, PEA, and PU esters. The typical structure of lipases is a protein structure covered by a lid-like structure. Similar to carboxylesterases, lipases need a hydrophobic surface to be activated since its natural function is the hydrolysis of lipids; increased lipase activity is observed when a hydrophobic substrate starts to form an emulsion due to its contact with a hydrophilic aqueous medium [482,485]. However, the difference respect to carboxylesterases is that lipases prefer to break down long chains (C > 10). Lipases are unable to hydrolyze ester bonds in intermediates that become water soluble [111,173]. However, Rizzarelli and Impallomeni reported some ability of lipases to hydrolyze dissolved esters in water solution [486]. Such findings indicate that the nature of the polymer could be more important than the stereo chemistry in the vicinity of the ester bond for substrate preference by enzymes with lid-like structures, such as lipases. Also, some works have reported that lipases act preferentially by random chain scission, showing an endo-type behavior where M_w reduction is highly affected in comparison to end chain scission [487].

In the case of lipases, the active site is found in a deep cavity of the protein structure. This is shielded by a lid-like α -helical structure that is reoriented when in contact with the substrate. The degree of freedom of polymer chains to move is a key factor in controlling the hydrolytic depolymerization of polyesters. This mobility ensures that the polymer chain can fold itself and fit in the active site of the lipase enzyme to carry out the depolymerization [482]. Hence, polyesters must be mobile enough to reach the active site of the lipase, making thermal and conformational properties other key factors for depolymerization since exposure temperature controls the mobility of polymer chains [173]. In general, lipases require a hydrophobic surface to reach full hydrolytic activity. For this reason, lipases are not likely to be observed developing high enzymatic activity for the PHAs family of aliphatic polyesters. The molecular weight of lipases from bacteria has been reported in the range of 20 to 77 kDa [482], thus their size allows activity on the surface of polymers.

2.15.3 Cutinases

Cutinases (3.1.1.74) are hydrolytic enzymes considered the smallest members of the α/β hydrolase superfamily (20–25 kDa) [488,489]. Enzymatic activity for cutinases has been reported for several biodegradable polymers. Cutinases are mainly produced and released by fungal pathogens and are able to degrade the polyester cutin, a natural crosslinked lipid polymer composed of n-C₁₇ and n-C₁₈ hydroxy and epoxy fatty acids, present in plant cell

walls and insoluble in water. However, some bacteria can also produce cutinases [490]. Cutinases are able to show enzymatic activity without needing interfacial activation like lipases and are capable of being active in both soluble and emulsified substrates [490], primarily due to the absence of the hydrophobic lid that covers the active site. Furthermore, the active site of cutinases is considered large enough to locate and catalyze even high M_w polyesters [491]. Cutinases act mostly against aliphatic polyesters; however, results for aliphatic-aromatic polyesters are scarce. A comparative study of five extracellular cutinases released by five species of microorganisms found that some cutinases are more stable and have higher activity towards polymer substrates than others; the higher stability and efficiency were related to additional disulfide bond formation [294]. In general, the presence of covalent disulfide bonds and neutral charge in the crowning area of the active site provides extra stability to the tertiary structure by linking regions of proteins. The presence of disulfide bonds in a cutinase was also reported by Liu et al. [305], and together with a favored catalytic triad resulted in improved activity, enhanced thermostability, and higher activity towards PCL. A cutinase (21.6 kDa) from the fungus Aspergillus oryzae was able to degrade PBS and PBSA and also showed low activity for PLA [183]. Furthermore, PCL was reported to be an optimal substrate for cutinases [303].

A study on the effect of pH on the surface charge of the area around the active site of cutinases reported that the active site becomes more positive as pH decreases from alkaline to acidic values, resulting in lower activity towards polymers such as PCL [294]. Electrostatic surface potentials generated by charged residues affect the enzyme/substrate interaction, transition stage stabilization, and efficiency during the product release stage [294]. Similar results were reported from the interaction of cutinase and PBS; the release of acidic monomers from PBS affected the pH and the activity of the cutinase, lowering the degradation rate of the PBS films [296,299]. The presence of the cofactors Ca²⁺, Na⁺, and K⁺

increased the activity of cutinases towards polymers such as PCL, PBS, and PBSA; however, the cofactors Mg^{2+} or Zn^{2+} did not show a significant effect or significantly inhibited the activity of the enzymes [295,297,298].

2.15.4 PHA, PHB depolymerases

PHA and PHB depolymerases (3.1.1.75 and 3.1.1.76) are produced by microorganisms and accumulate within the cells as an intracellular carbon and energy storage. Thus, they can undergo enzymatic degradation by functioning intra or/and extracellular. PHB depolymerases (3.1.1.75) show activity against short-chain length PHAs as PHB, PHV, and PHBV, while others (3.1.1.76) show more depolymerization of mediumchain length PHAs [489]. The primary structure of PHA depolymerases is formed by two functionally domains, a catalytic domain and a substrate binding domain, and is activated by the presence of Ca^{+2} and Mg^{+2} or inhibited by Cu^{+2} , Fe^{+2} , Mn^{+2} , and Hg^{+2} [489]. Inhibition of enzymatic hydrolysis due to the presence of detergents highlights the likely presence of a hydrophobic region near the active site of PHA depolymerases [489]. Furthermore, PHA depolymerases are reported to have exo and endo behavior since they were able to release monomers and oligomers [489]. The extended presence of hyphae due to the colonization of fungi on PHBV surface has been reported as evidence of enzymatic degradation by extracellular PHAs depolymerases released by fungus [449]. In terms of the structure, some PHA depolymerases are reported to belong to the serine esterases group due to the presence of lipase boxes [386]. Even though PHA depolymerases are specific for PHAs, enzymatic activity has been reported also for other polyesters.

The presence of additional carbon sources may reduce the enzymatic activity against polymers. For example, the reduction of PHA depolymerase produced by *Aspergillus* sp. showed a repression apparently influenced by the type of carbon source added to the media, which was indicative of a regulated behavior as a function of the available carbon source [452]. This finding is in accordance with the hypothesis that when abundant labile nutrients are present, the decomposition of more recalcitrant compounds is inhibited [107]

2.15.5 Peptidases (Proteinase K and Protease)

Peptidases (3.4), a group of enzymes acting on peptide bonds, are also commonly called proteases, generating some confusion in the literature. Peptidases or proteases hydrolyze peptide bonds that link amino acids in a protein. For example, Proteinase K (3.4.21.64) and proteases (3.4.21.112) belongs to the serine endo peptidases (3.4.21), enzymes that preferentially catalyze bond scission in the middle of the substrate chain. Also, it has been reported that enzymes belonging to the serine endo peptidases are able to hydrolyze polyesters like PLA. Proteinase K and proteases have been identified for major enzymatic activity on PLA. Lim et al. [293] reported the ability of Proteinase K to depolymerize PES, PEA, PBS, PBSA, and PCL but at lower levels of enzymatic activity than on PLA. More specifically, Proteinase K showed a higher enzymatic activity for PLLA (amorphous preferentially) than for PDLA and PDLLA.

The activity of these classes of enzymes towards PLA is still not fully understood in the sense that these enzymes are more prone to attack the scission of peptide bonds. Tokiwa and Jarerat [492] concluded that enzymes showing activity on PLA belong to the peptidases or protease-type group, and these enzymes are able to recognize the repeated L-lactic acid unit of PLA as the natural homologue L-alanine unit of silk fibroin, a natural protein present in silk. Later work by Lim et al. [293] reported the enzymatic ability of serine proteases on PLA, PHB, PES, PEA, PBS, PBSA, and PCL; in particular, alpha-chymotrypsin, a mammalian enzyme, showed preferential activity for PLA. In studies on PLA biodegradability, the incorporation of agents to the media, such as silk fibroin or gelatin, as a nitrogen source to induce production of protease, has resulted in increased enzymatic activity for PLA since proteases are more prone to interact with peptide bonds [424].

2.15.6 Amidases and ureases

In addition to esterases and proteases, PUs derived from esters can be enzymatically degraded by amidases (3.5.1.4) and ureases (3.5.1.5). Amidases attack the amide groups, proteases can attack amide and urethane bonds, esterases attack the ester bonds, and ureases catalyze the hydrolysis attack of the urea groups. However, the information in terms of amidases and ureases showing enzymatic activity towards PUs derived from esters in mesophilic environments is scarce [493–496].

2.15.7 Oxidoreductases PU and PVOH-oxidases

Oxidoreductases (1) have shown activity for PVOH and PU within the groups EC 1.1, EC 1.10, and EC 1.11. More specifically 1.1.3 (with O₂ as electron acceptor), polyvinyl-alcohol oxidase (1.1.3.30) or dehydrogenase have shown enzymatic activity on PVOH; these are enzymes that can act extra or intracellular. Laccase (EC 1.10.3.2) has been reported to show enzymatic activity against both PVOH and PU.

2.16 Biosurfactants and synthetic surfactants

Biosurfactants are amphipathic molecules with the capacity of reducing surface and interfacial tension between liquids, solids, and gases. They contain both hydrophilic and hydrophobic moieties that can improve the interaction between phases of different degree of polarity and hydrogen bonding [497,498]. Microorganisms are capable of synthesize and release biosurfactants such as glycolipids and phospholipids to emulsify the substrate and stimulate other functions as extracellular enzymatic activity [499].

Hydrophobins are a type of amphipathic surfactant secreted by fungi microorganisms, that besides other functions, they attach to the surface of biodegradable polymers and stimulate the hydrolysis by extracellular enzymes. They present a dual behavior with hydrophobic and hydrophilic parts (amphipathic proteins), and are adsorbed to the surface of the polymer, condensing, and stimulating its enzymatic hydrolysis by recruiting extracellular enzymes [500,501]. Hydrophobins are important to support the growth of fungal aerial structures (hyphae) and conidiospores by playing an important role for fungal adhesion to hydrophobic surfaces, development of a protective surface coating, and reduction of water tension [500,501].

Synthetic commercial surfactants are widely used during studies of extracellular enzymatic activity on polymers and are classified in function of the nature of their polar grouping. It has been reported the use of ionic and nonionic surfactants. Some common synthetic nonionic surfactants are added to culture studies for emulsification, like commercials ones as polysorbate 80 and polyoxyethylene type, to increase microbial activity on the polymer surface by increasing the hydrophilicity of the surface [109].

Interaction between surfactants and enzymes is still a subject of exploration. Holmberg mentioned that probably nonionic surfactant are more benign than ionic surfactants. The way in what ionic surfactants interact with enzymes can introduce significant changes in the conformational structure of the enzyme [502]. Detailed discussion about bio and commercial surfactants can be found elsewhere [497,498,502,503].

2.17 Polymers susceptible to biodegradation

The main group of biodegradable polymers susceptible to biodegradation in the mesophilic range are the aliphatic and aliphatic-aromatic polyesters; besides that, the soft segment of PUs derived from esters and PVOH are also considered biodegradable to some extent. Commercialized cellulose and starch-derived polymers, which are bio-based and naturally biodegradable, are also important to consider when discussing biodegradable polymers. **Figure 2.14** shows the tentative pathways of the most common bio- or fossil-based polymers reported to undergo depolymerization by specific microorganisms. These polymers are reviewed individually in this section.



Figure 2.14 Main tentative biodegradation pathways for biodegradable polymers in aerobic conditions.

2.17.1 Cellulose

Cellulose is a linear homopolymer of D-glucose units joined by β -1,4 glycosidic linkages, with a degree of polymerization ranging from several hundreds to over 10,000 [504]. Each glucose molecule is upside down in relation to the neighboring glucose molecule so that the repeating unit is cellobiose, consisting of two glucose molecules linked by a β -1,4 glycosidic bond. The fibrils of cellulose can have crystalline and amorphous regions. Depending on the origin and treatment, the crystallinity of cellulose can vary from fully amorphous to fully crystalline. Higher crystallinity makes cellulose resistant to chemical attacks. In the secondary wall of plant cells, cellulose forms several sheets organized as parallel microfibrils. These microfibrils are embedded in the matrix of hemicellulose and lignin. Pure cellulose is available in several forms, such as cotton and filter paper. Before 1950, cellulose-based polymers were one of the most important groups of polymers. Cellulose nitrate, the oldest plastic, was produced by replacing nitrates on all three hydroxyl groups of the cellulose glucose units. Several other cellulose ether and ester thermoplastics, such as cellulose acetate and cellulose butyrate, have been produced through the years [505]. The main cellulose ethers are: methylcellulose (MC) – non-thermoplastic, water-soluble with high O_2 barrier, generally used as filler and thickener agent; carboxymethyl cellulose (CMC) – hydrophilic, non-thermoplastic, generally used as a viscosity modifier and thickener; hydroxypropyl cellulose (HPC) – thermoplastic, with water barrier and grease resistance, generally used for coatings, and as binder and thickener; and hydroxypropyl methyl cellulose (HPMC) – non-thermoplastic, non-heat-sealable, generally used for coating purposes. Cellulose esters are thermoplastic, and they are produced by the reaction of organic or inorganic acid substituting the hydroxyls of the glucose unit. The prominent cellulose esters include cellulose acetate (CA), which is thermoplastic, used for molding and extrusion, and

can exist in several forms such as cellulose acetate butyrate (CAB), cellulose acetate propionate (CAP), and cellulose triacetate (CTA) [505].

Cellulolytic and non-cellulolytic mixed populations of microorganisms are present where cellulosic waste is present. These microorganisms interact synergistically to complete the biodegradation of cellulose, which is ultimately converted to CO₂ and H₂O in aerobic environments through the pathways shown in **Figure 2.14**, and to CO₂, CH₄, and H₂O in anaerobic environments.

Cellulose, as well as starch, is enzymatically hydrolyzed to glucose by extracellular enzymes, which are produced by bacteria and fungi (**Figure 2.15**). Natural polymers, such as cellulose and starch, are mostly attacked for enzymatic hydrolysis by cellulases and α/β amylases. In addition, oxidoreductases have been identified that can act prior to hydrolytic enzymes on cellulose [506].



Figure 2.15 Pathway for enzymatic degradation, bioassimilation and mineralization of the natural polymers, cellulose and starch.

After glucose is produced, glycolysis converts the glucose to pyruvic acid, which acts as the precursor for the TCA cycle. Glucose, together with the adenosine triphosphate (ATP), which is the molecule providing the energy source in the cell plus NAD⁺, and inorganic phosphate, breaks down into two pyruvates. In the pyruvic acid cycle, **Figure 2.16**, three main steps take place. First, a carbonyl group is removed from pyruvic acid, releasing CO_2 to the surrounding media and resulting in a two-carbon hydroxyethyl group bound to the enzyme pyruvate dehydrogenase. Second, the hydroxyethyl group is oxidized to an acetyl group and the electrons are picked up by the NAD⁺ (nicotinamide adenine dinucleotide), forming NADH. This electron will later be used by the cell to create energy through the ATP process. And third, the enzyme bound to the acetyl group is transferred to CoA, producing a molecule of acetyl CoA. This molecule is then further converted through the TCA cycle [86].



Figure 2.16 Pyruvic acid to Acetyl-CoA reaction pathway. Adapted from [86].

Cellulose biodegradation occurs primarily by cellulolytic microorganisms belonging to the bacteria and fungi. Aerobic biodegradation of cellulose occurs mostly by cellulolytic bacteria; several species identified in the genera *Cellulomona*, *Pseudomona*, *Thermomonospora*, and *Microbispora* have been shown to biodegrade cellulose [507].

Cellulose undergoes biodegradation in several environments. Amorphous forms of cellulose are used as positive controls for biodegradation studies due to their negligible chemical hydrolysis and rapid enzymatic hydrolysis rates and assimilation by microorganisms. In thermophilic and mesophilic environments, such as industrial composting or soil biodegradation, cellulose is widely used as a positive control, as stated in ASTM and ISO standards (**Table 2.8**). In marine environments, mineralization of cellulose powder was reported to reach ~95% after 450 days of testing at 25 °C [233], indicative of its high biodegradability in aquatic environments. Anunciado et al. [217] used cellulose in the form of a mulch paper, instead of powder, as a positive control in soil and composting conditions; after 365 days of testing mineralization values were in the range of 50 to 80% for samples in soil at 27 °C.

2.17.2 Starch

Low-cost starch, mainly obtained from crops not intended for human consumption, is a bio-based material that can be blended with other polymers to produce novel bio-based and biodegradable blends. Starch consists of two main molecules making up the constitutional unit: amylose (linear) and amylopectin (branched). Starches with high amylose content have been used to produce suitable blends and to improve the thermal, mechanical, and gas barrier properties of the resulting blends [508–512]. The T_g of pure starch is above its decomposition temperature, meaning the material does not soften and flow. To make it processable, starch needs to be combined with plasticizers such as glycerol, poly(ethylene glycol) (PEG), or sorbitol to obtain thermoplastic starch (TPS). The starch granules are plasticized by using plasticizers under heating, which provides a viscous melt that can then be processed using traditional methods such as extrusion foaming and injection molding [513]. TPS is highly hydrophilic, resulting in leaching of plasticizer during storage and poor dimensional stability and mechanical properties with time [514]. However, TPS can be used to blend with other bio-based polymers, improving O₂ barrier and elongation at break due to the presence of glycerol [515–522]. Since the properties of TPS by itself are not sufficient for producing polymeric structures for some applications, the possibility of blending TPS with other polymers to improve its mechanical and water barrier properties has opened a wide field for the development of novel TPS blends, with reactive functionalization as one of the suitable methods to enhance the compatibilization of TPS [512].

Since starch is sensitive to water, starch or the portion of blends containing TPS will mostly hydrolyze by enzymatic hydrolysis to glucose. The main extracellular enzymes involved during enzymatic degradation of starches are α/β -amylases (**Figure 2.14**). The general pathways for biodegradation and bioassimilation/mineralization of starch are shown in Figure 2.15 and Figure 2.16, respectively.

Starch, TPS, or TPS blends with other biodegradable polymers have shown high production of CO₂, which is indicative of the high biodegradability of TPS even in mesophilic environments. Ho and Pometto [118] reported values of mineralization of ~70% for starch at 28 at 40 °C in a soil environment under laboratory conditions after 180 days of testing. The main characteristic was rapid initial degradation at 40 °C, with a negligible abiotic phase of degradation, reaching the plateau stage at around day 60; lower activity was observed at 28 °C, reaching the plateau stage at around day 100.

2.17.3 Poly(glycolic acid) – PGA

PGA, the simplest aliphatic polyester, is a biodegradable and biocompatible thermoplastic, extensively used for many decades in the medical field for implants [523]. PGA can be synthesized using many mechanisms. Direct polycondensation polymerization of glycolic acid results in low M_w PGA ($M_w < 50$ kDa). Ring opening polymerization of glycolic acid results in high M_w PGA ($M_w > 50$ kDa). Solid-state polycondensation is used to increase the M_w by increasing the polymer chain lengths in the absence of heat and O₂, by constant removal of byproducts using inert gas or under vacuum [524,525].

PGA has a T_g in the range of 35 to 40 °C and T_m between 220 to 230 °C (**Table 2.5**) [526]. PGA displays good gas barrier properties due to its crystalline and stereochemistry structure [523]. PGA is also resistant to most organic solvents. In addition, the high density of c. 1.53 g/cm³ awards PGA good mechanical properties compared with other biodegradable polymers; however, the high cost associated with the PGA production process has hampered its entry into the consumer market as compared to other biodegradable polymers [523]. In general, PGA is blended with other polymers to improve their properties. For example, when PGA is blended with PLA, the result is better mechanical properties and improved flexural modulus of the PLA/PGA blend [527]. Due to its high O₂ and H₂O barrier properties, PGA can be used in packaging of products sensitive to O₂ [528]. PGA is widely used in biomedical applications such as sutures, drug delivery, and tissue engineering [529].

PGA degradation starts by abiotic degradation, and chemical hydrolysis is by a nonspecific chain scission of the ester backbone, with bulk erosion as the dominant mechanism (**Figure 2.14**) [50,523]. Therefore, water diffusion activated by temperature plays a crucial role in the initial hydrolysis of the ester backbone. The absence of asymmetrical methyl groups turns PGA more hydrophilic than PLA, increasing its bulk chemical hydrolysis rate. Currently, there is limited published information on PGA depolymerization by enzymatic activity in the mesophilic range in open environments. Since PGA has been used mainly for biomedical applications, most of the biodegradation data is from *in vivo* studies at 37 °C. Extracellular enzymes like esterases have been reported to have enzymatic activity on PGA sutures [310]. After the initial chemical and enzymatic hydrolysis, PGA is degraded into small oligomers and glycolic acid, which can be bioassimilated and oxidized to become a substrate for the TCA cycle, as shown in **Figure 2.17**.





In terms of CO₂ evolution and mineralization studies, biodegradation of PGA in a marine environment at around 30 °C, which is high for marine environments, showed a longer lag phase than for cellulose, but 75% mineralization was reached at 28 days for both PGA and cellulose [523]. At thermophilic conditions in a simulated industrial composting environment at 58 °C, PGA showed lower mineralization than cellulose; 70% mineralization was reached at around day 40 for cellulose and at around day 70 for PGA [523].

2.17.4 Poly(lactic acid) – PLA

PLA, a biodegradable aliphatic polyester, is a widely used alternative for conventional fossil-based plastics. In addition to PLA being biocompatible, biodegradable (compostable), its production from renewable resources results in energy savings and lower greenhouse gas (GHG) emissions [530]. The building block for PLA is lactic acid or lactide, which is derived from the fermentation of glucose obtained from varied sources such as corn and sugar cane. Lactic acid has two enantiomers: L-lactic and D-lactic acid [531]. Lactide can be produced in three stereochemical configurations: L, L-lactide; L, D-lactide, and D, D-lactide. High M_w PLA is obtained by ring opening polymerization of the different lactides and polycondensation of low M_w lactic acid [532,533]. PLA presents acceptable thermal, mechanical and barrier properties and its main applications include food and medical product packaging, medical devices, fibers, textiles, plasticulture, and automotive parts [532]. The ratio of L-lactic and D-lactic acid in a final PLA formulation plays a crucial role in its final properties and degradation rate [532–534].

The hydrolysable ester bonds in the backbone of PLA structure (**Table 2.5**) makes it susceptible to chemical hydrolysis. Several mechanistic, phenomenological, and probabilistic models have been developed for PLA and can be extended to other aliphatic polyesters, explaining how diffusion and geometric properties can modify the pathways and incentivize one or the other mechanism [75]. The mechanism proceeds in different stages, starting with water diffusion into the material, followed by the degradation of amorphous regions. After degradation of the amorphous regions, the random chain scission and cleavage of ester bonds results in the release of soluble oligomers and monomers [62], which can be used as substrates for bioassimilation (**Figure 2.18**). The hydrolysis rate of PLA, as well as other polyesters, is highly dependent on temperature (below or above T_g), pH, and several other properties of the polymer such as M_w and crystallinity, as reviewed elsewhere [64]. In the absence of other factors accelerating other mechanisms, chemical hydrolysis is the most important mechanism in the mesophilic range for the abiotic process.

The degradation activity of PLA by microorganisms has been monitored by different methods and correlated to different biodegradation stages. The crystal structure change or biofilm formation for PLA degraded in a compost environment was observed using SEM [239]. Weight loss indicating depolymerization of PLA was measured by size exclusion chromatography [424], the degree of biofragmentation of PLA fibers was monitored by X-ray diffraction (XRD) [290], and the generation of lactic acid was detected using an enzymatic bioanalysis kit [475].

The biotic degradation stage implies enzymatic activity and microbial assimilation. The enzymatic degradation of PLA involves interaction of the polymer with a reagent, such as water, in the hydrolysis reaction. Hydrolases, such as proteases and esterases, catalyze the hydrolysis reactions. When lactic acid becomes available for bioassimilation, it is transported through the semipermeable membrane and is oxidized to pyruvic acid through a dehydrogenization reaction, which then follows the pyruvic acid pathway (**Figure 2.18**), as previously described.



Figure 2.18 Biodegradation pathway for PLA in aerobic conditions.

Various bacteria, fungi and actinomycetes strains have been identified as having some ability to degrade PLA in different forms such as pellet, film, powder, and sheet. These microorganisms were isolated from different environments, such as soil, compost, wastewater sludge, by enrichment culture media, while some were procured from research facilities. The extracellular enzymes secreted by these microorganisms have been reported to preferentially degrade the amorphous regions of PLA, since the backbone chains are highly disordered and have higher mobility as compared to the crystalline region. This flexibility and mobility aids in the binding of the backbone chain to ensure a fit into the active site of the enzyme [323]. The extracellular enzymatic activity efficiency is dependent on the type of PLA (PLLA, PDLA, or PDLLA) as well as the temperature, crystallinity, and M_w of the PLA [239,535].

The enzymatic degradation of PLA involves the hydrolases, with esterases (3.1) and peptidases (3.4) as the main groups of enzymes. Carboxyl esterases ABO2449 and RPA1511 (3.1.1.1) have been reported to hydrolyze PLLA and PDLA with the highest activity for ABO2449 at 30 to 37 °C and for RPA1511 at 55 to 60 °C [292]. The analysis of the hydrolysis suggested that, like other hydrolases (e.g., nucleases and proteases) that are active in depolymerizing polymeric substrates, these enzymes can exhibit both exo and endo-esterase types of cleavage [292]. Several esterases (3.1) able to degrade PLA such as lipases, cutinases, and carboxyl esterases. Peptidases (3.4) have been reported to be able to degrade PLA in culture media. For example, Proteinase K (3.4.21.64) has been shown to be efficient during scission of polymer chains, favoring the hydrolysis of the amorphous region of PLLA [536]. The enzymatic degradation of PLA revealed the preferential activity of proteases for PLLA and for PDLA of lipase/cutinase/esterase type. The enzymatic activity of lipase on PLLA was affected by the addition of Na⁺ and K⁺ that increased the activity. However, Zn⁺², Mg⁺², Cu⁺², and Fe⁺² showed inhibition of the enzymatic activity [341]. Furthermore, the presence of anionic surfactant showed a significant inhibition of Proteinase K activity towards PLLA [293]. However, the presence of the same anionic surfactant showed a dual behavior during

enzymatic activity of α/β hydrolases on PDLLA, facilitating the binding of carboxylesterases on PDLLA and also reducing the hydrolytic activity by lipase-like esterases [292,537]. Nonionic surfactant also reduced the enzymatic activity towards PDLLA by lipase-like esterases [537]. Several studies have reported the importance of factors as stereochemistry, crystallinity, and hydrophilicity on the enzymatic degradation of PLLA, PDLA, and PDLLA mostly by action of Proteinase K [538–543].

Some biodegradation studies for PLA in the mesophilic range and different environments have reported high values of mineralization while other studies have reported low values of CO₂ evolution or mineralization. Ho and Pometto reported values of mineralization for three types of PLA films in soil environments under laboratory conditions; after 180 days of testing, mineralization values ranged from 10 to 45% at 28 °C and from 30 to 90% at 40 °C, depending on the film [118]. The effect of temperature also can be observed in the work of Muniyasamy et al., where mineralization values for PLA films in soil environment at c. 25 °C were a negligible 5% after 190 days of testing [225]. Biodegradation studies by Kim et al. in compost showed different mineralization values for PLA with different M_w and crystallinities after 40 days of testing [239]. The dependence of enzymatic activity on the initial M_w of PLA was evident, with mineralization values of c. 70% for low M_w PLA (5 kDa) and c. 30% for higher M_w PLA (34 kDa). The same study reported a similar trend for different levels of crystallinities at 30 °C, with reduced biodegradation rates for PLA with high crystallinity [239].

In aquatic environments, biodegradation of PLLA granules resulted in mineralization value of c. 10% after 180 days of testing at 25 and 37 °C [238]. After 50 days of testing, mineralization values had reached a similar plateau at both temperatures. Lower

mineralization values are indicative of limited chemical and enzymatic hydrolysis in aquatic conditions for PLA.

When PLA (initial M_w c. 188 kDa) powder was exposed to UV irradiation and studied by using the Sturm test and in compost for 40 days, the highest mineralization values were found in both the Sturm test (c. 20%) and compost (c. 45%) for samples treated for 8 hours; longer UV irradiation treatment times resulted in decreased mineralization values after 40 days of testing [242]. The authors stated that a Norrish reaction was not identified as the main effect for reduced biodegradation with longer UV irradiation time, leaving the presence of crosslinking as the most probable one.

Biodegradation of PLA sheets after 180 days of testing in soil at 28 °C resulted in c. 10% of mineralization [218]. Lower values obtained for PLA sheets in comparison to powder samples indicates the effect of shape and size as important factors decreasing the chemical hydrolysis and the mineralization rate.

Biostimulation and bioaugmentation of soil environments to improve PLA biodegradation under mesophilic conditions was studied by Satti et al. [216]. After 150 days of testing, improved results, with respect to natural biodegradation of PLA, were obtained for biostimulated soil with lactate and bioaugmented soil with previously isolated PLAdegrading bacteria strains. Techniques such as biostimulation and bioaugmentation to improve biotic conditions are increasingly considered as feasible alternatives to increase the biodegradation rate of polymers. In this sense, UV irradiated PLA sheets in soil, inoculated with *Pseudomonas geniculata* WS3 at 30 °C, showed maximum biodegradation values of c. 35% after 60 days of testing. However, in the case of soil non inoculated, the biodegradation was just about 15% after 60 days [220].

2.17.5 Polycaprolactone – PCL

PCL is a synthetic, aliphatic biodegradable polymer, semicrystalline in nature, and is obtained by the ring opening polymerization of caprolactone [544]. PCL has a T_g of around -60 °C and a T_m of 60 °C. Since the T_g is so low, PCL shows high molecular chain mobility due to its rubbery state [545]. Also, this low T_g provides good flexibility and malleability to PCL [544]. PCL is non-hazardous and biocompatible, so the polymer is often used in biomedical applications such as tissue engineering, drug delivery, and in the construction of scaffolds and sutures [546]. PCL displays excellent rheological and viscoelastic properties. Aside from the many listed advantages, the mechanical properties are less suited for rigid applications. The inferior mechanical advantage coupled with improved degradation rate warrants the use of fillers and incorporation of different polymers to attain the necessary mechanical properties. PCL is usually a raw material for production of polyurethanes, as polyol polyester-type [495,547].

The main abiotic degradation mechanism for PCL in the mesophilic range is chemical hydrolytic degradation through bulk erosion [548]. Furthermore, PCL can photodegrade when exposed to radiation via Norrish II reactions [549]. UV treatment also has been effective in increasing the degradation rate of PCL films, making it easier for microorganisms to attack during the biodegradation phase [550]. Due to its relatively low T_m (60 °C), PCL can undergo thermal degradation at conditions similar to the thermophilic conditions of the industrial composting process. A short abiotic lag phase was reported for PCL in home composting conditions, showing a biodegradation trend similar to readily biodegradable materials like cellulose or starch [115]. However, the initial M_w of the used PCL was low (M_w c. 50 kDa) in comparison to others polyesters evaluated such as PLA and PHAs [551].

A comparison of the hydrolysis mechanisms for PCL in water and phosphate buffer solutions revealed that, in general, enzymatic hydrolysis was faster than abiotic chemical hydrolysis in terms of mass loss, and that enzymatic hydrolysis is a surface erosion process whereas abiotic chemical hydrolysis is a bulk erosion process [338]. However, in more realworld conditions the enzymatic hydrolysis also could be affecting the chemical hydrolysis. For example, a comparison test for PCL abiotic degradation at 30 °C by *Aspergillus fumigatus* showed a different pattern; samples studied for chemical hydrolysis remained without surface changes, while samples in culture media showed an erosion pattern indicative of surface enzymatic degradation [178].

Carboxyl esterase (3.1.1.1), lipases (3.1.1.3), and cutinases (3.1.1.74) are able to degrade PCL. Also, low enzymatic activity by peptidases such as Proteinase K (3.4.21.64) has been observed. Cutinases from fungal phytopathogens are indicated as PCL depolymerases showing enzymatic activity [100]. Based on earlier works studying and identifying aerobic microorganisms able to biodegrade PCL, it has been reported that the natural polymers cutin and suberin are enzymatically degraded by lipases; since these materials are considered as natural analogous to PCL, enzymatic activity of PCL by lipases was potentially considered. Nishida et al. demonstrated that lipases are highly active in the degradation of PCL [552]. This finding was also indicative of potential microbial populations for PCL biodegradation being extensive in natural environments such as soil, home composting, and water. When enzymatic degradation of PCL by lipases and Proteinase K available in those environments was studied, lipase activity was reported but none for Proteinase K; the authors associated this result to the preferential specificity of lipases for ester bonds on hydrophobic substrates as in PCL [335].

Temperature and pH are key factors also identified in playing a main role in the degradation of PCL. The high stability of cutinases able to degrade PCL was associated with stabilization of the enzymes by neutral surfaces and additional disulfide bond formation [294]. Baker et al. compared cutinases for PCL degradation and showed that enzyme activity,

stability, and efficiency was affected by the type of microorganism that releases the extracellular enzyme and by temperature; the authors reported a similar residual activity for the enzymes at 25 °C, but reduced residual activity at 45 °C for some of the cutinases [294].

Li et al. reported the presence of 6-hydroxy-hexanoic acid instead of PCL oligomers during the enzymatic degradation of PCL by *Penicillium oxalicum*, indicative of an exo-type chain-end scission by the enzyme [354].

PCL degrading microorganisms and extracellular enzymes have been reported also in marine environments, showing relatively good activity in comparison to other aliphatic polyesters as PLA [553].

When 6-hydroxycaproic acid becomes available for bioassimilation after chemical and enzymatic depolymerization of PCL (**Figure 2.14**), it is transported through the semipermeable membrane, and then is converted to acetyl-CoA by β -oxidation of fatty acids, becoming available for the TCA cycle (**Figure 2.19**).



Figure 2.19 Tentative biodegradation pathway for PCL in aerobic conditions. Adapted from [544].

A few studies have demonstrated PCL biodegradation with production of CO₂ and mineralization at mesophilic conditions like soil or home composting. Ohtaki et al. reported a low mineralization value of c. 15% for PCL powder (M_w c. 100 kDa) in compost after 8 days of testing [138]. Modelli et al. reported ~102% mineralization for PCL in a soil environment in laboratory conditions at 22 °C after 270 days of testing [223]. Narancic et al. tested the biodegradation of PCL sheets in home composting and in marine (30 °C) and fresh water (21°C) environments [115]. In home composting, the PCL reached mineralization values of c. 90% after 180 days of testing, with a negligible abiotic degradation stage; however, PCL failed the marine (56 days) and freshwater tests, with mineralization values of c. 80 and 50%, respectively.

2.17.6 Poly(alkylene dicarboxylate)s

Poly(alkylene dicarboxylate)s are a family of biodegradable aliphatic polyesters derived from dicarboxylic acids and dihydroxy compounds [50]. This family includes PBA, PBS, PBSA, PBST, PBSe, PBSeT, PEA, PES, among others. Their general structures are presented in **Table 2.5**, and a general description of the main polymers of the family is provided here.

PBA is a biodegradable polyester that can be synthesized via polycondensation of adipic acid with 1,4-butanediol in the presence of a catalyst. Due to its low T_m (41–61 °C), PBA is generally copolymerized to obtain polyesters with improved mechanical properties. Potential applications for PBA are mainly in the medical area [50].

PBS, a biodegradable, linear, semi-crystalline, thermoplastic aliphatic polyester, is a result of the condensation polymerization of succinic acid (SA) and 1,4-butanediol (BDO). PBS can be 100% bio-based (bio-based SA and BDO), partially bio-based (bio-based SA and petrochemical BDO) or fossil-based (petrochemical SA and BDO), depending on the production route used [554]. SA is derived from maleic anhydride, which can be produced by the oxidation of butane or benzene or from the fermentation of carbohydrate sources such as glucose and starch [555]. BDO, on the other hand, can be derived via three routes: using petrochemicals, hydrogenation of SA, or fermentation of sugars [556]. PBS provides easy processability and mechanical properties comparable to LDPE and PP. The fact that PBS is flexible and not rigid and brittle like other biodegradable polymers (e.g., PLA, PBAT, and PHB) makes it a more viable and a cost-effective option for common applications [557]. The properties of PBS can be further fine-tuned for designated applications by blending with other polymers. For example, PBS/PLA blends have improved toughness and elongation at break with the help of random copolymers of poly(butylene succinate-*co*-lactic acid) as compared with neat blends [558]. PBS applications vary from food packaging, agriculture mulch films, hygiene products, fishing nets, plant pots, and coffee capsules, among others [559].

PBSA is obtained when adipic acid is added when synthesizing PBS. The addition of adipic acid decreases the crystallinity and increases the degradation rate [50]. In comparison to PBS, PBSA has a lower T_m of c. 95 °C (**Table 2.5**) and higher flexibility in terms of mechanical properties [489].

PBST is an aliphatic/aromatic polyester synthesized by direct esterification and polycondensation using titanium tetraisopropoxide as a catalyst. PBST has a potential development, and works about PBST biodegradation are limited [50].

PEA, an aliphatic polyester, is produced by the polycondensation of ethylene glycol and adipic acid or by the polycondensation of dimethyl adipate and ethylene glycol [560]. PEA has a T_g of c. -50 °C and T_m of c. 48 °C (**Table 2.4**). The polymer displays good flexibility due to the low T_g , but at the same time demonstrates low mechanical strength [561]. PEA is usually blended with other polymers. When blended with PLA, PEA helps in reducing the brittleness, improving the thermal stability, and has also been shown to increase the elongation at break compared with neat PLA [562]. PEA has application as a plasticizer, when low migration, good plasticity and better mechanical properties are desired for the copolymer blend [563].

PES is synthesized by ring opening polymerization of succinic anhydride with ethylene oxide or by polycondensation of succinic acid and ethylene glycol (**Table 2.5**). PES is highly permeable to O_2 [50].

In terms of abiotic degradation, the density of the ester bond is a factor affecting the chemical hydrolysis rate of poly(alkylene dicarboxylates). As reported for PES, a smaller ester bond reduces the hydrophilicity, affecting the overall process of hydrolysis [50]. For PBS, abiotic degradation generally occurs through chemical hydrolysis, with bulk erosion as the predominant mechanism [48]. In general, PBS is copolymerized with the aim of increasing the hydrolysis rate. For example, the addition of more hydrophilic components, such as PEG, has been reported to increase abiotic hydrolysis; however, adipic acid is the most common component added to obtain PBSA [50]. Hayase et al. reported a higher degradation of PBSA than PBS in the presence of *Bacillus pumilus*, which was attributed to the faster degradation of the adipate units [458].

Extracellular enzymes with activity for poly(alkylene dicarboxylate)s have been reported, mostly for PBA, PBS, PBSA, PEA, and PES (**Table 2.10**). In general, lipase activity data is scarce for PBA, PBS, and PBSA; however, the activity of cutinases is well reported for PBS and PBSA (**Table 2.10**). Cutinases are indicated as being more active for polyesters with chain lengths less than 10 C [489].

Fungi and bacteria have been shown to depolymerize PBS. For example, Ishii et al. [455] reported succinic acid and 1,4-butanediol as hydrolysis products due to the action of *Aspergillus fumigatus* strain NKCM1706. Li et al. [352] reported the action of an exo extracellular enzyme on PBS as an exo attack, since products identified by mass spectrometry were succinic acid and butylene succinate monomers rather than PBS oligomers. Furthermore, the presence of butylene succinate monomers and not 1,4-butanediol showed that the enzyme cut the polymer chain from the carboxyl end [352]. In the case of PBSA, 1,4-butanediol, succinic acid, and adipic acid were detected by HPLC during depolymerization by *Leptothrix* sp. TB-71 [314]. Enzymatic activity of *Rhizopus delemar* lipases against PEA

produced, besides ethylene glycol and adipic acid, PEA oligomers; indicative of an endo attack of the lipase extracellular enzyme [345].

When SA and BDO become available for bioassimilation (**Figure 2.20**) after chemical and enzymatic hydrolysis, they are transported through the semipermeable membrane and converted to succinyl-CoA, becoming available for the TCA cycle (**Figure 2.20**).



Figure 2.20 Tentative biodegradation pathway for PBS in aerobic conditions. Adapted from [564].

Some studies reported mineralization (or CO₂ production) for PBS, showing limited biodegradation in the mesophilic range, including home composting, soil, marine and freshwater environments. Narancic et al. tested the biodegradation of PBS sheets in soil, home composting, and in marine and freshwater environments, and found mineralization values lower than 20% after 365 days of testing in soil and home composting, whereas values were c. 20 and 5% in marine and freshwater after 56 days of testing, respectively [115]. However, other group evaluated PBS in powder form in soil environments at 25 and 37 °C, and mineralization values reached c. 85 and 80%, respectively, after 180 days of testing [210].

PBSe and PBSeT (films) were assessed in a marine environment at 25 °C under laboratory conditions, and similar mineralization values of c. 90% were obtained after 360 days of testing with stirring and without stirring the media containing the samples [211]. PBSe and PBSeT (films) in soil at 25 °C reached mineralization values of c. 90% after 360 days of testing [209]. Furthermore, when PBSe (powder) was evaluated in soil at 28 °C, mineralization improved for samples with higher available surface area; at day 140, the mineralization values were c. 55% for samples with 33 cm² surface area, and 80 to 95% for samples with 89 to 1650 cm² surface area [189].

PBSA films was evaluated in compost at 25 °C with values of mineralization of \sim 70% after 55 days of testing, with an abiotic phase duration of \sim 1 week [234].

2.17.7 Polyhydroxyalkanoates – PHAs

PHAs comprise a family of naturally occurring biodegradable aliphatic polyesters produced due to the fermentation of carbohydrate sources, such as sugar and lipids, by the action of a broad range of microorganisms [531,565]. PHAs are synthesized and stored as an intracellular energy resource for later metabolism under conditions of scarcity. PHAs can be classified according to the length of the side chain. The most common are short-chain-length PHAs, with 3 to 5 carbon atoms in their monomeric structure [50]. Poly(3-hydroxybutyrate) (PHB), poly(hydroxyvalerate) (PHV), and the copolymer poly(hydroxy-butyrate-*co*-valerate) (PHBV) are the most common, and PHB is abundantly manufactured. PHAs can be derived from renewable and non-renewable sources [566], and have excellent barrier and good thermo-mechanical properties [567]. However, drawbacks for PHAs in conventional thermal processing include a narrow processing window and high production costs. To improve the processability and ensure large-scale production, PHAs are often blended with other polymers. PHAs have been commonly used for cutlery, trays, food packaging, and cosmetics, and in the development of medical devices, surgical sutures, implants and tissue engineering, among others [568].

The most common PHAs undergo abiotic degradation by chemical hydrolysis scission of the ester bonds (**Figure 2.14**). A discussion is still open in the field whether PHAs go through a bulk or surface erosion process regardless of the thickness [48,50]. However, some studies have reported reduction of M_w , mass loss, and mechanical properties deterioration during degradation, indicative of surface erosion as the dominant mechanism [569,570]. Specifically, for the copolymer PHBV, a surface erosion process for both enzymatic and chemical hydrolysis was reported [571].

After depolymerization of PHB (as an example PHA), 3-hydroxybutyric acid is bioassimilated and, through a redox reaction, converted to acetyl-CoA, which feeds the TCA cycle (Figure 2.21).



Figure 2.21 Tentative biodegradation pathway for PHB in aerobic conditions.

Degradation of short-chain-length PHAs by enzymatic activity from bacteria and fungi has been extensively reported [48,50,489,572]. In general, an increase in side-chain length decreases the hydrolytic rate of the PHAs [359].

PHAs can be metabolized by intra or extracellular depolymerases depending on its location. In this sense, *in vivo* granules are amorphous PHAs that can be metabolized by intracellular enzymes. Denatured PHAs, after cell lysis, becomes semi-crystalline and can be depolymerized by microorganisms that release extracellular depolymerases [489,573].

PHB depolymerases (3.1.1.75) and PHA depolymerases (3.1.1.76) are the main enzymes reported as able to degrade PHB and other PHAs. A bacterial PHB depolymerase has been shown to have two functions during the hydrolysis of PHB films, which takes place via adsorption and hydrolysis, binding, and catalytic domains [366]. Investigations revealed that the binding domain of the enzyme is non-specific for binding to the surface of PHA films; however, the active site in a catalytic domain is specific for the hydrolysis of the PHB molecule [366]. The activity of extracellular PHB depolymerases in enzymatic depolymerization occurs initially on the surfaces of the polymer after biofilm formation, and the rate is dependent on the M_w , crystallinity, and microbial community [574]. PHA depolymerases can be described as serine hydrolases with protein progression formed by four regions. First is the signal sequence, second is a catalytic area which contains the lipase box, third is a substrate-binding domain where the adsorption of the polymer substrate takes place, and eventually a domain which connects the catalytic area to the substrate securing areas [100].

Stimulation activity for PHB depolymerase was observed in presence of Na⁺, K⁺, Ca²⁺, and Mg²⁺ [105,380]. However, Fe⁺², Hg⁺², Mn⁺², and Cu⁺² were reported as inhibitory of enzymatic PHB activity [399,489].

Nishida et al. reported the effect of crystallinity and amorphous fraction on microbial degradation [575]; showing that increased crystallinity repressed microbial activity.

PHB, PHV, and PHBV have been reported to be biodegradable in several mesophilic environments such as soils, composts, and natural waters (**Table 2.9**). A large microbial population has been identified as associated with biodegradation of PHB and the copolymer PHBV in mesophilic conditions [50]. Biodegradation in soil of PHBV films was reported as the combined action of fungi, bacteria and actinomycetes; however, the fungi population was identified as the dominant one due to the ability to increase the surface growth of hyphae [48,449]. Copolymer composition, crystallinity, microstructure, and surface morphology are factors reported to play an important role during biodegradation of PHBV in soil [221]. During degradation of PHBV in seawater, an increase in surface roughness was observed, which was reported as both surface erosion by chemical hydrolysis and enzymatic activity [48,576].

During biodegradation of PHBV (films) evaluated in soil at 25 °C, mineralization values of c. 25% were achieved after 190 days of testing [225]. Higher mineralization values
were reported for PHBV in marine environments. Thellen et al. [231] reported high mineralization for high M_w PHBV films with different contents of valerate in a marine environment at 30 °C; after 100 days of testing, mineralization values reached c. 100% for PHBV with 5 and 8% of valerate, and c. 90% for PHBV with 12% valerate. A recent study by Meereboer et al. [233] evaluated PHBV in powder form in a simulated marine environment at 25 °C; mineralization values were higher than 50% at day 190, and values reached c. 90% at the end of the test (450 days).

Mineralization of PHB in a soil environment was reported after 360 days of testing, with a degree of biodegradation of c. 95% at 25 °C [209]. High mineralization values were also reported by Narancic et al. [115]. After 136 days of testing in soil at 25 °C, mineralization values higher than 100%, showing priming effect, were reported; however, when evaluated in home composting, low mineralization values (less than 20%) were reported for PHB at 28 °C [115].

On the other hand, PHB in marine environment showed c. 70% of mineralization after 360 days of testing and c. 95% in the same test after 200 days, but with a stirring system; the difference in CO₂ evolution was attributed to the shortage of O₂ in the system without stirring [211]. Thellen et al. also reported high mineralization values (in the range of 80 to 90%) for high M_w and high crystallinity content PHB films in a simulated marine environment at 30 °C; this work was indicative of the high degradability of polymers from the PHA family even though the % crystallization and initial M_w was high for both PHB and PHBV [231]. Narancic et al. reported mineralization values of c. 90% for PHB sheets at day 56 of testing in a marine environment at 30 °C, whereas in a freshwater environment at 21 °C the values were c. 90% for PHB [115].

PHA biodegradation has been evaluated in soil. Gómez et al. reported mineralization values of c. 70% for PHA (injection molding samples) in soil after 650 days at 20 °C [215]. A

more recent study reported mineralization values of c. 90% for PHA (powder) in soil after 150 days at both 25 and 37 °C [210].

2.17.8 Poly(butylene adipate-co-terephthalate) – PBAT

PBAT is a co-polyester synthesized from 1,4-butanediol (BDO), adipic acid (AA), and dimethyl terephthalate by a polycondensation reaction. Adipic acid and BDO polymerize to produce their own polyester and water. Dimethyl terephthalate and BDO react to form their own polyester and methanol. The resulting polyester then reacts with the polyester of AA and BDO using tetrabutoxytitanium as a catalyst for the transesterification. The reactions are carried out at temperatures higher than 190 °C, under high vacuum, and usually require long times [577]. PBAT has a T_g of c. 30 °C and T_m of c. 106 °C. PBAT has good toughness and ductility, biodegradability, and is flexible. However, higher production costs, coupled with lower mechanical and heat resistance in comparison to common fossil-based plastics, has hindered PBAT development and acceptance in the consumer market [577]. These shortcomings can be overcome by blending PBAT with other biodegradable polymers. For example, blends of PBAT and PLA demonstrated higher yield stress, modulus, and rheological properties than those of neat PBAT [578]. PBAT is widely used for agricultural mulch films, and also for packaging applications including trash bags, shopping bags, wrapping films, and disposable food containers [579].

The reported abiotic mechanisms of degradation associated with PBAT are primarily mechanical, photodegradation, thermal, and hydrolysis. Mechanical degradation is, in general, associated with the entire spectra of biodegradable polymers; in the case of PBAT, erosion is a common situation due to its main application in agricultural mulch films. Photodegradation has been reported as the main abiotic mechanism of degradation for PBAT agricultural mulch films. A crosslinking effect as a result of exposure to sunlight has been reported; the effect is known to delay the biodegradation rate of PBAT by decreasing the chain mobility of the polymer [141,175,276].

The addition of aliphatic acids to aromatic polyesters improves the water uptake and hydrolysis of these polymers. For example, by adding adipic acid to poly(butylene terephthalate), PBAT is obtained and it can undergo faster hydrolytic degradation. However, PBAT still offers more resistance to chemical hydrolysis than aliphatic polyesters, such as PLA, due to the steric hindrance of the large aromatic ring repeating units.

PBAT is depolymerized through chemical and enzymatic hydrolysis into adipic acid, 1,4-butanediol, and terephthalic acid (**Figure 2.14**). Then each compound is bioassimilated or undergoes a redox reaction to feed the TCA cycle, as shown in **Figure 2.22**. The adipic acid pathway is through adipyl-CoA, and 1,4-butanediol is converted to succinic acid and to succinyl-CoA. Several bioassimilation pathways have been reported for terephthalic acid. The most probable, in the case of PBAT, seems to be the transport of terephthalic acid through the cell membrane, followed by degradation to protocatechuic acid and then through the pyruvic acid pathway to Acetyl-CoA to enter the TCA cycle [564].



Figure 2.22 Tentative biodegradation pathway for PBAT in aerobic conditions. Adapted from [564,580].

In terms of biodegradation and enzymatic activity at mesophilic conditions, PBAT has been reported to be degraded by cutinases [298], lipases [331], and PBAT hydrolases [260]. As for chemical hydrolysis, enzymatic activity is affected due to the presence of aromatic groups that make enzyme accessibility more difficult for scission of the ester bonds that are close to these groups [50,51]. The presence of the aromatic ring has been associated with a decrease of the enzymatic activity by creating a steric impediment to access the active site of the enzymes. Butanediol-terephthalate bonds have been reported to be hydrolyzed at a lower rate in comparison to adipate-butanediol bonds [81].

Low values of enzymatic activity reported for the actinobacteria *Rhodococcus fascians* in comparison to a mesophilic PBAT-degrading fungus show that both the type of enzyme and the microorganism producing the enzyme play major main roles in activity [259,260]. These results could be associated with the favorable conditions offered to microorganism populations in soil environments in the mesophilic range. The enzymatic hydrolysis of PBAT by a fungal strain generated terephthalic acid, adipic acid, and 1,4-butanediol, as identified by mass spectroscopy [260]. Furthermore, enzymatic activity of PBAT hydrolase by *Bacillus pumilus* on PBAT showed degradation products as adipate, 1,4-butanediol, and terephthalate [81].

Crosslinking due to exposure to UV-radiation treatment has also been shown to decrease the enzymatic activity against PBAT, due to the reduced flexibility of the polymer chains after crosslinking [581].

PBAT has been reported to be degraded in soil environments or soil in laboratory conditions (Table 2.9). In general, rates of biodegradation at mesophilic conditions are low. Biodegradation studies of PBAT showing CO₂ and mineralization in simulated and controlled media in the mesophilic range are limited [140]. Studies in more controlled environments like culture and/or buffer media are more commonly focused on identification of microbial activity and/or enzymatic activity towards PBAT. However, identification of extracellular enzymes able to degrade PBAT is relatively limited in comparison to those for common aliphatic polyesters. Most of the environments assessed for PBAT degradation are agricultural soils. CO₂ production from PBAT in soil environment media has been reported, with mineralization values of c. 10% after six weeks [140].

A novel approach by Zumstein et al. [140] demonstrated the mineralization of PBAT ¹³C to ¹³CO₂, with higher values of mineralization for ¹³C derived from depolymerization of the adipate structure, and lower values of mineralization associated with depolymerization of the aromatic terephthalate fraction. This finding is indicative of the increased complexity of aromatic polyesters towards depolymerization and assimilation. On the other hand, the presence of the aromatic component in the co-polyester was shown to improve the overall rate of biodegradation, even in the mesophilic range like in the soil environment evaluated [140].

PBAT films with 1% of a chain extender had low mineralization values of c. 20% after 180 days in soil at 28°C [218]. The effect of the chain extender on delaying M_w reduction and biodegradation was evident.

An interesting outcome of biodegradation studies for PBAT in a soil environment is that the degradation products have been shown to be harmless to the microbial population [582]. Although biodegradation can be a longer process in the mesophilic environment, and the formation of PBAT degradation products does not affect the quality and health of the soil and its microbial population, development of some microorganisms over others can be modified [583].

2.17.9 Poly(urethanes)s - PUs

PUs are synthetic plastics, insoluble in water, and produced by the condensation reaction of polyols and polyisocyanate having urethane bonds [49,584]. Polyisocyanates and polyols react with a chain extender to give polyurethane polymers with alternate soft and rigid segments. Polyol forms the soft segment and can be obtained from polyester or polyether polyols; whereas the rigid segment is derived from the isocyanate and chain extender, and has restricted mobility compared with the soft polyol segment (**Figure 2.23**) [495]. The rigid segment is considered the crystalline region and the soft segment the non-crystalline or amorphous region of PUs [493,585]. Depending on the polyol used, the resulting PU can be identified as polyester PU or polyether PU. The resulting properties and degradation behavior are dependent upon the selection and chemistry of the soft segment [586].



Figure 2.23 Soft and hard segment of the poly(urethane)s structure. Adapted from [495].

Poly(urethane)s are used in the medical, construction, and automotive fields, among others. Typically, products that contain PUs include furniture, paints, fibers, flexible foams, rigid foams, coatings, adhesives, synthetic skins, sutures, and tissue scaffolds [494,587,588]. Poly(urethane) elastomers (thermoplastic) are used in the medical field due to their high elasticity and toughness compared with other elastomers [584,587]. The good mechanical, thermal, and electrical properties of PUs allow these polymers to offer good adhesion for coatings, tensile strength, and abrasion resistance for several uses [587]. Poly(urethane) foam are a typically example of thermoset PUs [494].

Early studies demonstrated that PUs with long repeating units and hydrolytic groups were susceptible to biodegradation [589]. This review concentrates on polyester PUs. The ester bond of polyester PUs is susceptible to hydrolytic degradation and can be catalyzed with the help of extracellular enzymes. The extracellular enzymes for PU degradation have a hydrophobic area, which assists in attaching onto the polymer surface [585,587]. Microbial attack of PUs can occur by action of extracellular hydrolases such as ureases (3.5.1.5), amidases (3.5.1.4), proteases, and esterases (**Figure 2.14**). The cleavage site and the product of the breakdown is dependent on the type of the enzyme acting during depolymerization (**Figure 2.24**). Adipic acid and diethylene glycol were reported as degradation products by the action of extracellular enzymes on polyester PUs; however, no identification of the isocyanate hard segment byproducts was reported [315,316]. Later work by Shah et al. reported the probable presence of a hydrolyzed portion of the hard segment, detected by FTIR spectrum, when polyester PU was attacked by both *Bacillus subtilis* MZA-75 and *Pseudomonas aeruginosa* MZA-85 [246]. Furthermore, the mixing of esterase and amidase has been reported to hydrolyze the hard segment via the urethane bonds [437].

A bacterial esterase was identified to degrade ester PUs by acting in a two-step reaction: first, a hydrophobic adsorption of the enzyme on the surface of the PU; and second, the hydrolysis of the ester bonds of the PU [316]. Studies of enzymatic activity have shown that the rate of biodegradation decreases with decreasing ester content, indicating the impact of the esterase activity as relevant for PU depolymerization [584]. Fungal communities have been identified to degrade PU to some extent [481].



Figure 2.24 Sites of scission for urethane bonds in extracellular enzyme function. Adapted from [585].

A tentative route for metabolism of the soft segment (PEG) is presented in **Figure** 2.25 for PU derived from ester.



Figure 2.25 Tentative metabolism pathway for poly(urethane)s derived from esters. Adapted from [590].

Biodegradation studies showing CO₂ evolution or mineralization in mesophilic environments for PUs are limited. Most of the reported studies on PUs are for enzymatic activity of both fungi and bacteria. More investigations of the abiotic degradation process of PUs prior to the biotic degradation stage, such as hydrolysis or photodegradation, in the mesophilic range would help determine whether PUs derived from esters are fully biodegradable in soil, home composting, industrial composting, or water environments. The biodegradation of polyester-PUs studied under mesophilic composting conditions resulted in mineralization between 5 and 43% after 45 days of testing, and this wide range was attributed to the different chemical structures of the PUs [591]. A high content of the hard segment led to decreased biodegradation rates and mineralization, whereas biodegradation increased as the amount of diol carbon chains of the polyol (soft segment) increased. The hard segment composition in PUs was presented as a more dominant effect than the crystallinity or surface properties during PU biodegradation in composting [591]. The presence of aromatic diisocyanates decreased the rate of biodegradation in comparison to PUs with aliphatic diisocyanates [591].

Biodegradation of PU films during the Sturm test showed high CO₂ evolution at 30 °C for 28 days in comparison to the control [240]. Also, a Sturm test revealed the production of CO₂ during the enzymatic hydrolysis of PU films by *Bacillus subtilis* MZA-75 and *Pseudomonas aeruginosa* MZA-85, hydrolyzing the ester portion in 1,4-butanediol and adipic acid products [246–248]. This result indicated that *Bacillus subtilis* was able to hydrolyze and assimilate the intermediates as carbon sources with final mineralization. An interesting outcome of the reported enzymes attacking ester PUs is the evidence of the presence of membrane-bound enzymes, besides extracellular enzymes. For an esterase not secreted to the culture medium, its high hydrophobicity was reported as the most probably cause for its membrane-bound characteristic [547].

A new approach is the development of non-isocyanate PUs (NIPU). NIPU are a promising and more sustainable alternative for traditional PUs [592,593]. However, studies in this area looking at degradation and biodegradation are still limited. Production and biodegradation assessment of polyhydroxyurethane, a NIUP based on cyclic carbonate and polyamine, was reported by Ghasemlou et al. [229]. Mineralization values for film samples reached c. 40% after 120 days of testing in soil conditions.

2.17.10 Poly(vinyl alcohol) – PVOH

PVOH is a synthetic, water-soluble polymer produced by partial or complete hydrolysis of polyvinyl acetate. Unlike other polymers, PVOH is not synthesized from the polymerization of its monomer (vinyl alcohol), due to the unstable nature of the high density of hydroxyl groups in the monomer. Polyvinyl acetate is first synthesized by the polymerization of vinyl acetate and then subjected to saponification wherein the ester groups of vinyl acetate are replaced by hydroxyl groups in the presence of caustic soda [594]. Different grades and properties of PVOH are available, depending on the degree of hydrolysis and the variation in initial length of the vinyl acetate polymer. PVOH is odorless and nontoxic in nature, has excellent resistance to aroma and gases, is resistant to solvents and oil, has good optical and adhesive properties, and film forming capacity [595]. In terms of disadvantages, PVOH is expensive, and mechanical properties are highly conditioned by the presence of water or humidity so it needs to be blended with other polymers to achieve more desirable properties [596]. Due to its good adhesion to other hydrophilic surfaces, PVOH is used widely in emulsifiers, binders, and hydrogels for a broad range of industries, including textile, paper sizing, fabrics, and packaging films as a protective film for laundry and dish detergents. The applications are not limited and extend to the biomedical, cosmetic, and food packaging industries [597,598].

The degree of solubility of PVOH in water can be tailored, depending on the amount of OH groups and remaining acetate bonds.

Besides the abiotic mechanism of biodegradation, PVOH could be considered as partially biodegradable since the number of microorganisms and enzymes identified to biodegrade it is rather scarce in comparison to polyesters.

Biodegradation of PVOH has been reported to start from random chain scission where the action of oxidative enzymes catalyzes the break of the carbon backbone. Mostly dehydrogenases or oxidases are responsible for the carbon-carbon bond scission. Hydrolases or aldolases have been reported as responsible for the chain scission of the hydroxyl group (**Figure 2.14**). Furthermore, a two-step process has been proposed for the enzymatic degradation of PVOH: the first step, by action of PVOH oxidases, involves the oxidation of hydroxyl groups to form diketone or monoketone structures; and the second step involves hydrolysis of the carbonyl structure formed by oxidized PVOH hydrolases [599]. Since PVOH is a water-soluble polymer, its biodegradation has been studied mostly in aqueous media. The identified microorganisms and enzymes able to biodegrade PVOH are associated mainly with contaminated environments, such as waste sludge, which are common end-of-life scenarios for PVOH.

Abiotic degradation of PVOH by UV/chlorine oxidation via generation of active free radicals has been investigated; in acidic media the efficiency was higher due to the higher ratio of [HOCl]/[OCl-] [600]. The abiotic degradation of PVOH by photocatalytic oxidation or radiation and ozone also has been reported [478,601].

Published works have identified that microorganisms able to biodegrade PVOH are mostly from the genus *Pseudomonas* [602]. Also, many PVOH degradation pathways have been proposed for different bacteria such as *Alcaligenes* and *Pseudomonas* species [603]. These routes include scission of the polymer chain by an extracellular oxidase (dehydrogenase), followed by aldolase and hydrolase reactions, releasing compounds such as acetic acid and hydroxyl fatty acids that can be incorporated into the β -oxidation and TCA cycle, respectively [100]. In **Figure 2.26** is presented a tentative metabolization route for PVOH [439].

As stated, scarce mineralization was reported for PVOH films in water conditions, with $\sim 10\%$ after 100 days of testing at c. 30 °C [232].



Figure 2.26 Proposed pathway for metabolism of PVOH. Adapted from [109,603].

2.18 Final insights and remarks

Addressing plastic pollution has become one of the main concerns of our modern society. The impacts of plastic waste in terms of global climate change, health, and social effects, circular economy, sustainable use of resources and production, and improved waste management systems have garnered the attention of industry, government, and NGO stakeholders and the society in general. The development of new plastic waste pacts and commitments to curb the use of virgin plastic are ongoing globally, with targets for 2025 [21]. However, the damage to ecosystems has already created deleterious impacts, which will require forward-thinking actions to remediate, to mitigate, and to avoid permanent damage [9].

The development of biodegradable polymers derived from both bio- and fossil-based resources has transcended from the lab scale to commercial applications in the last two decades, and these polymers have become an option for packaging and consumer goods applications to mitigate the impact of plastic waste. However, as with any material created by society, biodegradable polymers also must reach a waste management end-of-life to avoid a rebound effect on creating additional pollution.

The degradation process for biodegradable polymers starts by the action of external abiotic and biotic factors. The main abiotic mechanisms of degradation associated with mesophilic environments are photodegradation, mechanical degradation, and chemical hydrolysis. Photodegradation in the presence of O_2 introduces modifications during the degradation of biodegradable polymers in specific environments such as agriculture soils, inducing a dual effect: chain scission that contributes to the degradation and also crosslinking that act to delay the process. The initial deterioration of the polymer structure enhances the mechanical degradation, generating micro and nano plastics but not guaranteeing biodegradable polymers since most of them contain ester bonds that are prone to water attack.

The formation of biofilms, the stage prior to the release of extracellular enzymes, affects the whole dynamic of the degradation process, as discussed. Since biofilms create an extra layer on the polymer surface and potentially affect water diffusion during chemical hydrolysis, a better understanding of biofilm formation and its effect on water diffusion and bulk erosion are needed.

Extracellular enzymes act at the surface level of polymers, making enzymatic activity a surface erosion process. As presented in this review, the main groups of enzymes reportedly able to break chemical bonds in polymers belong to the esterase group (amidases, cutinases, esterases, lipases, and PHA depolymerases), proteases (specific for PLLA), and oxidoreductases (for PVOH and PU). Recent advances in the identification of protein sequences and residues, structural domains, mechanisms of substrate binding, kinetic analysis, and the presence and effects of cofactors have provided a better understanding of enzymatic activity on biodegradable polymers. However, a better understanding of bioassimilation and mineralization is still needed at the biochemical level of monomer compounds produced from the chemical and enzymatic hydrolysis of biodegradable polymers.

In terms of polymer properties, the key bulk properties affecting biodegradation in the mesophilic range are stereochemistry, crystallinity, and M_w , which are tailored for each application. The amorphous region offers the optimal conditions for chemical hydrolysis due to the easy diffusion of water and also for exo and endo enzymatic attack by extracellular enzymes. Microorganisms start the assimilation process when low M_w compounds such as dimers and monomers are released. Since biofilm formation, microbial colonization, and enzymatic activity are surface related processes, the key surface properties of polymers impacting biofilm formation and colonization are hydrophobic/hydrophilic balance, roughness, and surface energy parameters.

This review has summarized the enzymes and microorganisms (e.g., bacteria, fungi, and actinomycetes) isolated from several environments and showing activity towards aliphatic, aliphatic-aromatic polyesters, PUs derived from esters, and PVOH. Usually the identification of microorganisms and/or enzymes involves techniques, such as culturing, where the polymer is the solely source of carbon for the biotic process. These studies provide unique insights on enzymatic activity and pathways of degradation. However, natural environments introduce far more complexities to the degradation process, creating a dynamic that undoubtedly affects the rate of degradation. Microbial consortia have demonstrated an increased efficiency for elimination of toxic metabolites in comparison to pure cultures. Studies showed that some microorganisms are directly involved in the degradation process, while other microorganisms showed activity towards eliminating toxic metabolites excreted by the first ones. However, besides symbiotic, mutual, and synergistic interactions, efficiency differs among microbial consortia. Complex tracking of microorganism population dynamics during biodegradation should provide better insights on the real pathways of degradation and assimilation of these polymers in actual environments. Research in the areas of biostimulation (addition of specific nutrients to the soil, compost, or water environments to stimulate the activity of naturally occurring microorganism populations), bioaugmentation (addition of specific microorganisms to increase the biodegradation rate of an indigenous microbial population), and engineering of enzymes (modification of enzymes to reach specific reactions) are needed to address the complexities associated with microbial consortia involved in the biodegradation process, extracellular enzymes, and biocatalytic cascades of enzymes.

Standards, methodologies, and techniques have been developed to assess the degradation of polymers in the environments as discussed. Some are more focused on evaluating the degradation of mechanical properties and the mass loss due to various factors of abiotic mechanisms. However, the assessment of CO_2 or O_2 and ultimate mineralization must be the definitive assessment to determine the extent of biodegradability in a specific environment. The use of complementary techniques, such as carbon tracking and M_w reduction, constitutes important tracking parameters that must be incorporated when evaluating biodegradation to the mineralization level.

Many works reported the CO_2 or mineralization for the biodegradable polymers available in the market in soil, home composting, and aquatic environments at mesophilic conditions using several standards. From the aliphatic polyesters group (e.g., PGA, PEA, PLA, PCL, and PBS), chemical hydrolysis has been reported to be the main controlling step. Other aliphatic polyesters, such as PCL, PBS, and PBSA, were consistently reported to biodegrade in soil conditions. For the aliphatic-aromatic polyesters (PBAT and PBST), reports of PBAT biodegradation in soil and marine environments are limited. The degradation of the natural polyesters PHB and PHBV in aquatic environments was reported extensively, showing the high level of biodegradability at mild conditions. For the PUs, the presence of the soft segment offers availability for enzymatic attack and biodegradation with mineralization at some extent; however, the bioassimilation pathway of the hard segment has not been well identified and/or described.

New, innovative methods to tailor the biodegradation of biodegradable polymers through creation of novel polymers with tailored biodegradation [604], biostimulation, bioaugmentation, and addition of natural enzymes [605–607] or modified enzymes [608] are opening new routes to accelerate the biodegradation process. However, these new methods must be connected to standards to fully track the biodegradation process and the end products in the environment, so that further insights on biodegradation pathways of polymers can be elucidated.

REFERENCES

- [1] Plastics Europe, Plastics the Facts 2021, 2021. https://plasticseurope.org/wpcontent/uploads/2021/12/Plastics-the-Facts-2021-web-final.pdf (accessed Jan. 31, 2022).
- [2] Chinthapalli R., Skoczinski P., Carus M., et al., Bio-based Building Blocks and Polymers – Global Capacities, Production and Trends 2020 – 2025, 2021. Accessed: Jan. 31, 2022. [Online]. Available: https://renewablecarbon.eu/publications/product/bio-based-building-blocks-and-polymers-globalcapacities-production-and-trends-2020-2025-short-version/.
- [3] Geyer R., Jambeck J. R., and Law K. L., Production, use, and fate of all plastics ever made, *Sci. Adv.*, vol. 3, no. 7, pp. 1–5, Jul. 2017, doi: 10.1126/sciadv.1700782.
- [4] Statista, Market size of plastics worldwide from 2016 to 2028, 2021. https://www.statista.com/statistics/1060583/global-market-value-of-plastic/ (accessed Jan. 31, 2022).
- [5] Grand view research, Plastic Market Size, Share & Trends Analysis Report By Product (PE, PP, PU, PVC, PET, Polystyrene, ABS, PBT, PPO, Epoxy Polymers, LCP, PC, Polyamide), By Application, By Region, And Segment Forecasts, 2020 - 2027, 2021. https://www.grandviewresearch.com/industry-analysis/global-plastics-market (accessed Jul. 04, 2020).
- [6] UNNE, Exploring the potential for adopting alternative materials to reduce marine plastic litter, *United Nations Environment Programme (2017)*, 2018. http://wedocs.unep.org/bitstream/handle/20.500.11822/25485/plastic_alternative.pdf? sequence=1&isAllowed=y (accessed Jan. 31, 2022).
- [7] The Pew Charitable Trusts and Systemiq, Breaking the Plastic Wave: A Comprehensive Assessment of Pathways Towards Stopping Ocean Plastic Pollution, 2020.
- [8] Lau W. W. Y., Shiran Y., Bailey R. M., et al., Evaluating scenarios toward zero plastic pollution, Science (80-.), vol. 369, no. 6510, pp. 1455–1461, Sep. 2020, doi: 10.1126/science.aba9475.
- [9] UNNE, From Pollution to Solution. A global assessment of marine litter and plastic pollution Nairobi, *United Nations Environment Programme*, 2021. https://www.unep.org/resources/pollution-solution-global-assessment-marine-litter-and-plastic-pollution (accessed Jan. 31, 2022).
- [10] McKinsey & Company Center for Business and Environment and Ocean Conservancy, Stemming the tide: Land-based strategies for a plastic-free ocean, 2015. https://oceanconservancy.org/wp-content/uploads/2017/04/full-report-stemmingthe.pdf (accessed Nov. 22, 2019).
- [11] Jambeck J. R., Geyer R., Wilcox C., et al., Plastic waste inputs from land into the

ocean, *Science (80-.).*, vol. 347, no. 6223, pp. 768–771, Feb. 2015, doi: 10.1126/science.1260352.

- [12] Borrelle S. B., Ringma J., Law K. L., *et al.*, Predicted growth in plastic waste exceeds efforts to mitigate plastic pollution, *Science (80-.).*, vol. 369, no. 6510, pp. 1515–1518, Sep. 2020, doi: 10.1126/science.aba3656.
- [13] Vethaak A. D. and Legler J., Microplastics and human health, *Science (80-.).*, vol. 371, no. 6530, pp. 672–674, Feb. 2021, doi: 10.1126/science.abe5041.
- [14] MacLeod M., Arp H. P. H., Tekman M. B., and Jahnke A., The global threat from plastic pollution, *Science (80-.).*, vol. 373, no. 6550, pp. 61–65, Jul. 2021, doi: 10.1126/science.abg5433.
- [15] Jâms I. B., Windsor F. M., Poudevigne-Durance T., et al., Estimating the size distribution of plastics ingested by animals, Nat. Commun., vol. 11, no. 1, p. 1594, Dec. 2020, doi: 10.1038/s41467-020-15406-6.
- [16] Schmidt C., Krauth T., and Wagner S., Export of Plastic Debris by Rivers into the Sea, *Environ. Sci. Technol.*, vol. 51, no. 21, pp. 12246–12253, Nov. 2017, doi: 10.1021/acs.est.7b02368.
- [17] Pahl S., Wyles K. J., and Thompson R. C., Channelling passion for the ocean towards plastic pollution, *Nat. Hum. Behav.*, vol. 1, no. 10, pp. 697–699, Oct. 2017, doi: 10.1038/s41562-017-0204-4.
- [18] Allen S., Allen D., Phoenix V. R., et al., Atmospheric transport and deposition of microplastics in a remote mountain catchment, Nat. Geosci., vol. 12, no. 5, pp. 339– 344, May 2019, doi: 10.1038/s41561-019-0335-5.
- [19] Bergmann M., Mützel S., Primpke S., et al., White and wonderful? Microplastics prevail in snow from the Alps to the Arctic, Sci. Adv., vol. 5, no. 8, pp. 1–11, Aug. 2019, doi: 10.1126/sciadv.aax1157.
- [20] Stubbins A., Law K. L., Muñoz S. E., *et al.*, Plastics in the Earth system, *Science (80-.)*, vol. 373, no. 6550, pp. 51–55, Jul. 2021, doi: 10.1126/science.abb0354.
- [21] The Ellen MacArthur Foundation, The Global Commitment 2021 Progress Report, 2021. https://ellenmacarthurfoundation.org/global-commitment/overview (accessed Jan. 31, 2022).
- [22] The Ellen MacArthur Foundation, Towards the Circular Economy, 2013. https://reports.weforum.org/toward-the-circular-economy-accelerating-the-scale-upacross-global-supply-chains/ (accessed Jan. 31, 2022).
- [23] World Economic Forum, The New Plastics Economy: Rethinking the future of plastics, 2016. https://www.weforum.org/reports/the-new-plastics-economy-rethinking-the-future-of-plastics (accessed Jan. 31, 2022).
- [24] European Environment Agency, Preventing plastic waste in Europe, 2019.

https://www.eea.europa.eu/publications/preventing-plastic-waste-in-europe (accessed Jan. 31, 2022).

- [25] U.S. Department of Energy, Bio-Optimized Technologies to keep Thermoplastics out of Landfills and the Environment (BOTTLETM), 2019. https://www.bottle.org/index.html (accessed Jan. 31, 2022).
- [26] Alliance to end plastic waste, Alliance to end plastic waste, 2022. https://endplasticwaste.org/ (accessed Jan. 31, 2022).
- [27] United Nations, Transforming Our World: The 2030 Agenda for Sustainable Development, 2015. https://sdgs.un.org/2030agenda (accessed Jan. 31, 2022).
- [28] Xanthos D. and Walker T. R., International policies to reduce plastic marine pollution from single-use plastics (plastic bags and microbeads): A review, *Mar. Pollut. Bull.*, vol. 118, no. 1–2, pp. 17–26, May 2017, doi: 10.1016/j.marpolbul.2017.02.048.
- [29] Natural Resources Council of Maine, Recycling Reform for Maine. Extended Producer Responsibility (EPR) for Packaging, 2022. https://www.nrcm.org/programs/sustainability/recycling-wastemanagement/recycling-reform-maine/ (accessed Jan. 31, 2022).
- [30] Staab A., Attention, Shoppers: Get Ready for NJ's Plastic-Bag Ban, 2021. https://njmonthly.com/articles/jersey-living/nj-plastic-bag-ban/ (accessed Jan. 31, 2022).
- [31] BioCycle, Oregon Second State To Pass Packaging EPR Law, 2021. https://www.biocycle.net/oregon-second-state-to-pass-packaging-eprlaw/?utm_source=BioCycle+CONNECT&utm_campaign=0398833ba8-EMAIL_CAMPAIGN_2020_03_20_08_35_COPY_01&utm_medium=email&utm_ter m=0_8396f01c15-0398833ba8-513876499 (accessed Jan. 31, 2022).
- [32] Braun S., 5 things to know about the EU single-use plastics ban, 2021. https://www.dw.com/en/5-things-to-know-about-the-eu-single-use-plastics-ban/a-58109909 (accessed Jan. 31, 2022).
- [33] Reuters, Indonesia's capital bans single-use plastic bags from markets and malls, 2020. https://www.reuters.com/article/us-indonesia-environment-plastic/indonesias-capital-bans-single-use-plastic-bags-from-markets-and-malls-idUSKBN1Z612H (accessed Jan. 31, 2022).
- [34] Library of Congress, China: Single-Use Plastic Straw and Bag Ban Takes Effect, 2021. https://www.loc.gov/item/global-legal-monitor/2021-03-23/china-single-use-plasticstraw-and-bag-ban-takes-effect/ (accessed Jan. 31, 2022).
- [35] Lindsey Parkinson, New Zealand and two Australian states phase out single-use plastics, *Food Packaging Forum*, 2021. https://www.foodpackagingforum.org/news/new-zealand-and-two-australian-states-phase-out-single-use-plastics (accessed Jan. 31, 2022).

- [36] Kaza, Silpa; Yao, Lisa C.; Bhada-Tata, Perinaz; Van Woerden F., What a Waste 2.0 A Global Snapshot of Solid Waste Management to 2050, Urban Development;. Washington, DC: World Bank. © World Bank., 2018. https://openknowledge.worldbank.org/handle/10986/30317 (accessed Jan. 31, 2022).
- [37] BBC News, How plastic bottles are paying for lessons in Nigeria, 2019. https://www.bbc.com/news/av/world-africa-48547893 (accessed Jan. 31, 2022).
- [38] World Economic Forum, These 4 methods can help solve Ghana's plastic dilemma, 2021. https://www.weforum.org/agenda/2021/09/4-ways-trade-ghana-transitioncircular-plastics-economy/ (accessed Jan. 31, 2022).
- [39] Hatti-Kaul R., Nilsson L. J., Zhang B., et al., Designing Biobased Recyclable Polymers for Plastics, Trends Biotechnol., vol. 38, no. 1, pp. 50–67, Jan. 2020, doi: 10.1016/j.tibtech.2019.04.011.
- [40] RameshKumar S., Shaiju P., O'Connor K. E., and P R. B., Bio-based and biodegradable polymers - State-of-the-art, challenges and emerging trends, *Curr. Opin. Green Sustain. Chem.*, vol. 21, pp. 75–81, Feb. 2020, doi: 10.1016/j.cogsc.2019.12.005.
- [41] Chen G.-Q. and Patel M. K., Plastics Derived from Biological Sources: Present and Future: A Technical and Environmental Review, *Chem. Rev.*, vol. 112, no. 4, pp. 2082– 2099, Apr. 2012, doi: 10.1021/cr200162d.
- [42] Weiss M., Haufe J., Carus M., et al., A Review of the Environmental Impacts of Biobased Materials, J. Ind. Ecol., vol. 16, no. S1, pp. S169–S181, Apr. 2012, doi: 10.1111/j.1530-9290.2012.00468.x.
- [43] Zheng J. and Suh S., Strategies to reduce the global carbon footprint of plastics, *Nat. Clim. Chang.*, vol. 9, no. 5, pp. 374–378, May 2019, doi: 10.1038/s41558-019-0459-z.
- [44] Spierling S., Knüpffer E., Behnsen H., et al., Bio-based plastics A review of environmental, social and economic impact assessments, J. Clean. Prod., vol. 185, pp. 476–491, Jun. 2018, doi: 10.1016/j.jclepro.2018.03.014.
- [45] U.S. Department of Energy, Roundtable on Chemical Upcycling of Polymers, 2019. https://science.osti.gov/bes/Community-Resources/Reports (accessed Jan. 31, 2022).
- [46] U.S. Department of Energy, Plastics for a Circular Economy Workshop: Summary Report, 2019. https://www.energy.gov/eere/bioenergy/downloads/plastics-circular-economy-workshop-summary-report (accessed Jan. 31, 2022).
- [47] Chandra R., Biodegradable polymers, Prog. Polym. Sci., vol. 23, no. 7, pp. 1273–1335, Nov. 1998, doi: 10.1016/S0079-6700(97)00039-7.
- [48] Laycock B., Nikolić M., Colwell J. M., et al., Lifetime prediction of biodegradable polymers, Prog. Polym. Sci., vol. 71, pp. 144–189, Aug. 2017, doi: 10.1016/j.progpolymsci.2017.02.004.

- [49] Shah A. A., Hasan F., Hameed A., and Ahmed S., Biological degradation of plastics: A comprehensive review, *Biotechnol. Adv.*, vol. 26, no. 3, pp. 246–265, May 2008, doi: 10.1016/j.biotechadv.2007.12.005.
- [50] Larrañaga A. and Lizundia E., A review on the thermomechanical properties and biodegradation behaviour of polyesters, *Eur. Polym. J.*, vol. 121, no. October, p. 109296, Dec. 2019, doi: 10.1016/j.eurpolymj.2019.109296.
- [51] Müller R.-J., Kleeberg I., and Deckwer W.-D., Biodegradation of polyesters containing aromatic constituents, J. Biotechnol., vol. 86, no. 2, pp. 87–95, Mar. 2001, doi: 10.1016/S0168-1656(00)00407-7.
- [52] Lucas N., Bienaime C., Belloy C., et al., Polymer biodegradation: Mechanisms and estimation techniques – A review, Chemosphere, vol. 73, no. 4, pp. 429–442, Sep. 2008, doi: 10.1016/j.chemosphere.2008.06.064.
- [53] Singh B. and Sharma N., Mechanistic implications of plastic degradation, *Polym. Degrad. Stab.*, vol. 93, no. 3, pp. 561–584, Mar. 2008, doi: 10.1016/j.polymdegradstab.2007.11.008.
- [54] Albright V. C. and Chai Y., Knowledge Gaps in Polymer Biodegradation Research, *Environ. Sci. Technol.*, vol. 55, no. 17, pp. 11476–11488, Sep. 2021, doi: 10.1021/acs.est.1c00994.
- [55] Ghosh K. and Jones B. H., Roadmap to Biodegradable Plastics—Current State and Research Needs, ACS Sustain. Chem. Eng., vol. 9, no. 18, pp. 6170–6187, May 2021, doi: 10.1021/acssuschemeng.1c00801.
- [56] Witt U., Yamamoto M., Seeliger U., et al., Biodegradable Polymeric Materials—Not the Origin but the Chemical Structure Determines Biodegradability, Angew. Chemie Int. Ed., vol. 38, no. 10, pp. 1438–1442, May 1999, doi: 10.1002/(SICI)1521-3773(19990517)38:10<1438::AID-ANIE1438>3.0.CO;2-U.
- [57] Gabbott S., Key S., Russell C., *et al.*, The geography and geology of plastics, in *Plastic Waste and Recycling*, Elsevier, 2020, pp. 33–63.
- [58] Law K. L. and Narayan R., Reducing environmental plastic pollution by designing polymer materials for managed end-of-life, *Nat. Rev. Mater.*, vol. 7, pp. 104–116, Oct. 2022, doi: 10.1038/s41578-021-00382-0.
- [59] European Bioplastics, Accountability is Key Environmental Communication Guide for Bioplastics, 2017. https://docs.europeanbioplastics.org/2016/publications/EUBP_environmental_communications_guide.pdf (accessed Jan. 31, 2022).
- [60] Kijchavengkul T. and Auras R., Compostability of polymers, *Polym. Int.*, vol. 57, no. 6, pp. 793–804, Jun. 2008, doi: 10.1002/pi.2420.
- [61] Narayan R., Carbon footprint of bioplastics using biocarbon content analysis and lifecycle assessment, *MRS Bull.*, vol. 36, no. 9, pp. 716–721, Sep. 2011, doi:

10.1557/mrs.2011.210.

- [62] Gorrasi G. and Pantani R., Hydrolysis and Biodegradation of Poly(lactic acid), in *Romanian Reports of Physics*, vol. 54, no. 3–4, 2017, pp. 119–151.
- [63] Göpferich A., Mechanisms of polymer degradation and erosion, *Biomaterials*, vol. 17, no. 2, pp. 103–114, Jan. 1996, doi: 10.1016/0142-9612(96)85755-3.
- [64] Tsuji H., Hydrolytic Degradation, in *Poly(Lactic Acid)*, R. Auras, L.-T. Lim, S. E. M. Selke, and H. Tsuji, Eds. Hoboken, NJ, USA: John Wiley & Sons, Inc., 2010, pp. 343–381.
- [65] Devi R., Kannan V., Natarajan K., et al., The Role of Microbes in Plastic Degradation, in Environmental Waste Management, R. Chandra, Ed. Boca Raton: CRC Press, 2016, pp. 355–384.
- [66] Rocca-Smith J. R., Chau N., Champion D., et al., Effect of the state of water and relative humidity on ageing of PLA films, Food Chem., vol. 236, pp. 109–119, Dec. 2017, doi: 10.1016/j.foodchem.2017.02.113.
- [67] de Jong S. ., Arias E. ., Rijkers D. T. ., et al., New insights into the hydrolytic degradation of poly(lactic acid): participation of the alcohol terminus, Polymer (Guildf)., vol. 42, no. 7, pp. 2795–2802, Mar. 2001, doi: 10.1016/S0032-3861(00)00646-7.
- [68] Román-Ramírez L. A., Mckeown P., Jones M. D., and Wood J., Poly(lactic acid) Degradation into Methyl Lactate Catalyzed by a Well-Defined Zn(II) Complex, ACS Catal., vol. 9, no. 1, pp. 409–416, Jan. 2019, doi: 10.1021/acscatal.8b04863.
- [69] Coates G. W. and Getzler Y. D. Y. L., Chemical recycling to monomer for an ideal, circular polymer economy, *Nat. Rev. Mater.*, vol. 5, no. 7, pp. 501–516, Jul. 2020, doi: 10.1038/s41578-020-0190-4.
- [70] Siracusa V., Microbial Degradation of Synthetic Biopolymers Waste, *Polymers* (*Basel*)., vol. 11, no. 6, p. 1066, Jun. 2019, doi: 10.3390/polym11061066.
- [71] Woodard L. N. and Grunlan M. A., Hydrolytic Degradation and Erosion of Polyester Biomaterials, ACS Macro Lett., vol. 7, no. 8, pp. 976–982, Aug. 2018, doi: 10.1021/acsmacrolett.8b00424.
- [72] George S. C. and Thomas S., Transport phenomena through polymeric systems, Prog. Polym. Sci., vol. 26, no. 6, pp. 985–1017, Aug. 2001, doi: 10.1016/S0079-6700(00)00036-8.
- [73] Hedenqvist M., Diffusion of small-molecule penetrants in semicrystalline polymers, *Prog. Polym. Sci.*, vol. 21, no. 2, pp. 299–333, 1996, doi: 10.1016/0079-6700(95)00022-4.
- [74] Pierre T. S. and Chiellini E., Review: Biodegradability of Synthetic Polymers for Medical and Pharmaceutical Applications: Part 2—Backbone Hydrolysis, J. Bioact.

Compat. Polym., vol. 2, no. 1, pp. 4-30, Jan. 1987, doi: 10.1177/088391158700200101.

- [75] Burkersroda F. von, Schedl L., and Göpferich A., Why degradable polymers undergo surface erosion or bulk erosion, *Biomaterials*, vol. 23, no. 21, pp. 4221–4231, Nov. 2002, doi: 10.1016/S0142-9612(02)00170-9.
- [76] Lyu, Schley J., Loy B., et al., Kinetics and Time-Temperature Equivalence of Polymer Degradation, *Biomacromolecules*, vol. 8, no. 7, pp. 2301–2310, Jul. 2007, doi: 10.1021/bm070313n.
- [77] Iñiguez-Franco F., Auras R., Ahmed J., et al., Control of hydrolytic degradation of Poly(lactic acid) by incorporation of chain extender: From bulk to surface erosion, Polym. Test., vol. 67, no. March, pp. 190–196, May 2018, doi: 10.1016/j.polymertesting.2018.02.028.
- [78] Höglund A., Odelius K., and Albertsson A.-C., Crucial Differences in the Hydrolytic Degradation between Industrial Polylactide and Laboratory-Scale Poly(L-lactide), ACS Appl. Mater. Interfaces, vol. 4, no. 5, pp. 2788–2793, May 2012, doi: 10.1021/am300438k.
- [79] Kijchavengkul T., Auras R., Rubino M., *et al.*, Biodegradation and hydrolysis rate of aliphatic aromatic polyester, *Polym. Degrad. Stab.*, vol. 95, no. 12, pp. 2641–2647, Dec. 2010, doi: 10.1016/j.polymdegradstab.2010.07.018.
- [80] Van Krevelen D. W. and Te Nijenhuis K., Polymer Properties, in *Properties of Polymers*, vol. 16, no. 2, Elsevier, 2009, pp. 3–5.
- [81] Muroi F., Tachibana Y., Soulenthone P., *et al.*, Characterization of a poly(butylene adipate- co-terephthalate) hydrolase from the aerobic mesophilic bacterium Bacillus pumilus, *Polym. Degrad. Stab.*, vol. 137, pp. 11–22, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.006.
- [82] Kijchavengkul T., Auras R., Rubino M., et al., Atmospheric and soil degradation of aliphatic–aromatic polyester films, Polym. Degrad. Stab., vol. 95, no. 2, pp. 99–107, Feb. 2010, doi: 10.1016/j.polymdegradstab.2009.11.048.
- [83] Heller J., Controlled drug release from poly(ortho esters) A surface eroding polymer, J. Control. Release, vol. 2, pp. 167–177, Nov. 1985, doi: 10.1016/0168-3659(85)90042-2.
- [84] Tamada J. A. and Langer R., Erosion kinetics of hydrolytically degradable polymers., *Proc. Natl. Acad. Sci.*, vol. 90, no. 2, pp. 552–556, Jan. 1993, doi: 10.1073/pnas.90.2.552.
- [85] Sackett C. K. and Narasimhan B., Mathematical modeling of polymer erosion: Consequences for drug delivery, Int. J. Pharm., vol. 418, no. 1, pp. 104–114, Oct. 2011, doi: 10.1016/j.ijpharm.2010.11.048.
- [86] Madigan M. T., Bender K. S., Buckley D., et al., Brock Biology of Microorganisms. 2006.

- [87] Haider T. P., Völker C., Kramm J., et al., Plastics of the Future? The Impact of Biodegradable Polymers on the Environment and on Society, Angew. Chemie Int. Ed., vol. 58, no. 1, pp. 50–62, Jan. 2019, doi: 10.1002/anie.201805766.
- [88] Flemming H.-C., Biodeterioration of synthetic materials A brief review, *Mater. Corros.*, vol. 61, no. 12, pp. 986–992, Dec. 2010, doi: 10.1002/maco.201005837.
- [89] Gu J.-D. and Mitchell R., Biodeterioration, in *The Prokaryotes*, vol. 9783642313, no. Ford 1993, Berlin, Heidelberg: Springer Berlin Heidelberg, 2013, pp. 309–341.
- [90] Flemming H., Biofilms, in *eLS*, Wiley, 2008, pp. 1–10.
- [91] Stoodley P., Sauer K., Davies D. G., and Costerton J. W., Biofilms as Complex Differentiated Communities, Annu. Rev. Microbiol., vol. 56, no. 1, pp. 187–209, Oct. 2002, doi: 10.1146/annurev.micro.56.012302.160705.
- [92] Rummel C. D., Jahnke A., Gorokhova E., et al., Impacts of Biofilm Formation on the Fate and Potential Effects of Microplastic in the Aquatic Environment, Environ. Sci. Technol. Lett., vol. 4, no. 7, pp. 258–267, Jul. 2017, doi: 10.1021/acs.estlett.7b00164.
- [93] Flemming H.-C. and Wingender J., The biofilm matrix, *Nat. Rev. Microbiol.*, vol. 8, no. 9, pp. 623–633, Sep. 2010, doi: 10.1038/nrmicro2415.
- [94] Flemming H.-C., Relevance of biofilms for the biodeterioration of surfaces of polymeric materials*, *Polym. Degrad. Stab.*, vol. 59, no. 1–3, pp. 309–315, Jan. 1998, doi: 10.1016/S0141-3910(97)00189-4.
- [95] Harding M. W., Marques L. L. R., Howard R. J., and Olson M. E., Can filamentous fungi form biofilms?, *Trends Microbiol.*, vol. 17, no. 11, pp. 475–480, Nov. 2009, doi: 10.1016/j.tim.2009.08.007.
- [96] Costerton J. W., Lewandowski Z., Caldwell D. E., *et al.*, Microbial Biofilms, *Annu. Rev. Microbiol.*, vol. 49, no. 1, pp. 711–745, Oct. 1995, doi: 10.1146/annurev.mi.49.100195.003431.
- [97] Flemming H.-C., Wingender J., Szewzyk U., et al., Biofilms: an emergent form of bacterial life, Nat. Rev. Microbiol., vol. 14, no. 9, pp. 563–575, Sep. 2016, doi: 10.1038/nrmicro.2016.94.
- [98] Rumbaugh K. P. and Sauer K., Biofilm dispersion, Nat. Rev. Microbiol., vol. 18, no. 10, pp. 571–586, Oct. 2020, doi: 10.1038/s41579-020-0385-0.
- [99] Center for Biofilm Engineering, What are biofilms?, 2022. https://biofilm.montana.edu/biofilm-basics/what_are_biofilms.html (accessed Jan. 31, 2022).
- [100] Shimao M., Biodegradation of plastics, Curr. Opin. Biotechnol., vol. 12, no. 3, pp. 242– 247, Jun. 2001, doi: 10.1016/S0958-1669(00)00206-8.
- [101] Krueger M. C., Harms H., and Schlosser D., Prospects for microbiological solutions to

environmental pollution with plastics, *Appl. Microbiol. Biotechnol.*, vol. 99, no. 21, pp. 8857–8874, Nov. 2015, doi: 10.1007/s00253-015-6879-4.

- [102] IUBMB, International union of Biochemistry and Molecular Biology, 2022. https://www.qmul.ac.uk/sbcs/iubmb/ (accessed Jan. 31, 2022).
- [103] Sheel A. and Pant D., Microbial Depolymerization, in *Waste Bioremediation*, S. J. Varjani, E. Gnansounou, B. Gurunathan, D. Pant, and Z. A. Zakaria, Eds. Singapore: Springer Singapore, 2018, pp. 61–103.
- [104] Mueller R.-J., Biological degradation of synthetic polyesters—Enzymes as potential catalysts for polyester recycling, *Process Biochem.*, vol. 41, no. 10, pp. 2124–2128, Oct. 2006, doi: 10.1016/j.procbio.2006.05.018.
- [105] Colak A. and Güner S., Polyhydroxyalkanoate degrading hydrolase-like activities by Pseudomonas sp. isolated from soil, *Int. Biodeterior. Biodegradation*, vol. 53, no. 2, pp. 103–109, Mar. 2004, doi: 10.1016/j.ibiod.2003.10.006.
- [106] Schimel J. P. and Schaeffer S. M., Microbial control over carbon cycling in soil, Front. Microbiol., vol. 3, no. SEP, pp. 1–11, 2012, doi: 10.3389/fmicb.2012.00348.
- [107] Allison S. D. and Vitousek P. M., Responses of extracellular enzymes to simple and complex nutrient inputs, *Soil Biol. Biochem.*, vol. 37, no. 5, pp. 937–944, May 2005, doi: 10.1016/j.soilbio.2004.09.014.
- [108] Luo L., Meng H., and Gu J.-D., Microbial extracellular enzymes in biogeochemical cycling of ecosystems, J. Environ. Manage., vol. 197, pp. 539–549, Jul. 2017, doi: 10.1016/j.jenvman.2017.04.023.
- [109] Wilkes R. A. and Aristilde L., Degradation and metabolism of synthetic plastics and associated products by Pseudomonas sp.: capabilities and challenges, J. Appl. Microbiol., vol. 123, no. 3, pp. 582–593, Sep. 2017, doi: 10.1111/jam.13472.
- [110] Lenz R. W., Biodegradable polymers, in *Biopolymers I*, vol. 85, no. 6, Berlin/Heidelberg: Springer-Verlag, 1993, pp. 1–40.
- [111] Herzog K., Müller R.-J., and Deckwer W.-D., Mechanism and kinetics of the enzymatic hydrolysis of polyester nanoparticles by lipases, *Polym. Degrad. Stab.*, vol. 91, no. 10, pp. 2486–2498, Oct. 2006, doi: 10.1016/j.polymdegradstab.2006.03.005.
- [112] Ghosh S., Qureshi A., and Purohit H. J., Microbial degradation of plastics: Biofilms and degradation pathways, in *Contaminants in Agriculture and Environment: Health Risks and Remediation*, no. January, Agro Environ Media - Agriculture and Ennvironmental Science Academy, Haridwar, India, 2019, pp. 184–199.
- [113] Castro-Aguirre E., Auras R., Selke S., *et al.*, Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.

- [114] Kumari A., Chaudhary D. R., and Jha B., Microbial Degradation of Plastics and Its Biotechnological Advancement, in *Environmental Biotechnology Vol. 3*, K. M. Gothandam, S. Ranjan, N. Dasgupta, and E. Lichtfouse, Eds. Cham: Springer International Publishing, 2021, pp. 1–30.
- [115] Narancic T., Verstichel S., Reddy Chaganti S., et al., Biodegradable Plastic Blends Create New Possibilities for End-of-Life Management of Plastics but They Are Not a Panacea for Plastic Pollution, Environ. Sci. Technol., vol. 52, no. 18, pp. 10441–10452, Sep. 2018, doi: 10.1021/acs.est.8b02963.
- [116] Napper I. E. and Thompson R. C., Environmental Deterioration of Biodegradable, Oxo-biodegradable, Compostable, and Conventional Plastic Carrier Bags in the Sea, Soil, and Open-Air Over a 3-Year Period, *Environ. Sci. Technol.*, vol. 53, no. 9, pp. 4775–4783, May 2019, doi: 10.1021/acs.est.8b06984.
- [117] Degli Innocenti F. and Breton T., Intrinsic Biodegradability of Plastics and Ecological Risk in the Case of Leakage, ACS Sustain. Chem. Eng., vol. 8, no. 25, pp. 9239–9249, Jun. 2020, doi: 10.1021/acssuschemeng.0c01230.
- [118] Ho K. L. G. and Pometto A. L., Temperature effects on soil mineralization of polylactic acid plastic in laboratory respirometers, J. Environ. Polym. Degrad., vol. 7, no. 2, pp. 101–108, 1999, doi: 10.1023/A:1021812418325.
- [119] Ohtaki A., Akakura N., and Nakasaki K., Effects of temperature and inoculum on the degradability of poly-ε-caprolactone during composting, *Polym. Degrad. Stab.*, vol. 62, no. 2, pp. 279–284, Nov. 1998, doi: 10.1016/S0141-3910(98)00008-1.
- [120] Pischedda A., Tosin M., and Degli-Innocenti F., Biodegradation of plastics in soil: The effect of temperature, *Polym. Degrad. Stab.*, vol. 170, p. 109017, Dec. 2019, doi: 10.1016/j.polymdegradstab.2019.109017.
- Bläsing M. and Amelung W., Plastics in soil: Analytical methods and possible sources, Sci. Total Environ., vol. 612, pp. 422–435, Jan. 2018, doi: 10.1016/j.scitotenv.2017.08.086.
- [122] Okoffo E. D., O'Brien S., Ribeiro F., et al., Plastic particles in soil: state of the knowledge on sources, occurrence and distribution, analytical methods and ecological impacts, *Environ. Sci. Process. Impacts*, vol. 23, no. 2, pp. 240–274, 2021, doi: 10.1039/D0EM00312C.
- [123] Zee M. Van Der, Biodegradability of biodegradable mulch film, *Wageningen Food & Biobased Research*, 2021. https://edepot.wur.nl/544211 (accessed Jan. 31, 2022).
- [124] SAPEA, Biodegradibility of Plastics in the Open Environment, 2020. https://www.sapea.info/topics/biodegradability-of-plastics/ (accessed Jan. 31, 2022).
- [125] Sander M., Biodegradation of Polymeric Mulch Films in Agricultural Soils: Concepts, Knowledge Gaps, and Future Research Directions, *Environ. Sci. Technol.*, vol. 53, no. 5, pp. 2304–2315, Mar. 2019, doi: 10.1021/acs.est.8b05208.

- [126] Kasirajan S. and Ngouajio M., Polyethylene and biodegradable mulches for agricultural applications: a review, Agron. Sustain. Dev., vol. 32, no. 2, pp. 501–529, Apr. 2012, doi: 10.1007/s13593-011-0068-3.
- [127] Nannipieri P., Soil Is Still an Unknown Biological System, *Appl. Sci.*, vol. 10, no. 11, p. 3717, May 2020, doi: 10.3390/app10113717.
- [128] Nannipieri P., Ascher J., Ceccherini M. T., et al., Microbial diversity and soil functions, Eur. J. Soil Sci., vol. 54, no. 4, pp. 655–670, Dec. 2003, doi: 10.1046/j.1351-0754.2003.0556.x.
- [129] USDA, From the Surface Down. An Introduction to Soil Surveys for Agronomic Use, 2010. https://nrcspad.sc.egov.usda.gov/DistributionCenter/product.aspx?ProductID=449 (accessed Jan. 31, 2022).
- [130] ISRIC, World Soil Information, 2022. https://www.isric.org/ (accessed Jan. 31, 2022).
- [131] Bastioli, Catia E., Handbook of Biodegradable Polymers (2nd Edition), no. July. 2016.
- [132] Hoffmann H., Soil classification, 2022. https://www.mathworks.com/matlabcentral/fileexchange/45468-soil_classificationsand-clay-t-varargin (accessed Feb. 02, 2022).
- [133] Delgado A. and Gómez J. A., The Soil. Physical, Chemical and Biological Properties, in *Principles of Agronomy for Sustainable Agriculture*, Cham: Springer International Publishing, 2016, pp. 15–26.
- [134] Wolf A. B., Vos M., de Boer W., and Kowalchuk G. A., Impact of Matric Potential and Pore Size Distribution on Growth Dynamics of Filamentous and Non-Filamentous Soil Bacteria, *PLoS One*, vol. 8, no. 12, p. e83661, Dec. 2013, doi: 10.1371/journal.pone.0083661.
- [135] Arnosti C., Bell C., Moorhead D. L., et al., Extracellular enzymes in terrestrial, freshwater, and marine environments: perspectives on system variability and common research needs, *Biogeochemistry*, vol. 117, no. 1, pp. 5–21, Jan. 2014, doi: 10.1007/s10533-013-9906-5.
- [136] USDA, Soil Fungi, 2022. https://www.nrcs.usda.gov/wps/portal/nrcs/detailfull/soils/health/biology/?cid=nrcs14 2p2_053864 (accessed Jan. 31, 2022).
- [137] WSU, Acid Soils: How Do They Interact with Root Diseases?, 2022. http://pubs.cahnrs.wsu.edu/publications/pubs/fs195e/ (accessed Mar. 13, 2021).
- [138] Kyrikou I. and Briassoulis D., Biodegradation of Agricultural Plastic Films: A Critical Review, J. Polym. Environ., vol. 15, no. 2, pp. 125–150, Apr. 2007, doi: 10.1007/s10924-007-0053-8.
- [139] Brodhagen M., Peyron M., Miles C., and Inglis D. A., Biodegradable plastic

agricultural mulches and key features of microbial degradation, *Appl. Microbiol. Biotechnol.*, vol. 99, no. 3, pp. 1039–1056, Feb. 2015, doi: 10.1007/s00253-014-6267-5.

- [140] Zumstein M. T., Schintlmeister A., Nelson T. F., et al., Biodegradation of synthetic polymers in soils: Tracking carbon into CO2 and microbial biomass, Sci. Adv., vol. 4, no. 7, Jul. 2018, doi: 10.1126/sciadv.aas9024.
- [141] Kijchavengkul T., Auras R., Rubino M., et al., Assessment of aliphatic-aromatic copolyester biodegradable mulch films. Part I: Field study, *Chemosphere*, vol. 71, no. 5, pp. 942–953, Mar. 2008, doi: 10.1016/j.chemosphere.2007.10.074.
- [142] Samantaray P. K., Little A., Wemyss A. M., et al., Design and Control of Compostability in Synthetic Biopolyesters, ACS Sustain. Chem. Eng., vol. 9, no. 28, pp. 9151–9164, Jul. 2021, doi: 10.1021/acssuschemeng.1c01424.
- [143] Vázquez M. A. and Soto M., The efficiency of home composting programmes and compost quality, *Waste Manag.*, vol. 64, pp. 39–50, Jun. 2017, doi: 10.1016/j.wasman.2017.03.022.
- [144] Mitaftsi O. and Smith S. R., Quantifying Household Waste Diversion from Landfill Disposal by Home Composting and Kerbside Collection, 2006.
- [145] The Bokashi Bucket, The Bokashi Bucket, 2022. http://thebokashibucket.com/ (accessed Jan. 31, 2022).
- [146] Bokashi, Bokashi One, 2022. https://www.bokashi.com.au/Bokashi+One/How+it+Works.html (accessed Jan. 31, 2022).
- [147] Farachi F., Bettas Ardisson G., and Degli Innocenti F., 3. Environmental fate and ecotoxicity assessment of biodegradable polymers, in *Handbook of Biodegradable Polymers*, De Gruyter, 2020, pp. 45–74.
- [148] Azim K., Soudi B., Boukhari S., et al., Composting parameters and compost quality: a literature review, Org. Agric., vol. 8, no. 2, pp. 141–158, Jun. 2018, doi: 10.1007/s13165-017-0180-z.
- [149] Adamcová D., Zloch J., Brtnický M., and Vaverková M. D., Biodegradation/Disintegration of Selected Range of Polymers: Impact on the Compost Quality, J. Polym. Environ., vol. 27, no. 4, pp. 892–899, Apr. 2019, doi: 10.1007/s10924-019-01393-3.
- [150] Pergola M., Persiani A., Palese A. M., et al., Composting: The way for a sustainable agriculture, Appl. Soil Ecol., vol. 123, no. October 2017, pp. 744–750, Feb. 2018, doi: 10.1016/j.apsoil.2017.10.016.
- [151] Graves R. E., Chapter 2 Composting, in *Environmental Engineering*, vol. Part 637, no. February, 2000, p. 88.
- [152] Stofella P. and Kahn B., Compost utilization in horticultural cropping systems. 2001.

- [153] USCC, STA Certified Compost Test Result Limits for Compost Manufacturers, 2022. https://www.compostingcouncil.org/page/TestResultLimits (accessed Jan. 31, 2022).
- [154] Kale G., Auras R., and Singh S. P., Degradation of Commercial Biodegradable Packages under Real Composting and Ambient Exposure Conditions, J. Polym. Environ., vol. 14, no. 3, pp. 317–334, Nov. 2006, doi: 10.1007/s10924-006-0015-6.
- [155] Association of Recyclers of Oregon, A Message from Composters Serving Oregon, 2020. https://oregonrecyclers.org/blog/composters-say-no-compostable-packagingand-serviceware (accessed Jan. 31, 2022).
- [156] Schwarz A. E., Ligthart T. N., Boukris E., and van Harmelen T., Sources, transport, and accumulation of different types of plastic litter in aquatic environments: A review study, *Mar. Pollut. Bull.*, vol. 143, no. April, pp. 92–100, Jun. 2019, doi: 10.1016/j.marpolbul.2019.04.029.
- [157] Law K. L., Plastics in the Marine Environment, Ann. Rev. Mar. Sci., vol. 9, no. 1, pp. 205–229, Jan. 2017, doi: 10.1146/annurev-marine-010816-060409.
- [158] Welden N. A. and Lusher A., Microplastics, in *Plastic Waste and Recycling*, vol. 32, no. 3, Elsevier, 2020, pp. 223–249.
- [159] Kingsford M. J., Marine ecosystem Encyclopedia Britannica, 2022. https://www.britannica.com/science/marine-ecosystem#ref70716 (accessed Jan. 31, 2022).
- [160] René M., The marine environment Encyclopedia of the Environment, 2019. https://www.encyclopedie-environnement.org/en/water/marine-environment/ (accessed Jan. 31, 2022).
- [161] Amaral-Zettler L. A., Zettler E. R., and Mincer T. J., Ecology of the plastisphere, *Nat. Rev. Microbiol.*, vol. 18, no. 3, pp. 139–151, Mar. 2020, doi: 10.1038/s41579-019-0308-0.
- [162] Wright R. J., Erni-Cassola G., Zadjelovic V., et al., Marine Plastic Debris: A New Surface for Microbial Colonization, Environ. Sci. Technol., vol. 54, no. 19, pp. 11657– 11672, Oct. 2020, doi: 10.1021/acs.est.0c02305.
- [163] Jacquin J., Cheng J., Odobel C., et al., Microbial Ecotoxicology of Marine Plastic Debris: A Review on Colonization and Biodegradation by the "Plastisphere," Front. Microbiol., vol. 10, no. APR, pp. 1–16, Apr. 2019, doi: 10.3389/fmicb.2019.00865.
- [164] Encyclopædia Britannica, Pelagic zone, 2022. https://www.britannica.com/science/pelagic-zone#/media/1/449062/10293 (accessed Jan. 31, 2022).
- [165] Ahmed T., Shahid M., Azeem F., et al., Biodegradation of plastics: current scenario and future prospects for environmental safety, *Environ. Sci. Pollut. Res.*, vol. 25, no. 8, pp. 7287–7298, Mar. 2018, doi: 10.1007/s11356-018-1234-9.

- [166] Iñiguez-Franco F., Auras R., Burgess G., et al., Concurrent solvent induced crystallization and hydrolytic degradation of PLA by water-ethanol solutions, Polymer (Guildf)., vol. 99, pp. 315–323, Sep. 2016, doi: 10.1016/j.polymer.2016.07.018.
- [167] Copinet A., Bertrand C., Govindin S., et al., Effects of ultraviolet light (315 nm), temperature and relative humidity on the degradation of polylactic acid plastic films, *Chemosphere*, vol. 55, no. 5, pp. 763–773, May 2004, doi: 10.1016/j.chemosphere.2003.11.038.
- [168] Selke S., Auras R., Nguyen T. A., et al., Evaluation of Biodegradation-Promoting Additives for Plastics, Environ. Sci. Technol., vol. 49, no. 6, pp. 3769–3777, Mar. 2015, doi: 10.1021/es504258u.
- [169] Drew M. C. and Lynch J. M., Soil Anaerobiosis, Microorganisms, and Root Function, Annu. Rev. Phytopathol., vol. 18, no. 1, pp. 37–66, Sep. 1980, doi: 10.1146/annurev.py.18.090180.000345.
- [170] Wilpiszeski R. L., Aufrecht J. A., Retterer S. T., et al., Soil Aggregate Microbial Communities: Towards Understanding Microbiome Interactions at Biologically Relevant Scales, Appl. Environ. Microbiol., vol. 85, no. 14, Jul. 2019, doi: 10.1128/AEM.00324-19.
- [171] Sailema-Palate G. P., Vidaurre A., Campillo-Fernández A. J., and Castilla-Cortázar I., A comparative study on Poly(ε-caprolactone) film degradation at extreme pH values, *Polym. Degrad. Stab.*, vol. 130, pp. 118–125, Aug. 2016, doi: 10.1016/j.polymdegradstab.2016.06.005.
- [172] Mergaert J., Webb A., Anderson C., et al., Microbial degradation of poly(3-hydroxybutyrate) and poly(3-hydroxybutyrate-co-3-hydroxyvalerate) in soils, Appl. Environ. Microbiol., vol. 59, no. 10, pp. 3233–3238, Oct. 1993, doi: 10.1128/aem.59.10.3233-3238.1993.
- [173] Marten E., Müller R.-J., and Deckwer W.-D., Studies on the enzymatic hydrolysis of polyesters. II. Aliphatic–aromatic copolyesters, *Polym. Degrad. Stab.*, vol. 88, no. 3, pp. 371–381, Jun. 2005, doi: 10.1016/j.polymdegradstab.2004.12.001.
- [174] Li Z., Lin H., Ishii N., et al., Study of enzymatic degradation of microbial copolyesters consisting of 3-hydroxybutyrate and medium-chain-length 3-hydroxyalkanoates, *Polym. Degrad. Stab.*, vol. 92, no. 9, pp. 1708–1714, Sep. 2007, doi: 10.1016/j.polymdegradstab.2007.06.001.
- [175] Kijchavengkul T., Auras R., Rubino M., et al., Formulation selection of aliphatic aromatic biodegradable polyester film exposed to UV/solar radiation, Polym. Degrad. Stab., vol. 96, no. 10, pp. 1919–1926, Oct. 2011, doi: 10.1016/j.polymdegradstab.2011.07.001.
- [176] Sonchaeng U., Iñiguez-Franco F., Auras R., et al., Poly(lactic acid) mass transfer properties, Prog. Polym. Sci., vol. 86, pp. 85–121, Nov. 2018, doi: 10.1016/j.progpolymsci.2018.06.008.

- [177] Fang X. and Vitrac O., Predicting diffusion coefficients of chemicals in and through packaging materials, *Crit. Rev. Food Sci. Nutr.*, vol. 57, no. 2, pp. 275–312, Jan. 2017, doi: 10.1080/10408398.2013.849654.
- [178] Eldsäter C., The biodegradation of amorphous and crystalline regions in film-blown poly(e-caprolactone), *Polymer (Guildf)*., vol. 41, no. 4, pp. 1297–1304, Feb. 2000, doi: 10.1016/S0032-3861(99)00278-5.
- [179] Bher A., Unalan I. U., Auras R., et al., Graphene modifies the biodegradation of poly(lactic acid)-thermoplastic cassava starch reactive blend films, Polym. Degrad. Stab., vol. 164, pp. 187–197, Jun. 2019, doi: 10.1016/j.polymdegradstab.2019.04.014.
- [180] Donlan R. M., Biofilms: Microbial Life on Surfaces, *Emerg. Infect. Dis.*, vol. 8, no. 9, pp. 881–890, Sep. 2002, doi: 10.3201/eid0809.020063.
- [181] Fukuda N. and Tsuji H., Physical properties and enzymatic hydrolysis of poly(Llactide)-TiO2 composites, J. Appl. Polym. Sci., vol. 96, no. 1, pp. 190–199, Apr. 2005, doi: 10.1002/app.21411.
- [182] Tsuji H., Ishida T., and Fukuda N., Surface hydrophilicity and enzymatic hydrolyzability of biodegradable polyesters: 1. effects of alkaline treatment, *Polym. Int.*, vol. 52, no. 5, pp. 843–852, May 2003, doi: 10.1002/pi.1199.
- [183] Maeda H., Yamagata Y., Abe K., et al., Purification and characterization of a biodegradable plastic-degrading enzyme from Aspergillus oryzae, Appl. Microbiol. Biotechnol., vol. 67, no. 6, pp. 778–788, Jun. 2005, doi: 10.1007/s00253-004-1853-6.
- [184] Tribedi P., Sarkar S., Mukherjee K., and Sil A. K., Isolation of a novel Pseudomonas sp from soil that can efficiently degrade polyethylene succinate, *Environ. Sci. Pollut. Res.*, vol. 19, no. 6, pp. 2115–2124, Jul. 2012, doi: 10.1007/s11356-011-0711-1.
- [185] Rosa D. ., Lotto N. ., Lopes D. ., and Guedes C. G. ., The use of roughness for evaluating the biodegradation of poly-8-(hydroxybutyrate) and poly-8-(hydroxybutyrate-co-8valerate), *Polym. Test.*, vol. 23, no. 1, pp. 3–8, Feb. 2004, doi: 10.1016/S0142-9418(03)00042-4.
- [186] Hallab N. J., Bundy K. J., O'Connor K., et al., Evaluation of Metallic and Polymeric Biomaterial Surface Energy and Surface Roughness Characteristics for Directed Cell Adhesion, *Tissue Eng.*, vol. 7, no. 1, pp. 55–71, Feb. 2001, doi: 10.1089/107632700300003297.
- [187] Mercier A., Gravouil K., Aucher W., et al., Fate of Eight Different Polymers under Uncontrolled Composting Conditions: Relationships Between Deterioration, Biofilm Formation, and the Material Surface Properties, *Environ. Sci. Technol.*, vol. 51, no. 4, pp. 1988–1997, Feb. 2017, doi: 10.1021/acs.est.6b03530.
- [188] Taylor R. L., Verran J., Lees G. C., and Ward A. J. P., The influence of substratum topography on bacterial adhesion to polymethyl methacrylate, *J. Mater. Sci. Mater. Med.*, vol. 9, no. 1, pp. 17–22, 1998, doi: 10.1023/A:1008874326324.

- [189] Chinaglia S., Tosin M., and Degli-Innocenti F., Biodegradation rate of biodegradable plastics at molecular level, *Polym. Degrad. Stab.*, vol. 147, no. October 2017, pp. 237– 244, Jan. 2018, doi: 10.1016/j.polymdegradstab.2017.12.011.
- [190] Komiyama K., Omura T., and Iwata T., Effect of morphology and molecular orientation on environmental water biodegradability of poly[(R)-3-hydroxybutyrateco-(R)-3-hydroxyvalerate], *Polym. Degrad. Stab.*, vol. 193, p. 109719, Nov. 2021, doi: 10.1016/j.polymdegradstab.2021.109719.
- [191] Philp J. C., Bartsev A., Ritchie R. J., et al., Bioplastics science from a policy vantage point, N. Biotechnol., vol. 30, no. 6, pp. 635–646, Sep. 2013, doi: 10.1016/j.nbt.2012.11.021.
- [192] Briassoulis D. and Dejean C., Critical Review of Norms and Standards for Biodegradable Agricultural Plastics Part I. Biodegradation in Soil, J. Polym. Environ., vol. 18, no. 3, pp. 384–400, Sep. 2010, doi: 10.1007/s10924-010-0168-1.
- [193] Wilde B. De, Mortier N., Briassoulis D., *et al.*, Report on current relevant biodegradation and ecotoxicity standards, 2013.
- [194] Briassoulis D., Dejean C., and Picuno P., Critical Review of Norms and Standards for Biodegradable Agricultural Plastics Part II: Composting, J. Polym. Environ., vol. 18, no. 3, pp. 364–383, Sep. 2010, doi: 10.1007/s10924-010-0222-z.
- [195] Ruggero F., Gori R., and Lubello C., Methodologies to assess biodegradation of bioplastics during aerobic composting and anaerobic digestion: A review, *Waste Manag. Res.*, vol. 37, no. 10, pp. 959–975, Oct. 2019, doi: 10.1177/0734242X19854127.
- [196] De Wilde B., 5. International and national norms on biodegradability and certification procedures, in *Handbook of Biodegradable Polymers*, De Gruyter, 2020, pp. 115–146.
- [197] Rudnik E., Biodegradability Testing of Compostable Polymer Materials, in *Handbook* of *Biopolymers and Biodegradable Plastics*, no. 2008, Elsevier, 2013, pp. 213–263.
- [198] Guo W., Tao J., Yang C., *et al.*, Introduction of Environmentally Degradable Parameters to Evaluate the Biodegradability of Biodegradable Polymers, *PLoS One*, vol. 7, no. 5, p. e38341, May 2012, doi: 10.1371/journal.pone.0038341.
- [199] Guo W., Tao J., Yang C., *et al.*, The rapid evaluation of material biodegradability using an improved ISO 14852 method with a microbial community, *Polym. Test.*, vol. 29, no. 7, pp. 832–839, Oct. 2010, doi: 10.1016/j.polymertesting.2010.07.004.
- [200] Kulkarni A. and Narayan R., Effects of Modified Thermoplastic Starch on Crystallization Kinetics and Barrier Properties of PLA, *Polymers (Basel).*, vol. 13, no. 23, p. 4125, Nov. 2021, doi: 10.3390/polym13234125.
- [201] Funabashi M., Ninomiya F., and Kunioka M., Biodegradation of Polycaprolactone Powders Proposed as Reference Test Materials for International Standard of Biodegradation Evaluation Method, J. Polym. Environ., vol. 15, no. 1, pp. 7–17, Feb. 2007, doi: 10.1007/s10924-006-0041-4.

- [202] Massardier-Nageotte V., Pestre C., Cruard-Pradet T., and Bayard R., Aerobic and anaerobic biodegradability of polymer films and physico-chemical characterization, *Polym. Degrad. Stab.*, vol. 91, no. 3, pp. 620–627, Mar. 2006, doi: 10.1016/j.polymdegradstab.2005.02.029.
- [203] Mezzanotte V., Bertani R., Innocenti F. D., and Tosin M., Influence of inocula on the results of biodegradation tests, *Polym. Degrad. Stab.*, vol. 87, no. 1, pp. 51–56, Jan. 2005, doi: 10.1016/j.polymdegradstab.2004.06.009.
- [204] Moura I., Machado A. V, Duarte F. M., and Nogueira R., Biodegradability assessment of aliphatic polyesters-based blends using standard methods, J. Appl. Polym. Sci., vol. 119, no. 6, pp. 3338–3346, Mar. 2011, doi: 10.1002/app.32966.
- [205] Cho H. S., Moon H. S., Kim M., et al., Biodegradability and biodegradation rate of poly(caprolactone)-starch blend and poly(butylene succinate) biodegradable polymer under aerobic and anaerobic environment, *Waste Manag.*, vol. 31, no. 3, pp. 475–480, Mar. 2011, doi: 10.1016/j.wasman.2010.10.029.
- [206] Tosin M., Weber M., Siotto M., et al., Laboratory Test Methods to Determine the Degradation of Plastics in Marine Environmental Conditions, Front. Microbiol., vol. 3, no. JUN, p. 225, 2012, doi: 10.3389/fmicb.2012.00225.
- [207] Iggui K., Le Moigne N., Kaci M., et al., A biodegradation study of poly(3hydroxybutyrate-co-3-hydroxyvalerate)/organoclay nanocomposites in various environmental conditions, *Polym. Degrad. Stab.*, vol. 119, pp. 77–86, Sep. 2015, doi: 10.1016/j.polymdegradstab.2015.05.002.
- [208] Jeszeová L., Puškárová A., Bučková M., et al., Microbial communities responsible for the degradation of poly(lactic acid)/poly(3-hydroxybutyrate) blend mulches in soil burial respirometric tests, World J. Microbiol. Biotechnol., vol. 34, no. 7, p. 101, Jul. 2018, doi: 10.1007/s11274-018-2483-y.
- [209] Briassoulis D., Mistriotis A., Mortier N., and Tosin M., A horizontal test method for biodegradation in soil of bio-based and conventional plastics and lubricants, J. Clean. Prod., vol. 242, p. 118392, Jan. 2020, doi: 10.1016/j.jclepro.2019.118392.
- [210] Šerá J., Serbruyns L., De Wilde B., and Koutný M., Accelerated biodegradation testing of slowly degradable polyesters in soil, *Polym. Degrad. Stab.*, vol. 171, p. 109031, Jan. 2020, doi: 10.1016/j.polymdegradstab.2019.109031.
- [211] Briassoulis D., Pikasi A., Papardaki N. G., and Mistriotis A., Aerobic biodegradation of bio-based plastics in the seawater/sediment interface (sublittoral) marine environment of the coastal zone – Test method under controlled laboratory conditions, *Sci. Total Environ.*, vol. 722, p. 137748, Jun. 2020, doi: 10.1016/j.scitotenv.2020.137748.
- [212] Srimalanon P., Prapagdee B., and Sombatsompop N., Soil Inoculation with Pseudomonas geniculata WS3 for Accelerating the Biodegradation Process of In Situ Compatibilized PBS/PLA Blends Doped with HPQM, J. Polym. Environ., vol. 28, no.

4, pp. 1138–1149, Apr. 2020, doi: 10.1007/s10924-020-01670-6.

- [213] Pan H., Hao Y., Zhao Y., et al., Improved mechanical properties, barrier properties and degradation behavior of poly(butylenes adipate-co-terephthalate)/poly(propylene carbonate) films, Korean J. Chem. Eng., vol. 34, no. 5, pp. 1294–1304, May 2017, doi: 10.1007/s11814-017-0066-5.
- [214] Tosin M., Pischedda A., and Degli-Innocenti F., Biodegradation kinetics in soil of a multi-constituent biodegradable plastic, *Polym. Degrad. Stab.*, vol. 166, pp. 213–218, Aug. 2019, doi: 10.1016/j.polymdegradstab.2019.05.034.
- [215] Gómez E. F. and Michel F. C., Biodegradability of conventional and bio-based plastics and natural fiber composites during composting, anaerobic digestion and long-term soil incubation, *Polym. Degrad. Stab.*, vol. 98, no. 12, pp. 2583–2591, Dec. 2013, doi: 10.1016/j.polymdegradstab.2013.09.018.
- [216] Satti S. M., Shah A. A., Marsh T. L., and Auras R., Biodegradation of Poly(lactic acid) in Soil Microcosms at Ambient Temperature: Evaluation of Natural Attenuation, Bioaugmentation and Bio-stimulation, J. Polym. Environ., vol. 26, no. 9, pp. 3848–3857, Sep. 2018, doi: 10.1007/s10924-018-1264-x.
- [217] Anunciado M. B., Hayes D. G., Astner A. F., et al., Effect of Environmental Weathering on Biodegradation of Biodegradable Plastic Mulch Films under Ambient Soil and Composting Conditions, J. Polym. Environ., vol. 29, no. 9, pp. 2916–2931, Sep. 2021, doi: 10.1007/s10924-021-02088-4.
- [218] Palsikowski P. A., Kuchnier C. N., Pinheiro I. F., and Morales A. R., Biodegradation in Soil of PLA/PBAT Blends Compatibilized with Chain Extender, J. Polym. Environ., vol. 26, no. 1, pp. 330–341, Jan. 2018, doi: 10.1007/s10924-017-0951-3.
- [219] Pérez-Arauz A. O., Aguilar-Rabiela A. E., Vargas-Torres A., et al., Production and characterization of biodegradable films of a novel polyhydroxyalkanoate (PHA) synthesized from peanut oil, Food Packag. Shelf Life, vol. 20, p. 100297, Jun. 2019, doi: 10.1016/j.fpsl.2019.01.001.
- [220] Pattanasuttichonlakul W., Sombatsompop N., and Prapagdee B., Accelerating biodegradation of PLA using microbial consortium from dairy wastewater sludge combined with PLA-degrading bacterium, *Int. Biodeterior. Biodegradation*, vol. 132, no. May, pp. 74–83, Aug. 2018, doi: 10.1016/j.ibiod.2018.05.014.
- [221] Arcos-Hernandez M. V., Laycock B., Pratt S., et al., Biodegradation in a soil environment of activated sludge derived polyhydroxyalkanoate (PHBV), Polym. Degrad. Stab., vol. 97, no. 11, pp. 2301–2312, Nov. 2012, doi: 10.1016/j.polymdegradstab.2012.07.035.
- [222] Lammi S., Gastaldi E., Gaubiac F., and Angellier-Coussy H., How olive pomace can be valorized as fillers to tune the biodegradation of PHBV based composites, *Polym. Degrad.* Stab., vol. 166, pp. 325–333, Aug. 2019, doi: 10.1016/j.polymdegradstab.2019.06.010.

- [223] Modelli A., Calcagno B., and Scandola M., Kinetics of aerobic polymer degradation in soil by means of the ASTM D 5988-96 standard method, J. Environ. Polym. Degrad., vol. 7, no. 2, pp. 109–116, 1999, doi: 10.1023/A:1021864402395.
- [224] Phua Y. J., Lau N. S., Sudesh K., et al., Biodegradability studies of poly(butylene succinate)/organo-montmorillonite nanocomposites under controlled compost soil conditions: Effects of clay loading and compatibiliser, *Polym. Degrad. Stab.*, vol. 97, no. 8, pp. 1345–1354, Aug. 2012, doi: 10.1016/j.polymdegradstab.2012.05.024.
- [225] Muniyasamy S., Ofosu O., John M. J., and Anandjiwala R. D., Mineralization of Poly(lactic acid) (PLA), Poly(3-hydroxybutyrate-co-valerate) (PHBV) and PLA/PHBV Blend in Compost and Soil Environments, J. Renew. Mater., vol. 4, no. 2, pp. 133–145, Apr. 2016, doi: 10.7569/JRM.2016.634104.
- [226] Apinya T., Sombatsompop N., and Prapagdee B., Selection of a Pseudonocardia sp. RM423 that accelerates the biodegradation of poly(lactic) acid in submerged cultures and in soil microcosms, *Int. Biodeterior. Biodegradation*, vol. 99, pp. 23–30, Apr. 2015, doi: 10.1016/j.ibiod.2015.01.001.
- [227] Trhlíková O., Vlčková V., Abbrent S., et al., Microbial and abiotic degradation of fully aliphatic polyurethane foam suitable for biotechnologies, Polym. Degrad. Stab., vol. 194, p. 109764, Dec. 2021, doi: 10.1016/j.polymdegradstab.2021.109764.
- [228] Gómez E. F., Luo X., Li C., et al., Biodegradability of crude glycerol-based polyurethane foams during composting, anaerobic digestion and soil incubation, *Polym. Degrad. Stab.*, vol. 102, no. 1, pp. 195–203, Apr. 2014, doi: 10.1016/j.polymdegradstab.2014.01.008.
- [229] Ghasemlou M., Daver F., Murdoch B. J., et al., Biodegradation of novel bioplastics made of starch, polyhydroxyurethanes and cellulose nanocrystals in soil environment, *Sci. Total Environ.*, vol. 815, p. 152684, Apr. 2022, doi: 10.1016/j.scitotenv.2021.152684.
- [230] Briassoulis D., Pikasi A., Briassoulis C., and Mistriotis A., Disintegration behaviour of bio-based plastics in coastal zone marine environments: A field experiment under natural conditions, *Sci. Total Environ.*, vol. 688, pp. 208–223, Oct. 2019, doi: 10.1016/j.scitotenv.2019.06.129.
- [231] Thellen C., Coyne M., Froio D., et al., A Processing, Characterization and Marine Biodegradation Study of Melt-Extruded Polyhydroxyalkanoate (PHA) Films, J. Polym. Environ., vol. 16, no. 1, pp. 1–11, Jan. 2008, doi: 10.1007/s10924-008-0079-6.
- [232] Thellen C., Cheney S., and Ratto J. A., Melt processing and characterization of polyvinyl alcohol and polyhydroxyalkanoate multilayer films, J. Appl. Polym. Sci., vol. 127, no. 3, pp. 2314–2324, Feb. 2013, doi: 10.1002/app.37850.
- [233] Meereboer K. W., Pal A. K., Cisneros-López E. O., et al., The effect of natural fillers on the marine biodegradation behaviour of poly(3-hydroxybutyrate-co-3hydroxyvalerate) (PHBV), Sci. Rep., vol. 11, no. 1, p. 911, Dec. 2021, doi:
10.1038/s41598-020-78122-7.

- [234] Altieri R., Seggiani M., Esposito A., et al., Thermoplastic Blends Based on Poly(Butylene Succinate-co-Adipate) and Different Collagen Hydrolysates from Tanning Industry—II: Aerobic Biodegradation in Composting Medium, J. Polym. Environ., vol. 29, no. 10, pp. 3375–3388, Oct. 2021, doi: 10.1007/s10924-021-02124-3.
- [235] Eubeler J. P., Zok S., Bernhard M., and Knepper T. P., Environmental biodegradation of synthetic polymers I. Test methodologies and procedures, *TrAC Trends Anal. Chem.*, vol. 28, no. 9, pp. 1057–1072, Oct. 2009, doi: 10.1016/j.trac.2009.06.007.
- [236] Zee M. van der, 1. Methods for evaluating the biodegradability of environmentally degradable polymers, in *Handbook of Biodegradable Polymers*, De Gruyter, 2020, pp. 1–22.
- [237] Niaounakis M., Definitions and Assessment of (Bio)degradation, in *Biopolymers Reuse, Recycling, and Disposal*, Elsevier, 2013, pp. 77–94.
- [238] Itävaara M., Karjomaa S., and Selin J.-F., Biodegradation of polylactide in aerobic and anaerobic thermophilic conditions, *Chemosphere*, vol. 46, no. 6, pp. 879–885, Feb. 2002, doi: 10.1016/S0045-6535(01)00163-1.
- [239] Kim M. N. and Park S. T., Degradation of poly(L-lactide) by a mesophilic bacterium, J. Appl. Polym. Sci., vol. 116, no. 5, p. NA-NA, 2010, doi: 10.1002/app.31950.
- [240] Osman M., Satti S. M., Luqman A., et al., Degradation of Polyester Polyurethane by Aspergillus sp. Strain S45 Isolated from Soil, J. Polym. Environ., vol. 26, no. 1, pp. 301–310, Jan. 2018, doi: 10.1007/s10924-017-0954-0.
- [241] Šerá J., Stloukal P., Jančová P., et al., Accelerated Biodegradation of Agriculture Film Based on Aromatic–Aliphatic Copolyester in Soil under Mesophilic Conditions, J. Agric. Food Chem., vol. 64, no. 28, pp. 5653–5661, Jul. 2016, doi: 10.1021/acs.jafc.6b01786.
- [242] Jeon H. J. and Kim M. N., Biodegradation of poly(l-lactide) (PLA) exposed to UV irradiation by a mesophilic bacterium, *Int. Biodeterior. Biodegradation*, vol. 85, pp. 289–293, Nov. 2013, doi: 10.1016/j.ibiod.2013.08.013.
- [243] Kim M., Lee A., Yoon J., and Chin I., Biodegradation of poly(3-hydroxybutyrate), Sky-Green® and Mater-Bi® by fungi isolated from soils, *Eur. Polym. J.*, vol. 36, no. 8, pp. 1677–1685, Aug. 2000, doi: 10.1016/S0014-3057(99)00219-0.
- [244] Deroiné M., César G., Le Duigou A., et al., Natural Degradation and Biodegradation of Poly(3-Hydroxybutyrate-co-3-Hydroxyvalerate) in Liquid and Solid Marine Environments, J. Polym. Environ., vol. 23, no. 4, pp. 493–505, Dec. 2015, doi: 10.1007/s10924-015-0736-5.
- [245] Greene J., Marine Biodegradation of PLA, PHA, and Bio-additive Polyethylene Based on ASTM D7081, 2012.

- [246] Shah Z., Gulzar M., Hasan F., and Shah A. A., Degradation of polyester polyurethane by an indigenously developed consortium of Pseudomonas and Bacillus species isolated from soil, *Polym. Degrad. Stab.*, vol. 134, pp. 349–356, Dec. 2016, doi: 10.1016/j.polymdegradstab.2016.11.003.
- [247] Shah Z., Hasan F., Krumholz L., et al., Degradation of polyester polyurethane by newly isolated Pseudomonas aeruginosa strain MZA-85 and analysis of degradation products by GC–MS, Int. Biodeterior. Biodegradation, vol. 77, pp. 114–122, Feb. 2013, doi: 10.1016/j.ibiod.2012.11.009.
- [248] Shah Z., Krumholz L., Aktas D. F., et al., Degradation of polyester polyurethane by a newly isolated soil bacterium, Bacillus subtilis strain MZA-75, *Biodegradation*, vol. 24, no. 6, pp. 865–877, Nov. 2013, doi: 10.1007/s10532-013-9634-5.
- [249] Han Y., Teng Y., Wang X., et al., Soil Type Driven Change in Microbial Community Affects Poly(butylene adipate- co-terephthalate) Degradation Potential, *Environ. Sci. Technol.*, vol. 55, no. 8, pp. 4648–4657, Apr. 2021, doi: 10.1021/acs.est.0c04850.
- [250] Shah A. A., Hasan F., Akhter J. I., et al., Degradation of polyurethane by novel bacterial consortium isolated from soil, Ann. Microbiol., vol. 58, no. 3, pp. 381–386, Sep. 2008, doi: 10.1007/BF03175532.
- [251] Lee S.-H. and Kim M.-N., Isolation of bacteria degrading poly(butylene succinate-cobutylene adipate) and their lip A gene, *Int. Biodeterior. Biodegradation*, vol. 64, no. 3, pp. 184–190, Jun. 2010, doi: 10.1016/j.ibiod.2010.01.002.
- [252] Jin H.-J., Lee B.-Y., Kim M.-N., and Yoon J.-S., Properties and biodegradation of poly(ethylene adipate) and poly(butylene succinate) containing styrene glycol units, *Eur. Polym. J.*, vol. 36, no. 12, pp. 2693–2698, Dec. 2000, doi: 10.1016/S0014-3057(00)00057-4.
- [253] Rattanapan S., Pasetto P., Pilard J.-F., and Tanrattanakul V., Preparation and properties of bio-based polyurethane foams from natural rubber and polycaprolactone diol, J. Polym. Res., vol. 23, no. 9, p. 182, Sep. 2016, doi: 10.1007/s10965-016-1081-7.
- [254] Kasuya K., Takagi K., Ishiwatari S., et al., Biodegradabilities of various aliphatic polyesters in natural waters, *Polym. Degrad. Stab.*, vol. 59, no. 1–3, pp. 327–332, Jan. 1998, doi: 10.1016/S0141-3910(97)00155-9.
- [255] Doi Y., Kasuya K., Abe H., *et al.*, Evaluation of biodegradabilities of biosynthetic and chemosynthetic polyesters in river water, *Polym. Degrad. Stab.*, vol. 51, no. 3, pp. 281– 286, Mar. 1996, doi: 10.1016/0141-3910(95)00178-6.
- [256] Rudnik E. and Briassoulis D., Comparative Biodegradation in Soil Behaviour of two Biodegradable Polymers Based on Renewable Resources, J. Polym. Environ., vol. 19, no. 1, pp. 18–39, Mar. 2011, doi: 10.1007/s10924-010-0243-7.
- [257] Olewnik-Kruszkowska E., Burkowska-But A., Tarach I., et al., Biodegradation of polylactide-based composites with an addition of a compatibilizing agent in different environments, Int. Biodeterior. Biodegradation, vol. 147, no. March 2019, p. 104840,

Feb. 2020, doi: 10.1016/j.ibiod.2019.104840.

- [258] Richert A. and Dąbrowska G. B., Enzymatic degradation and biofilm formation during biodegradation of polylactide and polycaprolactone polymers in various environments, *Int. J. Biol. Macromol.*, vol. 176, pp. 226–232, Apr. 2021, doi: 10.1016/j.ijbiomac.2021.01.202.
- [259] Soulenthone P., Tachibana Y., Muroi F., et al., Characterization of a mesophilic actinobacteria that degrades poly(butylene adipate-co-terephthalate), Polym. Degrad. Stab., vol. 181, p. 109335, Nov. 2020, doi: 10.1016/j.polymdegradstab.2020.109335.
- [260] Kasuya K., Ishii N., Inoue Y., et al., Characterization of a mesophilic aliphatic– aromatic copolyester-degrading fungus, *Polym. Degrad. Stab.*, vol. 94, no. 8, pp. 1190– 1196, Aug. 2009, doi: 10.1016/j.polymdegradstab.2009.04.013.
- [261] Nakayama A., Yamano N., and Kawasaki N., Biodegradation in seawater of aliphatic polyesters, *Polym. Degrad. Stab.*, vol. 166, pp. 290–299, Aug. 2019, doi: 10.1016/j.polymdegradstab.2019.06.006.
- [262] Zumstein M. T., Rechsteiner D., Roduner N., et al., Enzymatic Hydrolysis of Polyester Thin Films at the Nanoscale: Effects of Polyester Structure and Enzyme Active-Site Accessibility, *Environ. Sci. Technol.*, vol. 51, no. 13, pp. 7476–7485, Jul. 2017, doi: 10.1021/acs.est.7b01330.
- [263] Zumstein M. T., Kohler H.-P. E., McNeill K., and Sander M., Enzymatic Hydrolysis of Polyester Thin Films: Real-Time Analysis of Film Mass Changes and Dissipation Dynamics, *Environ. Sci. Technol.*, vol. 50, no. 1, pp. 197–206, Jan. 2016, doi: 10.1021/acs.est.5b04103.
- [264] Zumstein M. T., Kohler H.-P. E., McNeill K., and Sander M., High-Throughput Analysis of Enzymatic Hydrolysis of Biodegradable Polyesters by Monitoring Cohydrolysis of a Polyester-Embedded Fluorogenic Probe, *Environ. Sci. Technol.*, vol. 51, no. 8, pp. 4358–4367, Apr. 2017, doi: 10.1021/acs.est.6b06060.
- [265] Hou Y., Chen J., Sun P., et al., In situ investigations on enzymatic degradation of poly(ε-caprolactone), Polymer (Guildf)., vol. 48, no. 21, pp. 6348–6353, Oct. 2007, doi: 10.1016/j.polymer.2007.08.054.
- [266] Yamashita K., Kikkawa Y., Kurokawa K., and Doi Y., Enzymatic Degradation of Poly(L-lactide) Film by Proteinase K: Quartz Crystal Microbalance and Atomic Force Microscopy Study, *Biomacromolecules*, vol. 6, no. 2, pp. 850–857, Mar. 2005, doi: 10.1021/bm049395v.
- [267] Chamas A., Moon H., Zheng J., et al., Degradation Rates of Plastics in the Environment, ACS Sustain. Chem. Eng., vol. 8, no. 9, pp. 3494–3511, Mar. 2020, doi: 10.1021/acssuschemeng.9b06635.
- [268] Kijchavengkul T. and Auras R., Design Of Biodegradable Aliphatic Aromatic Polyester Films For Agricultural Applications Using Response Surface Methodology, 2010.

- [269] Shah A. A., Hasan F., and Hameed A., Degradation of poly(3-hydroxybutyrate-co-3hydroxyvalerate) by a newly isolated Actinomadura sp. AF-555, from soil, *Int. Biodeterior. Biodegradation*, vol. 64, no. 4, pp. 281–285, Jul. 2010, doi: 10.1016/j.ibiod.2009.10.012.
- [270] Numata K., Hirota T., Kikkawa Y., et al., Enzymatic Degradation Processes of Lamellar Crystals in Thin Films for Poly[(R)-3-hydroxybutyric acid] and Its Copolymers Revealed by Real-Time Atomic Force Microscopy, *Biomacromolecules*, vol. 5, no. 6, pp. 2186–2194, Nov. 2004, doi: 10.1021/bm0497670.
- [271] Murase T., Iwata T., and Doi Y., Direct Observation of Enzymatic Degradation Behavior of Poly[(R)-3-hydroxybutyrate] Lamellar Single Crystals by Atomic Force Microscopy, *Macromolecules*, vol. 34, no. 17, pp. 5848–5853, Aug. 2001, doi: 10.1021/ma001505c.
- [272] Nobes G. A. R., Marchessault R. H., Chanzy H., et al., Splintering of Poly(3hydroxybutyrate) Single Crystals by PHB-Depolymerase A from Pseudomonas lemoignei, *Macromolecules*, vol. 29, no. 26, pp. 8330–8333, Jan. 1996, doi: 10.1021/ma961219u.
- [273] Iwata T., Doi Y., Kasuya K., and Inoue Y., Visualization of Enzymatic Degradation of Poly[(R)-3-hydroxybutyrate] Single Crystals by an Extracellular PHB Depolymerase, *Macromolecules*, vol. 30, no. 4, pp. 833–839, Feb. 1997, doi: 10.1021/ma961352m.
- [274] Lu X. L., Du F. G., Ge X. C., et al., Biodegradability and thermal stability of poly(propylene carbonate)/starch composites, J. Biomed. Mater. Res. Part A, vol. 77A, no. 4, pp. 653–658, Jun. 2006, doi: 10.1002/jbm.a.30664.
- [275] Limsukon W., Auras R., and Selke S., Hydrolytic degradation and lifetime prediction of poly(lactic acid) modified with a multifunctional epoxy-based chain extender, *Polym. Test.*, vol. 80, no. September, pp. 106–108, Dec. 2019, doi: 10.1016/j.polymertesting.2019.106108.
- [276] Kijchavengkul T., Auras R., Rubino M., et al., Assessment of aliphatic-aromatic copolyester biodegradable mulch films. Part II: Laboratory simulated conditions, *Chemosphere*, vol. 71, no. 9, pp. 1607–1616, Apr. 2008, doi: 10.1016/j.chemosphere.2008.01.037.
- [277] Thakur K. A. M., Kean R. T., Hall E. S., et al., 1 H NMR Spectroscopy in the Analysis and Characterization of Poly(lactide), Int. J. Polym. Anal. Charact., vol. 4, no. 5, pp. 379–391, Sep. 1998, doi: 10.1080/10236669808009724.
- [278] Weng Y.-X., Wang L., Zhang M., et al., Biodegradation behavior of P(3HB,4HB)/PLA blends in real soil environments, *Polym. Test.*, vol. 32, no. 1, pp. 60–70, Feb. 2013, doi: 10.1016/j.polymertesting.2012.09.014.
- [279] Mbarki K., Fersi M., Louati I., et al., Biodegradation study of PDLA/cellulose microfibres biocomposites by Pseudomonas aeruginosa, *Environ. Technol.*, vol. 42, no. 5, pp. 731–742, Feb. 2021, doi: 10.1080/09593330.2019.1643926.

- [280] Augusta J., Müller R.-J., and Widdecke H., A rapid evaluation plate-test for the biodegradability of plastics, *Appl. Microbiol. Biotechnol.*, vol. 39, no. 4–5, pp. 673–678, Jul. 1993, doi: 10.1007/BF00205073.
- [281] Palmisano A. C. and Pettigrew C. A., Biodegradability of Plastics, *Bioscience*, vol. 42, no. 9, pp. 680–685, Oct. 1992, doi: 10.2307/1312174.
- [282] Urbanek A. K., Strzelecki M. C., and Mirończuk A. M., The potential of cold-adapted microorganisms for biodegradation of bioplastics, *Waste Manag.*, vol. 119, pp. 72–81, 2020, doi: 10.1016/j.wasman.2020.09.031.
- [283] Kale G., Auras R., Singh S. P., and Narayan R., Biodegradability of polylactide bottles in real and simulated composting conditions, *Polym. Test.*, vol. 26, no. 8, pp. 1049– 1061, Dec. 2007, doi: 10.1016/j.polymertesting.2007.07.006.
- [284] Sander M., Kohler H.-P. E., and McNeill K., Assessing the environmental transformation of nanoplastic through 13C-labelled polymers, *Nat. Nanotechnol.*, vol. 14, no. 4, pp. 301–303, Apr. 2019, doi: 10.1038/s41565-019-0420-3.
- [285] Batiste D. C., De Hoe G. X., Nelson T. F., et al., Site-Specific Mineralization of a Polyester Hydrolysis Product in Natural Soil, ACS Sustain. Chem. Eng., 2022, doi: 10.1021/acssuschemeng.1c07948.
- [286] Albertsson A.-C., Biodegradation of synthetic polymers. II. A limited microbial conversion of 14C in polyethylene to 14CO2 by some soil fungi, J. Appl. Polym. Sci., vol. 22, no. 12, pp. 3419–3433, Dec. 1978, doi: 10.1002/app.1978.070221207.
- [287] Erlandsson B., Karlsson S., and Albertsson A.-C., Correlation between molar mass changes and CO2 evolution from biodegraded14C-labeled ethylene-vinyl alcohol copolymer and ethylene polymers, *Acta Polym.*, vol. 49, no. 7, pp. 363–370, Jul. 1998, doi: 10.1002/(SICI)1521-4044(199807)49:7<363::AID-APOL363>3.0.CO;2-U.
- [288] Albertsson A.-C. and Ljungquist O., Degradable Polymers. I. Synthesis, Characterization, and Long-Term in Vitro Degradation of a 14 C-Labeled Aliphatic Polyester, J. Macromol. Sci. Part A - Chem., vol. 23, no. 3, pp. 393–409, Mar. 1986, doi: 10.1080/00222338608063402.
- [289] Albertsson A.-C., Báanhidi Z. G., and Beyer-Ericsson L.-L., Biodegradation of synthetic polymers. III. The liberation of 14CO2 by molds like fusarium redolens from 14C labeled pulverized high-density polyethylene, J. Appl. Polym. Sci., vol. 22, no. 12, pp. 3435–3447, Dec. 1978, doi: 10.1002/app.1978.070221208.
- [290] Lee S. H., Kim I. Y., and Song W. S., Biodegradation of polylactic acid (PLA) fibers using different enzymes, *Macromol. Res.*, vol. 22, no. 6, pp. 657–663, Jun. 2014, doi: 10.1007/s13233-014-2107-9.
- [291] Akutsu-Shigeno Y., Adachi Y., Yamada C., et al., Isolation of a bacterium that degrades urethane compounds and characterization of its urethane hydrolase, Appl. Microbiol. Biotechnol., vol. 70, no. 4, pp. 422–429, Apr. 2006, doi: 10.1007/s00253-005-0071-1.

- [292] Hajighasemi M., Nocek B. P., Tchigvintsev A., et al., Biochemical and Structural Insights into Enzymatic Depolymerization of Polylactic Acid and Other Polyesters by Microbial Carboxylesterases, *Biomacromolecules*, vol. 17, no. 6, pp. 2027–2039, Jun. 2016, doi: 10.1021/acs.biomac.6b00223.
- [293] Lim H.-A., Raku T., and Tokiwa Y., Hydrolysis of polyesters by serine proteases, *Biotechnol. Lett.*, vol. 27, no. 7, pp. 459–464, Apr. 2005, doi: 10.1007/s10529-005-2217-8.
- [294] Baker P. J., Poultney C., Liu Z., et al., Identification and comparison of cutinases for synthetic polyester degradation, Appl. Microbiol. Biotechnol., vol. 93, no. 1, pp. 229– 240, Jan. 2012, doi: 10.1007/s00253-011-3402-4.
- [295] Suzuki K., Sakamoto H., Shinozaki Y., et al., Affinity purification and characterization of a biodegradable plastic-degrading enzyme from a yeast isolated from the larval midgut of a stag beetle, Aegus laevicollis, Appl. Microbiol. Biotechnol., vol. 97, no. 17, pp. 7679–7688, Sep. 2013, doi: 10.1007/s00253-012-4595-x.
- [296] Shi K., Jing J., Song L., et al., Enzymatic hydrolysis of polyester: Degradation of poly(e-caprolactone) by Candida antarctica lipase and Fusarium solani cutinase, Int. J. Biol. Macromol., vol. 144, pp. 183–189, Feb. 2020, doi: 10.1016/j.ijbiomac.2019.12.105.
- [297] Mao H., Liu H., Gao Z., et al., Biodegradation of poly(butylene succinate) by Fusarium sp. FS1301 and purification and characterization of poly(butylene succinate) depolymerase, Polym. Degrad. Stab., vol. 114, pp. 1–7, Apr. 2015, doi: 10.1016/j.polymdegradstab.2015.01.025.
- [298] Suzuki K., Noguchi M. T., Shinozaki Y., et al., Purification, characterization, and cloning of the gene for a biodegradable plastic-degrading enzyme from Paraphomarelated fungal strain B47-9, Appl. Microbiol. Biotechnol., vol. 98, no. 10, pp. 4457– 4465, May 2014, doi: 10.1007/s00253-013-5454-0.
- [299] Pan W., Bai Z., Su T., and Wang Z., Enzymatic degradation of poly(butylene succinate) with different molecular weights by cutinase, *Int. J. Biol. Macromol.*, vol. 111, pp. 1040–1046, May 2018, doi: 10.1016/j.ijbiomac.2018.01.107.
- [300] Bai Z., Shi K., Su T., and Wang Z., Correlation between the chemical structure and enzymatic hydrolysis of Poly(butylene succinate), Poly(butylene adipate), and Poly(butylene suberate), *Polym. Degrad. Stab.*, vol. 158, pp. 111–118, Dec. 2018, doi: 10.1016/j.polymdegradstab.2018.10.024.
- [301] Shinozaki Y., Morita T., Cao X., et al., Biodegradable plastic-degrading enzyme from Pseudozyma antarctica: cloning, sequencing, and characterization, Appl. Microbiol. Biotechnol., vol. 97, no. 7, pp. 2951–2959, Apr. 2013, doi: 10.1007/s00253-012-4188-8.
- [302] Shinozaki Y., Kikkawa Y., Sato S., *et al.*, Enzymatic degradation of polyester films by a cutinase-like enzyme from Pseudozyma antarctica: surface plasmon resonance and atomic force microscopy study, *Appl. Microbiol. Biotechnol.*, vol. 97, no. 19, pp. 8591–

8598, Oct. 2013, doi: 10.1007/s00253-012-4673-0.

- [303] Murphy C. A., Cameron J. A., Huang S. J., and Vinopal R. T., Fusarium polycaprolactone depolymerase is cutinase, *Appl. Environ. Microbiol.*, vol. 62, no. 2, pp. 456–460, Feb. 1996, doi: 10.1128/aem.62.2.456-460.1996.
- [304] Adıgüzel A. O. and Tunçer M., Purification and characterization of cutinase from Bacillus sp. KY0701 isolated from plastic wastes, *Prep. Biochem. Biotechnol.*, vol. 47, no. 9, pp. 925–933, Oct. 2017, doi: 10.1080/10826068.2017.1365245.
- [305] Liu Z., Gosser Y., Baker P. J., et al., Structural and Functional Studies of Aspergillus oryzae Cutinase: Enhanced Thermostability and Hydrolytic Activity of Synthetic Ester and Polyester Degradation, J. Am. Chem. Soc., vol. 131, no. 43, pp. 15711– 15716, Nov. 2009, doi: 10.1021/ja9046697.
- [306] Seo H.-S., Um H.-J., Min J., et al., Pseudozyma jejuensis sp. nov., a novel cutinolytic ustilaginomycetous yeast species that is able to degrade plastic waste, FEMS Yeast Res., vol. 7, no. 6, pp. 1035–1045, Sep. 2007, doi: 10.1111/j.1567-1364.2007.00251.x.
- [307] Watanabe T., Suzuki K., Shinozaki Y., et al., A UV-induced mutant of Cryptococcus flavus GB-1 with increased production of a biodegradable plastic-degrading enzyme, *Process Biochem.*, vol. 50, no. 11, pp. 1718–1724, Nov. 2015, doi: 10.1016/j.procbio.2015.07.005.
- [308] Masaki K., Kamini N. R., Ikeda H., and Iefuji H., Cutinase-Like Enzyme from the Yeast Cryptococcus sp. Strain S-2 Hydrolyzes Polylactic Acid and Other Biodegradable Plastics, *Appl. Environ. Microbiol.*, vol. 71, no. 11, pp. 7548–7550, Nov. 2005, doi: 10.1128/AEM.71.11.7548-7550.2005.
- [309] Suzuki M., Tachibana Y., Oba K., et al., Microbial degradation of poly(e-caprolactone) in a coastal environment, *Polym. Degrad. Stab.*, vol. 149, no. October 2017, pp. 1–8, Mar. 2018, doi: 10.1016/j.polymdegradstab.2018.01.017.
- [310] Chu C. C. and Williams D. F., Effect of Gamma Irradiation on the Enzymatic Degradation of Polyglycolic Acid Sutures., *Trans. Annu. Meet. Soc. Biomater. conjunction with Interna*, vol. 6, p. 111, 1983.
- [311] Khan S., Nadir S., Shah Z. U., et al., Biodegradation of polyester polyurethane by Aspergillus tubingensis, *Environ. Pollut.*, vol. 225, pp. 469–480, Jun. 2017, doi: 10.1016/j.envpol.2017.03.012.
- [312] Arena M., Abbate C., Fukushima K., and Gennari M., Degradation of poly (lactic acid) and nanocomposites by Bacillus licheniformis, *Environ. Sci. Pollut. Res.*, vol. 18, no. 6, pp. 865–870, Jul. 2011, doi: 10.1007/s11356-011-0443-2.
- [313] Oceguera-Cervantes A., Carrillo-García A., López N., et al., Characterization of the Polyurethanolytic Activity of Two Alicycliphilus sp. Strains Able To Degrade Polyurethane and N -Methylpyrrolidone, Appl. Environ. Microbiol., vol. 73, no. 19, pp. 6214–6223, Oct. 2007, doi: 10.1128/AEM.01230-07.

- [314] Nakajima-Kambe T., Toyoshima K., Saito C., et al., Rapid monomerization of poly(butylene succinate)-co-(butylene adipate) by Leptothrix sp., J. Biosci. Bioeng., vol. 108, no. 6, pp. 513–516, Dec. 2009, doi: 10.1016/j.jbiosc.2009.05.018.
- [315] Nakajima-Kambe T., Onuma F., Kimpara N., and Nakahara T., Isolation and characterization of a bacterium which utilizes polyester polyurethane as a sole carbon and nitrogen source, *FEMS Microbiol. Lett.*, vol. 129, no. 1, pp. 39–42, Jun. 1995, doi: 10.1111/j.1574-6968.1995.tb07554.x.
- [316] Akutsu Y., Nakajima-Kambe T., Nomura N., and Nakahara T., Purification and Properties of a Polyester Polyurethane-Degrading Enzyme from Comamonas acidovorans TB-35, *Appl. Environ. Microbiol.*, vol. 64, no. 1, pp. 62–67, Jan. 1998, doi: 10.1128/AEM.64.1.62-67.1998.
- [317] Nakajima-Kambe T., Onuma F., Akutsu Y., and Nakahara T., Determination of the polyester polyurethane breakdown products and distribution of the polyurethane degrading enzyme of Comamonas acidovorans strain TB-35, *J. Ferment. Bioeng.*, vol. 83, no. 5, pp. 456–460, Jan. 1997, doi: 10.1016/S0922-338X(97)83000-0.
- [318] Crabbe J. R., Campbell J. R., Thompson L., et al., Biodegradation of a colloidal esterbased polyurethane by soil fungi, Int. Biodeterior. Biodegradation, vol. 33, no. 2, pp. 103–113, Jan. 1994, doi: 10.1016/0964-8305(94)90030-2.
- [319] Allen A. B., Hilliard N. P., and Howard G. T., Purification and characterization of a solublepolyurethane degrading enzyme from Comamonasacidovorans, *Int. Biodeterior. Biodegradation*, vol. 43, no. 1–2, pp. 37–41, Mar. 1999, doi: 10.1016/S0964-8305(98)00066-3.
- [320] Szumigaj J., Zakowska Z., Klimek L., *et al.*, Assessment of polylactide foil degradation as a result of filamentous fungi activity, *Polish J. Environ. Stud.*, vol. 17, no. 3, pp. 335–341, 2008.
- [321] Noor H., Satti S. M., Din S. ud, et al., Insight on esterase from Pseudomonas aeruginosa strain S3 that depolymerize poly(lactic acid) (PLA) at ambient temperature, Polym. Degrad. Stab., vol. 174, no. 2020, p. 109096, Apr. 2020, doi: 10.1016/j.polymdegradstab.2020.109096.
- [322] Tezuka Y., Ishii N., Kasuya K., and Mitomo H., Degradation of poly(ethylene succinate) by mesophilic bacteria, *Polym. Degrad. Stab.*, vol. 84, no. 1, pp. 115–121, Apr. 2004, doi: 10.1016/j.polymdegradstab.2003.09.018.
- [323] Fukuzaki H., Yoshida M., Asano M., and Kumakura M., Synthesis of copoly(d,l-lactic acid) with relatively low molecular weight and in vitro degradation, *Eur. Polym. J.*, vol. 25, no. 10, pp. 1019–1026, Jan. 1989, doi: 10.1016/0014-3057(89)90131-6.
- [324] Uchida H., Properties of a bacterium which degrades solid poly(tetramethylene succinate)-co-adipate, a biodegradable plastic, *FEMS Microbiol. Lett.*, vol. 189, no. 1, pp. 25–29, Aug. 2000, doi: 10.1016/S0378-1097(00)00246-9.
- [325] Nakajima-Kambe T., Edwinoliver N. G., Maeda H., et al., Purification, cloning and

expression of an Aspergillus niger lipase for degradation of poly(lactic acid) and poly(ɛ-caprolactone), *Polym. Degrad. Stab.*, vol. 97, no. 2, pp. 139–144, Feb. 2012, doi: 10.1016/j.polymdegradstab.2011.11.009.

- [326] Hermanová S., Omelková J., Voběrková S., et al., The Effect of Processing of Polycaprolactone Films on Degradation Process Initiated by Aspergillus Oryzae Lipase, Int. J. Polym. Anal. Charact., vol. 17, no. 6, pp. 465–475, Aug. 2012, doi: 10.1080/1023666X.2012.696402.
- [327] Ma Q., Shi K., Su T., and Wang Z., Biodegradation of Polycaprolactone (PCL) with Different Molecular Weights by Candida antarctica Lipase, J. Polym. Environ., vol. 28, no. 11, pp. 2947–2955, Nov. 2020, doi: 10.1007/s10924-020-01826-4.
- [328] Zhang M., Ma X., Li C., *et al.*, A correlation between the degradability of poly(butylene succinate)-based copolyesters and catalytic behavior with Candida antarctica lipase B, *RSC Adv.*, vol. 7, no. 68, pp. 43052–43063, 2017, doi: 10.1039/C7RA05553F.
- [329] Gautam R., Bassi A. S., and Yanful E. K., Candida rugosa lipase-catalyzed polyurethane degradation in aqueous medium, *Biotechnol. Lett.*, vol. 29, no. 7, pp. 1081–1086, May 2007, doi: 10.1007/s10529-007-9354-1.
- [330] Hoshino A. and Isono Y., Degradation of aliphatic polyester films by commercially available lipases with special reference to rapid and complete degradation of poly(Llactide) film by lipase PL derived from Alcaligenes sp., *Biodegradation*, vol. 13, no. 2, pp. 141–147, 2002, doi: 10.1023/A:1020450326301.
- [331] Aarthy M., Puhazhselvan P., Aparna R., et al., Growth associated degradation of aliphatic-aromatic copolyesters by Cryptococcus sp. MTCC 5455, Polym. Degrad. Stab., vol. 152, pp. 20–28, Jun. 2018, doi: 10.1016/j.polymdegradstab.2018.03.021.
- [332] Thirunavukarasu K., Purushothaman S., Gowthaman M. K., *et al.*, Utilization of fish meal and fish oil for production of Cryptococcus sp. MTCC 5455 lipase and hydrolysis of polyurethane thereof, *J. Food Sci. Technol.*, vol. 52, no. 9, pp. 5772–5780, Sep. 2015, doi: 10.1007/s13197-014-1697-8.
- [333] Khan I., Ray Dutta J., and Ganesan R., Lactobacillus sps. lipase mediated poly (ecaprolactone) degradation, *Int. J. Biol. Macromol.*, vol. 95, pp. 126–131, Feb. 2017, doi: 10.1016/j.ijbiomac.2016.11.040.
- [334] Tokiwa Y. and Suzuki T., Purification and Some Properties of Polyethylene Adipatedegrading Enzyme Produced by Penicillium sp. Strain 14–3, *Agric. Biol. Chem.*, vol. 41, no. 2, pp. 265–274, Feb. 1977, doi: 10.1080/00021369.1977.10862486.
- [335] Liu L., Li S., Garreau H., and Vert M., Selective Enzymatic Degradations of Poly(L -Lactide) and Poly("-Caprolactone) Blend Films, *Biomacromolecules*, vol. 1, no. 3, pp. 350–359, Sep. 2000, doi: 10.1021/bm000046k.
- [336] Gan Z., Yu D., Zhong Z., et al., Enzymatic degradation of poly(\(\epsilon\)-poly(dllactide) blends in phosphate buffer solution, Polymer (Guildf)., vol. 40, no. 10, pp. 2859–2862, May 1999, doi: 10.1016/S0032-3861(98)00549-7.

- [337] Papadimitriou S. A., Papageorgiou G. Z., and Bikiaris D. N., Crystallization and enzymatic degradation of novel poly(e-caprolactone-co-propylene succinate) copolymers, *Eur. Polym. J.*, vol. 44, no. 7, pp. 2356–2366, Jul. 2008, doi: 10.1016/j.eurpolymj.2008.05.004.
- [338] Castilla-Cortázar I., Más-Estellés J., Meseguer-Dueñas J. M., et al., Hydrolytic and enzymatic degradation of a poly(e-caprolactone) network, Polym. Degrad. Stab., vol. 97, no. 8, pp. 1241–1248, Aug. 2012, doi: 10.1016/j.polymdegradstab.2012.05.038.
- [339] Thirunavukarasu K., Purushothaman S., Sridevi J., et al., Degradation of poly(butylene succinate) and poly(butylene succinate-co-butylene adipate) by a lipase from yeast Cryptococcus sp. grown on agro-industrial residues, Int. Biodeterior. Biodegradation, vol. 110, pp. 99–107, May 2016, doi: 10.1016/j.ibiod.2016.03.005.
- [340] Murphy C. A., Cameron J. A., Huang S. J., and Vinopal R. T., A second polycaprolactone depolymerase from Fusarium, a lipase distinct from cutinase, *Appl. Microbiol. Biotechnol.*, vol. 50, no. 6, pp. 692–696, Dec. 1998, doi: 10.1007/s002530051352.
- [341] Wang Z., Wang Y., Guo Z., et al., Purification and characterization of poly(<scp>L</scp>-lactic acid) depolymerase from Pseudomonas sp. strain DS04-T, Polym. Eng. Sci., vol. 51, no. 3, pp. 454–459, Mar. 2011, doi: 10.1002/pen.21857.
- [342] Sekosan G. and Vasanthan N., Morphological changes of annealed poly-e-caprolactone by enzymatic degradation with lipase, J. Polym. Sci. Part B Polym. Phys., vol. 48, no. 2, pp. 202–211, Jan. 2010, doi: 10.1002/polb.21889.
- [343] Umare S. S. and Chandure A. S., Synthesis, characterization and biodegradation studies of poly(ester urethane)s, *Chem. Eng. J.*, vol. 142, no. 1, pp. 65–77, Aug. 2008, doi: 10.1016/j.cej.2007.11.017.
- [344] Gan Z., Liang Q., Zhang J., and Jing X., Enzymatic degradation of poly(ecaprolactone) film in phosphate buffer solution containing lipases, *Polym. Degrad. Stab.*, vol. 56, no. 2, pp. 209–213, May 1997, doi: 10.1016/S0141-3910(96)00208-X.
- [345] Tokiwa Y. and Suzuki T., Hydrolysis of polyesters by lipases, *Nature*, vol. 270, no. 5632, pp. 76–78, 1977, Accessed: Jun. 18, 2021. [Online]. Available: https://www.nature.com/articles/270076a0.pdf?origin=ppub.
- [346] kanwal A., Zhang M., Sharaf F., and Chengtao L., Screening and characterization of novel lipase producing Bacillus species from agricultural soil with high hydrolytic activity against PBAT poly (butylene adipate co terephthalate) co-polyesters, *Polym. Bull.*, no. 0123456789, Jan. 2022, doi: 10.1007/s00289-021-03992-4.
- [347] Tang J., Li L., Wang X., et al., Tailored crystallization behavior, thermal stability, and biodegradability of poly(ethylene adipate): Effects of a biocompatible diamide nucleating agent, Polym. Test., vol. 81, p. 106116, Jan. 2020, doi: 10.1016/j.polymertesting.2019.106116.
- [348] Jia H., Zhang M., Weng Y., et al., Degradation of poly(butylene adipate-co-

terephthalate) by Stenotrophomonas sp. YCJ1 isolated from farmland soil, *J. Environ. Sci.*, vol. 103, pp. 50–58, May 2021, doi: 10.1016/j.jes.2020.10.001.

- [349] Kanwal A., Zhang M., Sharaf F., and Li C., Enzymatic degradation of poly (butylene adipate co-terephthalate) (PBAT) copolymer using lipase B from Candida antarctica (CALB) and effect of PBAT on plant growth, *Polym. Bull.*, no. 0123456789, Oct. 2021, doi: 10.1007/s00289-021-03946-w.
- [350] Soulenthone P., Tachibana Y., Suzuki M., *et al.*, Characterization of a poly(butylene adipate-co-terephthalate) hydrolase from the mesophilic actinobacteria Rhodococcus fascians, *Polym. Degrad. Stab.*, vol. 184, p. 109481, Feb. 2021, doi: 10.1016/j.polymdegradstab.2021.109481.
- [351] Akutsu-Shigeno Y., Teeraphatpornchai T., Teamtisong K., et al., Cloning and Sequencing of a Poly(<scp>dl</scp> -Lactic Acid) Depolymerase Gene from Paenibacillus amylolyticus Strain TB-13 and Its Functional Expression in Escherichia coli, Appl. Environ. Microbiol., vol. 69, no. 5, pp. 2498–2504, May 2003, doi: 10.1128/AEM.69.5.2498-2504.2003.
- [352] Li F., Hu X., Guo Z., et al., Purification and characterization of a novel poly(butylene succinate)-degrading enzyme from Aspergillus sp. XH0501-a, World J. Microbiol. Biotechnol., vol. 27, no. 11, pp. 2591–2596, Nov. 2011, doi: 10.1007/s11274-011-0731-5.
- [353] Nawaz A., Hasan F., and Shah A. A., Degradation of poly(ε-caprolactone) (PCL) by a newly isolated Brevundimonas sp. strain MRL-AN1 from soil, *FEMS Microbiol. Lett.*, vol. 362, no. 1, pp. 1–7, Jan. 2015, doi: 10.1093/femsle/fnu004.
- [354] Li F., Yu D., Lin X., et al., Biodegradation of poly(ɛ-caprolactone) (PCL) by a new Penicillium oxalicum strain DSYD05-1, World J. Microbiol. Biotechnol., vol. 28, no. 10, pp. 2929–2935, Oct. 2012, doi: 10.1007/s11274-012-1103-5.
- [355] Oda Y., Polycaprolactone depolymerase produced by the bacterium Alcaligenes faecalis, *FEMS Microbiol. Lett.*, vol. 152, no. 2, pp. 339–343, Jul. 1997, doi: 10.1016/S0378-1097(97)00222-X.
- [356] Oda Y., Asari H., Urakami T., and Tonomura K., Microbial degradation of poly(3hydroxybutyrate) and polycaprolactone by filamentous fungi, *J. Ferment. Bioeng.*, vol. 80, no. 3, pp. 265–269, Jan. 1995, doi: 10.1016/0922-338X(95)90827-M.
- [357] Liang T.-W., Jen S.-N., Nguyen A., and Wang S.-L., Application of Chitinous Materials in Production and Purification of a Poly(l-lactic acid) Depolymerase from Pseudomonas tamsuii TKU015, *Polymers (Basel).*, vol. 8, no. 3, p. 98, Mar. 2016, doi: 10.3390/polym8030098.
- [358] Krajangsang S., Dechsresawut N., Lomthong T., and Samaimai S., Production of poly (l-lactide)-degrading enzyme by Actinomadura keratinilytica strain T16-1 under solid state fermentation using agricultural wastes as substrate, *3 Biotech*, vol. 11, no. 12, p. 512, Dec. 2021, doi: 10.1007/s13205-021-03060-8.

- [359] Kanesawa Y., Tanahashi N., Doi Y., and Saito T., Enzymatic degradation of microbial poly(3-hydroxyalkanoates), *Polym. Degrad. Stab.*, vol. 45, no. 2, pp. 179–185, Jan. 1994, doi: 10.1016/0141-3910(94)90135-X.
- [360] Mukai K., Yamada K., and Doi Y., Efficient hydrolysis of polyhydroxyalkanoates by Pseudomonas stutzeri YM1414 isolated from lake water, *Polym. Degrad. Stab.*, vol. 43, no. 3, pp. 319–327, Jan. 1994, doi: 10.1016/0141-3910(94)90002-7.
- [361] Wang Y., Inagawa Y., Saito T., et al., Enzymatic Hydrolysis of Bacterial Poly(3hydroxybutyrate- c o -3-hydroxypropionate)s by Poly(3-hydroxyalkanoate) Depolymerase from Acidovorax Sp. TP4, Biomacromolecules, vol. 3, no. 4, pp. 828– 834, Jul. 2002, doi: 10.1021/bm020019p.
- [362] Molitoris H. P., Moss S. T., de Koning G. J. M., and Jendrossek D., Scanning electron microscopy of polyhydroxyalkanoate degradation by bacteria, *Appl. Microbiol. Biotechnol.*, vol. 46, no. 5–6, pp. 570–579, Dec. 1996, doi: 10.1007/s002530050863.
- [363] Mukai K., Yamada K., and Doi Y., Enzymatic degradation of poly(hydroxyalkanoates) by a marine bacterium, *Polym. Degrad. Stab.*, vol. 41, no. 1, pp. 85–91, Jan. 1993, doi: 10.1016/0141-3910(93)90066-R.
- [364] Mao H., Jiang H., Su T., and Wang Z., Purification and characterization of two extracellular polyhydroxyalkanoate depolymerases from Pseudomonas mendocina, *Biotechnol. Lett.*, vol. 35, no. 11, pp. 1919–1924, Nov. 2013, doi: 10.1007/s10529-013-1288-1.
- [365] Mohanan N., Sharma P. K., and Levin D. B., Characterization of an intracellular poly(3-hydroxyalkanoate) depolymerase from the soil bacterium, Pseudomonas putida LS46, *Polym. Degrad. Stab.*, vol. 175, p. 109127, May 2020, doi: 10.1016/j.polymdegradstab.2020.109127.
- [366] Kasuya K., Inoue Y., and Doi Y., Adsorption kinetics of bacterial PHB depolymerase on the surface of polyhydroxyalkanoate films, *Int. J. Biol. Macromol.*, vol. 19, no. 1, pp. 35–40, Jul. 1996, doi: 10.1016/0141-8130(96)01097-5.
- [367] Kasuya K., Ohura T., Masuda K., and Doi Y., Substrate and binding specificities of bacterial polyhydroxybutyrate depolymerases, *Int. J. Biol. Macromol.*, vol. 24, no. 4, pp. 329–336, May 1999, doi: 10.1016/S0141-8130(99)00046-X.
- [368] Scherer T. M., Fuller R. C., Lenz R. W., and Goodwin S., Hydrolase activity of an extracellular depolymerase from Aspergillus fumigatus with bacterial and synthetic polyesters, *Polym. Degrad. Stab.*, vol. 64, no. 2, pp. 267–275, May 1999, doi: 10.1016/S0141-3910(98)00201-8.
- [369] Scherer T. M., Fuller R. C., Lenz R. W., and Goodwin S., Production, purification and activity of an extracellular depolymerase from Aspergillus fumigatus, J. Environ. Polym. Degrad., vol. 7, no. 3, pp. 117–125, 1999, doi: 10.1023/A:1022881204565.
- [370] Kasuya K., Doi Y., and Yao T., Enzymatic degradation of poly[(R)-3-hydroxybutyrate] by Comamonas testosteroni ATSU of soil bacterium, *Polym. Degrad. Stab.*, vol. 45, no.

3, pp. 379-386, Jan. 1994, doi: 10.1016/0141-3910(94)90208-9.

- [371] Sadocco P., Nocerino S., Dubini-Paglia E., et al., Characterization of a poly(3hydroxybutyrate) depolymerase from Aureobacterium saperdae: Active site and kinetics of hydrolysis studies, J. Environ. Polym. Degrad., vol. 5, no. 1, pp. 57–65, 1997, doi: 10.1007/BF02763569.
- [372] Jung H.-W., Yang M.-K., and Su R.-C., Purification, characterization, and gene cloning of an Aspergillus fumigatus polyhydroxybutyrate depolymerase used for degradation of polyhydroxybutyrate, polyethylene succinate, and polybutylene succinate, *Polym. Degrad. Stab.*, vol. 154, pp. 186–194, Aug. 2018, doi: 10.1016/j.polymdegradstab.2018.06.002.
- [373] Kim D. Y., Yun J. H., Kim H. W., et al., Purification and characterization of poly(3hydroxybutyrate) depolymerase from a fungal isolate, Emericellopsis minima W2, J. Microbiol., vol. 40, no. 2, pp. 129–133, 2002.
- [374] Sayyed R. Z., Wani S. J., Alyousef A. A., et al., Purification and kinetics of the PHB depolymerase of Microbacterium paraoxydans RZS6 isolated from a dumping yard, *PLoS One*, vol. 14, no. 6, p. e0212324, Jun. 2019, doi: 10.1371/journal.pone.0212324.
- [375] Zhou H., Wang Z., Chen S., et al., Purification and Characterization of Extracellular Poly(6-hydroxybutyrate) Depolymerase from Penicillium sp. DS9701-D2, Polym. Plast. Technol. Eng., vol. 48, no. 1, pp. 58–63, Dec. 2008, doi: 10.1080/03602550802539627.
- [376] Shah A. A., Hasan F., Hameed A., and Ahmed S., Isolation and characterisation of poly(3-hydroxybutyrate-co-3-hydroxyvalerate) degrading actinomycetes and purification of PHBV depolymerase from newly isolatedStreptoverticillium kashmirense AF1, Ann. Microbiol., vol. 57, no. 4, pp. 583–588, Dec. 2007, doi: 10.1007/BF03175359.
- [377] Kobayashi T., Sugiyama A., Kawase Y., et al., Biochemical and genetic characterization of an extracellular poly(3-hydroxybutyrate) depolymerase from Acidovorax sp. strain TP4, J. Environ. Polym. Degrad., vol. 7, no. 1, pp. 9–18, 1999, doi: 10.1023/A:1021885901119.
- [378] Asano Y. and Watanabe S., Isolation of Poly(3-Hydroxybutyrate) (PHB)-degrading Microorganisms and Characterization of PHB-depolymerase from Arthrobacter sp. strain W6, *Biosci. Biotechnol. Biochem.*, vol. 65, no. 5, pp. 1191–1194, Jan. 2001, doi: 10.1271/bbb.65.1191.
- [379] Shivakumar S., Poly- β -hydroxybutyrate (PHB) Depolymerase from Fusarium solani Thom, J. Chem., vol. 2013, pp. 1–9, 2013, doi: 10.1155/2013/406386.
- [380] Takaku H., Kimoto A., Kodaira S., et al., Isolation of a Gram-positive poly(3hydroxybutyrate) (PHB)-degrading bacterium from compost, and cloning and characterization of a gene encoding PHB depolymerase of Bacillus megaterium N-18-25-9, FEMS Microbiol. Lett., vol. 264, no. 2, pp. 152–159, Nov. 2006, doi:

10.1111/j.1574-6968.2006.00448.x.

- [381] Liu H., Zhang H., Chen S., et al., Purification and Properties of a Poly (βhydroxybutyrate) Depolymerase From Penicillium sp., J. Polym. Environ., vol. 14, no. 4, pp. 419–426, Nov. 2006, doi: 10.1007/s10924-006-0031-6.
- [382] Kasuya K., Takano T., Tezuka Y., et al., Cloning, expression and characterization of a poly(3-hydroxybutyrate) depolymerase from Marinobacter sp. NK-1, Int. J. Biol. Macromol., vol. 33, no. 4–5, pp. 221–226, Dec. 2003, doi: 10.1016/j.ijbiomac.2003.08.006.
- [383] Kasuya K. and Mitomo H., Identification of a Marine Benthic P (3HB) -Degrading Bacterium Isolate and Characterization of Its P (3HB) Depolymerase, pp. 194–201, 2000.
- [384] Ghanem N. B., Mabrouk M. E. S., Sabry S. A., and El-Badan D. E. S., Degradation of polyesters by a novel marine Nocardiopsis aegyptia sp. nov.: Application of Plackett-Burman experimental design for the improvement of PHB depolymerase activity, J. Gen. Appl. Microbiol., vol. 51, no. 3, pp. 151–158, 2005, doi: 10.2323/jgam.51.151.
- [385] Miyazaki S., Takahashi K., Shiraki M., et al., Properties of a poly(3-hydroxybutyrate) depolymerase from Penicillium funiculosum, J. Polym. Environ., vol. 8, no. 4, pp. 175– 182, 2000, doi: 10.1023/A:1015245710406.
- [386] Han J.-S. and Kim M.-N., Purification and Characterization of Extracellular Poly(3hydroxybutyrate) Depolymerase from Penicillium simplicissimum LAR13, J. *Microbiol.*, vol. 40, no. 1, pp. 20–25, 2002.
- [387] Kasuya K., Inoue Y., Tanaka T., *et al.*, Biochemical and molecular characterization of the polyhydroxybutyrate depolymerase of Comamonas acidovorans YM1609, isolated from freshwater, *Appl. Environ. Microbiol.*, vol. 63, no. 12, pp. 4844–4852, Dec. 1997, doi: 10.1128/aem.63.12.4844-4852.1997.
- [388] Uefuji M., Kasuya K., and Doi Y., Enzymatic degradation of poly[(R)-3hydroxybutyrate]: secretion and properties of PHB depolymerase from Pseudomonas stutzeri, *Polym. Degrad. Stab.*, vol. 58, no. 3, pp. 275–281, Dec. 1997, doi: 10.1016/S0141-3910(97)00058-X.
- [389] Nojima S., Mineki S., and Iida M., Purification and characterization of extracellular poly(3-hydroxybutyrate) depolymerases produced by Agrobacterium sp. K-03, J. *Ferment. Bioeng.*, vol. 81, no. 1, pp. 72–75, Jan. 1996, doi: 10.1016/0922-338X(96)83124-2.
- [390] Klingbeil B., Taxonomic identification of Streptomyces exfoliatus K10 and characterization of its poly(3-hydroxybutyrate) depolymerase gene, *FEMS Microbiol. Lett.*, vol. 142, no. 2–3, pp. 215–221, Sep. 1996, doi: 10.1016/0378-1097(96)00271-6.
- [391] Yamada K., Mukai K., and Doi Y., Enzymatic degradation of poly(hydroxyalkanoates) by Pseudomonas pickettii, *Int. J. Biol. Macromol.*, vol. 15, no. 4, pp. 215–220, Aug. 1993, doi: 10.1016/0141-8130(93)90040-S.

- [392] Jendrossek D., Backhaus M., and Andermann M., Characterization of the extracellular poly(3-hydroxybutyrate) depolymerase of Comamonas sp. and of its structural gene, vol. 41, pp. 160–169, 1995.
- [393] Kita K., Mashiba S., Nagita M., *et al.*, Cloning of poly ž 3-hydroxybutyrate / depolymerase from a marine bacterium , Alcaligenes faecalis AE122 , and characterization of its gene product, pp. 113–122, 1997.
- [394] Kita K., Ishimaru K., Teraoka M., et al., Properties of poly(3-hydroxybutyrate) depolymerase from a marine bacterium, Alcaligenes faecalis AE122, Appl. Environ. Microbiol., vol. 61, no. 5, pp. 1727–1730, May 1995, doi: 10.1128/aem.61.5.1727-1730.1995.
- [395] Iyer S., Shah R., Sharma A., *et al.*, Purification of Aspergillus fumigatus (Pdf1) Poly ([\]L -hydroxybutyrate) (PHB) Depolymerase Using a New, Single-Step Substrate Affinity Chromatography Method: Characterization of the PHB Depolymerase Exhibiting Novel Self-Aggregation Behavior, vol. 8, no. 4, 2002.
- [396] Tanio T., Fukui T., Shirakura Y., et al., An Extracellular Poly(3-Hydroxybutyrate) Depolymerase from Alcaligenes faecalis, Eur. J. Biochem., vol. 124, no. 1, pp. 71–77, May 1982, doi: 10.1111/j.1432-1033.1982.tb05907.x.
- [397] Sang B.-I., Lee W.-K., Hori K., and Unno H., Purification and Characterization of Fungal Poly(3-hydroxybutyrate) Depolymerase from Paecilomyces lilacinus F4-5 and Enzymatic Degradation of Poly(3-hydroxybutyrate-co-3-hydroxyvalerate) Film, World J. Microbiol. Biotechnol., vol. 22, no. 1, pp. 51–57, Jan. 2006, doi: 10.1007/s11274-005-5773-0.
- [398] Zhang T., Chaudhry M. T., and Liu Z.-P., Genetic and biochemical characterization of poly 3-hydroxybutyrate depolymerase from Diaphorobacter sp. PCA039, World J. Microbiol. Biotechnol., vol. 26, no. 10, pp. 1803–1811, Oct. 2010, doi: 10.1007/s11274-010-0361-3.
- [399] Bhatt R., Patel K. C., and Trivedi U., Purification and Properties of Extracellular Poly(3-hydroxybutyrate) Depolymerase Produced by Aspergillus fumigatus 202, J. Polym. Environ., vol. 18, no. 2, pp. 141–147, Jun. 2010, doi: 10.1007/s10924-010-0176-1.
- [400] Gowda U. S. V. and Shivakumar S., Poly(-8-hydroxybutyrate) (PHB) depolymerase PHAZ Pen from Penicillium expansum: purification, characterization and kinetic studies, *3 Biotech*, vol. 5, no. 6, pp. 901–909, Dec. 2015, doi: 10.1007/s13205-015-0287-4.
- [401] Mabrouk M. M. and Sabry S. A., Degradation of poly (3-hydroxybutyrate) and its copolymer poly (3-hydroxybutyrate-co-3-hydroxyvalerate) by a marine Streptomyces sp. SNG9, *Microbiol. Res.*, vol. 156, no. 4, pp. 323–335, 2001, doi: 10.1078/0944-5013-00115.
- [402] Jendrossek D., Knoke I., Habibian R. B., et al., Degradation of poly(3-

hydroxybutyrate), PHB, by bacteria and purification of a novel PHB depolymerase fromComamonas sp., *J. Environ. Polym. Degrad.*, vol. 1, no. 1, pp. 53–63, Jan. 1993, doi: 10.1007/BF01457653.

- [403] Brucato L., Extracellular Poly (3-hydroxybutyrate) Depolymerase from PeniciKum funicu / osum ~ General Characteristics and Active Site Studies, vol. 290, no. 2, 1991.
- [404] Oda Y., Osaka H., Urakami T., and Tonomura K., Purification and Properties of Poly(3-Hydroxybutyrate) Depolymerase from the Fungus Paecilomyces lilacinus D218, *Curr. Microbiol.*, vol. 34, no. 4, pp. 230–232, Apr. 1997, doi: 10.1007/s002849900174.
- [405] Ishii N., Inoue Y., Shimada K., et al., Fungal degradation of poly(ethylene succinate), Polym. Degrad. Stab., vol. 92, no. 1, pp. 44–52, Jan. 2007, doi: 10.1016/j.polymdegradstab.2006.09.014.
- [406] Nadhman A., Hasan F., Shah Z., et al., Production of poly(3-hydroxybutyrate-co-3hydroxyvalerate) depolymerase from Aspergillus sp. NA-25, Appl. Biochem. Microbiol., vol. 48, no. 5, pp. 482–487, Sep. 2012, doi: 10.1134/S0003683812050080.
- [407] Wang Z., Gao J., Li L., and Jiang H., Purification and characterization of an extracellular poly(3-hydroxybutyrate-co-3-hydroxyvalerate) depolymerase from Acidovorax sp. HB01, World J. Microbiol. Biotechnol., vol. 28, no. 6, pp. 2395–2402, Jun. 2012, doi: 10.1007/s11274-012-1048-8.
- [408] Akbar S., Hasan F., Nadhman A., et al., Production and Purification of Poly(3hydroxybutyrate-co-3-hydroxyvalerate) Degrading Enzyme from Streptomyces sp. AF-111, J. Polym. Environ., vol. 21, no. 4, pp. 1109–1116, Dec. 2013, doi: 10.1007/s10924-013-0600-4.
- [409] Schöber U., Thiel C., and Jendrossek D., Poly(3-Hydroxyvalerate) Depolymerase of Pseudomonas lemoignei, Appl. Environ. Microbiol., vol. 66, no. 4, pp. 1385–1392, Apr. 2000, doi: 10.1128/AEM.66.4.1385-1392.2000.
- [410] Jendrossek D., Frisse A., Behrends A., et al., Biochemical and molecular characterization of the Pseudomonas lemoignei polyhydroxyalkanoate depolymerase system, J. Bacteriol., vol. 177, no. 3, pp. 596–607, Feb. 1995, doi: 10.1128/jb.177.3.596-607.1995.
- [411] Rowe L. and Howard G. T., Growth of Bacillus subtilis on polyurethane and the purification and characterization of a polyurethanase-lipase enzyme, *Int. Biodeterior. Biodegradation*, vol. 50, no. 1, pp. 33–40, Jul. 2002, doi: 10.1016/S0964-8305(02)00047-1.
- [412] Howard G. T., Ruiz C., and Hilliard N. P., Growth of Pseudomonas chlororaphis on apolyester-polyurethane and the purification and characterization of a polyurethanase-esterase enzyme, *Int. Biodeterior. Biodegradation*, vol. 43, no. 1–2, pp. 7–12, Mar. 1999, doi: 10.1016/S0964-8305(98)00057-2.
- [413] Ruiz C., Main T., Hilliard N. P., and Howard G. T., Purification and characterization

of twopolyurethanase enzymes from Pseudomonas chlororaphis, *Int. Biodeterior*. *Biodegradation*, vol. 43, no. 1–2, pp. 43–47, Mar. 1999, doi: 10.1016/S0964-8305(98)00067-5.

- [414] Howard G. T. and Blake R. C., Growth of Pseudomonas fluorescens on a polyester– polyurethane and the purification and characterization of a polyurethanase–protease enzyme, *Int. Biodeterior. Biodegradation*, vol. 42, no. 4, pp. 213–220, Nov. 1998, doi: 10.1016/S0964-8305(98)00051-1.
- [415] Hung C.-S., Zingarelli S., Nadeau L. J., et al., Carbon Catabolite Repression and Impranil Polyurethane Degradation in Pseudomonas protegens Strain Pf-5, Appl. Environ. Microbiol., vol. 82, no. 20, pp. 6080–6090, Oct. 2016, doi: 10.1128/AEM.01448-16.
- [416] Howard G. T., Norton W. N., and Burks T., Growth of Acinetobacter gerneri P7 on polyurethane and the purification and characterization of a polyurethanase enzyme, *Biodegradation*, vol. 23, no. 4, pp. 561–573, Jul. 2012, doi: 10.1007/s10532-011-9533-6.
- [417] Ibrahim I. N., Maraqa A., Hameed K. M., et al., Polyester-polyurethane biodegradation by Alternaria solani, isolated from northern Jordan, Adv. Environ. Biol., vol. 3, no. 2, pp. 162–170, 2009.
- [418] Pérez-Lara L. F., Vargas-Suárez M., López-Castillo N. N., et al., Preliminary study on the biodegradation of adipate/phthalate polyester polyurethanes of commercial-type by Alicycliphilus sp. BQ8, J. Appl. Polym. Sci., vol. 133, no. 6, p. n/a-n/a, Feb. 2016, doi: 10.1002/app.42992.
- [419] Russell J. R., Huang J., Anand P., et al., Biodegradation of Polyester Polyurethane by Endophytic Fungi, Appl. Environ. Microbiol., vol. 77, no. 17, pp. 6076–6084, Sep. 2011, doi: 10.1128/AEM.00521-11.
- [420] Jarerat A., Tokiwa Y., and Tanaka H., Production of poly(l-lactide)-degrading enzyme by Amycolatopsis orientalis for biological recycling of poly(l-lactide), *Appl. Microbiol. Biotechnol.*, vol. 72, no. 4, pp. 726–731, Oct. 2006, doi: 10.1007/s00253-006-0343-4.
- [421] Jarerat A., Tokiwa Y., and Tanaka H., Microbial Poly(L-lactide)-Degrading Enzyme Induced by Amino Acids, Peptides, and Poly(L-amino Acids), J. Polym. Environ., vol. 12, no. 3, pp. 139–146, 2004.
- [422] Pranamuda H., Tsuchii A., and Tokiwa Y., Poly (L-lactide)-Degrading Enzyme Produced byAmycolatopsis sp., *Macromol. Biosci.*, vol. 1, no. 1, pp. 25–29, Jan. 2001, doi: 10.1002/1616-5195(200101)1:1<25::AID-MABI25>3.0.CO;2-3.
- [423] Penkhrue W., Khanongnuch C., Masaki K., et al., Isolation and screening of biopolymer-degrading microorganisms from northern Thailand, World J. Microbiol. Biotechnol., vol. 31, no. 9, pp. 1431–1442, Sep. 2015, doi: 10.1007/s11274-015-1895-1.
- [424] Bubpachat T., Sombatsompop N., and Prapagdee B., Isolation and role of polylactic acid-degrading bacteria on degrading enzymes productions and PLA biodegradability

at mesophilic conditions, *Polym. Degrad. Stab.*, vol. 152, pp. 75–85, Jun. 2018, doi: 10.1016/j.polymdegradstab.2018.03.023.

- [425] Tsuji H., Echizen Y., and Nishimura Y., Enzymatic Degradation of Poly(l-Lactic Acid): Effects of UV Irradiation, J. Polym. Environ., vol. 14, no. 3, pp. 239–248, Nov. 2006, doi: 10.1007/s10924-006-0023-6.
- [426] Bian H., Wang G., Cao M., et al., Improved biodegradation of polyvinyl alcohol by hybrid nanoflowers of degrading enzymes from Bacillus niacini, Korean J. Chem. Eng., vol. 37, no. 6, pp. 1020–1028, Jun. 2020, doi: 10.1007/s11814-020-0547-9.
- [427] Vaclavkova T., Ruzicka J., Julinova M., et al., Novel aspects of symbiotic (polyvinyl alcohol) biodegradation, Appl. Microbiol. Biotechnol., vol. 76, no. 4, pp. 911–917, Sep. 2007, doi: 10.1007/s00253-007-1062-1.
- [428] Yamatsu A., Matsumi R., Atomi H., and Imanaka T., Isolation and characterization of a novel poly(vinyl alcohol)-degrading bacterium, Sphingopyxis sp. PVA3, *Appl. Microbiol. Biotechnol.*, vol. 72, no. 4, pp. 804–811, Oct. 2006, doi: 10.1007/s00253-006-0351-4.
- [429] Watanabe Y., Hamada N., Morita M., and Tsujisaka Y., Purification and properties of a polyvinyl alcohol-degrading enzyme produced by a strain of Pseudomonas, Arch. Biochem. Biophys., vol. 174, no. 2, pp. 575–581, Jun. 1976, doi: 10.1016/0003-9861(76)90386-6.
- [430] Zhang Y., Li Y., Shen W., et al., A new strain, Streptomyces venezuelae GY1, producing a poly(vinyl alcohol)-degrading enzyme, World J. Microbiol. Biotechnol., vol. 22, no. 6, pp. 625–628, May 2006, doi: 10.1007/s11274-005-9081-5.
- [431] Qian D., Du G., and Chen J., Isolation and Culture Characterization of a New Polyvinyl Alcohol-Degrading Strain: Penicillium sp. WSH02-21, World J. Microbiol. Biotechnol., vol. 20, no. 6, pp. 587–591, 2004, doi: 10.1023/B:WIBI.0000043172.83610.08.
- [432] Matsumura S., Tomizawa N., Toki A., et al., Novel Poly(vinyl alcohol)-Degrading Enzyme and the Degradation Mechanism, *Macromolecules*, vol. 32, no. 23, pp. 7753– 7761, Nov. 1999, doi: 10.1021/ma990727b.
- [433] Nakamura K., Tomita T., Abe N., and Kamio Y., Purification and Characterization of an Extracellular Poly(<scp>l</scp> -Lactic Acid) Depolymerase from a Soil Isolate, *Amycolatopsis* sp. Strain K104-1, *Appl. Environ. Microbiol.*, vol. 67, no. 1, pp. 345– 353, Jan. 2001, doi: 10.1128/AEM.67.1.345-353.2001.
- [434] Shah A. A., Eguchi T., Mayumi D., et al., Purification and properties of novel aliphaticaromatic co-polyesters degrading enzymes from newly isolated roseateles depolymerans strain TB-87, Polym. Degrad. Stab., vol. 98, no. 2, pp. 609–618, 2013, doi: 10.1016/j.polymdegradstab.2012.11.013.
- [435] Shah A. A., Eguchi T., Mayumi D., *et al.*, Degradation of aliphatic and aliphaticaromatic co-polyesters by depolymerases from Roseateles depolymerans strain TB-87

and analysis of degradation products by LC-MS, *Polym. Degrad. Stab.*, vol. 98, no. 12, pp. 2722–2729, Dec. 2013, doi: 10.1016/j.polymdegradstab.2013.10.003.

- [436] Teeraphatpornchai T., Nakajima-Kambe T., Shigeno-Akutsu Y., et al., Isolation and characterization of a bacterium that degrades various polyester-based biodegradable plastics, *Biotechnol. Lett.*, vol. 25, no. 1, pp. 23–28, 2003, doi: 10.1023/A:1021713711160.
- [437] Magnin A., Pollet E., Perrin R., et al., Enzymatic recycling of thermoplastic polyurethanes: Synergistic effect of an esterase and an amidase and recovery of building blocks, Waste Manag., vol. 85, pp. 141–150, Feb. 2019, doi: 10.1016/j.wasman.2018.12.024.
- [438] Vega R. E., Main T., and Howard G. T., Cloning and expression in Escherichia coli of apolyurethane-degrading enzyme from Pseudomonasfluorescens, *Int. Biodeterior. Biodegradation*, vol. 43, no. 1–2, pp. 49–55, Mar. 1999, doi: 10.1016/S0964-8305(98)00068-7.
- [439] Mohamed H., Shah A. M., Nazir Y., et al., Biodegradation of poly (vinyl alcohol) by an orychophragmus rhizosphere-associated fungus Penicillium brevicompactum OVR-5, and its proposed PVA biodegradation pathway, World J. Microbiol. Biotechnol., vol. 38, no. 1, p. 10, Jan. 2022, doi: 10.1007/s11274-021-03197-x.
- [440] Ignat L., Ignat M., Ciobanu C., et al., Effects of flax lignin addition on enzymatic oxidation of poly(ethylene adipate) urethanes, Ind. Crops Prod., vol. 34, no. 1, pp. 1017–1028, Jul. 2011, doi: 10.1016/j.indcrop.2011.03.010.
- [441] Yin C., Xu Y., Deng S., et al., A Novel Esterase, DacA pva, from Comamonas sp. Strain NyZ500 with Deacetylation Activity for the Acetylated Polymer Polyvinyl Alcohol, Appl. Environ. Microbiol., vol. 87, no. 8, Mar. 2021, doi: 10.1128/AEM.03016-20.
- [442] Bayan R. and Karak N., Renewable resource modified polyol derived aliphatic hyperbranched polyurethane as a biodegradable and UV-resistant smart material, *Polym. Int.*, vol. 66, no. 6, pp. 839–850, Jun. 2017, doi: 10.1002/pi.5323.
- [443] Suzuki M., Tachibana Y., Takizawa R., et al., A novel poly(3-hydroxybutyrate)degrading actinobacterium that was isolated from plastisphere formed on marine plastic debris, *Polym. Degrad. Stab.*, vol. 183, p. 109461, Jan. 2021, doi: 10.1016/j.polymdegradstab.2020.109461.
- [444] Mathur G. and Prasad R., Degradation of Polyurethane by Aspergillus flavus (ITCC 6051) Isolated from Soil, *Appl. Biochem. Biotechnol.*, vol. 167, no. 6, pp. 1595–1602, Jul. 2012, doi: 10.1007/s12010-012-9572-4.
- [445] Zhao J.-H., Wang X.-Q., Zeng J., et al., Biodegradation of poly(butylene succinate-cobutylene adipate) by Aspergillus versicolor, *Polym. Degrad. Stab.*, vol. 90, no. 1, pp. 173–179, Oct. 2005, doi: 10.1016/j.polymdegradstab.2005.03.006.
- [446] Gautam R., Bassi A. S., Yanful E. K., and Cullen E., Biodegradation of automotive waste polyester polyurethane foam using Pseudomonas chlororaphis ATCC55729, *Int.*

Biodeterior. Biodegradation, vol. 60, no. 4, pp. 245–249, Jan. 2007, doi: 10.1016/j.ibiod.2007.03.009.

- [447] Zafar U., Houlden A., and Robson G. D., Fungal Communities Associated with the Biodegradation of Polyester Polyurethane Buried under Compost at Different Temperatures, *Appl. Environ. Microbiol.*, vol. 79, no. 23, pp. 7313–7324, Dec. 2013, doi: 10.1128/AEM.02536-13.
- [448] Volova T. G., Boyandin A. N., Vasiliev A. D., et al., Biodegradation of polyhydroxyalkanoates (PHAs) in tropical coastal waters and identification of PHAdegrading bacteria, *Polym. Degrad. Stab.*, vol. 95, no. 12, pp. 2350–2359, Dec. 2010, doi: 10.1016/j.polymdegradstab.2010.08.023.
- [449] Sang, K. Hori, Y. Tanji, H. Unno B.-I., Fungal contribution to in situ biodegradation of poly(3-hydroxybutyrate- co -3-hydroxyvalerate) film in soil, *Appl. Microbiol. Biotechnol.*, vol. 58, no. 2, pp. 241–247, Jan. 2002, doi: 10.1007/s00253-001-0884-5.
- [450] Abe M., Kobayashi K., Honma N., and Nakasaki K., Microbial degradation of poly(butylene succinate) by Fusarium solani in soil environments, *Polym. Degrad.* Stab., vol. 95, no. 2, pp. 138–143, Feb. 2010, doi: 10.1016/j.polymdegradstab.2009.11.042.
- [451] Tsujiyama S., Nitta T., and Maoka T., Biodegradation of polyvinyl alcohol by Flammulina velutipes in an unsubmerged culture, J. Biosci. Bioeng., vol. 112, no. 1, pp. 58–62, Jul. 2011, doi: 10.1016/j.jbiosc.2011.03.004.
- [452] Sanyal P., Samaddar P., and Paul A. K., Degradation of Poly(3-hydroxybutyrate) and Poly(3-hydroxybutyrate-co-3-hydroxyvalerate) by Some Soil Aspergillus spp., J. Polym. Environ., vol. 14, no. 3, pp. 257–263, Nov. 2006, doi: 10.1007/s10924-006-0022-7.
- [453] Wu C.-S., Characterization and biodegradability of polyester bioplastic-based green renewable composites from agricultural residues, *Polym. Degrad. Stab.*, vol. 97, no. 1, pp. 64–71, Oct. 2011, doi: 10.1016/j.polymdegradstab.2011.10.012.
- [454] Al Hosni A. S., Pittman J. K., and Robson G. D., Microbial degradation of four biodegradable polymers in soil and compost demonstrating polycaprolactone as an ideal compostable plastic, *Waste Manag.*, vol. 97, pp. 105–114, Sep. 2019, doi: 10.1016/j.wasman.2019.07.042.
- [455] Ishii N., Inoue Y., Tagaya T., et al., Isolation and characterization of poly(butylene succinate)-degrading fungi, Polym. Degrad. Stab., vol. 93, no. 5, pp. 883–888, May 2008, doi: 10.1016/j.polymdegradstab.2008.02.005.
- [456] Nakajima-Kambe T., Ichihashi F., Matsuzoe R., et al., Degradation of aliphaticaromatic copolyesters by bacteria that can degrade aliphatic polyesters, Polym. Degrad. Stab., vol. 94, no. 11, pp. 1901–1905, Nov. 2009, doi: 10.1016/j.polymdegradstab.2009.08.006.
- [457] Wu C.-S., Renewable resource-based composites of recycled natural fibers and

maleated polylactide bioplastic: Characterization and biodegradability, *Polym. Degrad. Stab.*, vol. 94, no. 7, pp. 1076–1084, Jul. 2009, doi: 10.1016/j.polymdegradstab.2009.04.002.

- [458] Hayase N., Yano H., Kudoh E., et al., Isolation and characterization of poly(butylene succinate-co-butylene adipate)-degrading microorganism, J. Biosci. Bioeng., vol. 97, no. 2, pp. 131–133, Jan. 2004, doi: 10.1016/S1389-1723(04)70180-2.
- [459] Cho J. Y., Lee Park S., Lee H.-J., et al., Polyhydroxyalkanoates (PHAs) degradation by the newly isolated marine Bacillus sp. JY14, *Chemosphere*, vol. 283, no. May, p. 131172, Nov. 2021, doi: 10.1016/j.chemosphere.2021.131172.
- [460] Sekiguchi T., Saika A., Nomura K., et al., Biodegradation of aliphatic polyesters soaked in deep seawaters and isolation of poly(ε-caprolactone)-degrading bacteria, *Polym. Degrad. Stab.*, vol. 96, no. 7, pp. 1397–1403, Jul. 2011, doi: 10.1016/j.polymdegradstab.2011.03.004.
- [461] Lipsa R., Tudorachi N., Darie-Nita R. N., et al., Biodegradation of poly(lactic acid) and some of its based systems with Trichoderma viride, Int. J. Biol. Macromol., vol. 88, pp. 515–526, Jul. 2016, doi: 10.1016/j.ijbiomac.2016.04.017.
- [462] Satti S. M., Shah A. A., Auras R., and Marsh T. L., Isolation and characterization of bacteria capable of degrading poly(lactic acid) at ambient temperature, *Polym. Degrad.* Stab., vol. 144, pp. 392–400, Oct. 2017, doi: 10.1016/j.polymdegradstab.2017.08.023.
- [463] Decorosi F., Exana M. L., Pini F., et al., The Degradative Capabilities of New Amycolatopsis Isolates on Polylactic Acid, *Microorganisms*, vol. 7, no. 12, p. 590, Nov. 2019, doi: 10.3390/microorganisms7120590.
- [464] Pranamuda H. and Tokiwa Y., Degradation of poly(L-lactide) by strains belonging to genus Amycolatopsis, *Biotechnol. Lett.*, vol. 21, no. 10, pp. 901–905, 1999, doi: 10.1023/A:1005547326434.
- [465] Ikura Y. and Kudo T., Isolation of a microorganism capable of degrading poly-(Llactide)., J. Gen. Appl. Microbiol., vol. 45, no. 5, pp. 247–251, 1999, doi: 10.2323/jgam.45.247.
- [466] Pranamuda H., Tokiwa Y., and Tanaka H., Polylactide Degradation by an Amycolatopsis sp, Appl. Environ. Microbiol., vol. 63, no. 4, pp. 1637–1640, Apr. 1997, doi: 10.1128/aem.63.4.1637-1640.1997.
- [467] Tokiwa Y., Konno M., and Nishida H., Isolation of Silk Degrading Microorganisms and Its Poly(L-lactide) Degradability, *Chem. Lett.*, vol. 28, no. 4, pp. 355–356, Apr. 1999, doi: 10.1246/cl.1999.355.
- [468] Boyandin A. N., Prudnikova S. V., Karpov V. A., et al., Microbial degradation of polyhydroxyalkanoates in tropical soils, *Int. Biodeterior. Biodegradation*, vol. 83, pp. 77–84, Sep. 2013, doi: 10.1016/j.ibiod.2013.04.014.

- [469] Chomchoei A., Pathom-aree W., Yokota A., et al., Amycolatopsis thailandensis sp. nov., a poly(l-lactic acid)-degrading actinomycete, isolated from soil, Int. J. Syst. Evol. Microbiol., vol. 61, no. 4, pp. 839–843, Apr. 2011, doi: 10.1099/ijs.0.023564-0.
- [470] Bonifer K. S., Wen X., Hasim S., et al., Bacillus pumilus B12 Degrades Polylactic Acid and Degradation Is Affected by Changing Nutrient Conditions, Front. Microbiol., vol. 10, no. November, pp. 1–13, Nov. 2019, doi: 10.3389/fmicb.2019.02548.
- [471] Jarerat A., Tokiwa Y., and Tanaka H., Poly(l-lactide) degradation by Kibdelosporangium aridum, *Biotechnol. Lett.*, vol. 25, no. 23, pp. 2035–2038, Dec. 2003, doi: 10.1023/B:BILE.0000004398.38799.29.
- [472] Jarerat A., Pranamuda H., and Tokiwa Y., Poly(L-lactide)-Degrading Activity in Various Actinomycetes, *Macromol. Biosci.*, vol. 2, no. 9, pp. 420–428, Dec. 2002, doi: 10.1002/mabi.200290001.
- [473] Konkit M., Jarerat A., Kamongnuch C., et al., Poly (lactide) Degradation By Pseudonocardia alni AS4.1531T, Chiang Mai J. Sci., vol. 39, no. 1, pp. 128–132, 2012.
- [474] Jarerat A. and Tokiwa Y., Poly(L-lactide) degradation by Saccharothrix waywayandensis, *Biotechnol. Lett.*, vol. 25, no. 5, pp. 401–404, 2003, doi: 10.1023/A:1022450431193.
- [475] Jarerat A. and Tokiwa Y., Degradation of Poly(L-lactide) by a Fungus, *Macromol. Biosci.*, vol. 1, no. 4, pp. 136–140, Jun. 2001, doi: 10.1002/1616-5195(20010601)1:4<136::AID-MABI136>3.0.CO;2-3.
- [476] Aladyshev A. M., Klyamkina A. N., Kostyuk S. V, et al., Biodegradation of Poly-εcaprolactones and Poly-L-lactides by Fungi, J. Polym. Environ., vol. 26, pp. 4350– 4359, 2018.
- [477] Karamanlioglu M., Houlden A., and Robson G. D., Isolation and characterisation of fungal communities associated with degradation and growth on the surface of poly(lactic) acid (PLA) in soil and compost, *Int. Biodeterior. Biodegradation*, vol. 95, no. PB, pp. 301–310, Nov. 2014, doi: 10.1016/j.ibiod.2014.09.006.
- [478] Ullah M., Li H., Sun S.-W., et al., Polyvinyl alcohol degradation by Bacillus cereus RA23 from oil sludge sample, 3 Biotech, vol. 9, no. 10, p. 350, Oct. 2019, doi: 10.1007/s13205-019-1882-6.
- [479] Zhang H. Z., Influence of pH and C/N Ratio on poly(vinyl alcohol) biodegradation in mixed bacterial culture, J. Polym. Environ., vol. 17, no. 4, pp. 286–290, 2009, doi: 10.1007/s10924-009-0151-x.
- [480] Deng Y., Wang C., Liu Y., et al., The first demonstration of a novel isolated fungus Eutypella sp. BJ associated with the biodegradation of polyvinyl alcohol, RSC Adv., vol. 9, no. 47, pp. 27398–27405, 2019, doi: 10.1039/C9RA04410H.
- [481] Cosgrove L., McGeechan P. L., Robson G. D., and Handley P. S., Fungal Communities Associated with Degradation of Polyester Polyurethane in Soil, *Appl. Environ*.

Microbiol., vol. 73, no. 18, pp. 5817-5824, Sep. 2007, doi: 10.1128/AEM.01083-07.

- [482] Jaeger K., Bacterial lipases, FEMS Microbiol. Rev., vol. 15, no. 1, pp. 29–63, Sep. 1994, doi: 10.1016/0168-6445(94)90025-6.
- [483] Marten E., Müller R.-J., and Deckwer W.-D., Studies on the enzymatic hydrolysis of polyesters I. Low molecular mass model esters and aliphatic polyesters, *Polym. Degrad. Stab.*, vol. 80, no. 3, pp. 485–501, Jan. 2003, doi: 10.1016/S0141-3910(03)00032-6.
- [484] Gricajeva A., Nadda A. K., and Gudiukaite R., Insights into polyester plastic biodegradation by carboxyl ester hydrolases, J. Chem. Technol. Biotechnol., vol. 97, no. 2, pp. 359–380, Feb. 2022, doi: 10.1002/jctb.6745.
- [485] Singh A. K. and Mukhopadhyay M., Overview of Fungal Lipase: A Review, Appl. Biochem. Biotechnol., vol. 166, no. 2, pp. 486–520, Jan. 2012, doi: 10.1007/s12010-011-9444-3.
- [486] Rizzarelli P., Impallomeni G., and Montaudo G., Evidence for Selective Hydrolysis of Aliphatic Copolyesters Induced by Lipase Catalysis, *Biomacromolecules*, vol. 5, no. 2, pp. 433–444, Mar. 2004, doi: 10.1021/bm034230s.
- [487] Bikiaris D. N., Papageorgiou G. Z., and Achilias D. S., Synthesis and comparative biodegradability studies of three poly(alkylene succinate)s, *Polym. Degrad. Stab.*, vol. 91, no. 1, pp. 31–43, Jan. 2006, doi: 10.1016/j.polymdegradstab.2005.04.030.
- [488] Longhi S. and Cambillau C., Structure-activity of cutinase, a small lipolytic enzyme, Biochim. Biophys. Acta - Mol. Cell Biol. Lipids, vol. 1441, no. 2–3, pp. 185–196, Nov. 1999, doi: 10.1016/S1388-1981(99)00159-6.
- [489] Urbanek A. K., Mirończuk A. M., García-Martín A., et al., Biochemical properties and biotechnological applications of microbial enzymes involved in the degradation of polyester-type plastics, *Biochim. Biophys. Acta - Proteins Proteomics*, vol. 1868, no. 2, p. 140315, Feb. 2020, doi: 10.1016/j.bbapap.2019.140315.
- [490] Dutta K., Sen S., and Veeranki V. D., Production, characterization and applications of microbial cutinases, *Process Biochem.*, vol. 44, no. 2, pp. 127–134, Feb. 2009, doi: 10.1016/j.procbio.2008.09.008.
- [491] Chen S., Su L., Chen J., and Wu J., Cutinase: Characteristics, preparation, and application, *Biotechnol. Adv.*, vol. 31, no. 8, pp. 1754–1767, Dec. 2013, doi: 10.1016/j.biotechadv.2013.09.005.
- [492] Tokiwa Y. and Jarerat A., Biodegradation of poly(l-lactide), *Biotechnol. Lett.*, vol. 26, no. 10, pp. 771–777, May 2004, doi: 10.1023/B:BILE.0000025927.31028.e3.
- [493] Cregut M., Bedas M., Durand M.-J., and Thouand G., New insights into polyurethane biodegradation and realistic prospects for the development of a sustainable waste recycling process, *Biotechnol. Adv.*, vol. 31, no. 8, pp. 1634–1647, Dec. 2013, doi: 10.1016/j.biotechadv.2013.08.011.

- [494] Skleničková K., Abbrent S., Halecký M., et al., Biodegradability and ecotoxicity of polyurethane foams: A review, Crit. Rev. Environ. Sci. Technol., vol. 52, no. 2, pp. 157– 202, Jan. 2022, doi: 10.1080/10643389.2020.1818496.
- [495] Magnin A., Pollet E., Phalip V., and Avérous L., Evaluation of biological degradation of polyurethanes, *Biotechnol. Adv.*, vol. 39, no. 2019, p. 107457, Mar. 2020, doi: 10.1016/j.biotechadv.2019.107457.
- [496] Liu J., He J., Xue R., *et al.*, Biodegradation and up-cycling of polyurethanes: Progress, challenges, and prospects, *Biotechnol. Adv.*, vol. 48, no. March, p. 107730, May 2021, doi: 10.1016/j.biotechadv.2021.107730.
- [497] Desai J. D. and Banat I. M., Microbial production of surfactants and their commercial potential, *Microbiol. Mol. Biol. Rev.*, vol. 61, no. 1, pp. 47–64, Mar. 1997, doi: 10.1128/mmbr.61.1.47-64.1997.
- [498] Ron E. Z. and Rosenberg E., Natural roles of biosurfactants. Minireview, *Environ. Microbiol.*, vol. 3, no. 4, pp. 229–236, Apr. 2001, doi: 10.1046/j.1462-2920.2001.00190.x.
- [499] Banat I. M., Franzetti A., Gandolfi I., et al., Microbial biosurfactants production, applications and future potential, Appl. Microbiol. Biotechnol., vol. 87, no. 2, pp. 427– 444, Jun. 2010, doi: 10.1007/s00253-010-2589-0.
- [500] Takahashi T., Tanaka T., Tsushima Y., et al., Ionic interaction of positive amino acid residues of fungal hydrophobin RolA with acidic amino acid residues of cutinase CutL1, Mol. Microbiol., vol. 96, no. 1, pp. 14–27, Apr. 2015, doi: 10.1111/mmi.12915.
- [501] Terauchi Y., Kim Y.-K., Tanaka T., et al., Asp30 of Aspergillus oryzae cutinase CutL1 is involved in the ionic interaction with fungal hydrophobin RolA, Biosci. Biotechnol. Biochem., vol. 81, no. 7, pp. 1363–1368, Jul. 2017, doi: 10.1080/09168451.2017.1321952.
- [502] Holmberg K., Interactions between surfactants and hydrolytic enzymes, Colloids Surfaces B Biointerfaces, vol. 168, pp. 169–177, Aug. 2018, doi: 10.1016/j.colsurfb.2017.12.002.
- [503] Banat I. M., Makkar R. S., and Cameotra S. S., Potential commercial applications of microbial surfactants, *Appl. Microbiol. Biotechnol.*, vol. 53, no. 5, pp. 495–508, May 2000, doi: 10.1007/s002530051648.
- [504] Moon R. J., Martini A., Nairn J., et al., Cellulose nanomaterials review: structure, properties and nanocomposites, Chem. Soc. Rev., vol. 40, no. 7, p. 3941, 2011, doi: 10.1039/c0cs00108b.
- [505] Klemm D., Heublein B., Fink H.-P., and Bohn A., Cellulose: Fascinating Biopolymer and Sustainable Raw Material, Angew. Chemie Int. Ed., vol. 44, no. 22, pp. 3358–3393, May 2005, doi: 10.1002/anie.200460587.
- [506] Vaaje-Kolstad G., Westereng B., Horn S. J., et al., An Oxidative Enzyme Boosting the

Enzymatic Conversion of Recalcitrant Polysaccharides, *Science (80-.).*, vol. 330, no. 6001, pp. 219–222, Oct. 2010, doi: 10.1126/science.1192231.

- [507] Béguin P. and Aubert J.-P., The biological degradation of cellulose, *FEMS Microbiol. Rev.*, vol. 13, no. 1, pp. 25–58, Jan. 1994, doi: 10.1111/j.1574-6976.1994.tb00033.x.
- [508] Detyothin S., Selke S. E. M., Narayan R., *et al.*, Effects of molecular weight and grafted maleic anhydride of functionalized polylactic acid used in reactive compatibilized binary and ternary blends of polylactic acid and thermoplastic cassava starch, *J. Appl. Polym. Sci.*, vol. 132, no. 28, pp. 1–15, 2015, doi: 10.1002/app.42230.
- [509] Carvalho A. J. F., Starch as Source of Polymeric Materials, in *Biopolymers*, Hoboken, NJ, USA: John Wiley & Sons, Inc., 2011, pp. 81–98.
- [510] Avérous L., Biodegradable Multiphase Systems Based on Plasticized Starch: A Review, J. Macromol. Sci. Part C Polym. Rev., vol. 44, no. 3, pp. 231–274, Dec. 2004, doi: 10.1081/MC-200029326.
- [511] Jiménez A., Fabra M. J., Talens P., and Chiralt A., Edible and Biodegradable Starch Films: A Review, *Food Bioprocess Technol.*, vol. 5, no. 6, pp. 2058–2076, Aug. 2012, doi: 10.1007/s11947-012-0835-4.
- [512] Bher A., Uysal Unalan I., Auras R., et al., Toughening of Poly(lactic acid) and Thermoplastic Cassava Starch Reactive Blends Using Graphene Nanoplatelets, Polymers (Basel)., vol. 10, no. 1, p. 95, Jan. 2018, doi: 10.3390/polym10010095.
- [513] Liu H., Xie F., Yu L., *et al.*, Thermal processing of starch-based polymers, *Prog. Polym. Sci.*, vol. 34, no. 12, pp. 1348–1368, Dec. 2009, doi: 10.1016/j.progpolymsci.2009.07.001.
- [514] Detyothin S., Production and characterization of thermoplastic cassava starch, functionalized poly(lactic acid), and their reactive blends, Michigan State University, 2012.
- [515] Chen M., Chen B., and Evans J. R. G., Novel thermoplastic starch–clay nanocomposite foams, *Nanotechnology*, vol. 16, no. 10, pp. 2334–2337, Oct. 2005, doi: 10.1088/0957-4484/16/10/056.
- [516] Santayanon R. and Wootthikanokkhan J., Modification of cassava starch by using propionic anhydride and properties of the starch-blended polyester polyurethane, *Carbohydr. Polym.*, vol. 51, no. 1, pp. 17–24, Jan. 2003, doi: 10.1016/S0144-8617(02)00109-1.
- [517] Forssell P., Mikkilä J., Suortti T., et al., Plasticization of Barley Starch with Glycerol and Water, J. Macromol. Sci. Part A, vol. 33, no. 5, pp. 703–715, May 1996, doi: 10.1080/10601329608010888.
- [518] Willett J. . and Shogren R. ., Processing and properties of extruded starch/polymer foams, *Polymer (Guildf)*., vol. 43, no. 22, pp. 5935–5947, 2002, doi: 10.1016/S0032-3861(02)00497-4.

- [519] Popa V. I. and Asachi G., Monomers, Polymers and Composites from Renewable Resources, .
- [520] Vieira M. G. A., da Silva M. A., dos Santos L. O., and Beppu M. M., Natural-based plasticizers and biopolymer films: A review, *Eur. Polym. J.*, vol. 47, no. 3, pp. 254– 263, Mar. 2011, doi: 10.1016/j.eurpolymj.2010.12.011.
- [521] Chen Q., Yu H., Wang L., et al., Recent progress in chemical modification of starch and its applications, RSC Adv., vol. 5, no. 83, pp. 67459–67474, 2015, doi: 10.1039/C5RA10849G.
- [522] Fringant C., Rinaudo M., Foray M. F., and Bardet M., Preparation of mixed esters of starch or use of an external plasticizer: two different ways to change the properties of starch acetate films, *Carbohydr. Polym.*, vol. 35, no. 1–2, pp. 97–106, Jan. 1998, doi: 10.1016/S0144-8617(97)00250-6.
- [523] Jem K. J. and Tan B., The development and challenges of poly (lactic acid) and poly (glycolic acid), Adv. Ind. Eng. Polym. Res., vol. 3, no. 2, pp. 60–70, Apr. 2020, doi: 10.1016/j.aiepr.2020.01.002.
- [524] Singh V. and Tiwari M., Structure-Processing-Property Relationship of Poly(Glycolic Acid) for Drug Delivery Systems 1: Synthesis and Catalysis, *Int. J. Polym. Sci.*, vol. 2010, pp. 1–23, 2010, doi: 10.1155/2010/652719.
- [525] Steinborn-Rogulska, Izabela Rokicki G., Solid-state polycondensation (SSP) as a method to obtain high molecular weight polymers. Part II. Synthesis of polylactide and polyglycolide via SSP, *Polimery*, vol. 58, no. 2, pp. 85–92, Feb. 2013, doi: 10.14314/polimery.2013.085.
- [526] Samantaray P. K., Little A., Haddleton D. M., et al., Poly(glycolic acid) (PGA): a versatile building block expanding high performance and sustainable bioplastic applications, Green Chem., vol. 22, no. 13, pp. 4055–4081, 2020, doi: 10.1039/D0GC01394C.
- [527] Takayama T., Daigaku Y., Ito H., and Takamori H., Mechanical properties of bioabsorbable PLA/PGA fiber-reinforced composites, J. Mech. Sci. Technol., vol. 28, no. 10, pp. 4151–4154, Oct. 2014, doi: 10.1007/s12206-014-0927-3.
- [528] Wu F., Misra M., and Mohanty A. K., Challenges and new opportunities on barrier performance of biodegradable polymers for sustainable packaging, *Prog. Polym. Sci.*, vol. 117, p. 101395, Jun. 2021, doi: 10.1016/j.progpolymsci.2021.101395.
- [529] Epple M. and Herzberg O., Polyglycolide with controlled porosity: an improved biomaterial, *R. Soc. Chem.*, vol. 7, pp. 1037–1042, 1997.
- [530] Karamanlioglu M., Preziosi R., and Robson G. D., Abiotic and biotic environmental degradation of the bioplastic polymer poly(lactic acid): A review, *Polym. Degrad. Stab.*, vol. 137, pp. 122–130, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.009.
- [531] Kabasci S., Biobased plastics, in *Plastic Waste and Recycling*, vol. 25, no. 11, Elsevier,

2020, pp. 67–96.

- [532] Castro-Aguirre E., Iñiguez-Franco F., Samsudin H., et al., Poly(lactic acid)—Mass production, processing, industrial applications, and end of life, Adv. Drug Deliv. Rev., vol. 107, pp. 333–366, Dec. 2016, doi: 10.1016/j.addr.2016.03.010.
- [533] Hamad K., Kaseem M., Ayyoob M., et al., Polylactic acid blends: The future of green, light and tough, Prog. Polym. Sci., vol. 85, pp. 83–127, Oct. 2018, doi: 10.1016/j.progpolymsci.2018.07.001.
- [534] Lim L.-T., Auras R., and Rubino M., Processing technologies for poly(lactic acid), Prog. Polym. Sci., vol. 33, no. 8, pp. 820–852, Aug. 2008, doi: 10.1016/j.progpolymsci.2008.05.004.
- [535] Kawai F., Polylactic Acid (PLA)-Degrading Microorganisms and PLA Depolymerases, in Green Polymer Chemistry: Biocatalysis and Biomaterials, vol. 1043, 2010, pp. 405– 414.
- [536] Williams D. F., Enzymic Hydrolysis of Polylactic Acid, *Eng. Med.*, vol. 10, no. 1, pp. 5– 7, Jan. 1981, doi: 10.1243/EMED_JOUR_1981_010_004_02.
- [537] Hajighasemi M., Tchigvintsev A., Nocek B., et al., Screening and Characterization of Novel Polyesterases from Environmental Metagenomes with High Hydrolytic Activity against Synthetic Polyesters, Environ. Sci. Technol., vol. 52, no. 21, pp. 12388–12401, Nov. 2018, doi: 10.1021/acs.est.8b04252.
- [538] Reeve M. S., McCarthy S. P., Downey M. J., and Gross R. A., Polylactide stereochemistry: effect on enzymic degradability, *Macromolecules*, vol. 27, no. 3, pp. 825–831, Jan. 1994, doi: 10.1021/ma00081a030.
- [539] MacDonald R. T., McCarthy S. P., and Gross R. A., Enzymatic Degradability of Poly(lactide): Effects of Chain Stereochemistry and Material Crystallinity, *Macromolecules*, vol. 29, no. 23, pp. 7356–7361, Jan. 1996, doi: 10.1021/ma960513j.
- [540] Cai H., Dave V., Gross R. A., and McCarthy S. P., Effects of physical aging, crystallinity, and orientation on the enzymatic degradation of poly(lactic acid), J. Polym. Sci. Part B Polym. Phys., vol. 34, no. 16, pp. 2701–2708, Nov. 1996, doi: 10.1002/(SICI)1099-0488(19961130)34:16<2701::AID-POLB2>3.0.CO;2-S.
- [541] Iwata T. and Doi Y., Morphology and Enzymatic Degradation of Poly(<scp>l</scp> lactic acid) Single Crystals, *Macromolecules*, vol. 31, no. 8, pp. 2461–2467, Apr. 1998, doi: 10.1021/ma980008h.
- [542] Li S. and McCarthy S., Influence of Crystallinity and Stereochemistry on the Enzymatic Degradation of Poly(lactide)s, *Macromolecules*, vol. 32, no. 13, pp. 4454– 4456, Jun. 1999, doi: 10.1021/ma990117b.
- [543] Kawai F., Nakadai K., Nishioka E., et al., Different enantioselectivity of two types of poly(lactic acid) depolymerases toward poly(l-lactic acid) and poly(d-lactic acid), *Polym. Degrad. Stab.*, vol. 96, no. 7, pp. 1342–1348, Jul. 2011, doi:

10.1016/j.polymdegradstab.2011.03.022.

- [544] Woodruff M. A. and Hutmacher D. W., The return of a forgotten polymer— Polycaprolactone in the 21st century, *Prog. Polym. Sci.*, vol. 35, no. 10, pp. 1217–1256, Oct. 2010, doi: 10.1016/j.progpolymsci.2010.04.002.
- [545] Tay B. Y., Zhang S. X., Myint M. H., et al., Processing of polycaprolactone porous structure for scaffold development, J. Mater. Process. Technol., vol. 182, no. 1–3, pp. 117–121, Feb. 2007, doi: 10.1016/j.jmatprotec.2006.07.016.
- [546] Malikmammadov E., Tanir T. E., Kiziltay A., et al., PCL and PCL-based materials in biomedical applications, J. Biomater. Sci. Polym. Ed., vol. 29, no. 7–9, pp. 863–893, Jun. 2018, doi: 10.1080/09205063.2017.1394711.
- [547] Nakajima-Kambe T., Shigeno-Akutsu Y., Nomura N., et al., Microbial degradation of polyurethane, polyester polyurethanes and polyether polyurethanes, Appl. Microbiol. Biotechnol., vol. 51, no. 2, pp. 134–140, Feb. 1999, doi: 10.1007/s002530051373.
- [548] Bartnikowski M., Dargaville T. R., Ivanovski S., and Hutmacher D. W., Degradation mechanisms of polycaprolactone in the context of chemistry, geometry and environment, *Prog. Polym. Sci.*, vol. 96, pp. 1–20, Sep. 2019, doi: 10.1016/j.progpolymsci.2019.05.004.
- [549] Tsuji H., Echizen Y., and Nishimura Y., Photodegradation of biodegradable polyesters: A comprehensive study on poly(l-lactide) and poly(ε-caprolactone), *Polym. Degrad. Stab.*, vol. 91, no. 5, pp. 1128–1137, May 2006, doi: 10.1016/j.polymdegradstab.2005.07.007.
- [550] Khatiwala V. K., Shekhar N., Aggarwal S., and Mandal U. K., Biodegradation of Poly(ɛ-caprolactone) (PCL) Film by Alcaligenes faecalis, J. Polym. Environ., vol. 16, no. 1, pp. 61–67, Jan. 2008, doi: 10.1007/s10924-008-0104-9.
- [551] Perstorp, CAPA 6500, 2022. https://www.perstorp.com (accessed Jan. 31, 2022).
- [552] Nishida H. and Tokiwa Y., Distribution of poly(?-hydroxybutyrate) and poly(?caprolactone)aerobic degrading microorganisms in different environments, J. *Environ. Polym. Degrad.*, vol. 1, no. 3, pp. 227–233, Jul. 1993, doi: 10.1007/BF01458031.
- [553] Suzuki M., Tachibana Y., and Kasuya K., Biodegradability of poly(3hydroxyalkanoate) and poly(e-caprolactone) via biological carbon cycles in marine environments, *Polym. J.*, vol. 53, no. 1, pp. 47–66, Jan. 2021, doi: 10.1038/s41428-020-00396-5.
- [554] Platnieks O., Gaidukovs S., Kumar Thakur V., et al., Bio-based poly (butylene succinate): Recent progress, challenges and future opportunities, Eur. Polym. J., vol. 161, no. September, p. 110855, Dec. 2021, doi: 10.1016/j.eurpolymj.2021.110855.
- [555] Xu J. and Guo B.-H., Poly(butylene succinate) and its copolymers: Research, development and industrialization, *Biotechnol. J.*, vol. 5, no. 11, pp. 1149–1163, Nov.

2010, doi: 10.1002/biot.201000136.

- [556] Patel M. K., Bechu A., Villegas J. D., et al., Second-generation bio-based plastics are becoming a reality – Non-renewable energy and greenhouse gas (GHG) balance of succinic acid- based plastic end products made from lignocellulosic biomass, *Biofuels*, *Bioprod. Biorefining*, vol. 12, pp. 426–441, 2018, doi: 10.1002/bbb.
- [557] Havstad M. R., Biodegradable plastics, in *Plastic Waste and Recycling*, Elsevier, 2020, pp. 97–129.
- [558] Spiegel S., Recent advances in applied polymer science, J. Appl. Polym. Sci., vol. 135, no. 24, p. 46279, Jun. 2018, doi: 10.1002/app.46279.
- [559] Bajpai P., Properties of Biobased Packaging Material, in *Biobased Polymers:* Properties and Applications in Packaging, 2019, pp. 25–110.
- [560] Monvisade P. and Loungvanidprapa P., Synthesis of poly(ethylene adipate) and poly(ethylene adipate-co-terephthalate) via ring-opening polymerization, *Eur. Polym.* J., vol. 43, no. 8, pp. 3408–3414, Aug. 2007, doi: 10.1016/j.eurpolymj.2007.05.009.
- [561] Okamoto K., Ichikawa T., Yokohara T., and Yamaguchi M., Miscibility, mechanical and thermal properties of poly(lactic acid)/polyester-diol blends, *Eur. Polym. J.*, vol. 45, no. 8, pp. 2304–2312, Aug. 2009, doi: 10.1016/j.eurpolymj.2009.05.011.
- [562] Liu J., Chen P., Li J., et al., Synthesis of poly(ethylene adipate-co-l-lactic acid) copolymers via ring opening polymerization, *Polym. Bull.*, vol. 66, no. 2, pp. 187–197, Jan. 2011, doi: 10.1007/s00289-010-0268-z.
- [563] He X. and Qiu Z., Effect of poly(ethylene adipate) with different molecular weights on the crystallization behavior and mechanical properties of biodegradable poly(Llactide), *Thermochim. Acta*, vol. 659, pp. 89–95, Jan. 2018, doi: 10.1016/j.tca.2017.11.010.
- [564] Tiso T., Winter B., Wei R., et al., The metabolic potential of plastics as biotechnological carbon sources – Review and targets for the future, *Metab. Eng.*, Dec. 2021, doi: 10.1016/j.ymben.2021.12.006.
- [565] Kumar M., Rathour R., Singh R., et al., Bacterial polyhydroxyalkanoates: Opportunities, challenges, and prospects, J. Clean. Prod., vol. 263, p. 121500, Aug. 2020, doi: 10.1016/j.jclepro.2020.121500.
- [566] Materials M., Mango Materials, 2022. https://www.mangomaterials.com (accessed Jan. 31, 2022).
- [567] Bugnicourt E., Cinelli P., Lazzeri A., and Alvarez V., Polyhydroxyalkanoate (PHA): Review of synthesis, characteristics, processing and potential applications in packaging, *eXPRESS Polym. Lett.*, vol. 8, no. 11, pp. 791–808, 2014, Accessed: Jul. 31, 2020. [Online]. Available: https://ri.conicet.gov.ar/bitstream/handle/11336/4563/CONICET_Digital_Nro.5438_A .pdf?sequence.

- [568] Kourmentza C., Plácido J., Venetsaneas N., et al., Recent Advances and Challenges towards Sustainable Polyhydroxyalkanoate (PHA) Production, *Bioengineering*, vol. 4, no. 4, p. 55, Jun. 2017, doi: 10.3390/bioengineering4020055.
- [569] Doi Y., Kanesawa Y., Kawaguchi Y., and Kunioka M., Hydrolytic degradation of microbial poly(hydroxyalkanoates), *Die Makromol. Chemie, Rapid Commun.*, vol. 10, no. 5, pp. 227–230, 1989, doi: 10.1002/marc.1989.030100506.
- [570] Deroiné M., Le Duigou A., Corre Y.-M., et al., Accelerated ageing and lifetime prediction of poly(3-hydroxybutyrate-co-3-hydroxyvalerate) in distilled water, Polym. Test., vol. 39, pp. 70–78, Oct. 2014, doi: 10.1016/j.polymertesting.2014.07.018.
- [571] Luo S. and Netravali A. ., A study of physical and mechanical properties of poly(hydroxybutyrate-co-hydroxyvalerate) during composting, *Polym. Degrad. Stab.*, vol. 80, no. 1, pp. 59–66, Jan. 2003, doi: 10.1016/S0141-3910(02)00383-X.
- [572] Meereboer K. W., Misra M., and Mohanty A. K., Review of recent advances in the biodegradability of polyhydroxyalkanoate (PHA) bioplastics and their composites, *Green Chem.*, vol. 22, no. 17, pp. 5519–5558, 2020, doi: 10.1039/D0GC01647K.
- [573] Jendrossek D. and Handrick R., Microbial Degradation of Polyhydroxyalkanoates, Annu. Rev. Microbiol., vol. 56, no. 1, pp. 403–432, Oct. 2002, doi: 10.1146/annurev.micro.56.012302.160838.
- [574] Gu J.-D., Microbiological deterioration and degradation of synthetic polymeric materials: recent research advances, *Int. Biodeterior. Biodegradation*, vol. 52, no. 2, pp. 69–91, Sep. 2003, doi: 10.1016/S0964-8305(02)00177-4.
- [575] Nishida H. and Tokiwa Y., Effects of higher-order structure of poly(3hydroxybutyrate) on its biodegradation. II. Effects of crystal structure on microbial degradation, J. Environ. Polym. Degrad., vol. 1, no. 1, pp. 65–80, Jan. 1993, doi: 10.1007/BF01457654.
- [576] Deroiné M., Le Duigou A., Corre Y.-M., et al., Seawater accelerated ageing of poly(3hydroxybutyrate-co-3-hydroxyvalerate), Polym. Degrad. Stab., vol. 105, no. 1, pp. 237– 247, Jul. 2014, doi: 10.1016/j.polymdegradstab.2014.04.026.
- [577] Jian J., Xiangbin Z., and Xianbo H., An overview on synthesis, properties and applications of poly(butylene-adipate-co-terephthalate)–PBAT, Adv. Ind. Eng. Polym. Res., vol. 3, no. 1, pp. 19–26, Jan. 2020, doi: 10.1016/j.aiepr.2020.01.001.
- [578] Zhao H., Liu H., Liu Y., and Yang Y., Blends of poly(butylene adipate- coterephthalate) (PBAT) and stereocomplex polylactide with improved rheological and mechanical properties, *RSC Adv.*, vol. 10, no. 18, pp. 10482–10490, 2020, doi: 10.1039/C9RA10827K.
- [579] Ferreira F. V., Cividanes L. S., Gouveia R. F., and Lona L. M. F., An overview on properties and applications of poly(butylene adipate- co-terephthalate)-PBAT based composites, *Polym. Eng. Sci.*, vol. 59, no. s2, pp. E7–E15, Mar. 2019, doi: 10.1002/pen.24770.

- [580] Yang J., Wei Y., Li G., et al., Computer-aided engineering of adipyl-CoA synthetase for enhancing adipic acid synthesis, *Biotechnol. Lett.*, vol. 42, no. 12, pp. 2693–2701, Dec. 2020, doi: 10.1007/s10529-020-02978-y.
- [581] De Hoe G. X., Zumstein M. T., Getzinger G. J., et al., Photochemical Transformation of Poly(butylene adipate- co -terephthalate) and Its Effects on Enzymatic Hydrolyzability, Environ. Sci. Technol., vol. 53, no. 5, pp. 2472–2481, Mar. 2019, doi: 10.1021/acs.est.8b06458.
- [582] Rychter P., Kawalec M., Sobota M., et al., Study of Aliphatic-Aromatic Copolyester Degradation in Sandy Soil and Its Ecotoxicological Impact, *Biomacromolecules*, vol. 11, no. 4, pp. 839–847, Apr. 2010, doi: 10.1021/bm901331t.
- [583] Muroi F., Tachibana Y., Kobayashi Y., et al., Influences of poly(butylene adipate-coterephthalate) on soil microbiota and plant growth, *Polym. Degrad. Stab.*, vol. 129, pp. 338–346, Jul. 2016, doi: 10.1016/j.polymdegradstab.2016.05.018.
- [584] Xie F., Zhang T., Bryant P., et al., Degradation and stabilization of polyurethane elastomers, Prog. Polym. Sci., vol. 90, pp. 211–268, Mar. 2019, doi: 10.1016/j.progpolymsci.2018.12.003.
- [585] Loredo-Treviño A., Gutiérrez-Sánchez G., Rodríguez-Herrera R., and Aguilar C. N., Microbial Enzymes Involved in Polyurethane Biodegradation: A Review, J. Polym. Environ., vol. 20, no. 1, pp. 258–265, Mar. 2012, doi: 10.1007/s10924-011-0390-5.
- [586] Vroman I. and Tighzert L., Biodegradable Polymers, *Materials (Basel).*, vol. 2, no. 2, pp. 307–344, Apr. 2009, doi: 10.3390/ma2020307.
- [587] Howard G. T., Biodegradation of polyurethane: a review, Int. Biodeterior. Biodegradation, vol. 49, no. 4, pp. 245–252, Jun. 2002, doi: 10.1016/S0964-8305(02)00051-3.
- [588] Mahajan N. and Gupta P., New insights into the microbial degradation of polyurethanes, RSC Adv., vol. 5, no. 52, pp. 41839–41854, 2015, doi: 10.1039/C5RA04589D.
- [589] Huang S. J. and Roby M. S., Biodegradable Polymers Poly(amide-urethanes) [1], J. Bioact. Compat. Polym., vol. 1, no. 1, pp. 61–71, Jan. 1986, doi: 10.1177/088391158600100106.
- [590] Mohanan N., Montazer Z., Sharma P. K., and Levin D. B., Microbial and Enzymatic Degradation of Synthetic Plastics, *Front. Microbiol.*, vol. 11, no. November, Nov. 2020, doi: 10.3389/fmicb.2020.580709.
- [591] Kim Y. D. and Kim S. C., Effect of chemical structure on the biodegradation of polyurethanes under composting conditions, *Polym. Degrad. Stab.*, vol. 62, no. 2, pp. 343–352, Nov. 1998, doi: 10.1016/S0141-3910(98)00017-2.
- [592] Nohra B., Candy L., Blanco J.-F., *et al.*, From Petrochemical Polyurethanes to Biobased Polyhydroxyurethanes, *Macromolecules*, vol. 46, no. 10, pp. 3771–3792, May

2013, doi: 10.1021/ma400197c.

- [593] Ghasemlou M., Daver F., Ivanova E. P., and Adhikari B., Bio-based routes to synthesize cyclic carbonates and polyamines precursors of non-isocyanate polyurethanes: A review, *Eur. Polym. J.*, vol. 118, no. June, pp. 668–684, Sep. 2019, doi: 10.1016/j.eurpolymj.2019.06.032.
- [594] Aslam M., Kalyar M. A., and Raza Z. A., Polyvinyl alcohol: A review of research status and use of polyvinyl alcohol based nanocomposites, *Polym. Eng. Sci.*, vol. 58, no. 12, pp. 2119–2132, Dec. 2018, doi: 10.1002/pen.24855.
- [595] Ben Halima N., Poly(vinyl alcohol): review of its promising applications and insights into biodegradation, RSC Adv., vol. 6, no. 46, pp. 39823–39832, 2016, doi: 10.1039/C6RA05742J.
- [596] Es-saheb M. and Elzatahry A., Post-Heat Treatment and Mechanical Assessment of Polyvinyl Alcohol Nanofiber Sheet Fabricated by Electrospinning Technique, Int. J. Polym. Sci., vol. 2014, pp. 1–6, 2014, doi: 10.1155/2014/605938.
- [597] Baker M. I., Walsh S. P., Schwartz Z., and Boyan B. D., A review of polyvinyl alcohol and its uses in cartilage and orthopedic applications, J. Biomed. Mater. Res. Part B Appl. Biomater., vol. 100B, no. 5, pp. 1451–1457, Jul. 2012, doi: 10.1002/jbm.b.32694.
- [598] Rolsky C. and Kelkar V., Degradation of Polyvinyl Alcohol in US Wastewater Treatment Plants and Subsequent Nationwide Emission Estimate, Int. J. Environ. Res. Public Health, vol. 18, no. 11, p. 6027, Jun. 2021, doi: 10.3390/ijerph18116027.
- [599] Chiellini E., Corti A., D'Antone S., and Solaro R., Biodegradation of poly (vinyl alcohol) based materials, *Prog. Polym. Sci.*, vol. 28, no. 6, pp. 963–1014, Jun. 2003, doi: 10.1016/S0079-6700(02)00149-1.
- [600] Ye B., Li Y., Chen Z., et al., Degradation of polyvinyl alcohol (PVA) by UV/chlorine oxidation: Radical roles, influencing factors, and degradation pathway, Water Res., vol. 124, pp. 381–387, Nov. 2017, doi: 10.1016/j.watres.2017.05.059.
- [601] Cataldo F. and Angelini G., Some aspects of the ozone degradation of poly(vinyl alcohol), *Polym. Degrad. Stab.*, vol. 91, no. 11, pp. 2793–2800, Nov. 2006, doi: 10.1016/j.polymdegradstab.2006.02.018.
- [602] Eubeler J. P., Bernhard M., and Knepper T. P., Environmental biodegradation of synthetic polymers II. Biodegradation of different polymer groups, *TrAC Trends Anal. Chem.*, vol. 29, no. 1, pp. 84–100, Jan. 2010, doi: 10.1016/j.trac.2009.09.005.
- [603] Kawai F. and Hu X., Biochemistry of microbial polyvinyl alcohol degradation, Appl. Microbiol. Biotechnol., vol. 84, no. 2, pp. 227–237, Aug. 2009, doi: 10.1007/s00253-009-2113-6.
- [604] Liu C., Liu F., Cai J., *et al.*, Polymers from Fatty Acids: Poly(ω-hydroxyl tetradecanoic acid) Synthesis and Physico-Mechanical Studies, *Biomacromolecules*, vol. 12, no. 9, pp. 3291–3298, Sep. 2011, doi: 10.1021/bm2007554.

- [605] Huang Q., Hiyama M., Kabe T., et al., Enzymatic Self-Biodegradation of Poly(L -lactic acid) Films by Embedded Heat-Treated and Immobilized Proteinase K, *Biomacromolecules*, vol. 21, no. 8, pp. 3301–3307, Aug. 2020, doi: 10.1021/acs.biomac.0c00759.
- [606] DelRe C., Jiang Y., Kang P., et al., Near-complete depolymerization of polyesters with nano-dispersed enzymes, *Nature*, vol. 592, no. 7855, pp. 558–563, Apr. 2021, doi: 10.1038/s41586-021-03408-3.
- [607] Huang Q., Kimura S., and Iwata T., Development of self-degradable aliphatic polyesters by embedding lipases via melt extrusion, *Polym. Degrad. Stab.*, vol. 190, p. 109647, Aug. 2021, doi: 10.1016/j.polymdegradstab.2021.109647.
- [608] Liu M., Zhang T., Long L., et al., Efficient enzymatic degradation of poly (εcaprolactone) by an engineered bifunctional lipase-cutinase, Polym. Degrad. Stab., vol. 160, pp. 120–125, Feb. 2019, doi: 10.1016/j.polymdegradstab.2018.12.020.

CHAPTER 3: BREAKING IT DOWN: HOW THERMOPLASTIC STARCH ENHANCES POLY(LACTIC ACID) BIODEGRADATION IN COMPOST - A COMPARATIVE ANALYSIS OF REACTIVE BLENDS

3.1 Abstract

Poly(lactic acid) (PLA) is a sustainable, bio-based, and industrial compostable polymer with a recalcitrant abiotic degradation phase limiting its organic recovery to well-managed industrial composting facilities. We present a methodology to biodegrade PLA in industrial and home composting settings fully. Thermoplastic starch (TPS) and PLA were reactively blended by adding a chemical modifier and peroxide radicals to obtain a PLA-g-TPS blend by twin screw extrusion and later processed into films by cast extrusion. Biodegradation of the films was investigated using a direct measurement respirometer (DMR) for 180 days by tracking the CO_2 evolution in compost media at 58 and 37 °C, and the molecular weight (M_n) reduction was measured by size exclusion chromatography. The hydrophilic nature of TPS and its role as a nutrient source accelerated the degradation of PLA in both abiotic and biotic phases of the composting process. The kinetic curve of M_n reduction showed the positive effect of TPS on accelerating PLA hydrolysis during the lag phase in both mesophilic and thermophilic conditions due to increased chain mobility. This work unlocks the capability of PLA-based films to be successfully composted in industrial and home composting without compromising its desired properties for applications in everyday life.

3.2 Introduction

Poly(lactic acid) (PLA), a bio-based polymer derived from renewable resources with end-of-life scenarios such as recyclability and industrial compostability, is considered a green and sustainable polymer and an alternative polymer to displace the ever-growing fossil-based plastic pollution [1]. PLA has been used in medical applications, but its use has mainly grown in food packaging applications [2,3]. PLA and its products are labeled and marketed as industrial compostable [4,5], so it can be collected with the organic fraction of municipal solid waste (OFMSW) routed to industrial composting. Marketing bio-based and compostable plastics, such as PLA, fuels the assumption that they can be discarded with the organic fraction and treated and recovered in the same manner [6,7]. However, some challenges arise when these polymers cannot fully disintegrate in the same time frame as readily biodegradable organic fraction wastes, such as food, starch, and cellulose, leaving fragments at the end of the compost cycle. Mostly, this situation occurs due to the slow abiotic degradation controlling phase, such as in the case of PLA, associated with inherent properties, such as high initial molecular weight (M_n) and crystallinity (X_c). Although composting of the organic fraction is accomplished in industrial composting facilities, the infrastructures have not been precisely designed to keep bio-based and compostable plastics in mind. Therefore, some composters are not keen on accepting these products [8–10].

Standards are available to delineate what conditions polymers should meet to be certified as compostable in industrial composting operations [11]. Conditions for home and industrial composting are different since temperatures and humidities encountered in these two environments differ drastically. Since the material volume, reached temperature, and abundance of microorganisms are lower in home composting, it is very challenging to maintain the necessary conditions for several days.

Regardless of whether the composting is done at home, in a community backyard, or in an industrial facility, the process must ensure a succession of microbial communities (i.e., mesophiles-thermophiles-mesophiles) and the corresponding temperature regimes to operate. These factors ensure and guarantee the safety and quality of the composting process and the final product. Furthermore, this process and high temperatures are crucial to degrade a polymer, such as PLA with a high glass transition temperature (T_g c. 58 °C) and whose initial breakdown is based on being able to traverse a rapid abiotic stage to reduce the weight average molecular weight number (M_n) low enough (≤ 10 kDa) so that the microorganisms can use PLA oligomers as a food source [12]. Furthermore, the industrial composting process is highly controlled. Efficient control of temperature, moisture, and airflow is exercised to achieve complete waste biodegradation. Any deviation from the required conditions in terms of turning frequency of the piles, consistent mixing to maintain adequate aeration and pH, and reduction of the composting process can lead to the formation of anaerobic pockets or increase the time needed for PLA to initially degrade through the abiotic reduction process and delay the complete breakdown of PLA, thereby leaving behind partially degraded or whole packages at the end of the composting cycle. Additionally, polymer fragments at the compost pile surface could take longer to degrade since they are not exposed to the same conditions. This is especially relevant for plastics that can constantly resurface during pile turning due to their lower density. So, incomplete biodegradation of PLA as well as other compostable plastics can impact the sale of compost by failing to abide by the agricultural compost requirements. To avoid this problem, some composting facilities are apprehensive about incorporating PLA with organic wastes [9]. Methods able to accelerate the biodegradation of PLA in industrial composting facilities and possibly expand its biodegradation efficiency in home/backyard composting are highly sought.

One potential way to improve PLA properties and its degradation rate in ambient conditions is by blending it with different biodegradable polymers to improve specific properties such as toughness, flexibility, and ductility and promote its complete biodegradation post-use [1,13–15]. One such polymer that can be used to blend with PLA to improve its biodegradability is starch, which is low-cost, renewable, non-toxic, readily available, and 100% bio-based. Thermoplastic starch (TPS) is a plasticized polymer used to
produce blends with different polymers. Starch undergoes rapid biodegradation by enzymatic hydrolysis to produce glucose, which is further assimilated to produce CO_2 and H_2O in compost or soil environments [16]. To process TPS efficiently and to blend and derive good interfacial adhesion between hydrophilic starch and hydrophobic PLA, compatibilizers are generally used [17–19]. Blending PLA and TPS helps improve the oxygen and water vapor barrier properties [20] and the elongation at break of the resulting matrix [21]. Furthermore, the hydrophilic feature of TPS can enhance PLA's sensitivity to humid environments and act as an excellent initial nutrient source for the microbes when introduced into the compost environment [22–24].

PLA is environmentally benign, to begin with, and the addition of starch derived from non-human crop consumption creates a safe polymer. Biodegradation of PLA-g-TPS ultimately will not leave behind traces of microplastics since degradation of the PLA evolves to lactic acid traces and degradation of the starch evolves to glucose, two naturally occurring and benign monomers that can be acted upon and utilized by the microorganisms. Adding TPS can further help accelerate the low abiotic degradation rate of PLA at mesophilic conditions, which otherwise takes longer. PLA-starch blends can open avenues for home/backyard composting and address the growing concern about microplastics [25–28] and their exo-toxicological effects [29,30] since their formation can be avoided.

In recent years, several studies have been conducted focusing on the biodegradation of PLA in soil environments [31–34] by introducing starch in the PLA matrix, where the biodegradation was determined by visual analysis, weight loss, or change in mechanical properties but not CO_2 evolution or M_n reduction. Biodegradation studies have also been conducted where additives such as wood flour [24], coir [35], and montmorillonite [36] have been blended with PLA, in addition to starch. Still, the additives failed to enhance properties, and the resulting biodegradation was attributed to the presence of starch. However, a detailed assessment of PLA and TPS mineralization and M_n evolution is still lacking, and there is a need to understand the role of starch in accelerating the biodegradation of PLA in compost. In this work, we investigated the role of starch in expediting the biodegradation of PLA under thermophilic and mesophilic conditions in a compost environment, and we monitored CO₂ evolution and quantified the changes in M_n throughout the test duration so as to replicate real-life scenarios encountered under home composting and industrial composting conditions.

3.3 Materials and methods

3.3.1 Materials

PLA resin, IngeoTM 2003D with L-lactic content of 96%, was provided by NatureWorks LLC (Minnetonka, MN, USA). Cassava starch was procured from Aldema LLC (Cooperativa Agricola e Industrial San Alberto Ltda., Puerto Rico, Misiones, Argentina), with an amylose content of $25 \pm 6\%$ wt/wt and moisture content of approximately 12%. Glycerol (>99.5%), maleic anhydride (MA), dicumyl peroxide (DCP), and cellulose (~20 µm particle size) were purchased from Sigma-Aldrich (Milwaukee, WI, USA). Except for the PLA resin, all materials were processed in the condition they were received in. PLA pellets were dried at 50 °C overnight at 67 kPa to avoid hydrolytic degradation while processing.

3.3.2 Preparation of PLA and thermoplastic starch (TPS) masterbatches and films

Cassava starch and glycerol (70/30% wt.) were mixed and held for 12 h before processing into the TPS masterbatch. The temperature profile for the TPS masterbatch was set at 25/100/105/110/115/120/120/120/115/115 °C for each zone from feed to the die, with a screw speed of 120 rpm. A ZSK-30 co-rotating twin screw extruder (Century, Traverse City, MI, USA) [37] was used to process the various masterbatches; the screw length was 1,260 mm and the diameter was 30 mm, with an L/D ratio of 42:1. PLA, MA, and DCP were mixed prior to processing; PLA-g-MA (PLA with 2% wt. MA and 0.65% wt. DCP, based on PLA weight) was processed in the twin screw extruder. The temperature profile for the PLA-g-MA masterbatch was set at 140/150/160/160/160/170/170/170/170/160 °C for each zone from feed to the die, with a screw speed of 120 rpm, feed rate of 70 g/min, and residence time of approximately 3 min. The extrudate from the die was fed through a water bath and pelletized using a BT 25 pelletizer (Scheer Bay Co., Bay City, MI, USA). The PLA-g-MA masterbatch was kept in an oven at 50 °C for 3 h to remove any traces of moisture and was later transferred to the freezer and stored at -15 °C until processed into films. The same procedure was followed to produce a PLA control masterbatch. To obtain PLA-g-TPS, PLA (56% wt.), TPS (30% wt.), and PLA-g-MA (14% wt.) were processed together to produce the final reactive blend. The temperature profile for the PLA-g-TPS masterbatch was set at 140/150/160/160/160/170/170/170/170/160 °C for each zone from feed to the die, with a screw speed of 120 rpm, feed rate of 70 g/min, and residence time of approximately 3 min, which was dried in a vacuum oven at 50 °C for 3 h to remove any residual traces of moisture. Figure S1 of the supporting information depicts the processing of PLA-g-MA, PLA and TPS masterbatches and the temperature profile for the twin screw extruder.

The masterbatch was introduced in a RCP-0625 microextruder (Randcastle Extrusion Systems, Inc., Cedar Grove, NJ, USA) to produce cast films; the screw diameter was 1.5875 cm, with an L/D ratio of 24/1 and volume of 34 cm³, and the temperature profile was set at 140/150/160/160/160/170/168 °C from feed to die. The screw speed, nip roller, and winding roller were set at 30 rpm, 50 rpm, and 12 rpm, respectively [38]. The same procedure was used to process PLA films. Figure S2 shows the processing of PLA-*g*-MA, PLA and TPS masterbatches into PLA-*g*-TPS film using the cast film extruder.

3.3.3 Biodegradation test in simulated compost conditions

The biodegradability of PLA and PLA-g-TPS films was investigated by analysis of evolved CO₂ in a simulated composting environment using an in-house built direct measurement respirometer (DMR) system [11,39]. The biodegradation tests were run at 37 \pm 2 °C and 58 \pm 2 °C to simulate mesophilic and thermophilic environmental conditions, respectively. The DMR chamber was maintained at specific temperatures, and a Li-COR[®] LI-820 non-dispersive infrared gas analyzer (Lincoln, NE, USA) was used to measure the evolving CO₂ concentration from each DMR bioreactor. The relative humidity was maintained at 50% \pm 5% and the optimal airflow rate was set at 40 \pm 2 sccm [40,41].

Compost was procured from the Michigan State University composting facility (East Lansing, MI, USA). The compost was filtered using a 10-mm screen to remove any larger lumps of material and then was conditioned at 37 ± 2 °C and 58 ± 2 °C for one week before the respective test use. The procured compost was characterized for its physicochemical parameters (presented in **Table A3.1**, Appendix 3B). When running the biodegradation test, deionized water was added to the compost to regulate and sustain the moisture content at around 50%. Vermiculite, an inert, inorganic hydrous phyllosilicate matrix, obtained from Sun Gro[®] Horticulture Distribution Inc. (Bellevue, WA, USA), was mixed 1:4 (wt/wt) with compost. Vermiculite does not affect the biodegradation test and helps provide better aeration, improving the accessible space, thus improving the microbial activity [42].

Each bioreactor was loaded with 400 g of compost which was then mixed with 8 g of film samples (cut into 1-cm² pieces) or 8 g of cellulose. Experiments with film samples in compost media were run in triplicate. Cellulose, a positive control reference, was used in the test due to its high biodegradation nature and as requested by ASTM D5338-15(2021) [11]. Three bioreactors were run with compost only (blank) and without any film samples or cellulose to identify the background CO_2 evolution. CO_2 -free air (≤ 30 ppm CO_2) was passed through each bioreactor, and the evolved CO_2 was measured for a set period. After completing measurements for each bioreactor, the whole system was purged using CO_2 -free air to ensure no residual CO_2 was disturbing the baseline test [41]. The mineralization formula shown in equation (3.1) was used to calculate the amount of carbon that was converted to CO_2 :

Mineralization % =
$$\frac{(CO_2)_t - (CO_2)_b}{M_t \, x \, C_t \, x \, \frac{44}{12}} \, x \, 100$$
 (3.1)

where $(CO_2)_t$ is the average cumulative mass of CO_2 evolved from the bioreactor containing the sample, $(CO_2)_b$ is the average cumulative mass of CO_2 evolved from the blank, M_t represents the total mass of the sample in the bioreactor, C_t is the total carbon content of the sample and is derived from CHN analysis, 44 is the molecular weight of CO_2 , and 12 is the atomic weight of carbon. The equation numerator represents the actual CO_2 evolved from the sample after accounting for the background compost activity, and the denominator depicts the maximum theoretical CO_2 the sample produces, i.e., when 100% of the sample carbon is converted to CO_2 .

3.3.4 Hydrolysis experiment

A hydrolysis test method adapted from ASTM D4754-18 [43] was run for PLA films at 58 ± 2 °C and 37 ± 2 °C to understand the hydrolytic degradation. The hydrolysis cell consisted of a stainless-steel wire, glass beads, and a glass vial with cap. PLA films were cut into small discs of 2-cm diameter, and ten such discs were strung into a stainless-steel wire and separated by glass beads. The vial was filled with 35 mL of HPLC-grade water (J.T. Baker, Center Valley, PA, USA). The water was preconditioned, and the hydrolysis cell was stored at the same temperature. Triplicates of the PLA films were retrieved at predetermined time intervals and dried before running size exclusion chromatography to assess the M_n reduction.

3.3.5 Size exclusion chromatography

The film samples were collected at specific intervals to determine the Mn, weight average molecular weight (M_w) , and molecular weight distribution (MWD). Samples were weighed and dissolved in tetrahydrofuran (THF) solvent in a ratio of 2:1. THF solvent was used as a mobile phase. A size exclusion chromatography (SEC) unit from Waters Corp. (Milford, MA, USA) was supplied with an isocratic pump (Waters[®] 1515), an autosampler (Waters[®] 717), a series of Styragel[®] columns (HR4, HR3, HR2), and a refractive index detector (Waters[®] 2414). The detector was maintained at 35 °C, and a 1 mL/min flow rate was applied for the mobile THF solvent. The Mark-Houwink constants of K = 0.000174 mL/g and α = 0.736 were used to determine the M_n and M_w of the PLA fraction in the samples. A detailed description of the technique can be found elsewhere [44]. The data was analyzed using the Waters BreezeTM 2 software.

3.3.6 Elemental analysis

The carbon, hydrogen, and nitrogen content of the PLA and PLA-g-TPS films was calculated using a PerkinElmer Series II CHNS/O Elemental Analyzer (PerkinElmer Inc., Shelton, CT, USA). Approximately 2 mg of sample was weighed in a small tin capsule and tested. Samples were tested in triplicates. The values are presented in **Table 3.1**.

Material	% Carbon	% Hydrogen	% Nitrogen
Cellulose	42.50 ± 0.34	6.53 ± 0.05	0.04 ± 0.01
PLA	49.72 ± 0.19	5.72 ± 0.04	0.11 ± 0.07
PLA-g-TPS	48.20 ± 0.05	6.00 ± 0.04	0.05 ± 0.01

Table 3.1 Percent of carbon, hydrogen, and nitrogen content by weight of cellulose and film samples.

3.3.7 Statistical analysis

The statistical analysis was conducted using MINITABTM software (Minitab Inc., State College Park, PA, USA). The statistical significance at p < 0.05 was evaluated using one-way ANOVA and Tukey-Kramer test. Data is reported as means ± standard deviation.

3.4 Results and discussion

PLA films have a recalcitrant abiotic degradation stage, limiting its deployment to well-managed industrial composting facilities. Facilities that rotate compost in a very short period ($c. \leq 2$ months) and do not follow a practice of allowing a full cycle of mesophilic, thermophilic, mesophilic, and mature phases are not able to fully biodegrade PLA, leaving residual at the end of the period [8]. Previously, we developed PLA-*g*-TPS reactive blend films with suitable thermal, mechanical, and barrier properties for several applications, including packaging and agriculture, to be recovered through composting [37,45]. Figure A3.3 b, Appendix 3C shows the SEM images for PLA-*g*-TPS film showing uniform dispersion of the TPS phase in the PLA matrix by reactive blending as opposed to physical blending. Here, we present the biodegradation of these PLA-*g*-TPS films in simulated composting at thermophilic and mesophilic conditions using a DMR unit and report the M_w reduction and thermal properties.

3.4.1 CO₂ evolution and mineralization of PLA and PLA-g-TPS films

PLA-*g*-TPS films were evaluated alongside PLA in simulated composting conditions at 58 ± 2 °C and 37 ± 2 °C to understand the effect of reactive blending TPS on PLA degradation. **Figure 3.1a and b** present the CO₂ evolution and % mineralization of the blank (compost only), cellulose, PLA films, and PLA-*g*-TPS films in compost at 58 ± 2 °C. The blank showed maximum CO₂ evolution of 28.3 g at day 180. The positive control cellulose achieved a 40.8 g for CO₂ evolution and showed maximum mineralization of *c*. 100%. No lag phase was observed for cellulose, as it is a readily available food source for microorganisms and easily biodegradable. Cellulose is degraded by the action of a battery of enzymes that work simultaneously and synergistically. Cellulases catalyze the hydrolysis of β -1,4-linkages in the cellulose [46]. The action of exoglucanases and endoglucanases on the ends and at random internal sections of cellulose's amorphous region produces varying lengths of cellooligosaccharides, which are then hydrolyzed by glucosidases to produce glucose [47]. Glucose is finally converted to CO₂ through a series of further cycles. Previous research has shown fungi, a few bacteria species, and actinomycetes in compost and the soil environment produce cellulase and are completely involved in the degradation of cellulose [48–52].

The biodegradation curve for PLA showed a lag phase of around 20 days, attributed to the initial abiotic hydrolysis phase. Ester bonds of PLA are cleaved during the hydrolytic abiotic degradation phase due to its susceptibility to water. As hydrolytic degradation proceeds, PLA chains are broken into smaller fragments, releasing low M_w oligomer populations. These lactic acid oligomers are available for microbial assimilation, releasing CO_2 and water, which can be observed during the biotic degradation phase. A maximum CO_2 evolution of 39 g and mineralization of *c*. 77% was observed for PLA over the test duration of 180 days at 58 °C.

PLA-g-TPS films showed a CO₂ evolution of 42.8 g and a maximum mineralization of 100% over the test duration of 180 days. PLA-g-TPS did not show any lag phase compared to the 20-day phase for PLA. This finding can be ascribed to the presence of TPS acting as an initial food supply for the microorganisms until the PLA undergoes hydrolysis, fragmentation, and is depolymerized into small fragments available for further assimilation. Starch, like cellulose, is a natural hydrophilic, biodegradable polymer and can be used up almost immediately as the microorganisms get acclimatized to the compost environment. The presence of hydroxyl groups in TPS contributes to and enhances the biodegradation of PLA by further enabling the disintegration process of PLA-g-TPS films. Several research studies have shown that introduction of starch increases the water absorption characteristics of a matrix. Maran et al. [53] demonstrated improved water absorption due to the presence of starch in specimens buried in soil for degradation studies. Also, the presence of starch increases the hygroscopic characteristics of films, which favors water absorption, thereby providing suitable conditions for fast hydrolysis and microbial invasion and colonization [34,54,55]. The presence of amylopectin branched hydroxylated chains improves the water penetration in PLA-g-TPS film matrices [33,56]. This sensitivity of starch to water helps in the enzymatic hydrolysis to glucose.



Figure 3.1 CO₂ evolution and % mineralization of PLA and PLA-*g*-TPS films in compost at 58 °C (a & b) and 37 °C (c & d). The shade in the background for each material represents the standard error between replicates.

Figure 3.1c and d depict the cumulative CO_2 evolution and % mineralization for cellulose, PLA films, and PLA-*g*-TPS films at 37 °C to evaluate the biodegradation behavior in mesophilic conditions. Cellulose achieved mineralization of *c*. 86% in 50 days and evolved around 37 g of CO_2 by the end of 180 days. There was no lag phase for cellulose, as also observed at 58 °C.

Bioreactors containing compost and PLA produced c. 22 g of CO_2 in compost around 180 days. In contrast, blank bioreactors evolved c. 26 g of CO_2 , which is translated to negative mineralization in the case of PLA. In a blank bioreactor, the microbes can easily utilize the organic matter in the compost, whereas in a bioreactor containing PLA, the presence of PLA leads to a reduction in the working efficiency, which translates to reduced carbon conversion as a collective. The negative mineralization values are generated as an artifact and indicate that the blank bioreactors produce more CO₂ than the bioreactors containing PLA samples. This could mean that PLA offers a physical hydrophobic barrier to water and air, and the microbes have difficulty accessing the carbon source as nutrients. It could also indicate that there is non-uniform mixing of the compost and PLA samples in the bioreactors, which could additionally hinder the microbes. Any change in the optimal conditions for the biological activity, either due to an insufficient supply of water (dryness of the compost), excess of water (agglomeration), or obstruction in the airflow passage, can create an unfavorable environment for microbial development. The negative mineralization values in the case of PLA do not necessarily imply the absence of hydrolytic degradation or enzymatic activity due to the action of extracellular enzymes secreted by the microbes but more like inhibition of the microbial activity due to the hydrophobic layer barrier created by the presence of the sample. The lower values of CO_2 evolution in the case of PLA indicate that the samples are still undergoing hydrolysis and are yet to be reduced to a point (≲10 kDa) needed to activate the

biodegradation stage where the microorganisms can start assimilating low M_w PLA, such as oligomers, dimers, and monomers for their biochemical processes. The hydrolytic degradation progresses very slowly when PLA is exposed to temperatures lower than T_g (c. 60 °C) due to scarce chain mobility of the amorphous fraction and low water diffusion in the crystalline fraction for effective bulk erosion [12,57,58]. PLA segments have little to no mobility and are not flexible below T_g , preventing diffusion or attack by water. Since the initial and ratelimiting step in PLA degradation is chemical hydrolysis, the lack of chain scissions, perhaps accompanied by simultaneous low surface colonization by microorganisms due to the hydrophobic surface [56], could have further prevented penetration by water molecules and slowed down the hydrolysis rate. Further attestation to this rationale is the lower CO₂ evolution values obtained for PLA (22 g) compared to the blank (26 g). The above reasons, as well as the lower chemical hydrolysis rate of PLA, can account for lower CO₂ evolution as compared to the blank.

In the case of PLA-g-TPS, c. 58% mineralization was observed by the end of the test (180 d) with an upward mineralization trend, since starch, like cellulose, is a readily available nutrient for the microbial species in the compost [59]. No lag phase was seen in the case of PLA-g-TPS films, which evolved 34.2 g of CO_2 by the end of the test. Ho and Pometto [22] documented mineralization of c. 70% for starch alone at 28 °C after 98 days and a higher mineralization rate with increasing temperature. The mineralization for PLA-g-TPS films at mesophilic conditions corroborates the research presented earlier [23,60,61]. The hydroxylated chains in TPS impart hydrophilicity to PLA, pave the way for water dispersion and hydrolysis in the blend matrix and support microbial development and growth [33,62]. The hydrophilic characteristics further stimulate microbial colonization and biofilm formation on the surface of PLA-g-TPS films. This is further corroborated by changes in the

roughness and contact angle for PLA and PLA-*g*-TPS film due to the addition of TPS, as seen in **Figure A3.4** and **Figure A3.5**, Appendix 3D.

Starch is enzymatically hydrolyzed by the action of extracellular enzymes, namely α/β -amylase belonging to the glycoside hydrolase family. Glucose produced as a result is then converted to pyruvic acid by glycolysis. Pyruvic acid is a precursor for the tricarboxylic acid cycle for energy production [63]. Once the microorganisms use starch as their food source, fragmented PLA films (holes and cracks) are left behind. These structural discrepancies in the form of macroscopic fractures aid in the biodegradation process. Starch biodegradation and PLA chemical hydrolysis processes complement each other, resulting in improved biodegradation of a starch-PLA mixture compared to PLA alone. Usually, at higher temperatures, such as industrial composting conditions, these complementary processes happen concurrently, and hence an improved degradation is seen for PLA. **Figure 3.1b and d** show that adding TPS to PLA not only accelerates the disintegration of PLA-g-TPS films in industrial composting but also opens a venue to make PLA blends biodegradable in home composting conditions if key parameters such as aeration and moisture are controlled.

3.4.2 Molecular weight reduction rate

The M_n was measured, and the reduction rate calculated during the abiotic hydrolysis stage was used to elucidate if the TPS fraction accelerated the hydrolysis of PLA besides also enhancing the biotic degradation. PLA and PLA-*g*-TPS films were retrieved at specific time intervals from the sampling bioreactors to track the reduction of M_w .

Figure 3.2 presents the MWD over time for PLA films and the PLA fraction in PLAg-TPS films in thermophilic and mesophilic conditions. For 58 ± 2 °C (**Figure 3.2a & b**), both films show shifting of the peaks depicting a decrease in M_n and broadening of the peaks, indicating an increase in dispersity (*D*) associated with hydrolytic degradation of the PLA chains. The shifting of peaks in the abiotic phase reveals that the chemical hydrolysis is due to random chain scission occurring in the bulk of the polymer and is not restricted to the surface [64,65]. The change in the peaks from monomodal to multimodal after day 15 is a characteristic feature of reconfiguration and crystallization of the newly crystalline region resulting from the regrouping of newly formed short polymer chains formed at $M_n \leq 10$ kDa. The amorphous region is subjected to hydrolysis, while the crystalline region remains stable [65]. Slight differences can be observed at day 20 on the MWD but may not be sufficient to impact the samples. Although the portion of PLA in PLA-g-TPS overall MWD follows the same path, no lag phase was observed in **Figure 3.1b**, supporting a more rapid disintegration of the samples.

For 37 ± 2 °C (**Figure 3.2c & d**), the PLA MWD shift occurred very slowly, as shown in **Figure 3.2c**, especially in the case of PLA. The peak amplitude remained the same, even at day 180, and a minimal shift of M_n is observed, indicating that chemical hydrolysis happened very slowly. Whereas for PLA-*g*-TPS films (**Figure 3.2d**), shifting and significant peak reduction are observed after day 90, indicating the reduction of the M_n and initial change in crystallinity (**Figure A3.6 and Table A3.2**, Appendix 3F). As biodegradation of the TPS component of the PLA-*g*-TPS film proceeded, so did hydrolysis of the PLA matrix, shown by the broadening of the peaks after day 60 and the peak reduction. This was accompanied by a stable increase in % mineralization, as shown in **Figure 3.1d**.



Figure 3.2 Molecular weight distribution of PLA and PLA-*g*-TPS films in compost at 58 °C (a & b) and 37 °C (c & d), respectively.

Figure 3.3 shows the reduction in M_n of PLA and PLA-g-TPS films in compost at 58 \pm 2 °C and 37 \pm 2 °C. In the thermophilic test, the samples could be retrieved only until day 20. For the mesophilic test, the samples were collected until day 180. **Figure 3.3** shows a more rapid reduction of the M_n at 58 \pm 2 °C than at 37 \pm 2 °C due to the higher hydrolysis rate at elevated temperatures. Furthermore, a larger deviation of the M_n of PLA-g-TPS seemed to occur at 180 days.



Figure 3.3 Normalized M_n reduction as a function of time for PLA and PLA-*g*-TPS films in compost at 58 °C and 37 °C. The experimental data were fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The data points indicate the original experimental triplicate values at specific times.

Hydrolysis experiments for PLA films were concurrently performed at 58 ± 2 °C and 37 ± 2 °C to decouple, simulate, and understand the abiotic phase (lag phase of biodegradation curve) dominated by chemical hydrolysis, as also reported elsewhere [44]. Hydrolysis in solid state (compost) and water are not exactly the same. We have previously conducted these experiments in sterilized vermiculite to show the difference; however, in this case, hydrolysis in water should be a sufficient proxy to model the abiotic stage. Additionally, the hydrolysis data obtained at 58 ± 2 °C was also used to predict the lifetime of PLA at 37 ± 2 °C using the master curve technique reported by Limsukon et al., who studied the hydrolytic degradation of PLA films over a series of temperatures (40 to 95 °C) and constructed a master curve to

predict the lifetime of PLA films at any temperature [57]. More details about this methodology can be found in section S5 of the Supporting Information. For this work, the fit obtained from the accelerated degradation testing at 58 ± 2 °C was used to predict the hydrolytic degradation at 37 ± 2 °C, as seen in **Figure 3.4**. To achieve the same reduction in M_n , *c*. 10 kDa or similar disintegration values for PLA, approximately 730 days were required at 37 ± 2 °C as compared to 25 days at 58 ± 2 °C.



Figure 3.4 Hydrolysis of PLA at 58 ± 2 °C (a) and master curve prediction for hydrolytic degradation at 37 ± 2 °C (b). a) represents the actual experimental data obtained at 58 ± 2 °C. The data points indicate the original experimental triplicate values at specific times.

As shown in **Figure 3.3** and **Figure 3.4**, PLA degradation follows a first-order reaction, where chemical hydrolysis is the dominant phase at high and lower temperatures and controls the degradation rate. So, the degradation rate of PLA and PLA in PLA-*g*-TPS in compost at 37 ± 2 °C was modeled to a first-order reaction. **Table 3.2** details the rate constant k (d⁻¹) for M_n reduction of PLA and PLA-*g*-TPS films for both conditions. No significant difference in M_n reduction among PLA and PLA-*g*-TPS films was observed in thermophilic conditions at 58 ± 2 °C. At the higher temperature, PLA is in the rubbery state, enabling easy diffusion of water and, thereby, more rapid hydrolysis. The presence of starch definitely aids in the initial disintegration and biodegradation of PLA [23], but at the higher temperature, the process of chemical hydrolysis is dominant and governs the reduction in M_n to differentiate any advantage associated with the presence of starch on the abiotic degradation of PLA.

Table 3.2 Rate constant (*k*) for PLA and PLA-*g*-TPS films evaluated in compost media at 58 \pm 2 °C and 37 \pm 2 °C.

Film	k (d ⁻¹) at 58 ± 2 °C	k (d ⁻¹) at 37 ± 2 °C
PLA	0.0885 ± 0.0010^{a}	0.0045 ± 0.0001^{a}
PLA-g-TPS	0.0900 ± 0.0167^{a}	0.0056 ± 0.0002^{b}

Values with different letters in a column are statistically different ($\alpha = 0.05$ Tukey-Kramer Test).

The chemical hydrolysis process proceeds at a lower rate at 37 ± 2 °C than at 58 ± 2 °C. Conversely, there is a significant effect on the reduction in M_n of the PLA-g-TPS films due to the presence of starch in the PLA matrix at 37 ± 2 °C, suggesting that the addition of starch leads to enhanced degradation at mesophilic conditions. The presence of starch increases the hydrophilicity of the PLA-g-TPS films, further promoting water absorption and microbial colonization and leading to changes in the structural integrity. This results in the accelerated degradation of PLA in PLA-g-TPS films as compared to neat PLA with a different (*k*) rate constant at 37 ± 2 °C.

Lv et al. conducted a biodegradation study in outdoor soil conditions to evaluate and account for the role starch plays in PLA degradation, and found higher degradation of PLA in PLA/starch composites as compared to pure PLA; the authors reported the theoretical and experimental values of PLA degraded in the presence of starch through weight loss and component content analysis determination [31,66]. Yu et al. also demonstrated higher biodegradability, using the mass loss technique, for PLA/starch blends as compared to PLA in soil conditions of 44 °C and high humidity due to the presence of acetylated starch [67]. Palai et al. recorded M_w reduction, depletion in mechanical properties, along with scanning electron microscopy morphological analysis, and reported higher degradation for PLA/starch blown films as compared to PLA and PLA/PBSA (poly butylene succinate-*co*-adipate) blend in a soil environment at 30 ± 2 °C due to the presence of TPS [68].

In this work, we demonstrated that reactive blending of starch with PLA presents an array of options for PLA to be easily accessed by water and microorganisms, thereby creating avenues for accelerated disintegration and rapid mineralization of PLA-g-TPS films at thermophilic conditions, which can increase the acceptance of PLA by industrial composting facilities. The blending of TPS and PLA also improves PLA biodegradation in mesophilic environments, thereby opening an opportunity for creation of PLA-based films that can be home composted. Overall, these findings can also help tackle the ever-growing problem of microplastics left behind after disintegration and incomplete biodegradation of plastics.

3.5 Conclusion

The biodegradation performance of a polyester-based reactive blend was evaluated in simulated composting conditions in both mesophilic and thermophilic conditions. PLA-g-TPS films showed the highest mineralization trend at 58 ± 2 °C by day 180, indicating that the presence of starch significantly affects the final biodegradation of PLA by reducing the initial lag phase and accelerating M_n reduction. Under mesophilic conditions (37 ± 2 °C), PLA-g-TPS showed improved degradation compared to PLA due to plasticized starch, achieving more than 57% mineralization. The fact that the amount of starch was around 30% of the blend indicates that PLA is being degraded and TPS influences M_n reduction of PLA as well. The presence of starch imparts hydrophilicity and surface roughness to PLA-g-TPS composites and creates favorable conditions for enhanced microbial activity. Furthermore, starch accelerates the biodegradation of pure PLA by acting as a food reservoir for the growth of microorganisms while PLA undergoes chemical and enzymatic hydrolysis. Starch can be used to improve PLA degradation in industrial composting conditions by reducing the lag phase of disintegration of PLA and, even at mild conditions as in home composting settings, by inducing biodegradation during the initial days.

3.6 Acknowledgments

P.M. thanks the undergraduate and graduate students at the School of Packaging, Michigan State University, for their help in running the biodegradation tests reported in this manuscript. P.M. also acknowledges the School of Packaging at Michigan State University for partial support; R.A. acknowledges the USDA National Institute of Food and Agriculture and Michigan State University AgBioResearch, Hatch project number MICL02665, for partial support of the study.

REFERENCES

- [1] Bher A., Castro-Aguirre E., and Auras R., END-OF-LIFE SCENARIOS FOR POLY(LACTIC ACID), in *Poly (Lactic Acid) Synthesis, Structures, Properties, Processing, Applications, and End of Life*, 2022, pp. 559–579.
- [2] Ncube L. K., Ude A. U., Ogunmuyiwa E. N., *et al.*, Environmental impact of food packaging materials: A review of contemporary development from conventional plastics to polylactic acid based materials, *Materials (Basel)*., vol. 13, no. 21, pp. 1–24, 2020, doi: 10.3390/ma13214994.
- [3] Muller J., González-Martínez C., and Chiralt A., Combination of Poly(lactic) Acid and Starch for Biodegradable Food Packaging, *Materials (Basel).*, vol. 10, no. 8, p. 952, Aug. 2017, doi: 10.3390/MA10080952.
- [4] ASTM, ASTM D6400 Standard Specification for Labeling of Plastics Designed to be Aerobically Composted in Municipal or Industrial Facilities, p. 3, 2019, doi: 10.1520/D6400-19.
- [5] ASTM Standard D6868, Standard Specification for Labeling of End Items that Incorporate Plastics and Polymers as Coatings or Additives with Paper and Other Substrates Designed to be Aerobically Composted in Municipal or Industrial Facilities, 2021. doi: 10.1520/D6868-21.
- [6] Paritosh K., Yadav M., Mathur S., *et al.*, Organic fraction of municipal solid waste: Overview of treatment methodologies to enhance anaerobic biodegradability, *Front. Energy Res.*, vol. 6, p. 75, Aug. 2018, doi: 10.3389/FENRG.2018.00075/BIBTEX.
- [7] Sun Y., Organic Waste Treatment Methods & Organic Waste Management, Nov. 2018. https://www.yessun.com/en/introduction-to-organic-waste-treatment-methods (accessed Sep. 21, 2022).
- [8] Bioplastics news, What Happens to Bioplastics at Industrial Composting Sites?, 2020. https://bioplasticsnews.com/2020/06/22/compostable-plastics-industrial-compostingsite/ (accessed Mar. 04, 2022).
- [9] EcoEnclose, Composters Don't Want Your Packaging, Especially Your Bioplastic, 2021. https://www.ecoenclose.com/blog/composters-dont-want-your-packaging-especiallyyour-bioplastic-/ (accessed Mar. 31, 2022).
- [10] Association of Recyclers of Oregon, A Message from Composters Serving Oregon, 2020. https://oregonrecyclers.org/blog/composters-say-no-compostable-packaging-andserviceware (accessed Jan. 31, 2022).
- [11] ASTM International, ASTM D5338-15(2021): Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials Under Controlled Composting Conditions, Incorporating Thermophilic Temperatures, in ASTM Standards, 2021, pp. 1–7.
- [12] Castro-Aguirre E., Auras R., Selke S., *et al.*, Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions,

Polym. Degrad. Stab., vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.

- [13] Detyothin S., Production and characterization of thermoplastic cassava starch, functionalized poly(lactic acid), and their reactive blends, Michigan State University, 2012.
- [14] Nazrin A., Sapuan S. M., Zuhri M. Y. M., et al., Nanocellulose Reinforced Thermoplastic Starch (TPS), Polylactic Acid (PLA), and Polybutylene Succinate (PBS) for Food Packaging Applications, *Front. Chem.*, vol. 8, p. 213, Apr. 2020, doi: 10.3389/FCHEM.2020.00213/BIBTEX.
- [15] Rosli N. A., Karamanlioglu M., Kargarzadeh H., and Ahmad I., Comprehensive exploration of natural degradation of poly(lactic acid) blends in various degradation media: A review, *Int. J. Biol. Macromol.*, vol. 187, no. July, pp. 732–741, 2021, doi: 10.1016/j.ijbiomac.2021.07.196.
- [16] Pokhrel S., A review on introduction and applications of starch and its biodegradable polymers, *Int. J. Environ.*, vol. 4, no. 4, pp. 114–125, Dec. 2015, doi: 10.3126/IJE.V4I4.14108.
- [17] Khan B., Bilal Khan Niazi M., Samin G., and Jahan Z., Thermoplastic Starch: A Possible Biodegradable Food Packaging Material—A Review, J. Food Process Eng., vol. 40, no. 3, p. e12447, Jun. 2017, doi: 10.1111/JFPE.12447.
- [18] Humphreys R., Thermoplastic Starch: A Renewable, Biodegradable Bioplastic, *Polymer Innovation*, 2013. https://polymerinnovationblog.com/thermoplastic-starch-arenewable-biodegradable-bioplastic/ (accessed Mar. 29, 2022).
- [19] Surendren A., Mohanty A. K., Liu Q., and Misra M., A review of biodegradable thermoplastic starches, their blends and composites: recent developments and opportunities for single-use plastic packaging alternatives, *Green Chem.*, vol. 24, no. 8606, 2022, doi: 10.1039/d2gc02169b.
- [20] Kulkarni A. and Narayan R., Effects of Modified Thermoplastic Starch on Crystallization Kinetics and Barrier Properties of PLA, *Polymers (Basel).*, vol. 13, no. 23, p. 4125, Nov. 2021, doi: 10.3390/polym13234125.
- [21] Yusoff N. H., Pal K., Narayanan T., and de Souza F. G., Recent trends on bioplastics synthesis and characterizations: Polylactic acid (PLA) incorporated with tapioca starch for packaging applications, *J. Mol. Struct.*, vol. 1232, p. 129954, May 2021, doi: 10.1016/J.MOLSTRUC.2021.129954.
- [22] Ho K. L. G. and Pometto A. L., Temperature effects on soil mineralization of polylactic acid plastic in laboratory respirometers, J. Environ. Polym. Degrad., vol. 7, no. 2, pp. 101–108, 1999, doi: 10.1023/A:1021812418325.
- [23] Wilfred O., Tai H., Marriott R., et al., Biodegradation of Polylactic Acid and Starch Composites in Compost and Soil, Int. J. Nano Res., vol. 1, no. 2, pp. 1–11, 2018, Accessed: Mar. 28, 2022. [Online]. Available: www.innovationinfo.org.

- [24] Lv S., Zhang Y., Gu J., and Tan H., Biodegradation behavior and modelling of soil burial effect on degradation rate of PLA blended with starch and wood flour, *Colloids Surfaces B Biointerfaces*, vol. 159, pp. 800–808, Nov. 2017, doi: 10.1016/J.COLSURFB.2017.08.056.
- [25] Bergmann M., Mützel S., Primpke S., *et al.*, White and wonderful? Microplastics prevail in snow from the Alps to the Arctic, *Sci. Adv.*, vol. 5, no. 8, pp. 1–11, Aug. 2019, doi: 10.1126/sciadv.aax1157.
- [26] Vethaak A. D. and Legler J., Microplastics and human health, Science (80-.)., vol. 371, no. 6530, pp. 672–674, Feb. 2021, doi: 10.1126/science.abe5041.
- [27] Welden N. A. and Lusher A., Microplastics, in *Plastic Waste and Recycling*, vol. 32, no. 3, Elsevier, 2020, pp. 223–249.
- [28] Allen S., Allen D., Phoenix V. R., et al., Atmospheric transport and deposition of microplastics in a remote mountain catchment, Nat. Geosci., vol. 12, no. 5, pp. 339– 344, May 2019, doi: 10.1038/s41561-019-0335-5.
- [29] Ng E. L., Huerta Lwanga E., Eldridge S. M., et al., An overview of microplastic and nanoplastic pollution in agroecosystems, Sci. Total Environ., vol. 627, pp. 1377–1388, 2018, doi: 10.1016/j.scitotenv.2018.01.341.
- [30] de Souza Machado A. A., Kloas W., Zarfl C., *et al.*, Microplastics as an emerging threat to terrestrial ecosystems, *Glob. Chang. Biol.*, vol. 24, no. 4, pp. 1405–1416, 2018, doi: 10.1111/gcb.14020.
- [31] Lv S., Zhang Y., Gu J., and Tan H., Soil burial-induced chemical and thermal changes in starch/poly (lactic acid) composites, *Int. J. Biol. Macromol.*, vol. 113, pp. 338–344, 2018, doi: 10.1016/j.ijbiomac.2018.02.139.
- [32] Zamir S. S., Fathi B., Ajji A., et al., Biodegradation of modified starch/poly lactic acid nanocomposite in soil, Polym. Degrad. Stab., vol. 199, p. 109902, 2022, doi: 10.1016/j.polymdegradstab.2022.109902.
- [33] Shogren R. L., Doane W. M., Garlotta D., et al., Biodegradation of starch/polylactic acid/poly(hydroxyester-ether) composite bars in soil, *Polym. Degrad. Stab.*, vol. 79, no. 3, pp. 405–411, Mar. 2003, doi: 10.1016/S0141-3910(02)00356-7.
- [34] Phetwarotai W., Potiyaraj P., and Aht-Ong D., Biodegradation of Polylactide and Gelatinized Starch Blend Films Under Controlled Soil Burial Conditions, J. Polym. Environ., vol. 21, no. 1, pp. 95–107, Mar. 2013, doi: 10.1007/S10924-012-0530-6/FIGURES/8.
- [35] Iovino R., Zullo R., Rao M. A., et al., Biodegradation of poly(lactic acid)/starch/coir biocomposites under controlled composting conditions, Polym. Degrad. Stab., vol. 93, pp. 147–157, 2008, doi: 10.1016/j.polymdegradstab.2007.10.011.
- [36] Jalalvandi E., Majid R. A., Ghanbari T., and Ilbeygi H., Effects of montmorillonite (MMT) on morphological, tensile, physical barrier properties and biodegradability of

polylactic acid/starch/ MMT nanocomposites, J. Thermoplast. Compos. Mater., vol. 28, no. 4, pp. 496–509, 2015, doi: 10.1177/0892705713486129.

- [37] Bher A., Auras R., and Schvezov C. E., Improving the toughening in poly(lactic acid)thermoplastic cassava starch reactive blends, J. Appl. Polym. Sci., 2017, doi: 10.1002/app.46140.
- [38] Bher A., Uysal Unalan I., Auras R., *et al.*, Toughening of Poly(lactic acid) and Thermoplastic Cassava Starch Reactive Blends Using Graphene Nanoplatelets, *Polymers (Basel).*, vol. 10, no. 1, p. 95, Jan. 2018, doi: 10.3390/polym10010095.
- [39] ISO 14855-1: 2012, Determination of the ultimate aerobic biodegradability of plastic materials under controlled composting conditions Method by analysis of evolved carbon dioxide Part 1: General method, 2012.
- [40] Castro E., Design And Construction Of A Medium-Scale Automated Direct Measurement Respirometric System To Assess Aerobic Biodegradation Of Polymers, 2013.
- [41] Kijchavengkul T., Auras R., Rubino M., et al., Development of an automatic laboratoryscale respirometric system to measure polymer biodegradability, *Polym. Test.*, vol. 25, no. 8, pp. 1006–1016, 2006, doi: 10.1016/j.polymertesting.2006.06.008.
- [42] Bellia G., Tosin M., Floridi G., and Degli-Innocenti F., Activated vermiculite, a solid bed for testing biodegradability under composting conditions, *Polym. Degrad. Stab.*, vol. 66, no. 1, pp. 65–79, 1999, doi: 10.1016/S0141-3910(99)00053-1.
- [43] ASTM International, ASTM Standard D4754-18, Standard Test Method for Two-Sided Liquid Extraction of Plastic Materials Using FDA, 2018. doi: 10.1520/D4754-98R03.2.
- [44] Mayekar P. C., Castro-Aguirre E., Auras R., et al., Effect of Nano-Clay and Surfactant on the Biodegradation of Poly(Lactic Acid) Films, Polym. 2020, Vol. 12, Page 311, vol. 12, no. 2, p. 311, Feb. 2020, doi: 10.3390/POLYM12020311.
- [45] Detyothin S., Selke S. E. M., Narayan R., et al., Effects of molecular weight and grafted maleic anhydride of functionalized polylactic acid used in reactive compatibilized binary and ternary blends of polylactic acid and thermoplastic cassava starch, J. Appl. Polym. Sci., vol. 132, no. 28, pp. 1–15, 2015, doi: 10.1002/app.42230.
- [46] Wilson D. B., Microbial diversity of cellulose hydrolysis, Curr. Opin. Microbiol., vol. 14, pp. 259–263, 2011, doi: 10.1016/j.mib.2011.04.004.
- [47] Lakhundi S., Siddiqui R., and Khan N. A., Cellulose degradation: A therapeutic strategy in the improved treatment of Acanthamoeba infections, *Parasites and Vectors*, vol. 8, no. 1, pp. 1–16, Jan. 2015, doi: 10.1186/S13071-015-0642-7/FIGURES/8.
- [48] Leppänen I., Vikman M., Harlin A., and Orelma H., Enzymatic Degradation and Pilot-Scale Composting of Cellulose-Based Films with Different Chemical Structures, J. Polym. Environ., vol. 28, pp. 458–470, 1234, doi: 10.1007/s10924-019-01621-w.

- [49] Béguin P. and Aubert J.-P., The biological degradation of cellulose, *FEMS Microbiol. Rev.*, vol. 13, no. 1, pp. 25–58, Jan. 1994, doi: 10.1111/j.1574-6976.1994.tb00033.x.
- [50] Sukumaran R. K., Singhania R. R., and Pandey A., Microbial cellulases Production, applications and challenges, *J. Sci. Ind. Res.*, vol. 64, pp. 832–844, 2005.
- [51] Doi R. H., Cellulases of Mesophilic Microorganisms Cellulosome and Noncellulosome Producers, Ann. N. Y. Acad. Sci., vol. 1125, no. 1, pp. 267–279, 2008, doi: 10.1196/annals.1419.002.
- [52] Eida M. F., Nagaoka T., Wasaki J., and Kouno K., Isolation and Characterization of Cellulose-decomposing Bacteria Inhabiting Sawdust and Coffee Residue Composts, *Microbes Environ.*, vol. 27, no. 3, pp. 226–233, 2012, doi: 10.1264/JSME2.ME11299.
- [53] Maran J. P., Sivakumar V., Thirugnanasambandham K., and Sridhar R., Degradation behavior of biocomposites based on cassava starch buried under indoor soil conditions, *Carbohydr. Polym.*, vol. 101, no. 1, pp. 20–28, 2014, doi: 10.1016/j.carbpol.2013.08.080.
- [54] Phetwarotai W., Potiyaraj P., Aht-Ong D., and Library W. O., Characteristics of Biodegradable Polylactide/Gelatinized Starch Films: Effects of Starch, Plasticizer, and Compatibilizer, J Appl Polym Sci, vol. 126, pp. 162–172, 2012, doi: 10.1002/app.36736.
- [55] Hulleman S. H. D., Janssen F. H. P., and Feil H., The role of water during plasticization of native starches, *Polymer (Guildf)*, vol. 39, no. 10, pp. 2043–2048, May 1998, doi: 10.1016/S0032-3861(97)00301-7.
- [56] Rodrigues C. A., Tofanello A., Nantes I. L., and Rosa D. S., Biological Oxidative Mechanisms for Degradation of Poly(lactic acid) Blended with Thermoplastic Starch, ACS Sustain. Chem. Eng., vol. 3, no. 11, pp. 2756–2766, 2015, doi: 10.1021/acssuschemeng.5b00639.
- [57] Limsukon W., Auras R., and Selke S., Hydrolytic degradation and lifetime prediction of poly(lactic acid) modified with a multifunctional epoxy-based chain extender, *Polym. Test.*, vol. 80, no. September, pp. 106–108, Dec. 2019, doi: 10.1016/j.polymertesting.2019.106108.
- [58] Iñiguez-Franco F., Auras R., Ahmed J., et al., Control of hydrolytic degradation of Poly(lactic acid) by incorporation of chain extender: From bulk to surface erosion, Polym. Test., vol. 67, no. March, pp. 190–196, May 2018, doi: 10.1016/j.polymertesting.2018.02.028.
- [59] del Rosario Salazar-Sánchez M., Campo-Erazo S. D., Villada-Castillo H. S., and Solanilla-Duque J. F., Structural changes of cassava starch and polylactic acid films submitted to biodegradation process, *Int. J. Biol. Macromol.*, vol. 129, pp. 442–447, May 2019, doi: 10.1016/J.IJBIOMAC.2019.01.187.
- [60] Yu M., Zheng Y., and Tian J., Study on the biodegradability of modified starch/polylactic acid (PLA) composite materials, *RSC Adv.*, vol. 10, no. 44, pp. 26298– 26307, Jul. 2020, doi: 10.1039/D0RA00274G.

- [61] Šerá J., Stloukal P., Jančová P., et al., Accelerated Biodegradation of Agriculture Film Based on Aromatic–Aliphatic Copolyester in Soil under Mesophilic Conditions, J. Agric. Food Chem., vol. 64, no. 28, pp. 5653–5661, Jul. 2016, doi: 10.1021/acs.jafc.6b01786.
- [62] Iovino R., Zullo R., Rao M. A., et al., Biodegradation of poly(lactic acid)/starch/coir biocomposites under controlled composting conditions, Polym. Degrad. Stab., vol. 93, no. 1, pp. 147–157, 2007, doi: 10.1016/j.polymdegradstab.2007.10.011.
- [63] Bher A., Mayekar P. C., Auras R. A., and Schvezov C. E., Biodegradation of Biodegradable Polymers in Mesophilic Aerobic Environments, *Int. J. Mol. Sci.*, vol. 23, no. 20, 2022, doi: 10.3390/ijms232012165.
- [64] Tsuji H. and Ikada Y., Blends of crystalline and amorphous poly(lactide) .3. Hydrolysis of solution-cast blend films, J. Appl. Polym. Sci., vol. 63, pp. 855–863, 1997, doi: papers://590F92D9-0B76-4B88-8729-9AF064BE5AC8/Paper/p4495.
- [65] Iñiguez-Franco F., Auras R., Burgess G., et al., Concurrent solvent induced crystallization and hydrolytic degradation of PLA by water-ethanol solutions, *Polymer* (*Guildf*)., vol. 99, pp. 315–323, Sep. 2016, doi: 10.1016/j.polymer.2016.07.018.
- [66] Lv S., Zhang Y., Gu J., and Tan H., Physicochemical evolutions of starch/poly (lactic acid) composite biodegraded in real soil, *J. Environ. Manage.*, vol. 228, pp. 223–231, 2018, doi: 10.1016/j.jenvman.2018.09.033.
- [67] Yu M., Zheng Y., and Tian J., Study on the biodegradability of modified starch/polylactic acid (PLA) composite materials, *RSC Adv.*, vol. 10, no. 44, pp. 26298–26307, Jul. 2020, doi: 10.1039/D0RA00274G.
- [68] Palai B., Mohanty S., and Sanjay, Nayak K., A Comparison on Biodegradation Behaviour of Polylactic Acid (PLA) Based Blown Films by Incorporating Thermoplasticized Starch (TPS) and Poly (Butylene Succinate-co-Adipate) (PBSA) Biopolymer in Soil, J. Polym. Environ., vol. 29, pp. 2772–2788, 2021, doi: 10.1007/s10924-021-02055-z.
- [69] Nabar Y., Narayan R., and Schindler M., Twin-Screw Extrusion Production and Characterization of Starch Foam Products for Use in Cushioning and Insulation Applications, *Polym. Eng. Sci.*, vol. 46, no. 4, pp. 438–451, 2006, doi: 10.1002/pen.20292.
- [70] Bher A., Auras R., and Schvezov C. E., Improving the toughening in poly(lactic acid)thermoplastic cassava starch reactive blends, J. Appl. Polym. Sci., vol. 135, no. 15, p. 46140, Apr. 2018, doi: 10.1002/APP.46140.
- [71] Bher A., Uysal Unalan I., Auras R., *et al.*, Toughening of Poly(lactic acid) and Thermoplastic Cassava Starch Reactive Blends Using Graphene Nanoplatelets, *Polymers (Basel).*, vol. 10, no. 1, p. 95, 2018, doi: 10.3390/polym10010095.
- [72] Schuster J. M., Schvezov C. E., and Rosenberger M. R., Construction and calibration of a goniometer to measure contact angles and calculate the surface free energy in

solids with uncertainty analysis, *Int. J. Adhes. Adhes.*, vol. 87, pp. 205–215, Dec. 2018, doi: 10.1016/J.IJADHADH.2018.10.012.

- [73] Limsukon W., Auras R., and Selke S., Hydrolytic degradation and lifetime prediction of poly(lactic acid) modified with a multifunctional epoxy-based chain extender, *Polym. Test.*, vol. 80, Dec. 2019, doi: 10.1016/J.POLYMERTESTING.2019.106108.
- [74] Xu D., Auras R. A., Sonchaeng U., et al., The effect of alcoholic solutions on the thermomechanical properties of immersed poly(lactic acid) films, J. Appl. Polym. Sci., no. August, pp. 1–10, Dec. 2022, doi: 10.1002/app.53489.
- [75] Lyu S., Schley J., Loy B., *et al.*, Kinetics and Time Temperature Equivalence of Polymer Degradation, pp. 2301–2310, 2007.

APPENDIX 3A: MATERIAL PROCESSING

Maleic anhydride (MA) and dicumyl peroxide (DCP) along with PLA was used to produce PLA-g-MA masterbatch. Similarly, glycerol and starch were processed together to produce a thermoplastic starch (TPS) masterbatch. PLA pellets were processed to produce a PLA masterbatch. The different zones in the co-rotating twin screw extruder namely the feeding-melting zone, large kneading/mixing zone, conveying zone, short kneading zone, and conveying zone provide the necessary shear in the form of mechanical and thermal energy to produce TPS [69]. **Figures A3.1** and **A3.2** show the processing of PLA, PLA-g-MA, and TPS masterbatches and PLA-g-TPS films respectively.



Figure A3.1 Processing of PLA-*g*-MA, TPS, and PLA masterbatches in a twin screw extruder and temperature profile from feeder to the die of the twin screw extruder.

The masterbatches produced from the twin screw extruder were blended in specific proportion (56% PLA, 14% PLA-g-MA, and 30% TPS) in a cast film extruder to produce PLA-g-TPS films.



Figure A3.2 Processing of PLA-*g*-MA, TPS and PLA masterbatches into PLA-*g*-TPS film by cast film extrusion.

APPENDIX 3B: PHYSICOCHEMICAL CHARACTERISTICS

Some compost was collected and sent to the Soil and Plant Nutrient Laboratory at Michigan State University (East Lansing, MI, USA) to evaluate its physicochemical parameters (dry solids, volatile solids, and C/N ratio) as previously described elsewhere [12]. The physicochemical parameters are reported below in **Table A3.1**.

Parameter	Compost
Dry solids, %	42.5
Volatile solids, %	41.7
pH	8.0
C/N ratio	10.1
Carbon, %	24.2
Nitrogen, %	2.42
Phosphorus, %	1.21
Potassium, %	3.15
Calcium, %	5.07
Magnesium, %	2.82
Sodium, %	0.58
Sulfur, %	0.58
Iron, ppm	9878
Zinc, ppm	480
Manganese, ppm	413
Copper, ppm	107
Boron, ppm	41
Aluminum, ppm	6751

Table A3.1 Physicochemical parameters and total nutrient analysis of compost used in the biodegradation test.

APPENDIX 3C: MORPHOLOGICAL CHARACTERIZATION OF PLA-g-TPS REACTIVE BLEND

The PLA-TPS physical blend and PLA-g-TPS reactive blend were characterized by SEM to observe the morphological changes [70]. **Figure A3.3 a** shows the non-uniform dispersion of the TPS phase in the PLA matrix for PLA-TPS physical blend while **Figure A3.3 b** shows good compatibilization and distribution of TPS domain in the PLA matrix.



Figure A3.3 SEM images for (a) PLA-TPS physical blend and (b) PLA-g-TPS reactive blend.

APPENDIX 3D: ROUGHNESS AND CONTACT ANGLE MEASUREMENTS

The roughness of the PLA film and PLA-g-TPS films in **Figure A3.4 a** and **b** respectively was conducted using a CypherTM atomic force microscope (Oxford Instruments Asylum Research, Inc., Santa Barbara, CA, USA) in the contact mode. Roughness parameters, calculated as the root mean square (Rq) and average roughness (Ra), were determined for each type of film and were calculated from the Htr mode image. Images were obtained in the Dfr mode [71]. The film area for the determination of roughness was 900 µm2.



Figure A3.4 Surface roughness of (a) PLA and (b) PLA-g-TPS films as measured by atomic force microscopy. Roughness parameters, calculated as the root mean square (Rq) and average roughness (Ra), were determined for each type of film and were calculated from the Htr mode image. Within columns, values followed by a different letter are significantly different at $p \le 0.05$ (Tukey's test).

The contact angle of the films was measured by the sessile drop technique using an in-house built goniometer equipped with a diffuse light source and a digital camera. Film samples of 7 cm length and 2 cm wide were attached to microscope slides. Next, a drop of HPLC-grade water (3 µL) was deposited on each film surface, and the contact angle was determined by using the tangent method as seen in **Figure A3.5 a** and **b** [72].



Figure A3.5 Water contact angle measurement on (a) PLA and (b) PLA-g-TPS films.

APPENDIX 3E: APPLICATION OF THE TIME-TEMPERATURE SUPERPOSITION PRINCIPLE FOR PREDICTING HYDROLYTIC DEGRADATION IN MESOPHILIC CONDITIONS

In a previous work, Limsukon et al. [73] studied the hydrolytic degradation of PLA film at temperatures from 40 to 95 °C and presented the lifetime prediction methods at the lower temperature of interest using the time-temperature superposition principle. The Williams-Landel-Ferry (WLF) equation was used to fit the hydrolysis of PLA film over the wide range of experiment temperatures crossing the T_g . The WLF equation is given as follows:

$$\log \alpha_{\rm T} = -\frac{C_1(T - T_{\rm i})}{C_2 + (T - T_{\rm i})}$$
(A3.1)

 C_1 and C_2 can be expressed as:

$$C_1 = \frac{E/R}{2.303 (T_i - T_s)}$$
 and $C_2 = (T_i - T_s)$ (A3.2)

where T and T_i are experimental temperature and the temperature of interest. T_s is the temperature at which the conformation entropy induced by segmental motion approaches zero (is about 50 K below the T_g of PLA immersed in water which is 51.6 °C or 324.75K [74]). E is the pseudo activation energy over a temperature range from below to above T_g. R is the gas constant (8.314 J/mol·K). a_T is the factor in shifting the data obtained at temperature T to overlap one at T_i .

To predict the hydrolysis of PLA in the mesophilic condition, a_r value was calculated by substituting C₁ and C₂ into S1 to shift the experimental data curves of 58 °C and construct the master curve at 37 °C using E of the hydrolysis of neat PLA film measured within the temperature range crossing the T_g reported as 2720 J/mol ; a_r can be determined as 0.03205. Finally, the master curve of the hydrolysis at 37 °C can be constructed by multiplying the hydrolysis time range by $1/a_r$ and plotting as a function of the prediction line of hydrolysis at 58 °C. A detailed description of the calculation can be found elsewhere [73,75].

APPENDIX 3F: CRYSTALLINITY MEASUREMENTS

The films retrieved from the sampling bioreactors were analyzed for the evolution of crystallinity throughout the test for PLA and PLA-g-TPS films at both 37 °C and 58 °C, respectively. The crystallinity (X_c) of the samples was tracked using a DSC Q100 (TA Instruments, New Castle, DE, USA). Samples weighing between 5 and 10 mg were sealed in aluminum pans and subjected to a cycle ranging from -5 °C to 210 °C. The samples were cooled to -5 °C and then heated to 210 °C at a ramp rate of 10 °C/min under a nitrogen atmosphere where the purge flow was maintained at 70 mL/min. The data obtained was analyzed using the software Thermal Universal Analysis 2000, V4.5 (TA Instruments). The % X_c was calculated using the equation S3:

$$X_{c}(\%) = \frac{(\Delta H_{m} - \Delta H_{c})}{\Delta H_{m}^{\circ} \left(1 - \frac{\% w t_{filler}}{100}\right)} \times 100$$
(A3.3)

where ΔH_m is the heat of fusion, ΔH_c is the enthalpy of cold crystallization, ΔH_m° is the enthalpy fusion for 100% pure crystalline PLA (93 J/g), and wt_{filler} is the weight fraction of TPS (30%).

Table A	43.2 Crys	stallinity evo	lution f	for PLA	and PLA-g-	TPS films	at 58°C	and 37°	С
---------	------------------	----------------	----------	---------	------------	-----------	---------	---------	---

Material	$58 \pm$	2 °C	37 ± 2 °C		
	Day 0	Day 15	Day 0	Day 90	
PLA	3.1 ± 0.6	35.5 ± 0.2	28.2 ± 3.5	32.3 ± 1.8	
PLA-g-TPS	7.0 ± 0.6	39.8*	7.0 ± 0.6	18.5 ± 2.2	

* Only one measurement is noted for the sample due to the fragile state of the film.



Figure A3.6 DSC thermograms showing the evolution of crystallinity of a) PLA and b) PLAg-TPS films in compost at 37 °C.
CHAPTER 4: ACCELERATING BIODEGRADATION: ENHANCING POLY(LACTIC ACID) BREAKDOWN AT MESOPHILIC ENVIRONMENTAL CONDITIONS WITH BIOSTIMULANTS

4.1 Abstract

Poly(lactic acid) – PLA – has garnered interest due to its low environmental footprint and ability to replace conventional polymers and be disposed of in industrial composting environments. Although PLA is compostable when subjected to a suitable set of conditions (i.e., aerobic thermophilic conditions for an extended period), its broader acceptance in industrial composting facilities has been affected adversely due to longer degradation timeframes than the readily biodegradable organic waste fraction. PLA must be fully exposed to thermophilic conditions for prolonged periods to biodegrade, which has restricted its adoption and hindered its acceptance in industrial composting facilities, also negating its home composting potential. Thus, enhancing PLA biodegradation is crucial to expand its acceptance. PLA's biodegradability was investigated in a compost matrix under mesophilic conditions at 37°C for 180 days by biostimulating the compost environment with skim milk, gelatin, and ethyl lactate to enhance the different stages of PLA biodegradation. The evolved CO₂, number average molecular weight, and crystallinity evolution were tracked. To achieve Mn ≤ 10 kDa for PLA, biodegradation rate was accelerated by 15% by adding skim milk, 25% by adding gelatin, and 22% by adding ethyl lactate. This work shows potential techniques to help biodegrade PLA in compost by adding biostimulants.

4.2 Introduction

Poly(lactic acid) (PLA) is a well-known biodegradable thermoplastic polymer and has garnered a lot of interest due to its ability to exhibit reasonable properties as compared to fossil-based conventional polymers [1]. In addition, PLA has turned out to be a key player leading the global bioplastic segment as per the market research conducted by nova-Institute which estimates an increase in bioplastics production from 2.18 million tonnes in 2023 to 7.43 million tonnes in 2028 [2]. Due to its biodegradable nature instead of contributing to plastic waste in landfills, PLA can be rerouted to composting facilities [3]. PLA is known to undergo chemical hydrolytic degradation during composting, significantly reducing its molecular weight [4]. This reduction in molecular weight precedes the assimilation by microorganisms and final mineralization to CO₂. However, it is difficult to degrade PLA at lower temperatures (e.g., backyard or home composting environments – mostly mesophilic conditions) because of its dependence on high temperatures ($\geq 60^{\circ}$ C) to undergo chemical hydrolysis. Additionally, PLA degradation in natural environments (usually associated with lower temperatures) is slow due to the sparse distribution of PLA-degrading microbes compared with other more aggressive thermophilic degradable environments such as industrial composting [5,6]. A detailed discussion about the compostability and biodegradation of PLA can be found elsewhere [7,8].

Biostimulation and bioaugmentation techniques are included under the bioremediation domain and are used to eradicate hazardous pollutants/waste materials that may be toxic to the environment. These techniques use enhanced settings or biological workers/microorganisms to eliminate pollutants. They also are designed to make up for the lack of factors that can speed up the removal process of pollutants in an eco-friendly approach. Biostimulation is an approach that addresses limiting factors such as nutrients (in the form of enzyme inducers), [9,10] electron donor or acceptor compounds, [11,12] nitrogensupplying compounds, [13,14] and compounds that can activate the biochemical processes by providing necessary resources or chemicals to stimulate the environment [15,16]. The biostimulation technique has been used extensively to revitalize the native microbial communities to boost the degradation of heavy and toxic metals,[11,17,18] diesel oil,[19,20] chlorinated hydrocarbons,[21] and petroleum hydrocarbons,[22,23] since these compounds are hard to biodegrade by native microorganism populations.

PLA biodegradation proceeds much faster at temperatures higher than its glass transition temperature $(T_g) \sim 58^{\circ}$ C, commonly reached in industrial composting conditions. Chemical hydrolysis, the dominant mechanism responsible for the significant reduction in molecular weight of PLA, takes place quickly at higher temperatures [24]. The resulting low molecular weight PLA is then available for further assimilation by the microorganisms.

Under thermophilic conditions (i.e., 45–60°C), the chemical hydrolysis of PLA proceeds quickly, efficiently, and in a suitable time frame compared to mesophilic conditions (i.e., 20–45°C). Accelerating PLA degradation, particularly hydrolytic degradation in compost, is exceptionally challenging because of the limitation of the solid complex matrix to be at high temperatures to activate the biodegradation process. Several research studies have focused on including metal compounds in PLA to catalyze its degradation [25–29]. But most of these metal compounds are included in PLA as blends/nanocomposites at lower concentrations, significantly altering physical properties such as number and weight average molecular weight, melting temperature, and T_g [30–33]. These final properties do not accurately represent the post-consumer PLA discarded packages ending up in compost, soil, or landfill scenarios.

Apart from the chemical hydrolysis, the biodegradation of PLA also entails enzymatic degradation, the release of extracellular enzymes by the microbes, which simultaneously break down PLA into low molecular weight compounds. The extracellular depolymerases released by the microorganisms in the case of PLA degradation belong mainly to the hydrolase class of enzymes, primarily proteases, and some also belong to lipase and cutinase [34–36]. The serine proteases produced by microbes disintegrate the ester bonds of PLA [37–

39]. Several studies have used the biostimulation technique to expedite the biodegradation of bio-based polymers. But most of these studies were conducted with PLA as the sole carbon source, in liquid media with specific microbial strains, and usually at higher temperatures [40–43]. So, there is a gap between these experiments conducted in highly controlled incubated culture media and the real-world unchecked environment in solid media such as soil, industrial composting, and home composting. The biological activity in such natural environments is highly complex due to the diversity of microorganisms competing or working together to consume organic resources.

This study aimed to evaluate the addition of different biostimulant compounds, which can be added externally in the compost media (solid) to accelerate the biodegradation of PLA in mesophilic conditions. The compounds selected as biostimulants were screened through multiple factors and were expected to meet various criteria: not toxic towards the microorganisms present, not antibacterial, able to disintegrate, consumable by the microorganisms, biodegradable in a suitable timeframe, cost-effective, and readily available or accessible.

Hydrolases (EC 3) are the hydrolytic enzymes responsible for degrading hydrolyzable polymers such as esters, carbonate, and amide groups. The ester bonds in the backbone of polyesters act as hydrolyzable linkages and are acted upon by esterases (EC 3.1) and proteases (EC 3.4), which are released by the PLA-degrading microbes. To induce the activity of these enzymes by microbes, skim milk and gelatin have been reported as suitable compounds [44–47]. So, skim milk and gelatin can incite proteolytic activity for enzymatic degradation, resulting in the cleavage of PLA polymer chains.

To stimulate the lactate-utilizing microbial species in compost, ethyl lactate can be used. Ethyl lactate belongs to the lactate esters family. It has been used as a food source for dechlorinators to remediate heavy metals in soil under anaerobic conditions [48–50]. It is considered a green solvent with numerous advantages such as favorable low toxicity and low cost, and is readily available. It is environmentally benign, highly biodegradable, and has a strong solvency power [51,52]. Ethyl lactate undergoes hydrolysis to produce ethanol and lactate, where both can act as electron donor compounds for reductive degradation. Lactate has been shown to act as an electron donor compound in the case of anaerobic degradation [53,54]. Hydrolysis of ethyl lactate is a slow process ensuring a constant, long-term supply of hydrogen as the electron donor for the microbial redox process. The use of ethyl lactate as a biostimulant for polymer degradation has not yet been reported.

Thus, this study aimed to investigate the effectiveness of biostimulating the compost environment with compounds that may enhance enzymatic degradation (skim milk and gelatin), and with an electron donor compound (ethyl lactate) to improve the aerobic biodegradation of PLA in compost at mesophilic conditions under which it is more difficult to degrade PLA. The difference in the CO_2 evolution and changes in the number average molecular weight (M_n) and crystallinity (X_c) of PLA degradation with and without biostimulants was tracked to account for the activity of biostimulants in compost.

4.3 Materials and methods

4.3.1 Materials

Commercial PLA resin, IngeoTM 2003D, with l-lactic acid content of 96%, with weight average molecular weight (M_w of $2.23 \pm 0.04 \ge 10^5$ Da) and number average molecular weight (M_n of $1.14 \pm 0.07 \ge 10^5$ Da) was obtained from NatureWorks LLC (Minnetonka, MN, USA). Skim milk powder was procured from the local grocery store. Gelatin of the brand McCormick & Co. (Hunt Valley, MD, USA) was purchased online. Ethyl lactate was procured from Sigma Aldrich (St. Louis, MO, USA). Cellulose of ~20 µm particle size was procured from from Sigma-Aldrich (Milwaukee, WI, USA).

4.3.2 Characterization of PLA and the biostimulants

The carbon, hydrogen, and nitrogen contents of PLA resin, skim milk, gelatin, and ethyl lactate were determined using a CHNS/O Elemental Analyzer, PerkinElmer 2400 Series II (Shelton, CT, USA), and are presented in **Table 4.1**. The methodology was provided in our previous work [55].

Table 4.1 Percent of carbon, hydrogen, and nitrogen content by weight of cellulose and film samples.

Material	% Carbon ^a	% Hydrogenª	% Nitrogen ^a		
PLA	49.72 ± 0.19	5.72 ± 0.04	0.11 ± 0.07		
Skim milk	41.28 ± 0.13	6.33 ± 0.03	5.75 ± 0.06		
Gelatin	44.80 ± 0.30	7.00 ± 0.10	16.4 ± 0.10		
Ethyl lactate	31.93 ± 0.98	5.46 ± 0.34	0.05 ± 0.03		
a: Percentage by weight.					

4.3.3 Biodegradation in compost

The biodegradation of PLA and the effect of biostimulant addition on the degradation of PLA were evaluated under aerobic mesophilic conditions using an in-house direct measurement respirometric (DMR) system for analysis of evolved CO₂ under simulated composting conditions. The test was adapted to ASTM D53338-15 and ISO 14855-12, as currently there are no standards in place for evaluating the aerobic biodegradation of plastic materials under controlled composting conditions at mesophilic temperatures [56–58]. The system was equipped with a non-dispersive infrared gas analyzer (NDIR) Li-COR[®] LI-820 (Lincoln, NE, USA), which measures the CO₂ concentration. The temperature and relative humidity (RH) of the chamber was maintained at $37 \pm 2^{\circ}$ C and $50\% \pm 5\%$ RH, and the airflow rate was controlled at 40 ± 2 cm³/min. Detailed information about the DMR chamber can be found elsewhere [24].

Manure compost was obtained from the Michigan State University (MSU) Composting Facility (East Lansing, MI, USA). The compost was sieved using a 10-mm screen to remove any large debris or chunks that might be present, and post screening was conditioned at 37 \pm 2°C. Deionized water was added to adjust the moisture content to 50%. Vermiculite, a hydrous phyllosilicate inorganic material of premium grade, was obtained from Sun Gro Horticulture Distribution Inc. (Bellevue, WA, USA). The vermiculite was mixed 1:4 with compost (dry weight basis). Samples of the compost were analyzed by the Soil and Plant Laboratory at MSU (East Lansing, MI, USA) for physicochemical parameters such as dry solids, volatile solids, pH, and C/N ratio, as presented in **Table 4.2**.

The bioreactors were packed with 400 g of compost, and then 8 g of PLA sample and 8 g of the selected biostimulant were added, and all the samples were tested in triplicate.

Parameter	Compost	Parameter	Compost
Dry solids, %	42.5	Magnesium, %	2.82
Volatile solids, %	41.7	Sodium, %	0.58
pH	8.0	Sulfur, %	0.58
C/N ratio	10.1	Iron, ppm	9878
Carbon, %	24.2	Zinc, ppm	480
Nitrogen, %	2.42	Manganese, ppm	413
Phosphorus, %	1.21	Copper, ppm	107
Potassium, %	3.15	Boron, ppm	41
Calcium, %	5.07	Aluminum, ppm	6751

Table 4.2 Physicochemical parameters and total nutrient analysis of compost used in the biodegradation test.

4.3.4 Biostimulation

Ethyl lactate, skim milk, or gelatin were added at 8 g to separate bioreactors and mixed thoroughly with the compost matrix containing the 8 g of PLA pellets to ensure uniform distribution. The bioreactor used to study the biodegradation of PLA samples is shown in **Figure 4.1**.



Figure 4.1 Bioreactor used to study the biodegradation of samples.

Moisture content was adjusted, and the optimal conditions were maintained by injecting the required amount of deionized water weekly into all bioreactors. CO_2 -free air (<30 ppm) was supplied to each bioreactor and the CO_2 evolved was measured for a specific time. The measurement system was purged after every measurement to get rid of any traces of CO_2 from the previous bioreactor and to ensure that the baseline was achieved. Mineralization, which is defined as the total amount of carbon converted to CO_2 molecules, was calculated according to equation (4.1)

$$Mineralization \% = \frac{(CO_2)_t - (CO_2)_b}{M_t \, x \, C_t \, x \, \frac{44}{12}} \, x \, 100 \tag{4.1}$$

The numerator expresses the CO_2 evolved from the sample, which is calculated by subtracting the average cumulative mass of CO_2 evolved from the blank $(CO_2)_b$ from the average cumulative mass of CO_2 evolved from the bioreactor containing the sample $(CO_2)_t$. The denominator represents the theoretical amount of CO_2 that can be produced by the sample and is calculated as a product of the total mass of the sample in the bioreactor (M_t) , and the total carbon content of the sample as derived from CHN analysis (C_t) . 44 is the molecular weight of CO_2 , and 12 is the atomic weight of carbon.

4.3.5 Hydrolysis experiment

A hydrolysis test method adapted from ASTM D4754-18 [59] was run for PLA films at 58 ± 2 °C and 37 ± 2 °C to understand the hydrolytic degradation. The hydrolysis cell consisted of a stainless-steel wire, glass beads, and a glass vial with cap. PLA films were cut into small discs of 2-cm diameter, and ten such discs were strung into a stainless-steel wire and separated by glass beads. The vial was filled with 35 mL of HPLC-grade water (J.T. Baker, Center Valley, PA, USA). The water was preconditioned, and the hydrolysis cell was stored at the same temperature. Triplicates of the PLA films were retrieved at predetermined time intervals and dried before running size exclusion chromatography to assess the M_n reduction.

4.3.6 Size exclusion chromatography

A size exclusion chromatography (SEC) instrument was used to quantify the weight average (M_w), number average (M_n) molecular weight and molecular weight distribution (MWD) of the PLA treated with and without biostimulants. SEC system from Waters Corp. (Milford, MA, USA) was furnished with an autosampler (Waters® 717), a refractive index detector (Waters® 2414), an isocratic pump (Waters® 1515), and a series of Styragel® columns (HR4, HR3, HR2). The flow rate and temperature was maintained at 1mL/min and 35°C respectively. The Mark-Houwink constants of K = 0.000174 dL/g and α = 0.736 were used for PLA dissolved in tetrahydrofuran solvent. Waters BreezeTM2 software was used to examine the data obtained.

4.3.7 Differential scanning calorimetry

A differential scanning calorimeter (DSC), model Q100 (TA Instruments, New Castle, DE, USA), was used to determine the glass transition temperature (T_s) , melting temperature (T_m) , crystallization temperature (T_c) , and crystallinity (X_c) for the PLA samples retrieved from biostimulated compost. Samples weighing between 5–10 mg were sealed in aluminum pans and subjected to a heating cycle to understand the evolution of crystallinity. The samples were cooled down to -5° C, and then heated to 210°C at a ramp rate of 10°C/min. The cooling was achieved using a nitrogen cooling system that maintained the purge flow rate at 70 mL/min. Thermograms were analyzed using the Thermal Universal Analysis 2000 software, V4.5 (TA Instruments). The degree of crystallinity was calculated using equation (4.2)

$$\chi_c \% = \frac{\Delta H_m - \Delta H_c}{\Delta H_m^0} x \ 100 \tag{4.2}$$

where ΔH_m is the heat of fusion, ΔH_c is cold crystallization enthalpy, and ΔH_m^0 is the heat of fusion for 100% crystalline pure PLA (93 J/g).

4.3.8 Statistical analysis

All the statistical analyses was conducted using MINITABTM 18 software (Minitab Inc., State College Park, PA, USA). One-way ANOVA and Tukey's test were used to evaluate statistical significance at p < 0.05. All the values are reported as mean ± standard deviation.

4.4 Results and discussion

The biodegradation of PLA in biostimulated compost with skim milk, gelatin, or ethyl lactate under mesophilic conditions (37°C) was evaluated for 180 days to understand the effect of these compounds on stimulating home composting operations and understanding the

potential effects in industrial composting. To decouple the effects of these compounds, the CO_2 evolution, M_n , and X_c of the samples were tracked.

4.4.1 Effect of skim milk on cellulose and PLA degradation

Figure 4.2a and **b** present the CO₂ evolution and % mineralization of the blank (compost only), cellulose, skim milk, PLA, cellulose+skim milk (Cell+skm), and PLA+skim milk (PLA+skm) in compost at 37°C. Cellulose evolved around 36.8 g of CO₂ and reached a mineralization of 87.7%, whereas skim milk showed around 39.2 g of CO₂ evolution and attained over 100% of its carbon conversion over 180 days. The priming effect is observed in the case of skim milk, due to over-degradation of the indigenous carbon present in the compost. Since skim milk and cellulose are readily biodegradable and can be easily utilized as a carbon source by the microorganisms present in the compost, no lag phase was observed. Skim milk was added to the compost to induce the protease activity of the microbes present, since serine protease (3.4.21.112), belonging to the peptidases, is in the class of extracellular enzymes able to hydrolyze the peptide bonds linking to the amino acids in the protein structure.

Skim milk comprises lactose, casein, and whey protein, making it a good precursor for the protease enzymatic activity by the microorganisms in the compost. Since cellulose and skim milk constitute excellent carbon sources when both are present in the compost, they are expected to show a much higher CO₂ evolution. However, the CO₂ evolution of the bioreactor containing both cellulose and skim milk was around 49.1 g, which is close to the CO₂ evolution of cellulose and skim milk separately and shows 96.6% biodegradation.

To better understand the interaction and isolate the degradation behavior of cellulose in the presence of skim milk (Cell_{skm}), we estimated the mineralization value wherein the background signal from the bioreactor containing skim milk is subtracted. As seen in **Figure 4.2b**, the mineralization % depicting the degradation behavior of cellulose reaches a maximum value of 95.9%. The value is higher than for cellulose alone, as skim milk and cellulose are easily accessible to microorganisms.

PLA pellets in compost showed around 22.5 g of CO₂ evolution, whereas the blank produced around 26.1 g of CO₂, implying that no carbon from PLA was degraded. The negative mineralization values indicate that the blank bioreactors produced more CO₂ than the bioreactors containing PLA samples. In a blank bioreactor, the microbes are easily able to assimilate the organic matter in the compost, whereas the presence of PLA in a bioreactor reduces the working efficiency, which in turns shows diminished CO₂ production. PLA offers an initial physical hydrophobic barrier to water, making it difficult for the microorganisms to utilize it as a carbon source and establish a macro colony able to break down PLA. We did not see any mineralization in PLA due to the low temperature of 37°C, which is opposite to the degradation of PLA in thermophilic temperatures [1,24].



Figure 4.2 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, skim milk, cellulose + skim milk (Cell+skm), PLA + skim milk (PLA+skm) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated by skim milk. The experimental data was fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$ Tukey-Kramer Test). (d) and (e) depict DSC thermograms for PLA and PLA in compost biostimulated by skim milk, respectively. (f) and (g) show the MWD of PLA in compost and compost biostimulated with skim milk, respectively.

Chemical hydrolysis is the initial and primary driving mechanism in the biodegradation of PLA. It dramatically reduces molecular weight and aids in PLA assimilation by microorganisms [60–62]. The T_g of PLA is much closer to the industrial

composting conditions, which reduces the rigidity of its polymer chains, creating more free volume and rearranging the polymer chains for easy diffusion of water into the matrix.[63,64] The polymer hydrophilicity is also elevated at higher temperatures, further promoting microbial attachment besides the chemical hydrolysis [65,66]. Apart from this, the initial M_n of PLA plays a crucial role in the biodegradation of PLA [67]. The bioreactor containing PLA and skim milk (PLA+skm) shows CO_2 evolution of 42.1 g and maximum mineralization of 60.1% by the end of the test duration. This result indicates improved mineralization compared to PLA alone, where no mineralization was observed. To isolate the degradation behavior of PLA in the presence of skim milk, mineralization of PLA_{skm} (subtracting blank + skim milk) was calculated, as described earlier, and is shown in Figure 4.2b. We observed positive mineralization, indicating PLA's enzymatic degradation due to the presence of skim milk. To further corroborate the biostimulation activity of skim milk, samples of PLA were retrieved from the bioreactor at specific intervals, and their $M_{\rm n}$ was evaluated. The reduction in M_n of PLA was tracked until the end of the test, as seen in **Figure 4.2c**. For the skim milk treatment, the kinetic reduction rate of biostimulated PLA was higher than for PLA. There was a significant difference in the degradation rates for PLA alone and PLA treated with skim milk, as indicated by the k values shown in **Figure 4.2c** inset (PLA $k = 0.0045 \pm 0.0001$ and PLA+skim milk $k = 0.0053 \pm 0.0003$). This increase in k can be interpreted as a final reduction of around 75 days when PLA biostimulated with skim milk reaches $M_n \leq 10$ kDa at 420 days, where microorganism assimilation of PLA n-mers accelerates biodegradation [24]. On the other hand, PLA needs at least 494 days to reach the same M_n conditions –an effective 15% reduction of time.

The PLA samples were also evaluated to understand crystallinity degradation and evolution at 37°C. The X_c increased from 28.3% to 31.4% for PLA and from 28.3% to 39.5%

for PLA samples biostimulated with skim milk, as seen in **Figure 4.2d** and **e**, and the values are tabulated in **Table 4.3** respectively. The significant difference between the two indicates the more rapid degradation of PLA samples occurring in the presence of skim milk since the lowest M_n can reconfigure and recrystallize even though the test was conducted below but close to $T_{\mathcal{S}}$. **Figure 4.2f** and **g** present the MWD for PLA in compost and compost biostimulated with skim milk, respectively. The peak amplitude for PLA in **Figure 4.2f** remains approximately the same throughout the test duration. No broadening of the peak and the negligible shift indicate that the chemical hydrolysis proceeded slowly at a mesophilic temperature of 37°C. Whereas for PLA biostimulated with skim milk, a significant shift and broadening of the peak are observed in **Figure 4.2g**, depicting the reduction in M_n , rearrangement of the PLA molecular chains and change in X_c . The presence of skim milk initiates protease activity and concurrent chemical hydrolysis, produces a significant peak shift.

ammuy	evolution for FI	LA and FLA DI	<u>ostimulated wit</u> i	1 SK
	Material	Day 0	Day 180	
	PLA	$28.2 \pm 3.5^{\mathrm{a}}$	31.4 ± 1.8^{a}	
-	PLA + skim milk	$28.2 \pm 3.5^{\mathrm{a}}$	$39.5 \pm 1.5^{\rm b}$	

Table 4.3 Crystallinity evolution for PLA and PLA biostimulated with skim milk.

Values with different letters within columns are statistically different ($\alpha = 0.05$ Tukey-Kramer test).

4.4.2 Effect of gelatin on cellulose and PLA degradation

Gelatin was used as another candidate to stimulate the protease activity of microbial strains in compost. Gelatin has been reported as a precursor for protease activity [10,34,40,43]. Figure 4.3a and b show the CO_2 evolved and % mineralization of the blank, cellulose, gelatin, PLA, cellulose+gelatin (Cell+gel), and PLA+gelatin (PLA+gel) in compost at 37°C. Gelatin shows around 41.5 g of CO_2 evolution and mineralization of 116% in 180 days. To further understand the biostimulation of compost by gelatin, it was combined with

cellulose. The CO₂ evolution in this case (Cell+gel) was 58.5 g, which is as expected and higher compared to the individual values for cellulose and gelatin, and mineralization of 130.4%. To better understand the interaction and account for the degradation behavior of cellulose in the presence of gelatin (Cell_{gel}), the mineralization value was estimated at 180.5% by subtracting the background signal from the gelatin bioreactor. This higher value depicts that gelatin does not affect cellulose degradation, and both are used up as carbon sources by the microorganisms. Additionally, a significant priming effect is observed in the case of bioreactor containing cellulose biostimulated with gelatin. To understand the influence of gelatin on PLA degradation, PLA was introduced in the gelatin-amended compost. The bioreactor containing both PLA and gelatin (PLA+gel) shows CO₂ evolution of around 41.8 g and maximum mineralization of 56.4% by the end of the test (Figure 4.3a). Improved mineralization is observed as opposed to no CO_2 evolution for PLA alone without any biostimulation of compost. The effect of gelatin on PLA degradation is calculated and subtracted by plotting the mineralization of PLA_{gel} (subtracting blank + gelatin), as mentioned earlier. The positive mineralization indicates that gelatin's protease activity helps in PLA's enzymatic degradation (Figure 4.3b). This is validated by the significant difference observed in the kinetic rate of degradation for PLA in compost, with and without any biostimulation by gelatin, as seen in Figure 4.3c inset (PLA $k = 0.0045 \pm 0.0001$ and PLA+gelatin $k = 0.0060 \pm 0.0002$). This change in k indicates a final reduction of around 124 days when PLA biostimulated with gelatin reaches an $M_n \leq 10$ kDa at 371 days compared to PLA, which needs at least 494 days to reach the same $M_{\rm n}$ conditions – an effective 25% reduction of time. The significant difference in the evolution of X_c from 28.3% to 31.4% for PLA, and from 28.3% to 41.9% for PLA samples biostimulated with gelatin, as seen in Figure



4.3d and **e**, and as seen in **Table 4.4** respectively, further shows the improvement in the enzymatic degradation of PLA due to the presence of gelatin.

Figure 4.3 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, gelatin, cellulose + gelatin (Cell+gel), PLA + gelatin (PLA+gel) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated by gelatin. The experimental data was fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$ Tukey-Kramer Test). (d) and (e) depict DSC thermograms for PLA and PLA in compost biostimulated by gelatin, respectively. (f) and (g) show the MWD of PLA in compost and compost biostimulated with gelatin, respectively.

Materia	1 Day 0	Day 180
PLA	$28.2 \pm 3.5^{\mathrm{a}}$	31.4 ± 1.8^{a}
PLA + gela	atin 28.2 ± 3.5^{a}	$41.9\pm1.9^{\rm b}$

Table 4.4 Crystallinity evolution for PLA and PLA biostimulated with gelatin.

Note: Values with different letters within columns are statistically different ($\alpha = 0.05$ Tukey-Kramer test).

The broadening and change in the amplitude of the MWD peaks through the test duration of 180 days, as seen in **Figure 4.3g** for PLA in compost stimulated with gelatin compared to **Figure 4.3f** for PLA alone, provide compelling evidence and confirm enhanced degradation for PLA with gelatin.

4.4.3 Effect of ethyl lactate on cellulose and PLA degradation

Ethyl lactate was used to stimulate the lactate-utilizing microbial species in the compost to improve PLA degradation at 37°C. **Figure 4.4a** and **b** show the CO_2 evolved and % mineralization of the blank, cellulose, ethyl lactate, PLA, cellulose+ ethyl lactate (Cell+el), and PLA+ethyl lactate (PLA+el) in compost at 37°C. Ethyl lactate evolves around 51.5 g of CO_2 , the corresponding mineralization reaches around 270%, and no lag phase is observed. As noted earlier with skim milk and gelatin, a priming effect is observed with ethyl lactate. The CO_2 evolution of the bioreactor containing both cellulose and ethyl lactate (Cell+el) was only around 66.2 g and there was a corresponding mineralization of 182.4%, as both are readily available to be used as carbon sources by the microbial population. The higher value can be attributed to the lactate-utilizing microbial community, increasing the microbial population. The cellulose degradation in the presence of ethyl lactate (Cell_{el}) was decoupled by subtracting the CO_2 evolution from ethyl lactate – maximum mineralization of 116.7% was observed.

To understand how ethyl lactate affects the degradation of PLA, PLA was introduced into the bioreactor biostimulated by ethyl lactate. PLA in the presence of ethyl lactate (PLA+el) evolved around 51.6 g of CO₂, followed a similar trend as ethyl lactate over 180 days,



Figure 4.4 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, ethyl lactate, cellulose + ethyl lactate (Cell-gel), PLA + ethyl lactate (PLA-gel) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated by ethyl lactate. The experimental data was fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$ Tukey-Kramer Test). (d) and (e) depict DSC thermograms for PLA and PLA in compost biostimulated by ethyl lactate, respectively. (f) and (g) show the MWD of PLA in compost and compost biostimulated with ethyl lactate, respectively.

and showed maximum mineralization of 105%. A positive mineralization behavior is seen for PLA when the effect of ethyl lactate is accounted for (PLA_{el}), indicating the lactatestimulating activity of ethyl lactate, but the biostimulation is lower than with skim milk and gelatin.

This finding is substantiated by a significant difference in the kinetic rate of degradation for PLA with and without the biostimulation effect of ethyl lactate, as indicated in **Figure 4.4c** inset (PLA $k = 0.0045 \pm 0.0001$ and PLA-ethyl lactate $k = 0.0058 \pm 0.0002$). This change in k indicates a final reduction of around 111 days when PLA biostimulated with ethyl lactate reaches an $M_{\rm n} \lesssim 10$ kDa at 384 days compared to PLA, which needs at least 494 days to reach the same M_n conditions – an effective 22% reduction of time. The PLA samples were also evaluated to understand crystallinity degradation and evolution at 37°C. The X_c increased from 28.3% to 31.4% for PLA, and from 28.3% to 38.4% for PLA samples biostimulated with ethyl lactate, as seen in Figure 4.4d and e, respectively. The significant difference between the two indicates the degradation of samples as seen from Table 4.5 occurring in the presence of ethyl lactate. Figure 4.4f and g present the MWD for PLA in compost and compost biostimulated with ethyl lactate, respectively. The peak amplitude for PLA in **Figure 4.4f** remains approximately the same throughout the test duration. No broadening of the peak, accompanied by a negligible shift, indicates that the chemical hydrolysis proceeded slowly at a mesophilic temperature of 37°C. Whereas for PLA biostimulated with ethyl lactate, a significant shift and broadening of the peak are observed in **Figure 4.4g**, depicting the reduction in M_n as previously shown for skim milk and gelatin.

Material	Day 0	Day 180
PLA	$28.2\pm3.5^{\rm a}$	31.4 ± 1.8^{a}
PLA + ethyl lactate	$28.2 \pm 3.5^{\rm a}$	38.4 ± 2.1^{b}

Table 4.5 Crystallinity evolution for PLA and PLA biostimulated with ethyl lactate.

Values with different letters within columns are statistically different ($\alpha = 0.05$ Tukey-Kramer test).

The addition of biostimulants in the compost media accelerated the enzymatic degradation of PLA in mesophilic environments, which could be scaled to home composting settings. The biostimulants tested in the study are easily available and can be sourced locally, thus encouraging the practice of home composting, especially for contaminated PLA-based food packages post use. These practices can be applied and carried out by consumers in the backyard, and by university campuses diverting cafeteria and dining hall food waste from ending up in landfill. Additionally, the improvement in PLA degradation because of the presence of skim milk, gelatin, and ethyl lactate can further improve acceptance of PLA in thermophilic industrial composting environments, as these biostimulants can help improve the turnover time as compared to organic fraction such as food waste. In future work, we will be focusing on analyzing the effect of adding biostimulants periodically on accelerating PLA degradation. Understanding how the periodic addition of biostimulant impacts the mechanism of PLA's enzymatic degradation can further open avenues to investigate incorporation of enzymes directly to effectively biodegrade PLA.

4.5 Conclusion

The study investigated the biodegradation behavior of PLA using the biostimulation treatment under an aerobic simulated mesophilic setting. Different compounds – skim milk, gelatin, and ethyl lactate – were used to study the biostimulation effect by targeting different steps in PLA biodegradation. As expected, without any treatment, a very long abiotic lag phase was observed for PLA at mesophilic conditions (37°C), indicating and reaffirming a very slow hydrolysis phase. This is depicted as negative mineralization over the test duration of 180 days. Improved CO_2 evolution was observed for PLA when the compost was biostimulated with the addition of skim milk, gelatin, or ethyl lactate. This finding was corroborated by tracking crystallinity evolution and molecular weight changes. PLA biodegradation was accelerated to reach $M_n \leq 10$ kDa (where the biotic phase is taken over to consume the PLA n-mers) by 15% by adding skim milk, 25% by adding gelatin, and 22% by adding ethyl lactate. This work opens the door to help biodegrade PLA in compost conditions by adding selected biostimulants.

4.6 Acknowledgements

The authors thank Anibal Bher for helping with the test, the data acquisition, and discussion about biodegradation. P.M. acknowledges the School of Packaging at Michigan State University; R.A. acknowledges the USDA National Institute of Food and Agriculture and Michigan State University AgBioResearch, Hatch project number MICL02665, for partial study support.

REFERENCES

- Mayekar P. C., Castro-Aguirre E., Auras R., *et al.*, Effect of nano-clay and surfactant on the biodegradation of poly(lactic acid) films, *Polymers (Basel).*, vol. 12, no. 2, pp. 1– 16, 2020, doi: 10.3390/polym12020311.
- [2] European Bioplastics and Nova-Institute, Bioplastics Market Development Update 2023, 2023. [Online]. Available: http://www.europeanbioplastics.org/news/publications/%0Ahttp://www.europeanbioplastics.org/news/publications/%0AEuropean.
- [3] Zaaba N. F. and Jaafar M., A review on degradation mechanisms of polylactic acid: Hydrolytic, photodegradative, microbial, and enzymatic degradation, *Polym. Eng. Sci.*, vol. 60, no. 9, pp. 2061–2075, Sep. 2020, doi: 10.1002/pen.25511.
- [4] Codari F., Lazzari S., Soos M., et al., Kinetics of the hydrolytic degradation of poly(lactic acid), Polym. Degrad. Stab., vol. 97, no. 11, pp. 2460-2466, 2012, doi: 10.1016/j.polymdegradstab.2012.06.026.
- [5] Al Hosni A. S., Pittman J. K., and Robson G. D., Microbial degradation of four biodegradable polymers in soil and compost demonstrating polycaprolactone as an ideal compostable plastic, *Waste Manag.*, vol. 97, pp. 105–114, Sep. 2019, doi: 10.1016/j.wasman.2019.07.042.
- [6] Pranamuda H., Tokiwa Y., and Tanaka H., Polylactide Degradation by an Amycolatopsis sp, Appl. Environ. Microbiol., vol. 63, no. 4, pp. 1637–1640, Apr. 1997, doi: 10.1128/aem.63.4.1637-1640.1997.
- [7] Samantaray P. K., Little A., Wemyss A. M., et al., Design and Control of Compostability in Synthetic Biopolyesters, ACS Sustain. Chem. Eng., vol. 9, no. 28, pp. 9151–9164, Jul. 2021, doi: 10.1021/acssuschemeng.1c01424.
- [8] Bher A., Mayekar P. C., Auras R. A., and Schvezov C. E., Biodegradation of Biodegradable Polymers in Mesophilic Aerobic Environments, *Int. J. Mol. Sci.*, vol. 23, no. 20, 2022, doi: 10.3390/ijms232012165.
- [9] Tokiwa Y. and Calabia B. P., Biodegradability and biodegradation of poly(lactide), *Appl. Microbiol. Biotechnol.*, vol. 72, no. 2, pp. 244–251, 2006, doi: 10.1007/s00253-006-0488-1.
- [10] Jarerat A., Tokiwa Y., and Tanaka H., Poly(l-lactide) degradation by Kibdelosporangium aridum, *Biotechnol. Lett.*, vol. 25, no. 23, pp. 2035–2038, Dec. 2003, doi: 10.1023/B:BILE.0000004398.38799.29.
- [11] Watson D. B., Wu W.-M., Mehlhorn T., et al., In Situ Bioremediation of Uranium with Emulsified Vegetable Oil as the Electron Donor, *Environ. Sci. Technol.*, vol. 47, no. 12, pp. 6440–6448, 2013, doi: 10.1021/es3033555.
- [12] Yang Q. I., Shang H.-T., Wang X.-L., *et al.*, Anaerobic Degradation of Tetrachloroethylene Using Different Co-substrates as Electron Donors, *Biomed.*

Environ. Sci., vol. 19, pp. 73-76, 2006.

- [13] Simpanen S., Dahl M., Gerlach M., *et al.*, Biostimulation proved to be the most efficient method in the comparison of in situ soil remediation treatments after a simulated oil spill accident, *Environ. Sci. Pollut. Res.*, vol. 23, no. 24, pp. 25024–25038, 2016, doi: 10.1007/s11356-016-7606-0.
- [14] Boonluksiri Y., Prapagdee B., and Sombatsompop N., Promotion of polylactic acid biodegradation by a combined addition of PLA-degrading bacterium and nitrogen source under submerged and soil burial conditions, *Polym. Degrad. Stab.*, vol. 188, 2021, doi: 10.1016/j.polymdegradstab.2021.109562.
- [15] Cosgrove L., McGeechan P. L., Handley P. S., and Robson G. D., Effect of biostimulation and bioaugmentation on degradation of polyurethane buried in soil, *Appl. Environ. Microbiol.*, vol. 76, no. 3, pp. 810–819, Feb. 2010, doi: 10.1128/AEM.00534-09.
- [16] Tribedi P., Goswami M., Chakraborty P., et al., Bioaugmentation and biostimulation: a potential strategy for environmental remediation, J. Microbiol. Exp., vol. 6, no. 5, pp. 223–231, Nov. 2018, doi: 10.15406/jmen.2018.06.00219.
- [17] Wu W.-M., Carley J., Gentry T., et al., Pilot-Scale in Situ Bioremedation of Uranium in a Highly Contaminated Aquifer. 2. Reduction of U(VI) and Geochemical Control of U(VI) Bioavailability, Environ. Sci. Technol., vol. 40, no. 12, pp. 3986–3995, 2006, doi: 10.1021/es051960u.
- [18] Chen A., Shang C., Shao J., et al., The application of iron-based technologies in uranium remediation: A review, Sci. Total Environ., vol. 575, pp. 1291–1306, Jan. 2017, doi: 10.1016/J.SCITOTENV.2016.09.211.
- [19] Bento F. M., Camargo F. A. O., Okeke B. C., and Frankenberger W. T., Comparative bioremediation of soils contaminated with diesel oil by natural attenuation, biostimulation and bioaugmentation, *Bioresour. Technol.*, vol. 96, pp. 1049–1055, 2005, doi: 10.1016/j.biortech.2004.09.008.
- [20] Nwankwegu A. S., Orji M. U., and Onwosi C. O., Studies on organic and in-organic biostimulants in bioremediation of diesel-contaminated arable soil, *Chemosphere*, vol. 162, pp. 148–156, Nov. 2016, doi: 10.1016/J.CHEMOSPHERE.2016.07.074.
- [21] Wang X., Xin J., Yuan M., and Zhao F., Electron competition and electron selectivity in abiotic, biotic, and coupled systems for dechlorinating chlorinated aliphatic hydrocarbons in groundwater: A review, *Water Res.*, vol. 183, p. 116060, Sep. 2020, doi: 10.1016/J.WATRES.2020.116060.
- [22] Wu M., Dick W. A., Li W., et al., Bioaugmentation and biostimulation of hydrocarbon degradation and the microbial community in a petroleum-contaminated soil, Int. Biodeterior. Biodegradation, vol. 107, pp. 158–164, Feb. 2016, doi: 10.1016/J.IBIOD.2015.11.019.
- [23] Koshlaf E. and Ball A. S., Soil bioremediation approaches for petroleum hydrocarbon

polluted environments, *AIMS Microbiol.*, vol. 3, no. 1, p. 25, 2017, doi: 10.3934/MICROBIOL.2017.1.25.

- [24] Castro-Aguirre E., Auras R., Selke S., et al., Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.
- [25] Fan Y., Nishida H., Mori T., et al., Thermal degradation of poly(L-lactide): Effect of alkali earth metal oxides for selective L,L-lactide formation, *Polymer (Guildf)*., vol. 45, no. 4, pp. 1197–1205, 2004, doi: 10.1016/J.POLYMER.2003.12.058.
- [26] Botvin V., Karaseva S., Khasanov V., and Filimoshkin A., Kinetic study of depolymerization of lactic and glycolic acid oligomers in the presence of oxide catalysts, *Polymers (Basel).*, vol. 12, no. 10, pp. 1–19, 2020, doi: 10.3390/polym12102395.
- [27] Hall E. A., Redfern L. R., Wang M. H., and Scheidt K. A., Lewis Acid Activation of a Hydrogen Bond Donor Metal–Organic Framework for Catalysis, ACS Catal., vol. 6, no. 5, pp. 3248–3252, 2016, doi: 10.1021/acscatal.6b00424.
- [28] Qu M., Tu H., Amarante M., et al., Zinc oxide nanoparticles catalyze rapid hydrolysis of poly(lactic acid) at low temperatures, J. Appl. Polym. Sci., vol. 131, no. 11, pp. 1–7, 2014, doi: 10.1002/app.40287.
- [29] Wang X. jie, Huang Z., Wei M. yu, *et al.*, Catalytic effect of nanosized ZnO and TiO2 on thermal degradation of poly(lactic acid) and isoconversional kinetic analysis, *Thermochim. Acta*, vol. 672, pp. 14–24, Feb. 2019, doi: 10.1016/J.TCA.2018.12.008.
- [30] Anžlovar A., Kržan A., and Žagar E., Degradation of PLA/ZnO and PHBV/ZnO composites prepared by melt processing, Arab. J. Chem., vol. 11, no. 3, pp. 343–352, Mar. 2018, doi: 10.1016/J.ARABJC.2017.07.001.
- [31] Luo Y., Lin Z., and Guo G., Biodegradation Assessment of Poly (Lactic Acid) Filled with Functionalized Titania Nanoparticles (PLA/TiO2) under Compost Conditions, Nanoscale Res. Lett., vol. 14, no. 56, pp. 1–10, 2019, doi: 10.1186/s11671-019-2891-4.
- [32] Luo Y. B., Wang X. L., and Wang Y. Z., Effect of TiO2 nanoparticles on the long-term hydrolytic degradation behavior of PLA, *Polym. Degrad. Stab.*, vol. 97, no. 5, pp. 721– 728, May 2012, doi: 10.1016/J.POLYMDEGRADSTAB.2012.02.011.
- [33] Lizundia E., Ruiz-Rubio L., Vilas J. L., and León L. M., Towards the development of eco-friendly disposable polymers: ZnO-initiated thermal and hydrolytic degradation in poly(l-lactide)/ZnO nanocomposites, *RSC Adv.*, vol. 6, no. 19, pp. 15660–15669, 2016, doi: 10.1039/c5ra24604k.
- [34] Jarerat A. and Tokiwa Y., Degradation of Poly(L-lactide) by a Fungus, *Macromol. Biosci.*, vol. 1, no. 4, pp. 136–140, Jun. 2001, doi: 10.1002/1616-5195(20010601)1:4<136::AID-MABI136>3.0.CO;2-3.
- [35] Masaki K., Kamini N. R., Ikeda H., and Iefuji H., Cutinase-Like Enzyme from the

Yeast Cryptococcus sp. Strain S-2 Hydrolyzes Polylactic Acid and Other Biodegradable Plastics, *Appl. Environ. Microbiol.*, vol. 71, no. 11, pp. 7548–7550, Nov. 2005, doi: 10.1128/AEM.71.11.7548-7550.2005.

- [36] Karamanlioglu M. and Robson G. D., The influence of biotic and abiotic factors on the rate of degradation of poly(lactic) acid (PLA) coupons buried in compost and soil, *Polym. Degrad. Stab.*, vol. 98, no. 10, pp. 2063–2071, 2013, doi: 10.1016/j.polymdegradstab.2013.07.004.
- [37] Lim H.-A., Raku T., and Tokiwa Y., Hydrolysis of polyesters by serine proteases, Biotechnol. Lett., vol. 27, no. 7, pp. 459–464, Apr. 2005, doi: 10.1007/s10529-005-2217-8.
- [38] Nakamura K., Tomita T., Abe N., and Kamio Y., Purification and Characterization of an Extracellular Poly(<scp>l</scp> -Lactic Acid) Depolymerase from a Soil Isolate, *Amycolatopsis* sp. Strain K104-1, *Appl. Environ. Microbiol.*, vol. 67, no. 1, pp. 345–353, Jan. 2001, doi: 10.1128/AEM.67.1.345-353.2001.
- [39] Pranamuda H., Tsuchii A., and Tokiwa Y., Poly (L-lactide)-Degrading Enzyme Produced byAmycolatopsis sp., *Macromol. Biosci.*, vol. 1, no. 1, pp. 25–29, Jan. 2001, doi: 10.1002/1616-5195(200101)1:1<25::AID-MABI25>3.0.CO;2-3.
- [40] Apinya T., Sombatsompop N., and Prapagdee B., Selection of a Pseudonocardia sp. RM423 that accelerates the biodegradation of poly(lactic) acid in submerged cultures and in soil microcosms, *Int. Biodeterior. Biodegradation*, vol. 99, pp. 23–30, Apr. 2015, doi: 10.1016/j.ibiod.2015.01.001.
- [41] Jarerat A., Tokiwa Y., and Tanaka H., Microbial Poly(L-lactide)-Degrading Enzyme Induced by Amino Acids, Peptides, and Poly(L-amino Acids), J. Polym. Environ., vol. 12, no. 3, pp. 139–146, 2004.
- [42] Bonifer K. S., Wen X., Hasim S., et al., Bacillus pumilus B12 Degrades Polylactic Acid and Degradation Is Affected by Changing Nutrient Conditions, Front. Microbiol., vol. 10, no. November, pp. 1–13, Nov. 2019, doi: 10.3389/fmicb.2019.02548.
- [43] Jarerat A. and Tokiwa Y., Poly(L-lactide) degradation by Saccharothrix waywayandensis, *Biotechnol. Lett.*, vol. 25, no. 5, pp. 401–404, 2003, doi: 10.1023/A:1022450431193.
- [44] Decorosi F., Exana M. L., Pini F., et al., The Degradative Capabilities of New Amycolatopsis Isolates on Polylactic Acid, *Microorganisms*, vol. 7, no. 12, p. 590, Nov. 2019, doi: 10.3390/microorganisms7120590.
- [45] Penkhrue W., Khanongnuch C., Masaki K., et al., Isolation and screening of biopolymer-degrading microorganisms from northern Thailand, World J. Microbiol. Biotechnol., vol. 31, no. 9, pp. 1431–1442, Sep. 2015, doi: 10.1007/s11274-015-1895-1.
- [46] Rupali D., Screening and Isolation of Protease Producing Bacteria from Soil Collected from Different Areas of Burhanpur Region (MP) India, *Int. J. Curr. Microbiol. Appl. Sci.*, vol. 4, no. 8, pp. 597–606, 2015, Accessed: Apr. 06, 2022. [Online]. Available:

http://www.ijcmas.com.

- [47] Tokiwa Y. and Calabia B. P., Biodegradability and biodegradation of polyesters, J. *Polym. Environ.*, vol. 15, no. 4, pp. 259–267, 2007, doi: 10.1007/s10924-007-0066-3.
- [48] Misra G., Thompson S., Sartain H., and Riggenbach J., Lactic Acid as a Barrier to Migrating Chlorinated Solvent Plumes, in *Remediation of Chlorinated and Recalcitrant Compounds*, 2006, pp. 1–7, Accessed: Apr. 06, 2022. [Online]. Available: www.battelle.org/bookstore.
- [49] Gohil H., Ogram A., and Thomas J., Stimulation of anaerobic biodegradation of DDT and its metabolites in a muck soil: laboratory microcosm and mesocosm studies, *Biodegradation*, vol. 25, no. 5, pp. 633–642, 2014, doi: 10.1007/s10532-014-9687-0.
- [50] Dyer M., Marnette E., Schuren C., and van den Brink K., Full-scale stimulation of reductive dechlorination using the Liner® technique, *Eng. Geol.*, vol. 85, no. 1–2, pp. 39–45, May 2006, doi: 10.1016/J.ENGGEO.2005.09.047.
- [51] Sun Y., Ji L., Wang W., et al., Simultaneous Removal of Polycyclic Aromatic Hydrocarbons and Copper from Soils using Ethyl Lactate-Amended EDDS Solution, J. Environ. Qual., vol. 38, no. 4, pp. 1591–1597, Jul. 2009, doi: 10.2134/jeq2008.0374.
- [52] Dolzhenko A. V, Ethyl lactate and its aqueous solutions as sustainable media for organic synthesis, Sustain. Chem. Pharm., vol. 18, p. 100322, 2020, doi: 10.1016/j.scp.2020.100322.
- [53] Jiang T., Gao C., Ma C., and Xu P., Microbial lactate utilization: Enzymes, pathogenesis, and regulation, *Trends Microbiol.*, vol. 22, no. 10, pp. 589–599, 2014, doi: 10.1016/j.tim.2014.05.008.
- [54] Farhana A. and Lappin S. L., Biochemistry, Lactate Dehydrogenase, *StatPearls*, 2021. https://www.ncbi.nlm.nih.gov/books/NBK557536/.
- [55] Mayekar P. C., Limsukon W., Bher A., and Auras R., Breaking It Down: How Thermoplastic Starch Enhances Poly(lactic acid) Biodegradation in Compost—A Comparative Analysis of Reactive Blends, ACS Sustain. Chem. Eng., vol. 11, no. 26, pp. 9729–9737, 2023, doi: 10.1021/acssuschemeng.3c01676.
- [56] ASTM International, ASTM D5338-15(2021): Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials Under Controlled Composting Conditions, Incorporating Thermophilic Temperatures, in ASTM Standards, 2021, pp. 1–7.
- [57] International Standard ISO/FDIS 14855-1:2012, Determination of the ultimate aerobic biodegradability of plastic materials under controlled composting conditions Method by analysis of evolved carbon dioxide, Part 1: General method, 2012.
- [58] Kijchavengkul T., Auras R., Rubino M., *et al.*, Development of an automatic laboratoryscale respirometric system to measure polymer biodegradability, *Polym. Test.*, vol. 25, no. 8, pp. 1006–1016, 2006, doi: 10.1016/j.polymertesting.2006.06.008.

- [59] ASTM International, ASTM Standard D4754-18, Standard Test Method for Two-Sided Liquid Extraction of Plastic Materials Using FDA, 2018. doi: 10.1520/D4754-98R03.2.
- [60] Karamanlioglu M., Preziosi R., and Robson G. D., Abiotic and biotic environmental degradation of the bioplastic polymer poly(lactic acid): A review, *Polym. Degrad. Stab.*, vol. 137, pp. 122–130, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.009.
- [61] Qi X., Ren Y., and Wang X., New advances in the biodegradation of Poly(lactic) acid, Int. Biodeterior. Biodegrad., vol. 117, pp. 215–223, 2017, doi: 10.1016/j.ibiod.2017.01.010.
- [62] Teixeira S., Eblagon K. M., Miranda F., et al., Towards Controlled Degradation of Poly(lactic) Acid in Technical Applications, C J. Carbon Res., vol. 7, no. 42, pp. 1–43, 2021, doi: 10.1007/978-0-8176-4829-9_1.
- [63] Iñiguez-Franco F., Auras R., Burgess G., et al., Concurrent solvent induced crystallization and hydrolytic degradation of PLA by water-ethanol solutions, Polymer (Guildf)., vol. 99, pp. 315–323, Sep. 2016, doi: 10.1016/j.polymer.2016.07.018.
- [64] Ahmed T., Shahid M., Azeem F., et al., Biodegradation of plastics: current scenario and future prospects for environmental safety, *Environ. Sci. Pollut. Res.*, vol. 25, no. 8, pp. 7287–7298, Mar. 2018, doi: 10.1007/s11356-018-1234-9.
- [65] Södergård A., Selin J. F., and Näsman J. H., Hydrolytic degradation of peroxide modified poly(L-lactide), *Polym. Degrad. Stab.*, vol. 51, no. 3, pp. 351–359, 1996, doi: 10.1016/0141-3910(95)00271-5.
- [66] Siparsky G. L., Voorhees K. J., Dorgan J. R., and Schilling K., Water Transport in Polylactic Acid (PLA), PLA/ Polycaprolactone Copolymers, and PLA/Polyethylene Glycol Blends, J. Environ. Polym. Degrad., vol. 5, no. 3, 1997.
- [67] Siracusa V., Microbial Degradation of Synthetic Biopolymers Waste, *Polymers (Basel).*, vol. 11, no. 6, p. 1066, Jun. 2019, doi: 10.3390/polym11061066.

CHAPTER 5: SPEEDING IT UP: DUAL EFFECTS OF BIOSTIMULANTS AND IRON ON BIODEGRADATION OF POLY(LACTIC ACID) AT MESOPHILIC CONDITIONS

5.1 Abstract

Plastic pollution presents a growing concern, and various solutions have been proposed to address it. One such solution involves the development of new plastics that match the properties of traditional polymers while exhibiting enhanced biodegradability when disposed of in a suitable environment. Poly(lactic acid) (PLA) is a biobased, compostable polymer known for its low environmental impact and ability to break down into harmless components within a specified timeframe. However, its degradation in industrial composting facilities poses challenges, and it cannot degrade in home composting. In this study, we investigated the biodegradability of PLA within a biostimulated compost matrix at mesophilic conditions (37°C) over 180 days. The compost environment was enhanced with Fe₃O₄ nanopowder, skim milk, gelatin, and ethyl lactate, individually and in combination, to target different stages of the PLA biodegradation process. We monitored key indicators, CO₂ evolution, number average molecular weight, and crystallinity, to assess the impact of the various biostimulants and iron. The results demonstrated that the most effective treatment for degrading PLA at mesophilic conditions was adding gelatin and Fe₃O₄. Gelatin accelerated PLA biodegradation by 25%, Fe_3O_4 by 17%, and a combination of gelatin and Fe₃O₄ by c. 30%. The effect of skim milk and ethyl lactate is also reported. This research introduces novel pathways to enhance PLA biodegradation in home composting scenarios, offering promising solutions to address the plastic pollution challenge.

5.2 Introduction

Poly(lactic acid) – PLA – is a biobased, biodegradable polymer that is an eco-friendly substitute for fossil-based polymers for a circular and sustainable economy. PLA is derived from natural resources and quickly degrades and breaks down in suitable waste management environments, such as industrial composting. PLA is a versatile biobased polymer because it has properties comparable to conventional polymers, is cost-effective, and can provide an additional disposal scenario, namely composting, at the end of a contaminated package life cycle [1]. These benefits, combined with the growing consumer awareness of plastic littering and white pollution, have propelled PLA to the forefront as the face of the green, biobased plastic movement [2].

Although PLA is industrially compostable, its practical and rapid biodegradation depends on reaching temperatures in the thermophilic range, 45 to 60°C, to undergo chemical hydrolysis and significantly reduce its molecular weight in a shorter period so that microorganisms present in the compost can use PLA oligomers as a food source. This constraint makes it difficult to degrade PLA at lower and ambient temperatures (i.e., mesophilic range, 20 to 45°C) [3]. Chemical hydrolysis is a crucial step in the PLA degradation mechanism, involving the breakdown of high into low molecular-weight polymer chains, such as oligomers, dimers, and monomers, which are easily assimilated by microorganisms [4]. However, at the lower temperatures commonly encountered in the home or backyard composting environments (herein referred to as backyard composting), chemical hydrolysis proceeds at a prolonged pace [5]. Boosting the hydrolytic degradation of PLA, particularly in backyard composting, is extremely difficult due to the limitation of high temperature essential to activate the biodegradation process [3].

Several research studies have included nanocomposites and metal compounds within the PLA matrix at lower concentrations to enhance its depolymerization and degradation [6– 8]. Including these metal oxides and nanoparticles significantly alters physical properties, such as number average molecular weight (M_n), melting temperature (T_m), and glass transition temperature (T_g) [9–11] of the resulting PLA or PLA blends, and may not be practical for food contact or single-use PLA discarded packages.

Enzymatic hydrolysis involving the release of extracellular enzymes is also essential for PLA degradation when low molecular weight PLA chains are available. PLA's degrading enzymes include the hydrolase class of enzymes, primarily proteases [3]. The serine proteases released by the microorganisms in response to the presence of amino acid compounds in compost have also been shown to cleave the ester bonds in PLA [12,13]. So, the introduction of enzymes to compost may assist in the biodegradation of PLA.

Adding various components besides enzymes, such as nutrients, electron donor/acceptor compounds, or compounds essential to trigger the biochemical processes of microorganisms in the given environment, is termed biostimulation [14–16]. Biostimulation techniques to enhance PLA biodegradation have been reported [17]. However, most of these studies were conducted in restricted settings where PLA was the only carbon source, in liquid media with specific microbial strains, and usually at higher temperatures, which do not replicate the conditions encountered during backyard composting [18,19].

In a previous study, we assessed the change in PLA biodegradation at a mesophilic temperature of 37°C in a solid composting matrix by adding biostimulants to the compost. The compounds selected as biostimulants in that study were screened through multiple criteria: anticipated to have no toxicity towards the microorganisms present, be able to degrade, be biodegradable in a suitable timeframe, be consumed, be cost-effective, and be readily available. The main goal was to introduce compounds enhancing the biotic degradation stage. Skim milk and gelatin were selected to trigger proteolytic activity. Ethyl lactate, belonging to the lactate esters family, was used to stimulate the lactate-utilizing microbial species in backyard compost. All these compounds effectively enhanced the biodegradation of PLA in simulated backyard composting by at least 15%, as determined by the accelerated reduction of M_n [20].

In this study, we focused on evaluating the effect of adding a metal compound to catalyze the chemical hydrolysis of PLA and combine that with the previous demonstrated biotic enhancement. Metals can be added externally to the compost media rather than within PLA for humification purposes, as previously reported [21,22] and screening through the earlier criteria. **Table 5.1** presents the permissible limits of metal compounds as derived from regulatory standards for heavy metals in agricultural soils (mg/kg) [23], since the resulting amended compost may be applied to agricultural land, lawns, or home gardens. The primary standards include EEA 2007 [24], TMS 2007 [25], BPI 2021 [26], GB 15168-2018 [27], OMOE 2011 [28], and NZME 2012 [29].

Country	As	Cd	\mathbf{Cr}	Cu	Hg	Ni	Pb	Zn
Australia	20	3	50	100	1	60	300	200
Canada	20	3	250	150	0.8	100	200	500
C1 ·	20 -	0.3 -	150-	50-	0.5-	60-	70-	200-
China	40	0.6	300	200	3.4	190	240	300
Germany	50	5	500	200	5	200	1000	600
Tanzania	1	1	100	200	2	100	200	150
Netherlands	76	13	180	190	36	100	530	720
New Zealand	17	3	290	> 10 ⁴	200	N/A	160	N/A
UK	43	1.8	N/A	N/A	26	230	N/A	N/A
US	6.5	1.5	105	50	0.4	31	75	250

Table 5.1 Critical limits of heavy metals in agricultural soils (mg/kg), adapted from [23].

Considering the criteria mentioned before and the information from Table 5.1 only a few metals could be selected for further consideration due to limitations placed by permissible limits. The selected compounds were further scrutinized (Table 5.2) for their antibacterial properties, as reported in the literature, to evaluate their use directly in compost.

Element	Antibacterial nature	References
Zn	Yes	[30,31]
Fe	No	[30,32,33]
Cu	Yes	[30]
Ni	Yes	[30]
Ti	Yes	[34,35]
Cl	Yes	[30,36]
Ag	Yes	[36,37]

Table 5.2 List of elements screened for their antimicrobial properties.

Table 5.2 indicates that iron was an acceptable compound that could be used to target chemical hydrolysis. It was demonstrated that introduction of a Lewis acid, FeCl₃, can speed up the hydrolysis of PLA in an alkali solution [38]. However, due to chlorine's antimicrobial behavior, it was impossible to introduce FeCl₃ into compost. Alternatively, iron oxides, such as FeO, FeO₂, Fe₃O₄, or Fe₂O₃, could be the primary option and are in a form that is not toxic to the microbes in the compost [32,33]. Several of these iron forms are present in soil worldwide [39].

Thus, this study aimed to investigate the effectiveness of biostimulating the compost environment with compounds that may be able to enhance the chemical hydrolysis (Fe₃O₄), the enzymatic degradation (skim milk and gelatin), and electroconductivity (ethyl lactate as electron donor compound) during the aerobic biodegradation of PLA in compost at mesophilic conditions. Differences in CO₂ evolution, changes in M_n , and the crystallinity (X_c) of PLA degradation with and without biostimulants were monitored to account for the activity of biostimulants in compost.

5.3 Materials and methods

5.3.1 Materials

PLA Ingeo[™] 2003D resin, with L-lactic acid content of 96%, was obtained from NatureWorks LLC (Minnetonka, MN, USA). Iron oxide nano-powder (Fe₃O₄) was obtained from US Research Nanomaterials, Inc. (Houston, TX, USA). Skim milk powder was procured from a local store (Walmart, Lansing, MI, USA). Gelatin of the brand McCormick & Co. (Hunt Valley, MD, USA) was purchased on Amazon LLC. Ethyl lactate was procured from Sigma Aldrich[™] (St. Louis, MO, USA).

5.3.2 Characterization of PLA and the biostimulants

The carbon, hydrogen, and nitrogen compositions of PLA resin, skim milk, gelatin, and ethyl lactate were determined using elemental analysis, (CHNS/O Elemental Analyzer, PerkinElmer 2400 Series II) (Shelton, CT, USA), and are presented in Table 5.3.

Material	% Carbon ^a	% Hydrogen ^a	% Nitrogen ^a
PLA	49.72 ± 0.19	5.72 ± 0.04	0.11 ± 0.07
Skim milk	41.28 ± 0.13	6.33 ± 0.03	5.75 ± 0.06
Gelatin	44.80 ± 0.30	7.00 ± 0.10	16.4 ± 0.10
Ethyl lactate	31.93 ± 0.98	5.46 ± 0.34	0.05 ± 0.03

Table 5.3 Carbon, hydrogen, and nitrogen content (percentage by weight) of the tested materials.

[a] Percentage by weight.

5.3.3 Biodegradation in compost

The biodegradation of PLA and the effectiveness of introducing biostimulants in compost on the degradation of PLA were evaluated under aerobic mesophilic conditions using a direct measurement respirometric (DMR) system [40–42]. The system included a non-dispersive infrared gas analyzer (NDIR) (Li-COR[®] LI-820, Lincoln, NE, USA) which measures the CO₂ concentration. The DMR system chamber was maintained at temperature of 37 ± 2 °C and relative humidity (RH) of $50\% \pm 5\%$. A flow rate of CO₂-free air (concentration <30 ppm to establish a low baseline) was controlled at 40 ± 2 cm³/min. Detailed information about the DMR equipment can be found in other source [43].

The mature compost obtained from MSU composting facility was sifted using a 10mm screen to get rid of any huge debris or chunks present, and then conditioned at 37°C until use. Deionized water was used to adjust the moisture content of compost to 50%. Saturated vermiculite (Sun Gro Horticulture Distribution Inc., Bellevue, WA, USA), was mixed with compost in 1:4 parts (dry weight). Samples of the resulting compost-vermiculite mixture were later sent to the Soil and Plant Laboratory at MSU for determining the physicochemical parameters. Data regarding the nutrient analysis is presented in **Table A5.1** (Appendix 5A) [44]. The bioreactors were packed with 400 g of compost, 8 g of PLA sample and the selected biostimulant, and all the samples were tested in triplicate. Blank (only compost) and positive control (cellulose) were also tested.

Biostimulation: Fe_3O_4 nanopowder at 17 g was mixed with the compost matrix in the bioreactor containing PLA to target the hydrolysis step. Ethyl lactate, skim milk, and gelatin were added at 8 g individually per bioreactor and mixed with the compost matrix thoroughly with the PLA pellets in it to ensure uniform distribution.

5.3.4 Size exclusion chromatography

The M_n and molecular weight distribution (*MWD*) of PLA for the control and each treatment with the biostimulants were measured using SEC (Waters Corp., Milford, MA, USA) as described elsewhere [4]. PLA samples weighing approximately 10 mg were retrieved at predetermined time intervals and dissolved in 5 mL of tetrahydrofuran (THF) solvent. A temperature of 35°C and a 1 mL/min flow rate were maintained during testing. The Mark-Houwink constants of K = 0.000174 dL/g and a = 0.736 were used to determine M_n , M_w and MWD of the PLA samples. Data analysis was carried out using Waters BreezeTM2 software.

5.3.5 Differential scanning calorimetry

A DSC model Q100 (TA Instruments, New Castle, DE, USA), was used to determine the T_g , T_m , crystallization temperature (T_c), and crystallinity (X_c) for the PLA samples retrieved from the regular and biostimulated compost. PLA samples weighing between 5 - 10 mg were packed in aluminum pans and cooled down to -5° C and then subjected to a heating cycle to reach 210°C at a ramp rate of 10°C/min . This helped to evaluate the evolution of X_c . The cooling was achieved using a nitrogen cooling system that maintained the purge flow rate at 70 mL/min. The degree of crystallinity was estimated using equation (5.1):

$$\chi_c \% = \frac{\Delta H_m - \Delta H_c}{\Delta H_m^0} x \ 100 \tag{5.1}$$

where ΔH_m is the heat of fusion, ΔH_c is cold crystallization enthalpy, and ΔH_m^0 is the heat of fusion for 100% crystalline pure PLA (93 J/g) [45].

5.4 Results and discussion

The CO_2 evolution of PLA samples in compost and biostimulated with Fe_3O_4 nanopowder and the combination of gelatin, skim milk, and ethyl lactate without and with Fe_3O_4 nanopowder was tracked over a test duration of 180 days at mesophilic conditions
(37°C). Samples were retrieved at specific times to evaluate the M_n and X_c evolution and determine the kinetic degradation rate.

5.4.1 Effect of Fe₃O₄ on cellulose and PLA degradation

Figure 5.1a and b shows the CO_2 evolution and mineralization, respectively, of cellulose, cellulose in compost biostimulated with Fe₃O₄ nanopowder (hereafter referred to as cellulose + Fe), PLA, and PLA in compost biostimulated with Fe_3O_4 nanopowder (hereafter referred to as PLA + Fe) at 37°C. Control compost (blank) evolved around 26.1 g of CO₂, and compost biostimulated with Fe_3O_4 nanopowder (hereafter referred to as blank + Fe) evolved around 27.2 g of CO₂. The minor difference can be attributed to the difference in weight of the compost introduced in the bioreactor, and the levels are not significantly different (p > p)0.05). Cellulose in compost evolved around 36.8 g of CO₂, and reached a mineralization of 87.7%, whereas cellulose + Fe evolved around 44.6 g of CO₂, depicting a mineralization of 137.7%. The primary reason for the priming effect (>100% mineralization) observed in the case of cellulose + Fe may be attributed to the over-deterioration of the endemic carbon present in the compost [4]. PLA in compost showed around 22.5 g of CO_2 evolution, whereas blank produced around 26.1 g of CO_2 , implying that no carbon from PLA was degraded. The negative mineralization values indicate more CO_2 production in the blank bioreactors than in the PLA bioreactors. PLA offers a physical hydrophobic barrier to water, making it difficult for the microorganisms to utilize it as a carbon source at the beginning of the test and until day 180 due to the low contribution of chemical hydrolysis at mesophilic temperatures. Overall, we did not see any mineralization in PLA due to the low temperature of 37°C, which is insufficient to activate the biotic stage. These values are very low compared to the degradation of PLA at thermophilic temperatures [4,44] but are similar to earlier reported values [5,46].

In the case of PLA and PLA + Fe, we observed the difference in the CO₂ evolution right from the start of the test. PLA + Fe evolved around 25.9 g of CO₂ compared with 22.5 g of CO₂ evolution for PLA. Similar to the case of cellulose + Fe, the reason for the difference in the CO₂ evolution for PLA can be attributed to the presence of Fe₃O₄. Fe₃O₄ promotes microbial activity in soil and enhances the nitrification potential [47]. Fe₃O₄ is also known to induce changes by enhancing enzymatic activity and microbial growth [32]. This characteristic can be corroborated by the kinetic degradation rate (k), as seen in **Figure 5.1c** inset (PLA $k = 0.0045 \pm 0.0001 d^{-1}$ and PLA+ Fe $k = 0.0052 \pm 0.0002 d^{-1}$). The significant difference can be credited to the presence of Fe₃O₄. **Figure 5.1d and e** present the molecular weight distribution (MWD) for PLA in compost and compost biostimulated with Fe₃O₄. The peak amplitude for PLA in Figure 5.1dremained approximately the same throughout the test duration. No broadening of the peak and negligible shift indicates that the chemical hydrolysis proceeded slowly at a mesophilic temperature of 37°C. In contrast, for PLA + Fe, a significant shift to low M_w and broadening of the peak are observed in Figure 5.1e, depicting the reduction in M_a as shown in **Figure 5.1c**.



Figure 5.1 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, blank + Fe₃O₄ (Blank + Fe), cellulose + Fe₃O₄ (Cellulose + Fe), PLA + Fe₃O₄ (PLA + Fe) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated with Fe₃O₄. The experimental data was fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$, Tukey-Kramer test). (d) and (e) show the MWD of PLA in compost and compost biostimulated with Fe₃O₄, respectively.

5.4.2 M_n reduction for PLA with biostimulants in compost

PLA samples were retrieved separately from the control compost and compost biostimulated with skim milk, gelatin, and ethyl lactate. **Figure 5.2** shows the reduction in M_n of PLA and biostimulated PLA tracked until the end of the test (180 days). A significant difference was observed in the kinetic reduction rates of PLA with biostimulant treatment compared with no biostimulation treatment. The CO₂ evolution values for PLA, PLA + skim milk, PLA + gelatin, and PLA + ethyl lactate are provided in Appendix 5B. Skim milk was added to the compost to induce protease activity by the microbes [48– 50]. Serine protease (3.4.21.112) belongs to the peptidases and is the class of extracellular enzymes able to hydrolyze the peptide bonds linked to amino acids in the protein structure. Skim milk is composed of different proteins, such as lactose, casein, and whey protein, making it a good precursor for enzymatic activity, as mentioned earlier. Other researchers have previously used skim milk to demonstrate extracellular protease synthesis. The microorganisms present in compost secrete protease to hydrolyze the milk protein. This protease is used by microorganisms capable of PLA degradation to depolymerize PLA [51]. This increase in k can be deduced as a final reduction of around 75 days on the biodegradation time when PLA in compost biostimulated with skim milk reaches $M_n \leq 10$ kDa at 420 days (**Figure 5.2**). Microorganisms assimilate the PLA n-mers at this stage, accelerating the biodegradation stage [44]. On the other hand, PLA needs at least 494 days to reach the same $M_n -$ an effective 15% reduction of time.

Gelatin is composed of protein and amino acids, and is a precursor for protease activity [19,52–54]. The addition of gelatin to compost produced an acceleration of PLA with an enhancement of k translated to a final reduction of around 124 days when PLA is biostimulated with gelatin, reaching an $M_n \leq 10$ kDa at 371 days compared with at least 494 days for PLA alone – an effective 25% reduction of time (Figure 5.2).

Ethyl lactate, on the other hand, was used to stimulate the lactate utilizing microbes in the compost. Ethyl lactate undergoes hydrolysis to produce ethanol and lactate, where both can act as a constant long-term supply of hydrogen sources as electron donor compounds for reductive degradation and microbial redox process. Lactate has been shown to act as an electron donor compound in the case of anaerobic degradation [55,56]. Lactate has previously been used for anaerobic degradation, trichloroethane dichlorination, and sulfate reduction [57,58]. PLA + ethyl lactate resulted in a change in k, which can be translated to around 111 days when PLA is biostimulated with ethyl lactate, reaching an $M_n \leq 10$ kDa at 384 days compared to PLA alone – an effective 22% reduction of time (**Figure 5.2**).

A detailed discussion of the effect of gelatin, skim milk, and ethyl lactate on the biodegradation of PLA is provided in our previous work [20].



Figure 5.2 Normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated with skim milk, gelatin, and ethyl lactate. The experimental data was fitted using a first-order reaction of the form $M_n / M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$, Tukey-Kramer test).

5.4.3 Effect of Fe_3O_4 on cellulose and PLA degradation with gelatin as a

biostimulant

Since gelatin resulted in the most significant M_n reduction for PLA, it was selected to discuss the effect of a combination of biostimulants (i.e., Fe₃O₄ nanopowder and gelatin). The data for Fe₃O₄ nanopowder, skim milk, and ethyl lactate are provided in Appendix 5B. When Fe₃O₄ nanopowder was introduced in compost amended with gelatin, the gelatin acted as a precursor for the protease enzyme secretion by the microbes present in the compost and the Fe₃O₄ nanopowder provided the metal to catalyze the hydrolysis.

Figure 5.3a and b show the CO_2 evolution and mineralization of cellulose + Fe, gelatin + Fe, PLA + Fe, cellulose + gelatin + Fe (hereafter referred to as cell + gel + Fe), and PLA + gel + Fe in compost at 37°C. Gelatin + Fe resulted in around 60.1 g of CO_2 evolution and mineralization of 247.8% in 180 days. Gelatin was combined with Fe to target the chemical hydrolysis and enzymatic degradation steps. The CO_2 evolution in this case (Cell+gel+Fe 1) was 64.6 g, which is as expected and higher compared with the individual values for cellulose + Fe and gelatin + Fe and a mineralization of 97.7%. To better understand the interaction and to account for the degradation behavior of cellulose in the presence of gelatin (Cell+gel+Fe 2), the mineralization value was estimated at 80.5% by subtracting the background signal from the bioreactor containing gelatin + Fe. This higher value indicates that cellulose degradation was not affected by gelatin, and both are used up by the microorganisms as carbon sources.



Figure 5.3 Cumulative CO₂ evolution (a) and mineralization (b) of blank + Fe, cellulose + Fe, PLA + Fe, gelatin + Fe, cellulose + gelatin + Fe (Cell+gel+Fe), PLA + gelatin + Fe (PLA+gel+Fe) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated with gelatin, and gelatin + Fe. The experimental data was fitted using a first-order reaction of the form $M_n/M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values; values with different lowercase letters are statistically different ($\alpha = 0.05$, Tukey-Kramer test). (d) and (e) depict DSC thermograms for PLA + Fe and PLA + Fe in compost biostimulated with gelatin. (f) and (g) show the MWD of PLA + Fe in compost and compost biostimulated with gelatin.

To understand the influence of gelatin + Fe on PLA degradation, PLA was introduced in the compost amended with gelatin + Fe. The bioreactor containing both PLA and gelatin + Fe (PLA+gel+Fe 1) generated CO₂ evolution of around 40.5 g and maximum mineralization of 56.5% by the end of the test. Improved mineralization was observed as opposed to no CO_2 evolution for PLA alone without any biostimulation of the compost. The effect of gelatin + Fe on PLA degradation is calculated by plotting the mineralization of PLA+gel+Fe 2 (subtracting gelatin + Fe), as mentioned earlier. Negative mineralization does not necessarily indicate the absence of gelatin's protease activity in PLA's enzymatic degradation. This finding is validated by the significant difference observed in the kinetic rate of degradation for PLA in compost, with and without any biostimulation with gelatin, as seen in Figure 5.3c inset (PLA+Fe $k = 0.0052 \pm 0.0001$ and PLA+Gel+Fe $k = 0.0076 \pm 0.0002$). The significant difference in the evolution of X_c from 28.3% to 31.4% for PLA, and from 28.3% to 40.9% for PLA biostimulated with gelatin + Fe, as seen in Figure 5.3d and e, respectively, further shows the improvement in the enzymatic degradation of PLA due to the presence of gelatin + Fe. Gelatin acts as a precursor for the microbes to release protease enzyme, aiding in the enzymatic degradation of PLA. The broadening and change in the amplitude of the MWD peaks through the test duration of 180 days, as seen in Figure 5.3g for PLA in compost biostimulated with gelatin + Fe compared to PLA + Fe alone in Figure 5.3g, show compelling evidence for enhanced degradation for PLA in the presence of gelatin and Fe. Iron is an essential micronutrient, necessary for life-sustaining processes, and plays a critical role in cell growth of microbes [59]. Iron also functions as a cofactor, promoting and increasing enzymatic activity. Iron plays important roles in various biological processes such as respiration, oxido-reduction mechanism, nitrogen fixation, tricarboxylic acid cycle, and electron transport [33]. S. He et al. demonstrated that a soil matrix amended by Fe nanoparticles shifted the microbial community composition and stimulated the metabolic activity of the bacterial community present by enhancing their growth rate [33]. The soil nitrification potential of the Fe-amended soil was improved by 10% to 19% compared to control soil, indicating that adding Fe aided in the increase of biomass capacity and eventually enhanced and boosted carbon cycling.

Zhang et al. further showed that the addition of Fe nanoparticles promoted the degradation of organic matter and amplified the dehydrogenase and urease activities, significantly improving the overall microbial activity and nitrogen mineralization [47]. Thus, adding Fe_3O_4 nanopowder improves the microbial metabolic activity, nitrification potential, and microbial population. When supplemented with the enzymatic activity associated with gelatin, these changes improve the degradation of PLA in compost compared with that of control PLA with no biostimulants present. Y. He et al. showed enhanced enzymatic and nitrification activity for organic matter degradation in a food-waste composting system due to compost amendment with Fe-carbon particles [60]. In addition, the bacterial and fungal communities exhibited significant improvement in the composting process due to the presence of iron, which can explain the improved PLA degradation found in the presence of Fe_3O_4 nanopowder.

Overall, the changes obtained in k indicate a final reduction of around 148 days when PLA stimulated with gelatin reaches an $M_n \leq 10$ kDa at 346 days compared to at last 494 days for PLA – an effective 25% reduction of time. Similarly, time reductions of 17% or 30% were observed when PLA + Fe₃O₄ or gelatin and Fe₃O₄ nanopowder, respectively, were included in the compost.

As organic waste disposal is becoming more stringent worldwide and landfill disposal bans are increasing, several food industries are being impacted and need to find alternative end-of-life scenarios. The gelatin industry produces a large amount of sludge, resulting in a tremendous amount of waste generated, which includes collagen fibers, bone residues, and other inorganic materials. This gelatin sludge usually ends up in landfill or waste management treatment without any pretreatment and creates several problems such as water contamination, greenhouse gas emission, and health risks for local habitats [61]. So, this waste could be diverted from landfills and used to produce mature compost along with the organic fraction of municipal solid waste. Gelatin sludge is high in nitrogen and organic matter content and can act as a valuable plant nutrient [61]. The nutritional value of the compost generated from treating gelatin waste and Fe₃O₄ nanopowder, considering the benefits mentioned earlier, complement each other to improve soil fertility once the compost is applied to agricultural land. The selection and combination of specific compounds can open a new route to accelerate the degradation of compostable polymers in industrial and home composting operations.

5.5 Conclusion

We investigated the role of different compounds—skim milk, gelatin, and ethyl lactate in combination with Fe₃O₄ nanopowder—on PLA degradation at 37°C by biostimulating the compost media. The different compounds were selected to target different stages of the biodegradation process. Without any biostimulant compounds, PLA continued to undergo a long abiotic lag phase affirming a slow hydrolysis phase, which was seen as a negative mineralization for the test duration of 180 days, whereas a boost in CO₂ evolution for PLA was observed in compost amended by Fe₃O₄ nanopowder in combination with gelatin, skim milk, or ethyl lactate. This observation was verified by the molecular weight change and crystallinity evolution. PLA biodegradation was accelerated by 30% to reach the biotic phase of $M_n \leq 10$ kDa by the addition of gelatin and Fe₃O₄ nanopowder. The addition of biostimulants opens new avenues to improve PLA biodegradation in home composting conditions.

5.6 Acknowledgments

P.M. acknowledges the School of Packaging at Michigan State University; R.A. acknowledges the USDA National Institute of Food and Agriculture and Michigan State University AgBioResearch, Hatch project number MICL02665, for partial study support.

REFERENCES

- [1] Auras R., Lim L.-T., Selke S. E. M., and Tsuji H., *Poly(lactic acid): Synthesis, Structures, Properties, Processing, Applications, and End of Life.* 2022.
- [2] Ghosh K. and Jones B. H., Roadmap to Biodegradable Plastics—Current State and Research Needs, ACS Sustain. Chem. Eng., vol. 9, no. 18, pp. 6170–6187, May 2021, doi: 10.1021/acssuschemeng.1c00801.
- [3] Bher A., Mayekar P. C., Auras R. A., and Schvezov C. E., Biodegradation of Biodegradable Polymers in Mesophilic Aerobic Environments, *Int. J. Mol. Sci.*, vol. 23, no. 20, 2022, doi: 10.3390/ijms232012165.
- [4] Mayekar P. C., Castro-Aguirre E., Auras R., et al., Effect of nano-clay and surfactant on the biodegradation of poly(lactic acid) films, *Polymers (Basel).*, vol. 12, no. 2, pp. 1– 16, 2020, doi: 10.3390/polym12020311.
- [5] Mayekar P. C., Limsukon W., Bher A., and Auras R., Breaking It Down: How Thermoplastic Starch Enhances Poly(lactic acid) Biodegradation in Compost—A Comparative Analysis of Reactive Blends, ACS Sustain. Chem. Eng., vol. 11, no. 26, pp. 9729–9737, 2023, doi: 10.1021/acssuschemeng.3c01676.
- [6] Botvin V., Karaseva S., Khasanov V., and Filimoshkin A., Kinetic study of depolymerization of lactic and glycolic acid oligomers in the presence of oxide catalysts, *Polymers (Basel).*, vol. 12, no. 10, pp. 1–19, 2020, doi: 10.3390/polym12102395.
- [7] Qu M., Tu H., Amarante M., et al., Zinc oxide nanoparticles catalyze rapid hydrolysis of poly(lactic acid) at low temperatures, J. Appl. Polym. Sci., vol. 131, no. 11, pp. 1–7, 2014, doi: 10.1002/app.40287.
- [8] Luo Y., Lin Z., and Guo G., Biodegradation Assessment of Poly (Lactic Acid) Filled with Functionalized Titania Nanoparticles (PLA/TiO2) under Compost Conditions, Nanoscale Res. Lett., vol. 14, no. 56, pp. 1–10, 2019, doi: 10.1186/s11671-019-2891-4.
- [9] Anžlovar A., Kržan A., and Žagar E., Degradation of PLA/ZnO and PHBV/ZnO composites prepared by melt processing, Arab. J. Chem., vol. 11, no. 3, pp. 343–352, Mar. 2018, doi: 10.1016/J.ARABJC.2017.07.001.
- [10] Luo Y. B., Wang X. L., and Wang Y. Z., Effect of TiO2 nanoparticles on the long-term hydrolytic degradation behavior of PLA, *Polym. Degrad. Stab.*, vol. 97, no. 5, pp. 721– 728, May 2012, doi: 10.1016/J.POLYMDEGRADSTAB.2012.02.011.
- [11] Lizundia E., Ruiz-Rubio L., Vilas J. L., and León L. M., Towards the development of eco-friendly disposable polymers: ZnO-initiated thermal and hydrolytic degradation in poly(l-lactide)/ZnO nanocomposites, *RSC Adv.*, vol. 6, no. 19, pp. 15660–15669, 2016, doi: 10.1039/c5ra24604k.
- [12] Lim H.-A., Raku T., and Tokiwa Y., Hydrolysis of polyesters by serine proteases, Biotechnol. Lett., vol. 27, no. 7, pp. 459–464, Apr. 2005, doi: 10.1007/s10529-005-2217-8.

- [13] Pranamuda H., Tsuchii A., and Tokiwa Y., Poly (L-lactide)-Degrading Enzyme Produced byAmycolatopsis sp., *Macromol. Biosci.*, vol. 1, no. 1, pp. 25–29, Jan. 2001, doi: 10.1002/1616-5195(200101)1:1<25::AID-MABI25>3.0.CO;2-3.
- [14] Tokiwa Y. and Calabia B. P., Biodegradability and biodegradation of poly(lactide), *Appl. Microbiol. Biotechnol.*, vol. 72, no. 2, pp. 244–251, 2006, doi: 10.1007/s00253-006-0488-1.
- [15] Watson D. B., Wu W.-M., Mehlhorn T., et al., In Situ Bioremediation of Uranium with Emulsified Vegetable Oil as the Electron Donor, *Environ. Sci. Technol.*, vol. 47, no. 12, pp. 6440–6448, 2013, doi: 10.1021/es3033555.
- [16] Cosgrove L., McGeechan P. L., Handley P. S., and Robson G. D., Effect of biostimulation and bioaugmentation on degradation of polyurethane buried in soil, *Appl. Environ. Microbiol.*, vol. 76, no. 3, pp. 810–819, Feb. 2010, doi: 10.1128/AEM.00534-09.
- [17] Boonluksiri Y., Prapagdee B., and Sombatsompop N., Promotion of polylactic acid biodegradation by a combined addition of PLA-degrading bacterium and nitrogen source under submerged and soil burial conditions, *Polym. Degrad. Stab.*, vol. 188, 2021, doi: 10.1016/j.polymdegradstab.2021.109562.
- [18] Bonifer K. S., Wen X., Hasim S., et al., Bacillus pumilus B12 Degrades Polylactic Acid and Degradation Is Affected by Changing Nutrient Conditions, Front. Microbiol., vol. 10, no. November, pp. 1–13, Nov. 2019, doi: 10.3389/fmicb.2019.02548.
- [19] Apinya T., Sombatsompop N., and Prapagdee B., Selection of a Pseudonocardia sp. RM423 that accelerates the biodegradation of poly(lactic) acid in submerged cultures and in soil microcosms, *Int. Biodeterior. Biodegradation*, vol. 99, pp. 23–30, Apr. 2015, doi: 10.1016/j.ibiod.2015.01.001.
- [20] Mayekar P. C. and Auras R., Accelerating Biodegradation : Enhancing Poly(lactic acid) Breakdown at Mesophilic Environmental Conditions with Biostimulants, *Macromol. Rapid Commun.*, vol. 2300641, pp. 1–10, 2023.
- [21] Zhang S., Wei Z., Zhao M., et al., Influence of malonic acid and manganese dioxide on humic substance formation and inhibition of CO2 release during composting, *Bioresour. Technol.*, vol. 318, no. August, p. 124075, 2020, doi: 10.1016/j.biortech.2020.124075.
- [22] Mei J., Ji K., Su L., et al., Effects of FeSO4 dosage on nitrogen loss and humification during the composting of cow dung and corn straw, *Bioresour. Technol.*, vol. 341, no. August, p. 125867, 2021, doi: 10.1016/j.biortech.2021.125867.
- [23] He Z., Shentu, Yang X., *et al.*, Heavy Metal Contamination of Soils: Sources, Indicators, and Assessment, *J. Environ. Indic.*, vol. 9, pp. 17–18, 2015.
- [24] European Environmental Agency, Progress in management of contaminated sites (CSI 015/LSI 003), 2007. https://www.eea.europa.eu/data-and-maps/indicators/progress-in-management-of-contaminated-sites-3/assessment.

- [25] Tanzania Minister of State (TMS), The environmental management (soil quality standards) regulations, 2007.
- [26] Biodegradable Products Institute, BPI Heavy Metals, 2021. .
- [27] Ministry of Environment Protection of the People's Republic of China, Soil Environmental Quality Risk Control Standard for Soil Contamination of Agricultural Land, 2018.
- [28] Ontario Ministry of the Environment (OMOE), Soil, ground water and sediment standards for use under Part XV.1 of the Ontario Protection Act. 2011.
- [29] New Zealand Ministry for the Environment (NZME), Users' Guide National Environmental Standard for Assessing and Managing Contaminants in Soil to Protect Human Health, 2012.
- [30] Yasuyuki M., Kunihiro K., Kurissery S., et al., Antibacterial properties of nine pure metals: a laboratory study using Staphylococcus aureus and Escherichia coli, *Biofouling*, vol. 26, no. 7, pp. 851–858, 2010, doi: 10.1080/08927014.2010.527000.
- [31] Hernández-García E., Vargas M., González-Martínez C., and Chiralt A., Biodegradable antimicrobial films for food packaging: Effect of antimicrobials on degradation, *Foods*, vol. 10, no. 6, 2021, doi: 10.3390/foods10061256.
- [32] Zhang L., Dong H., Zhu Y., *et al.*, Evolutions of different microbial populations and the relationships with matrix properties during agricultural waste composting with amendment of iron (hydr)oxide nanoparticles, *Bioresour. Technol.*, vol. 289, p. 121697, Oct. 2019, doi: 10.1016/J.BIORTECH.2019.121697.
- [33] He S., Feng Y., Ni J., et al., Different responses of soil microbial metabolic activity to silver and iron oxide nanoparticles, *Chemosphere*, vol. 147, pp. 195–202, Mar. 2016, doi: 10.1016/J.CHEMOSPHERE.2015.12.055.
- [34] Pachapur V. L., Dalila Larios A., Cledón M., et al., Behavior and characterization of titanium dioxide and silver nanoparticles in soils, Sci. Total Environ., vol. 563–564, pp. 933–943, Sep. 2016, doi: 10.1016/J.SCITOTENV.2015.11.090.
- [35] Kaseem M., Hamad K., and Rehman Z. U., Review of Recent Advances in Polylactic Acid/TiO2 Composites, *Materials (Basel).*, vol. 12, pp. 1–16, 2019.
- [36] Kim J., Pitts B., Stewart P. S., et al., Comparison of the Antimicrobial Effects of Chlorine, Silver Ion, and Tobramycin on Biofilm, Antimicrob. Agents Chemother., vol. 52, no. 4, p. 1446, Apr. 2008, doi: 10.1128/AAC.00054-07.
- [37] Gorrasi G., Sorrentino A., and Pantani R., Modulation of Biodegradation Rate of Poly(lactic acid) by Silver Nanoparticles, J. Polym. Environ., vol. 23, pp. 316–320, 2015, doi: 10.1007/s10924-015-0720-0.
- [38] Li X., Gong S., Yang L., *et al.*, Study on the degradation behavior and mechanism of Poly(lactic acid) modification by ferric chloride, *Polymer (Guildf).*, vol. 188, p. 121991,

Feb. 2020, doi: 10.1016/J.POLYMER.2019.121991.

- [39] EPA, Ecological soil screening level for iron interim final, US Environ. Prot. Agency -Off. Solid Waste Emerg., no. November, p. 211, 2005, [Online]. Available: https://rais.ornl.gov/documents/eco-ssl_iron.pdf.
- [40] ASTM International, ASTM D5338-15(2021): Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials Under Controlled Composting Conditions, Incorporating Thermophilic Temperatures, in ASTM Standards, 2021, pp. 1–7.
- [41] International Standard ISO/FDIS 14855-1:2012, Determination of the ultimate aerobic biodegradability of plastic materials under controlled composting conditions Method by analysis of evolved carbon dioxide, Part 1: General method, 2012.
- [42] Kijchavengkul T., Auras R., Rubino M., et al., Development of an automatic laboratoryscale respirometric system to measure polymer biodegradability, *Polym. Test.*, vol. 25, no. 8, pp. 1006–1016, 2006, doi: 10.1016/j.polymertesting.2006.06.008.
- [43] Castro E., Design And Construction Of A Medium-Scale Automated Direct Measurement Respirometric System To Assess Aerobic Biodegradation Of Polymers, 2013.
- [44] Castro-Aguirre E., Auras R., Selke S., *et al.*, Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.
- [45] Mayekar P. C., Castro-Aguirre E., Auras R., et al., Effect of Nano-Clay and Surfactant on the Biodegradation of Poly(Lactic Acid) Films, Polym. 2020, Vol. 12, Page 311, vol. 12, no. 2, p. 311, Feb. 2020, doi: 10.3390/POLYM12020311.
- [46] Satti S. M., Shah A. A., Marsh T. L., and Auras R., Biodegradation of Poly(lactic acid) in Soil Microcosms at Ambient Temperature: Evaluation of Natural Attenuation, Bioaugmentation and Bio-stimulation, J. Polym. Environ., vol. 26, no. 9, pp. 3848–3857, Sep. 2018, doi: 10.1007/s10924-018-1264-x.
- [47] Zhang L., Zhu Y., Zhang J., et al., Impacts of iron oxide nanoparticles on organic matter degradation and microbial enzyme activities during agricultural waste composting, *Waste Manag.*, vol. 95, pp. 289–297, 2019, doi: 10.1016/j.wasman.2019.06.025.
- [48] Mistry A. N., Kachenchart B., Wongthanaroj A., et al., Rapid biodegradation of high molecular weight semi-crystalline polylactic acid at ambient temperature via enzymatic and alkaline hydrolysis by a defined bacterial consortium, *Polym. Degrad. Stab.*, vol. 202, Aug. 2022, doi: 10.1016/J.POLYMDEGRADSTAB.2022.110051.
- [49] Rupali D., Screening and Isolation of Protease Producing Bacteria from Soil Collected from Different Areas of Burhanpur Region (MP) India, *Int. J. Curr. Microbiol. Appl. Sci.*, vol. 4, no. 8, pp. 597–606, 2015, Accessed: Apr. 06, 2022. [Online]. Available: http://www.ijcmas.com.

- [50] Penkhrue W., Khanongnuch C., Masaki K., et al., Isolation and screening of biopolymer-degrading microorganisms from northern Thailand, World J. Microbiol. Biotechnol., vol. 31, no. 9, pp. 1431–1442, Sep. 2015, doi: 10.1007/s11274-015-1895-1.
- [51] Decorosi F., Exana M. L., Pini F., et al., The Degradative Capabilities of New Amycolatopsis Isolates on Polylactic Acid, *Microorganisms*, vol. 7, no. 12, p. 590, Nov. 2019, doi: 10.3390/microorganisms7120590.
- [52] Jarerat A., Tokiwa Y., and Tanaka H., Poly(l-lactide) degradation by Kibdelosporangium aridum, *Biotechnol. Lett.*, vol. 25, no. 23, pp. 2035–2038, Dec. 2003, doi: 10.1023/B:BILE.0000004398.38799.29.
- [53] Jarerat A. and Tokiwa Y., Degradation of Poly(L-lactide) by a Fungus, *Macromol. Biosci.*, vol. 1, no. 4, pp. 136–140, Jun. 2001, doi: 10.1002/1616-5195(20010601)1:4<136::AID-MABI136>3.0.CO;2-3.
- [54] Jarerat A. and Tokiwa Y., Poly(L-lactide) degradation by Saccharothrix waywayandensis, *Biotechnol. Lett.*, vol. 25, no. 5, pp. 401–404, 2003, doi: 10.1023/A:1022450431193.
- [55] Jiang T., Gao C., Ma C., and Xu P., Microbial lactate utilization: Enzymes, pathogenesis, and regulation, *Trends Microbiol.*, vol. 22, no. 10, pp. 589–599, 2014, doi: 10.1016/j.tim.2014.05.008.
- [56] Farhana A. and Lappin S. L., Biochemistry, Lactate Dehydrogenase, *StatPearls*, 2021. https://www.ncbi.nlm.nih.gov/books/NBK557536/.
- [57] Li J., Hu A., Bai S., *et al.*, Characterization and performance of lactate-feeding consortia for reductive dechlorination of trichloroethene, *Microorganisms*, vol. 9, no. 4, pp. 1–18, 2021, doi: 10.3390/microorganisms9040751.
- [58] Santos A. M. dos, Costa J. M., Braga J. K., et al., Lactate as an effective electron donor in the sulfate reduction: impacts on the microbial diversity, *Environ. Technol. (United Kingdom)*, vol. 43, no. 20, pp. 3149–3160, 2022, doi: 10.1080/09593330.2021.1916092.
- [59] Colombo C., Palumbo G., He J.-Z., et al., Review on iron availability in soil: interaction of Fe minerals, plants, and microbes, J. Soils Sediments Sediments, vol. 14, pp. 538– 548, 2014, doi: 10.1007/s11368-013-0814-z.
- [60] He Y., Huang X., Zhang H., et al., Insights into the effect of iron-carbon particle amendment on food waste composting: Physicochemical properties and the microbial community, *Bioresour. Technol.*, vol. 351, no. March, p. 126939, 2022, doi: 10.1016/j.biortech.2022.126939.
- [61] Awasthi M. K., Pandey A. K., Bundela P. S., et al., Co-composting of gelatin industry sludge combined with organic fraction of municipal solid waste and poultry waste employing zeolite mixed with enriched nitrifying bacterial consortium, *Bioresour*. *Technol.*, vol. 213, pp. 181–189, 2016, doi: 10.1016/j.biortech.2016.02.026.
- [62] Mayekar P. C. and Auras R., Accelerating Biodegradation : Enhancing Poly(lactic acid)

Breakdown at Mesophilic Environmental Conditions with Biostimulants, *Macromol. Rapid Commun.*, vol. 2300641, pp. 1–10, 2023, doi: 10.1002/marc.202300641.

APPENDIX 5A: PHYSICOCHEMICAL CHARACTERISTICS

Some compost was collected and sent to the Soil and Plant Nutrient Laboratory at Michigan State University (East Lansing, MI, USA) to evaluate its physicochemical parameters (dry solids, volatile solids, and C/N ratio) as previously described elsewhere [44]. The physicochemical parameters are reported below in **Table A5.1**.

Parameter	Compost		
Dry solids, %	42.5		
Volatile solids, %	41.7		
pH	8.0		
C/N ratio	10.1		
Carbon, %	24.2		
Nitrogen, %	2.42		
Phosphorus, %	1.21		
Potassium, %	3.15		
Calcium, %	5.07 2.82 0.58		
Magnesium, %			
Sodium, %			
Sulfur, %	0.58		
Iron, ppm	9878		
Zinc, ppm	480		
Manganese, ppm	413		
Copper, ppm	107		
Boron, ppm	41		
Aluminum, ppm	6751		

Table A5.1 Physicochemical parameters and total nutrient analysis of compost used in the biodegradation test.

APPENDIX 5B: CO₂ EVOLUTION AND MINERALIZATION OF PLA IN THE PRESENCE OF SKIM MILK, GELATIN, AND ETHYL LACTATE

Figure A5.1 a and b shows the CO₂ evolution and % mineralization of cellulose, skim milk, PLA, cellulose + skim milk, and PLA + skim milk. Cellulose reached mineralization of 87.7% whereas skim milk attained over 100 % of its carbon conversion over a period of 180 days. Since skim milk and cellulose are readily biodegradable and can be easily utilized as a carbon source by the microorganisms present in the compost, no lag phase was observed. Skim milk was added to the compost with the goal of inducing the protease activity of the microbes present. PLA shows similar CO₂ evolution when compared to blank (compost only). This indicates that PLA is still undergoing chemical hydrolysis and is yet to breakdown to M_n of 10k Da, where it can be assimilated by microorganisms.

Due to the compost amendment with skim milk, PLA (PLA + skm1) shows a mineralization of approximately 60%. In order to account for the effect of skim milk, a separate mineralization plot (PLA + skm2) is derived. Around 35% mineralization is observed for the same depicting that the enzymatic degradation of PLA is enhanced due to the presence of skim milk. This is corroborated by the molecular weight analysis (PLA $k = 0.0045 \pm 0.0001$ and PLA-skim milk $k = 0.0053 \pm 0.0003$).



Figure A5.1 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, skim milk, cellulose + skim milk (Cell+skm), PLA + skim milk (PLA+skm) in compost at 37°C.



Figure A5.2 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, gelatin, cellulose + gelatin (Cell+gel), PLA + gelatin (PLA+gel) in compost at 37°C.

Figure A5.2 a and b shows the CO₂ evolution and % mineralization of cellulose, gelatin, PLA, cellulose + gelatin, and PLA + gelatin. Cellulose reached mineralization of 87.7% whereas gelatin attained over 100 % of its carbon conversion over a period of 180 days. Since gelatin and cellulose are readily biodegradable and can be easily utilized as a carbon source by the microorganisms present in the compost, no lag phase was observed.

Gelatin is composed of protein which the microorganisms in compost use for their biochemical process. The microorganisms secrete protease enzyme to digest gelatin which is the same mechanism when PLA is introduced in gelatin amended compost. Due to the compost amendment with skim milk, PLA (PLA + gel1) shows a mineralization of approximately 60%. In order to account for the effect of gelatin, a separate mineralization plot (PLA + gel2) is derived. Though there seems to be negative mineralization, the molecular weight analysis [62] shows that gelatin helps in the enzymatic degradation of PLA. This coupled with the chemical hydrolysis of PLA [5], produces a significant difference with respect to the kinetic rate of degradation (PLA $k = 0.0045 \pm 0.0001$ and PLA-gelatin $k = 0.0060 \pm 0.0002$) [62].



Figure A5.3 Cumulative CO₂ evolution (a) and mineralization (b) of blank, cellulose, PLA, ethyl lactate, cellulose + ethyl lactate (Cell+el), PLA + ethyl lactate (PLA+el) in compost at 37°C.

Figure A5.3 a and b shows the CO₂ evolution and % mineralization of cellulose, ethyl lactate, PLA, cellulose + ethyl lactate, and PLA + ethyl lactate. Cellulose reached mineralization of 87.7% whereas ethyl lactate attained over 100 % of its carbon conversion over a period of 180 days. Since ethyl lactate and cellulose are readily biodegradable and can be easily utilized as a carbon source by the microorganisms present in the compost, no lag phase was observed. Ethyl lactate was used to stimulate the lactate utilizing microbial community present in the compost.

Ethyl lactate evolves around 51.5 g of CO₂, and the corresponding mineralization reaches around 270%. The CO₂ evolution of the amended compost containing both cellulose and ethyl lactate (Cell-el 1) was only around 66.2 g and there was a corresponding mineralization of 182.4%. PLA in the presence of ethyl lactate (PLA-el 1) evolved around 51.6 g of CO₂, followed a similar trend as ethyl lactate over 180 days, and showed maximum mineralization of 105%. A positive mineralization behavior is seen for PLA when the effect of ethyl lactate is accounted for (PLA-el 2), indicating the lactate-stimulating activity of ethyl lactate. This is further confirmed by kinetic rates for PLA alone and PLA in compost amended with ethyl lactate (PLA $k = 0.0045 \pm 0.0001$ and PLA-ethyl lactate $k = 0.0058 \pm 0.0002$) [62].

PRESENCE OF FE₃O₄ NANOPOWDER, SKIM MILK + FE, AND ETHYL



LACTATE + FE

Figure A5.4 Cumulative CO2 evolution (a) and Mineralization (b) of blank + Fe, cellulose + Fe, PLA + Fe, skim milk + Fe , cellulose + skim milk + Fe (Cell+skm+Fe), PLA + skim milk + Fe (PLA+skm+Fe) in compost at 37°C. (c) represents the normalized Mn reduction as a function of time for PLA in control compost and compost biostimulated by gelatin, and gelatin + Fe. The experimental data was fitted using a first-order reaction of the form Mn /Mno = e(-kt), where Mno is the initial Mn, k is the rate constant, and t is the time. The inset shows the k-fitted values. Values in the column with different lowercase letters are statistically different (α = 0.05 Tukey-Kramer Test). (d) and (e) depict DSC thermograms for PLA + Fe and PLA + Fe in compost biostimulated by skim milk. (f) and (g) shows the MWD of PLA + Fe in compost and compost biostimulated with skim milk.

Figure A5.4 a and b shows the CO₂ evolution and mineralization of cellulose + Fe, skim milk + Fe, PLA + Fe, cell + skm + Fe, and PLA + skm + Fe in compost at 37°C. Skm + Fe shows around 60.1 g of CO₂ evolution and mineralization of 247.8 % in 180 days. Skim milk was combined with Fe to target chemical hydrolysis and enzymatic degradation steps. The CO₂ evolution in this case (Cell+skm+Fe) sees a higher production of 54.5 g, which is as expected and higher compared to the individual values for cellulose + Fe (44.5 g). Cell+skm+Fe 1 shows a mineralization of 121.6% whereas Cell+skm+Fe 2 after accounting for skm+Fe shows a mineralization of 99.4% indicating that the presence of skim milk in no way affects the degradation of cellulose.

To understand the influence of skim milk + Fe on PLA degradation, PLA was introduced in the compost amended with skim milk and Fe. The bioreactor containing both PLA and skim milk + Fe (PLA+skm+Fe 1) shows CO₂ evolution of around 36.4 g and maximum mineralization of $35.3 \$ by the end of the test. Improved mineralization is observed as opposed to no CO₂ evolution for PLA alone without any biostimulation of compost. The effect of skim milk + Fe on PLA degradation is calculated by plotting the mineralization of PLA+skm+Fe 2 (subtracting skim milk + Fe). The negative mineralization does not necessarily indicate the absence of skim milk's protease activity in PLA's enzymatic degradation. The significant difference in the evolution of X_c from 28.3% to 31.4% for PLA, and from 28.3% to 37.9% for PLA samples biostimulated with skim milk + Fe, as seen in Figure S4 d and e, respectively further shows the improvement in the enzymatic degradation of PLA due to the presence of skim milk + Fe. Skim milk acts as a precursor for protease activity and the broadening and change in the intensity of peaks as seen in **Figure A5.4** f and g enforces that the addition of skim milk and Fe does enhance PLA degradation.



Figure A5.5 Cumulative CO₂ evolution (a) and Mineralization (b) of blank + Fe, cellulose + Fe, PLA + Fe, ethyl lactate + Fe , cellulose + ethyl lactate + Fe (Cell+el+Fe), PLA + ethyl lactate + Fe (PLA+el+Fe) in compost at 37°C. (c) represents the normalized M_n reduction as a function of time for PLA in control compost and compost biostimulated by gelatin, and gelatin + Fe. The experimental data was fitted using a first-order reaction of the form M_n / $M_{no} = e^{(-kt)}$, where M_{no} is the initial M_n , k is the rate constant, and t is the time. The inset shows the k-fitted values. Values in the column with different lowercase letters are statistically different ($\alpha = 0.05$ Tukey-Kramer Test). (d) and (e) depict DSC thermograms for PLA + Fe in compost biostimulated by ethyl lactate. (f) and (g) shows the MWD of PLA + Fe in compost and compost biostimulated with ethyl lactate.

Figure A5.5 a and b shows the CO_2 evolution and mineralization of cellulose + Fe, ethyl lactate + Fe, PLA + Fe, cell + el + Fe, and PLA + el + Fe in compost at 37°C. El + Fe shows around 54.5 g of CO_2 evolution and mineralization of 288.1 % in 180 days. Ethyl lactate was combined with Fe to target chemical hydrolysis and lactate utilizing microbes in compost. The CO_2 evolution in this case (Cell+el+Fe) sees a higher production of 62.2 g, which is as expected and higher compared to the individual values for cellulose + Fe (44.5 g). Cell+el+Fe 1 shows a mineralization of 156.7% whereas Cell+el+Fe 2 after accounting for el+Fe shows a mineralization of 75.1% indicating that the presence of ethyl lactate does not affect the degradation of cellulose.

To understand the influence of ethyl lactate + Fe on PLA degradation, PLA was introduced in the compost amended with ethyl lactate and Fe. The bioreactor containing both PLA and ethyl lactate + Fe (PLA+el+Fe 1) shows CO₂ evolution of around 46.5 g and maximum mineralization of 88.8 % by the end of the test. Enhanced mineralization is observed as opposed to no CO₂ evolution for PLA alone without any biostimulation of compost. The effect of ethyl lactate + Fe on PLA degradation is calculated by plotting the mineralization of PLA+el+Fe 2 (subtracting ethyl lactate + Fe). The negative mineralization does not necessarily indicate the absence of ethyl lactate's lactate stimulating microbial activity in PLA's enzymatic degradation. The significant difference in the evolution of X_c from 28.3% to 31.4% for PLA, and from 28.3% to 39.9% for PLA samples biostimulated with ethyl lactate + Fe, as seen in Figure S5 d and e, respectively further shows the improvement in the enzymatic degradation of PLA due to the presence of ethyl lactate + Fe. As seen in Figure S5 f and g ethyl lactate stimulates the lactate utilizing microbial community in the compost which aids in the degradation of PLA.

CHAPTER 6: ENHANCING BIODEGRADATION OF POLY(LACTIC ACID) IN MESOPHILIC AND THERMOPHILIC ENVIRONMENTAL CONDITIONS: THE ROLE OF PROTEINASE K AS A PRETREATMENT

6.1 Abstract

Plastic pollution poses a significant environmental challenge, prompting a shift towards biodegradable and biobased polymers such as poly(lactic acid) (PLA) as sustainable alternatives. Despite PLA's potential, a slow abiotic process hinders its degradation in composting environments, influenced by factors like high number molecular weight and crystallinity. Enzymatic degradation, particularly by PLA depolymerase like proteinase K, offers a promising solution to accelerate the degradation of PLA during composting if used as pretreatment. This study aimed to investigate the enzymatic degradation of PLA films pretreated with proteinase K and degraded in simulated home and industrial composting conditions. PLA films were pretreated with proteinase K at 37°C and 58°C at different times. PLA films treated in underwater conditions at 37°C were used as control. The accelerated CO_2 evolution and mineralization of the pretreated PLA films highlight the effectiveness of proteinase K pretreatment. This study shows the potential of enzymatic pretreatment to enhance PLA biodegradability at mesophilic and thermophilic temperatures, offering insights into sustainable waste management strategies to reduce the biodegradation rate of PLA films and packaging.

6.2 Introduction

Plastic pollution in the environment is a looming crisis. Given the crucial role that plastics play in creating single-use plastic packaging and the generation of white pollution, using biodegradable polymers in day-to-day life to combat mismanaged waste is seen as a welcoming solution. So, the commercial plastic industry is adopting biobased and biodegradable plastics as an environmentally friendly alternative to fossil-based conventional polymers. Poly(lactic acid) - PLA - is one such polymer and is considered a major substitute for conventional fossil polymers because of its designed biodegradable nature when exposed to the proper set of composting conditions. So, the use of PLA has increased rapidly to support the growth of the circular sustainable bioeconomy, accompanied by the ongoing development in its production process.

Although PLA and its products are labeled as industrial compostable, it is essential to note that the time frame for the degradation process is longer as compared to the readily biodegradable organic waste fraction such as food, starch, and cellulose when PLA is collected together as a part of municipal solid waste and directed to an industrial composting facility [1]. This scenario arises because of the slow abiotic phase, which determines its degradation rate and is further influenced by molecular weight (M_n) and crystallinity (X_c). PLA is subjected to higher temperatures of 58°C-70°C in industrial composting facilities, which makes it able to degrade. However, its degradation is recalcitrant in soil and home composting environments where temperatures are much lower in the mesophilic range.

In environments such as soil and home composting, chemical hydrolysis proceeds at a prolonged rate due to the lower temperatures encountered, and biodegradation is mainly governed by biotic enzymatic degradation [2]. Abiotic degradation proceeds by reducing the M_n to values lower enough that the biotic enzymatic degradation can take over and accelerate the breakdown process. Biotic enzymatic degradation involves the breakdown of polymer chains into small molecules that the microorganisms can quickly assimilate. The biodegradation at such mesophilic conditions relies heavily on the depolymerization stage, wherein the microorganisms release the enzymes. For hydrolyzable polymers such as PLA, the main enzymes belong to the hydrolases (EC 3) class of extracellular enzymes reported for depolymerizing aliphatic polyesters. Depending on the substrate specificity, the PLA

depolymerases are categorized as protease and lipase types. Both classes of PLA depolymerases employ the serine hydrolase catalytic mechanisms, yet their different stereochemistry at the catalytic sites provides a structural basis for different specificities for substrates [3].

Proteinase K (3.4.21.64) and proteases (3.4.21.112) belong to the serine endo peptidases (3.4.21), which catalyze bond scission in the middle of the substrate chain and are also known to hydrolyze polyesters such as PLA. Proteinase K and proteases show enzymatic activity specifically for PLA containing L-lactate compared to PDLA and PDLLA. These enzymes can recognize PLA's repeating L-lactic acid unit as structurally homologous to the proteins composed of L-aminoacids [4]. Williams [5] first reported the hydrolysis of PLA by proteinase K sourced from *Tritirachium album*. Since then, the degradation of PLA by different enzymes has been studied substantially [2]. The commercially available proteinase K has been used extensively to study PLA enzymatic degradation.

Unlike mechanical and chemical recycling, enzymatic degradation consumes less energy, needs fewer chemicals, or generates less harmful compounds threatening the environment. So, enzymatic degradation can assist in further developing the degradation of PLA at lower temperatures. This study aimed to evaluate the enzymatic degradation of PLA by pretreating PLA with proteinase K enzyme in a buffer solution at 37°C and 58°C for different time intervals before introducing it into a simulated compost environment at 37°C to replicate home composting and at 58°C to simulate industrial composting. The CO_2 evolution of the samples was recorded, and the changes in M_n and X_c throughout the test duration were measured to understand how the pretreatment with proteinase K modified the biodegradation of PLA in the different composting conditions.

6.3 Materials and methods

6.3.1 Materials

Crystalline PLA films ranging 25 microns in thickness were obtained from EarthFirst (Columbus, OH, USA). The crystalline PLA films were heated at 180°C, for 10 mins and fast quenched in dry ice for 10 minutes to remove any residual crystallinity. Proteinase K from *Tritirachium album* (29.3 kDa) from Syd labs (Hopkinton, MA) was procured and used as received for enzymatic degradation. HPLC grade Tetrahydrofuran (THF) was obtained from (Sigma-Aldrich, St. Louis, MO, USA). All the chemicals were used as received.

6.3.2 Preparation of buffer and enzyme solution

Buffer powder (MSU, East Lansing, MI) was mixed in distilled water to prepare the buffer solution. NaOH pellets were mixed in the distilled water and added to the buffer solution until the pH reached 8.5. The proteinase K enzyme was added to the distilled water to get an aqueous solution of proteinase K (500 $\mu g/mL$). Based on the previous reports, the concentration was calculated and set to 500 $\mu g/mL$ [6,7].

6.3.3 Enzymatic degradation by externally adding Proteinase K

The enzymatic degradation of PLA films was performed by externally adding the proteinase K solution and buffer solution to the PLA films (1 cm x 1 cm squares). The reaction solution was incubated at 37°C for 7 and 10 days and at 58°C for 2 and 5 days with constant stirring to determine the effect of exposing PLA films to Proteinase K at different intervals.

6.3.4 Hydrolysis experiment

A hydrolysis test was also run as a control to understand the hydrolytic degradation of the PLA films. The test procedure was adapted from ASTM D4754-18 [8]. The PLA films were cut into 1 cm x 1 cm and introduced into the beaker with HPLC-grade water (VWR, Radnor, PA, USA) at 37°C for 10 days. At the end of the test, the samples were dried and stored for further analysis.

6.3.5 Biodegradation in vermiculite

The biodegradation of PLA and the effectiveness of pretreating PLA films with proteinase K in inoculated vermiculite on the degradation of PLA were evaluated under aerobic simulated mesophilic and thermophilic conditions using two direct measurement respirometric (DMR) systems [9–11]. Shortly, the system included a non-dispersive infrared gas analyzer (NDIR) (Li-COR[®] LI-820, Lincoln, NE, USA) which measures the CO₂ concentration. The DMR system chamber was maintained at a temperature of 37 ± 2 °C and 58 ± 2 °C and provided with air at a relative humidity (RH) of $50\% \pm 5\%$. A flow rate of CO₂-free air (concentration <30 ppm to establish a low baseline) was controlled at 40 ± 2 cm³/min. Additional detailed information about the DMR equipment can be found in another source [12].

Mature compost obtained from the MSU composting facility was sifted using a 10-mm screen to get rid of any considerable debris or chunks present and then conditioned at 37 ± 2 °C and 58 ± 2 °C until use. This compost extract was then amalgamated with a mineral solution in a 1:1 ratio, resulting in the inoculum solution. Detailed information regarding the preparation of the mineral solution can be found elsewhere [13]. Inoculated vermiculite provides the benefit of very low CO₂ evolution from the blank used as the baseline. Deionized water was used to adjust the moisture content of the vermiculite during testing to 50%. The resulting inoculated vermiculite mixture was sent to the Soil and Plant Testing Laboratory at the University of Missouri (Columbia, MO, USA) to determine the physicochemical parameters of the media. Data regarding the solid analysis is presented in **Table A6.1**, Appendix 6A. The bioreactors were packed with 400 g of inoculated vermiculite, and 8 g of all the samples were tested in triplicate and positive control (cellulose). Blank (only compost) was used as the baseline.

6.3.6 Size exclusion chromatography (SEC)

As described elsewhere, the M_n and molecular weight distribution (MWD) of PLA for the control and each biostimulant treatment were measured using SEC (Waters Corp., Milford, MA, USA) [14]. Shortly, PLA samples weighing approximately 10 mg were retrieved at predetermined intervals and dissolved in 5 mL of THF. A temperature of 35°C and a 1 mL/min flow rate were maintained during testing. The Mark-Houwink constants of K =0.000174 dL/g and a = 0.736 were used to determine the absolute M_n , M_w , and MWD of the PLA samples. Data analysis was carried out using Waters BreezeTM2 software from Waters.

6.3.7 Differential scanning calorimetry

A DSC model Q100 (TA Instruments, New Castle, DE, USA), was used to determine the glass transition temperature (T_g) , (T_m) , crystallization temperature (T_c) , and crystallinity (X_c) for the PLA samples retrieved from the regular and biostimulated compost. PLA samples weighing 5 - 10 mg were packed in aluminum pans, cooled to -5° C, and then subjected to a heating cycle to reach 210°C at a ramp rate of 10°C/min. This information was used to evaluate the evolution of X_c . The cooling was achieved using a nitrogen cooling system that maintained the purge flow rate at 70 mL/min. The degree of crystallinity was estimated using equation (6.1):

$$\chi_c \% = \frac{\Delta H_m - \Delta H_c}{\Delta H_m^0} x \ 100 \tag{6.1}$$

where ΔH_m is the heat of fusion, ΔH_c is cold crystallization enthalpy and ΔH_m^0 is the heat of fusion for 100% crystalline pure PLA (93 J/g) [15].

6.3.8 Statistical analysis

The statistical analysis was conducted using MINITABTM software (Minitab Inc., State College Park, PA, USA). The statistical significance at p < 0.05 was evaluated using one-way ANOVA and Tukey-Kramer test. Data is reported as means ± standard deviation.

6.4 Results and discussion

The CO₂ evolution of PLA control (PLA), PLA hydrolyzed films at 37°C for 10 days (PLA 37 hydro 10 D), and enzymatically pretreated PLA samples over 7 and 10 days at 37°C (PLA-37 proteinase K 7D and PLA-37 proteinase K 10D) and PLA samples over 2 and 5 days at 58°C (PLA-58 proteinase K 2D and PLA-58 proteinase K 5D) were tracked at simulated mesophilic conditions (37°C) and thermophilic (58°C) conditions using inoculated vermiculite as media, respectively.

6.4.1 M_n reduction and crystallinity evolution for PLA and PLA pretreated films

The M_n and X_c of PLA control, PLA hydrolyzed films at 37°C for 10 days, PLA pretreated with proteinase K at 37°C for 7 and 10 days, and PLA pretreated with proteinase K at 58°C for 2 and 5 days was determined to understand the effect of proteinase K pretreatment. **Table 6.1** presents the values for the same before introducing the samples in the DMR chamber for further biodegradation testing.

Films	Treatment time (days)	37 ± 2 °C		58 ± 2 °C	
		Molecular weight (<i>M</i> _n , kDa)	Crystallinity (Xc)	Molecular weight (<i>M</i> _n , kDa)	Crystallinity (Xc)
PLA Control	0	$79 \pm 3.0^{\mathrm{a}}$	0	$79 \pm 3.0^{\mathrm{a}}$	0
PLA hydrolysis	10	$75 \pm 2.8^{\mathrm{a}}$	2.5 ± 2.0^{a}	$75 \pm 2.8^{\mathrm{a}}$	2.5 ± 2.0^{a}
PLA proteinase K	7	$38 \pm 1.0^{\mathrm{b}}$	$7.1 \pm 1.5^{\mathrm{b}}$	-	-
	10	$17 \pm 0.9^{\circ}$	$12 \pm 1.3^{\circ}$	-	-
	2	-	-	$61 \pm 3.0^{\mathrm{b}}$	$23 \pm 3.5^{\mathrm{b}}$
	5	-	-	$47 \pm 2.9^{\circ}$	$43 \pm 3.0^{\circ}$

Table 6.1 Initial characterization of PLA, PLA hydrolyzed films, and PLA films treated with proteinase K at 37°C and 58°C.

A significant difference was seen for M_n for PLA control, PLA hydrolyzed films, and PLA films pretreated with proteinase K at both 37°C and 58°C. Apart from PLA and PLA hydrolyzed films, again, a significant difference was seen for X_c for PLA proteinase K pretreated films. This indicates that proteinase K enzymatically hydrolyzes PLA and reduces it to a lower M_n . The values differ for the different pretreatment time intervals, as seen in **Table 6.1** for 37°C and 58°C. The difference can be associated with the optimal activity of proteinase K at 37°C. The PLA hydrolyzed films do not show any difference in M_n and X_c _{compared} to PLA control films because of the slower chemical hydrolysis at 37°C.

6.4.2 CO₂ evolution and mineralization of PLA and PLA pretreated films in inoculated vermiculite

Figure 6.1 a and b present the CO₂ evolution and % mineralization of the blank (vermiculite only), cellulose, PLA control films, PLA hydrolyzed films at 37°C for 10 days, and PLA pretreated with proteinase K enzyme for 7 days and 10 days, respectively in inoculated vermiculite at 37 ± 2 °C. Blank showed CO₂ evolution of 2.4 g by the end of day 100.


Figure 6.1 CO₂ evolution and % mineralization of PLA, PLA 37hydro 10D, PLA-37 proteinase K 7D, and PLA-37 proteinase K 10D films in compost 37 °C (a & b) and PLA, PLA 37hydro 10D, PLA-58 proteinase K 2D, PLA-58 proteinase K 5D films in compost at 58 °C (c & d). The shade in the background for each material represents the standard error between replicates.

Cellulose showed CO_2 evolution of 14.3 g and mineralization of *c*. 94% by day 100. PLA films subjected to hydrolysis at 37°C for 10 days showed a very low CO_2 evolution of 2.5 g, whereas PLA control films showed only 2.7 g. A lag phase is observed for both PLA control and PLA hydrolyzed films, which could mean that PLA offers a physical hydrophobic barrier to water and air before hydrolysis, and the microbes have difficulty accessing the carbon source as nutrients. The absence of mineralization does not necessarily imply the absence of hydrolytic degradation or enzymatic activity due to the action of extracellular enzymes secreted by the microbes but more like inhibition of the microbial activity due to the hydrophobic layer barrier created by the presence of the high M_w PLA. The lower values for both indicate that PLA is still undergoing hydrolysis since PLA is exposed to temperatures lower than its T_g (c. 60 °C). PLA segments have little to no mobility and are not flexible below T_g , preventing diffusion or attack by water. Since the initial and rate-limiting step in PLA degradation is chemical hydrolysis, the lack of chain scission accompanied by the hydrophobic barrier provided by PLA can explain the lack of mineralization both for PLA control and PLA hydrolyzed films.

For PLA films pretreated with proteinase K enzyme for 7 days, CO₂ evolution of 6.6 g and mineralization of c. 28% was seen. In contrast, for PLA films pretreated with proteinase K enzyme for 10 days, 7.7 g of CO_2 was evolved, and c. 35% mineralization was observed by day 100. No lag phase was observed for proteinase K pretreated films, indicating that the pretreatment with proteinase K reduces PLA to M_n of ~ 10,000 Da, wherein the biotic degradation stage kicks in, and the microorganisms can utilize PLA for their biochemical processes. Figure 6.1c and d present the CO₂ evolution and % mineralization of the blank (vermiculite only), cellulose, PLA control films, PLA hydrolyzed films at 37°C for 10 days, and PLA pretreated with proteinase K enzyme for 2 days and 5 days, respectively in inoculated vermiculite at 58 ± 2 °C. The blank showed a maximum CO₂ evolution of 2.6 g at day 60. The positive control, cellulose showed CO_2 evolution of 13.1 g and mineralization of c. 84% at day 60. No lag phase is observed since cellulose is readily biodegradable and a food source for microorganisms. Cellulose is degraded by the action of a battery of enzymes that work simultaneously and synergistically. Cellulases catalyze the hydrolysis of 6-1,4-linkages in the cellulose [1,16]. The action of exoglucanases and endoglucanases on the ends and at random internal sections of cellulose's amorphous region produces varying lengths of cellooligosaccharides, which are then hydrolyzed by glucosidases to produce glucose [17]. Glucose

is finally converted to CO₂ through a series of further cycles. Previous research has shown fungi, a few bacteria species, and actinomycetes in compost and the soil environment produce cellulase, which are involved in the degradation of cellulose [18–22].

PLA control films show a lag phase of around 35 days because of the initial abiotic hydrolysis phase, wherein PLA ester bonds are broken down mainly due to the hydrolysis by water. The high M_n PLA chains are cleaved to produce low M_n oligometric used by the microorganisms in the inoculated vermiculite, releasing CO_2 and water, which can be observed during the biotic degradation phase. A CO_2 evolution of 7.0 g and mineralization of c. 28.5% was observed for PLA at 58 °C. PLA films subjected to hydrolysis at 37°C for 10 days (Figure 6.1b) showed a CO₂ evolution of 5.7 g and mineralization of c. 20.5%. PLA hydrolyzed films followed a similar trend to PLA control films; however, the mineralization is slightly lower, maybe due to the higher initial crystallinity of PLA 37 hydro 10D films. PLA films pretreated with proteinase K enzyme at 37°C for 10 days (PLA-37 proteinase K 10D) evolved 9.3 g of CO₂ and showed a mineralization of c. 44.3% at 58°C. The improved degradation can be attributed to the pretreatment with proteinase K at 37°C for 10 days. Proteinase K belongs to serine endo peptidases that preferentially catalyze bond scission in the middle of the substrate chain. The enzymes showing activity towards PLA belong to the protease type group or peptidases. They can recognize the repeating L -lactic acid unit of PLA as the natural homolog L -alanine unit of silk fibroin [2,4]. Though proteinase K is primarily known to cleave peptide bonds, it can exhibit a certain level of non-specific hydrolytic activity towards PLA.

The PLA films were also pretreated with proteinase K enzyme at 58°C for 2 and 5 days, respectively (PLA-58 proteinase K 2D and PLA-58 proteinase K 5D) to evaluate the effect of temperature and time on the efficiency of proteinase K's enzymatic activity. PLA films pretreated for 2 days showed CO₂ evolution of 12.2 g and mineralization of *c*. 62.3%, whereas PLA films pretreated for 5 days evolved 7.4 g of CO₂ and exhibited a mineralization of *c*. 31%.

All the films, when pretreated with proteinase K, followed similar trends and showed considerable mineralization in comparison to the PLA control and PLA hydrolyzed films.

6.5 Conclusion

PLA films were pretreated with proteinase K to evaluate the effect of pretreatment on the overall biodegradation of PLA films in simulated compost conditions (inoculated vermiculite.) Subjecting PLA films to proteinase K pretreatment showed significant improvements in CO₂ evolution and mineralization rates, indicating accelerated degradation compared to untreated PLA films. These findings suggest that enzymatic pretreatment offers a promising approach to overcoming the challenges faced with PLA degradation, particularly in composting facilities operating at lower temperatures. Furthermore, the study highlights the potential of enzymatic hydrolysis as a sustainable and environmentally friendly method for managing PLA waste. Future research efforts could optimize pretreatment conditions and explore the scalability of enzymatic degradation processes for large-scale PLA waste management applications.

6.6 Acknowledgments

P.M. acknowledges the School of Packaging at Michigan State University; R.A. acknowledges the USDA National Institute of Food and Agriculture and Michigan State University AgBioResearch, Hatch project number MICL02665, for partial study support.

REFERENCES

- [1] Mayekar P. C., Limsukon W., Bher A., and Auras R., Breaking It Down: How Thermoplastic Starch Enhances Poly(lactic acid) Biodegradation in Compost–A Comparative Analysis of Reactive Blends, ACS Sustain. Chem. Eng., vol. 11, no. 26, pp. 9729–9737, 2023, doi: 10.1021/acssuschemeng.3c01676.
- [2] Bher A., Mayekar P. C., Auras R. A., and Schvezov C. E., Biodegradation of Biodegradable Polymers in Mesophilic Aerobic Environments, *Int. J. Mol. Sci.*, vol. 23, no. 20, 2022, doi: 10.3390/ijms232012165.
- [3] Sourkouni G., Jeremić S., Kalogirou C., *et al.*, Study of PLA pre-treatment, enzymatic and model-compost degradation, and valorization of degradation products to bacterial nanocellulose, *World J. Microbiol. Biotechnol.*, vol. 39, no. 6, pp. 1–13, Jun. 2023, doi: 10.1007/S11274-023-03605-4/FIGURES/8.
- [4] Tokiwa Y. and Jarerat A., Biodegradation of poly(l-lactide), *Biotechnol. Lett.*, vol. 26, no. 10, pp. 771–777, May 2004, doi: 10.1023/B:BILE.0000025927.31028.e3.
- [5] Williams D. F., Enzymic Hydrolysis of Polylactic Acid, *Eng. Med.*, vol. 10, no. 1, pp. 5– 7, Jan. 1981, doi: 10.1243/EMED_JOUR_1981_010_004_02.
- [6] Huang Q., Hiyama M., Kabe T., et al., Enzymatic Self-Biodegradation of Poly(L -lactic acid) Films by Embedded Heat-Treated Immobilized and Proteinase K. Biomacromolecules, vol. 21,no. 8, pp. 3301-3307, Aug. 2020,doi: 10.1021/acs.biomac.0c00759.
- [7] Yamashita K., Kikkawa Y., Kurokawa K., and Doi Y., Enzymatic Degradation of Poly(L-lactide) Film by Proteinase K: Quartz Crystal Microbalance and Atomic Force Microscopy Study, *Biomacromolecules*, vol. 6, no. 2, pp. 850–857, Mar. 2005, doi: 10.1021/bm049395v.
- [8] ASTM International, ASTM Standard D4754-18, Standard Test Method for Two-Sided Liquid Extraction of Plastic Materials Using FDA, 2018. doi: 10.1520/D4754-98R03.2.
- [9] ASTM International, ASTM D5338-15(2021): Standard Test Method for Determining Aerobic Biodegradation of Plastic Materials Under Controlled Composting Conditions, Incorporating Thermophilic Temperatures, in *ASTM Standards*, 2021, pp. 1–7.
- [10] International Standard ISO/FDIS 14855-1:2012, Determination of the ultimate aerobic biodegradability of plastic materials under controlled composting conditions Method by analysis of evolved carbon dioxide, Part 1: General method, 2012.
- Kijchavengkul T., Auras R., Rubino M., *et al.*, Development of an automatic laboratory-scale respirometric system to measure polymer biodegradability, *Polym. Test.*, vol. 25, no. 8, pp. 1006–1016, 2006, doi: 10.1016/j.polymertesting.2006.06.008.
- [12] Castro E., Design And Construction Of A Medium-Scale Automated Direct Measurement Respirometric System To Assess Aerobic Biodegradation Of Polymers, 2013.

- [13] Castro-Aguirre E., Auras R., Selke S., et al., Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.
- [14] Mayekar P. C., Castro-Aguirre E., Auras R., et al., Effect of nano-clay and surfactant on the biodegradation of poly(lactic acid) films, *Polymers (Basel)*., vol. 12, no. 2, pp. 1– 16, 2020, doi: 10.3390/polym12020311.
- [15] Mayekar P. C., Castro-Aguirre E., Auras R., et al., Effect of Nano-Clay and Surfactant on the Biodegradation of Poly(Lactic Acid) Films, Polym. 2020, Vol. 12, Page 311, vol. 12, no. 2, p. 311, Feb. 2020, doi: 10.3390/POLYM12020311.
- [16] Wilson D. B., Microbial diversity of cellulose hydrolysis, Curr. Opin. Microbiol., vol. 14, pp. 259–263, 2011, doi: 10.1016/j.mib.2011.04.004.
- [17] Lakhundi S., Siddiqui R., and Khan N. A., Cellulose degradation: A therapeutic strategy in the improved treatment of Acanthamoeba infections, *Parasites and Vectors*, vol. 8, no. 1, pp. 1–16, Jan. 2015, doi: 10.1186/S13071-015-0642-7/FIGURES/8.
- [18] Leppänen I., Vikman M., Harlin A., and Orelma H., Enzymatic Degradation and Pilot-Scale Composting of Cellulose-Based Films with Different Chemical Structures, J. Polym. Environ., vol. 28, pp. 458–470, 1234, doi: 10.1007/s10924-019-01621-w.
- [19] Béguin P. and Aubert J.-P., The biological degradation of cellulose, FEMS Microbiol. Rev., vol. 13, no. 1, pp. 25–58, Jan. 1994, doi: 10.1111/j.1574-6976.1994.tb00033.x.
- [20] Sukumaran R. K., Singhania R. R., and Pandey A., Microbial cellulases Production, applications and challenges, *J. Sci. Ind. Res.*, vol. 64, pp. 832–844, 2005.
- [21] Doi R. H., Cellulases of Mesophilic Microorganisms Cellulosome and Noncellulosome Producers, Ann. N. Y. Acad. Sci., vol. 1125, no. 1, pp. 267–279, 2008, doi: 10.1196/annals.1419.002.
- [22] Eida M. F., Nagaoka T., Wasaki J., and Kouno K., Isolation and Characterization of Cellulose-decomposing Bacteria Inhabiting Sawdust and Coffee Residue Composts, *Microbes Environ.*, vol. 27, no. 3, pp. 226–233, 2012, doi: 10.1264/JSME2.ME11299.
- [23] Castro-Aguirre E., Auras R., Selke S., et al., Insights on the aerobic biodegradation of polymers by analysis of evolved carbon dioxide in simulated composting conditions, *Polym. Degrad. Stab.*, vol. 137, pp. 251–271, Mar. 2017, doi: 10.1016/j.polymdegradstab.2017.01.017.

APPENDIX 6A: PHYSICOCHEMICAL CHARACTERISTICS

Some inoculated vermiculite was collected and sent to the Soil and Plant Testing Laboratory at the University of Missouri (Columbia, MO, USA) to evaluate its physicochemical parameters (dry solids, volatile solids, and C/N ratio) as described elsewhere [23]. The physicochemical parameters are reported below in **Table A6.1**.

Parameter	Compost
Dry solids, %	20.0
Volatile solids, %	4.53
pH	6.0-8.0
C/N ratio	3.57
Carbon, %	0.18
Nitrogen, %	0.05
Phosphorus, %	0.06
Potassium, %	3.15
Calcium, %	0.24
Magnesium, %	2.44
Sodium, %	0.15
Sulfur, %	0.07
Iron, ppm	6811
Zinc, ppm	17
Manganese, ppm	60
Copper, ppm	59
Boron, ppm	24
Aluminum, ppm	4.05

Table A6.1 Physicochemical parameters and total nutrient analysis of compost used in the biodegradation test.

CHAPTER 7: CONCLUSIONS AND RECOMMENDATIONS FOR FUTURE WORK

7.1 Overall conclusion

Biobased, biodegradable polymers have emerged as a sustainable alternative to the ever-growing escalating waste management and disposal problems due to the increased fossil-based plastic production. Poly(lactic acid) - PLA, has garnered significant attention due to its biobased renewable origin, low environmental footprint, compostable nature, designable biodegradability, and because of its competitive price and expansion in different sectors such as packaging, plasticulture, and medical industries. Despite being biodegradable and compostable, the longer biodegradation timeframes associated with its degradation compared to the readily degradable organic fractions such as starch and cellulose have hindered PLA acceptance widely.

This dissertation addresses this issue to enhance PLA biodegradation under mesophilic and thermophilic conditions, to expand its application to different environments such as soil, and home/backyard composting, and to improve/ achieve a similar degradation timeframe with that of organic matter encountered in industrial composting settings.

Chapter 2 of this dissertation provides an in-depth literature review of the biodegradation of biodegradable polymers under mesophilic conditions, the different steps involved in the process, the role of enzymes, the different test methods used to determine degradation, key factors affecting the kinetic degradation rate. It also summarizes the different approaches and combines various factors to provide a holistic understanding of the complex biodegradation phenomena of biodegradable polymers, including PLA.

Chapter 3 of this dissertation investigated the reactive blending of PLA with thermoplastic starch (TPS) as one of the methods to accelerate biodegradation under mesophilic conditions, thereby simulating the home composting setting. The results revealed that reactive blending ensured uniform dispersion of TPS into the PLA matrix. The presence of TPS as a food source for the microorganisms in the compost eliminated the extended lag phase caused by the chemical hydrolysis process wherein the breakdown of PLA chains to oligomers, dimers, and monomers, and a major reduction in the molecular weight is achieved. Since TPS could be used immediately, structural imperfections were created in PLA, further enhancing its biotic degradation rate. Without any blending, PLA underwent a long lag phase, which lasted until the end of the test, which was 180 days. The test data from 58°C biodegradation was also compared to illustrate and highlight the role of TPS in improving the overall degradation of PLA at thermophilic conditions.

Chapters 4 and 5 explain the use of a biostimulation technique to enhance the enzymatic biodegradation of PLA at 37°C. Different biostimulants, Fe₃O₄ nano-powder, skim milk, gelatin, and ethyl lactate, were identified to target the chemical hydrolysis and the biotic enzymatic degradation steps to facilitate and accelerate PLA degradation. Fe₃O₄ nano-powder, skim milk, gelatin, and ethyl lactate were introduced into the compost media at 37°C, and CO₂ evolution, M_n , and X_c of PLA were monitored through 180 days test duration. The introduction of individual biostimulants and a combination showed improved enzymatic degradation of PLA compared to no using biostimulants. Adding biostimulants mainly enhanced the enzymatic biotic phase, reflected in the improved kinetic degradation rate.

Chapter 6 focused on a pretreatment approach that involved pretreating PLA with proteinase K enzyme to investigate the effect of enzymes in accelerating enzymatic degradation. For the most part, in the reported research, enzymatic degradation has been conducted for a minimal amount of PLA in a liquid media, wherein PLA was the only isolated carbon source. In this chapter, PLA films were first enzymatically treated with proteinase K at 37°C and 58°C for different intervals to achieve different M_n and X_c . After pretreatment, the PLA films were introduced in an inoculated vermiculite solid matrix at 37° and 58°C to replicate the home and industrial composting environment. The results depicted improved CO₂ evolution and a shortened lag phase for the treated samples compared to the untreated PLA films.

The different approaches mentioned above offered practical strategies to accelerate PLA biodegradation and advance the understanding of the complex biodegradation process at lower temperatures, which are encountered in real end-of-life scenarios. The approaches utilized above also provide a safe way of degrading PLA, unlike incineration and landfilling, which bring many problems, including generating toxic gases and using higher temperatures, thereby increasing pollution, energy consumption, and the leaching of microplastics into the environment.

7.2 **Recommendations for future work**

The biodegradation of biodegradable polymers in different environments is a complex phenomenon that requires an array of expertise to investigate, understand, and advance the complex involved processes.

Starch has been extensively researched and studied in terms of polymer. Though adding starch enhances the biodegradation of PLA, several setbacks are encountered while processing starch into the PLA matrix on a larger scale. The current study only focused on including starch as one of the additives and focused more on the biodegradation aspect of the resulting blend. But overall, the blend should be able to provide the necessary properties expected of a package or film (e.g., food packaging), until the end-of-life cycle. Different additives and properties enhancement should be further considered and studied to provide a holistic approach.

One of the main uncontrollable parameters encountered in the home composting process is temperature due to the unreliable process control. The change from higher to lower temperatures as we move from industrial to home composting hugely influences the biodegradation phenomena. The dominant and rate-determining chemical hydrolysis phase for PLA proceeds very slowly. Different biostimulants could be explored and studied to improve the hydrolytic phase without altering or modifying the PLA structure. In the case of the biostimulation approach, further delving and fine-tuning the optimal concentrations of biostimulants is essential to maximize their effectiveness while minimizing the associated costs and environmental impacts of accelerating PLA biodegradation. Systematic studies must be conducted by varying the concentrations of biostimulants to identify the optimal concentration and maximize the biodegradation efficiency. To provide a realistic and economic standpoint, laboratory-level studies should be scaled up to assess the feasibility and compatibility of the novel methodologies with the existing home infrastructure in place, specifically when compared to the industrial composting plants. Further testing of the phylogenetic makeup and microbial interactions due to adding biostimulants during the composting process should be conducted. This will help better understand the changes happening in the compost since compostability is a desired end-of-life scenario for contaminated PLA.

This dissertation used CO₂ evolution tracking and molecular weight determination to determine the acceleration of the biodegradation process, but further tests need to be conducted to understand the changes taking place at the molecular level and to comprehend better the process governing the interaction between the biostimulants, native microbial population in the compost, and PLA as a polymer model system. Carbon 13 label of polymer could also be a more expanded methodology to evaluate the biodegradation of PLA. Different scientific methods, such as enzymatic activity assays, spectroscopy analysis, and scanning electron microscopy imaging, can be complemented with computational modeling techniques to examine further and unravel complex pathways involved in PLA biodegradation. Machine learning can also be used to extract and analyze large existing microorganism datasets, identify patterns, and develop prediction models for optimizing PLA biodegradation before running compostability tests.

The enzymatic pretreatment of PLA with proteinase K shows promising results, but different enzymes, pretreatment conditions, and durations should be further explored. This should be primarily conducted in a solid matrix such as compost as the current studies are limited to liquid media, a small amount of PLA tested, and isolated with specific microbial strains that do not replicate real-life disposal compost scenarios.

This dissertation has provided three different approaches that can be used to enhance the biodegradation rate of PLA. From a broader perspective, future efforts should focus on using these novel methodologies in collaboration with industries, governments, and composting facilities to test the implications and improve PLA recovery.